Philosophische Fakultät Institut für Sprache und Information

Bachelorarbeit

Integrating Supertag Features into Neural Discontinuous Constituent Parsing

Erstgutachten: Dr. Kilian Evang Zweitgutachten: Univ.-Prof. Dr. Laura Kallmeyer

> Vorgelegt von: Lukas Michael Mielczarek, 2756129 lukas.mielczarek@uni-duesseldorf.de Computerlinguistik, BA

> > Abgabe: Düsseldorf, 14.09.2023

Contents

1	Intr	roduction 1
	1.1	Motivation
	1.2	Thesis Aims
	1.3	Related Work
	1.4	Structure
n	The	nameasing Contact Decorage
4	1ra	Demographics 7
	2.1	Ingrifficiance of Context Free Comments
	2.2	Defining LOEDC-
	2.3	Denning LOFRSS
	2.4	IreeDanks
3	Par	sing LCFRS 19
	3.1	CYK Parser
		3.1.1 Prerequisites
		3.1.2 Deduction Rules
		3.1.3 Chart Parsing
		3.1.4 Algorithm
		3.1.5 Soundness and Completeness
		3.1.6 Complexity
		3.1.7 Related Work
	3.2	Transition-Based Parsing
		3.2.1 Transition Systems
		3.2.2 Comparing Chart Parsing and Transition-Based Parsing
		3.2.3 Standard Shift-Reduce Projective Parsing 31
		3.2.4 Reordering with the SWAP Action
		3.2.5 Reordering with a Split Stack and the GAP Action
		3.2.6 Lexicalisation
		3.2.7 Scorers
4	Don	sing with a Stack Free Transition System 57
4	1 ai	Configurations 57
	4.2	Transitions 57
	4.3	Oracle 59
	110	4.3.1 Static Oracle 59
		4.3.2 Dynamic Oracle 60
		4.3.3 Oracle Correctness 64
	44	Complexity 65
	4.5	Implementation 65
	1.0	4 5 1 Token Representations 65
		4.5.2 Set Representations 68
		4.5.3 Action Scorer 70
		454 POS Tagger 71
		$455 \text{Objective Function} \qquad \qquad$
		4.5.6 Dealing with Unknown Words
		457 Training 79
		4.5.8 Evaluation

5	Exp	laining	g Supertags	74		
	5.1	Lexica	lised Grammar Formalisms	75		
	5.2	Combi	inatory Categorial Grammar	75		
		5.2.1	CCG Treebanks	79		
		5.2.2	Supertaggers and Parsers	79		
	5.3	Discor	atinuous Constituents and CCG	80		
		5.3.1	Wh-movement	81		
		5.3.2	Fronted Quotations	85		
		5.3.3	Extraposed Dependents	86		
		5.3.4	Circumpositioned Quotations	89		
		5.3.5	It-Extrapositions	91		
		5.3.6	Subject-Verb Inversion	91		
		5.3.7	Summary	94		
0	T /			0.0		
6	Inte	egratin	g Supertags	96		
	6.1	Pipelii	ne Approach	96		
		0.1.1	Supertag Representation	96		
	<u> </u>	6.1.2	Hyperparameters	97		
	6.2	Auxilia	ary Approach	98		
		6.2.1		98		
		6.2.2	Residual Connections	101		
		6.2.3	Increasing Model Width	103		
		6.2.4	Feature Bootstrapping	104		
		6.2.5	Head-Dependency Structure	107		
		6.2.6	Other Auxiliary Tasks	108		
		6.2.7	Hyperparameters	110		
	6.3	Experi	iments	111		
		6.3.1	Experimental Protocol	112		
		6.3.2	General Effects of Supertag Integration	112		
		6.3.3	Per-Phenomenon Evaluation	118		
		6.3.4	Error Analysis	121		
		6.3.5	Sample Analysis	123		
		6.3.6	External Comparison	132		
7	Cor	nclusio	n	135		
	7.1	Open	Questions	136		
		~ P.011		100		
Re	efere	nces		150		
\mathbf{St}	Statutory Declaration 151					

List of Figures

1.1	Projective derivation tree for the man took the book
1.2	A tree from Penn Treebank for Areas of the factory were particularly dusty where
	the crocidolite was used
1.3	A tree from NeGra Corpus for Darüber muß nachgedacht werden $\ldots \ldots \ldots 3$
1.4	LTAG elementary tree for <i>took</i>
2.1	CFG derivation tree for $abcba$
2.2	Domain of locality of CFG and LCFRS
2.3	LCFRS derivation tree for <i>abccab</i>
3.1	Deduction principle
3.2	Weighted CYK deduction system for PLCFRSs
3.3	CYK parse for word <i>aaa</i> and LCFRS G
3.4	Shift-reduce as deductions
3.5	Illustration of standard shift-reduce configuration
3.6	Shift-Reduce parse for <i>the man took the book</i> producing the tree from Figure 1.1 33
3.7	Two tree-cuts in a projective tree
3.8	SWAP transition
3.9	Example parse for SWAP transition
3.10	SWAP transition standard condition
3.11	Example for projective ordering by post-order traversal
3.12	Strict eager SWAP parse
3.13	Projective ordering for the tree over Darüber muß nachgedacht werden $\ldots \ldots 40$
3.14	Minimal example for the motivation of lazy swap
3.15	Tree with projective, fully projective and maximal fully projective constituents
	marked
3.16	Tree without maximal fully projective nodes
3.17	Parse using lazier swap
3.18	Worst case tree for SWAP 44
3.19	Implicit binarisation via MERGE and LABEL-X
3.20	Illustration of the ML-GAP configuration
3.21	Illustration of the FSA for the action sequences allowed in ML-GAP 49
3.22	Deduction transitions for GAP-ML
3.23	Example non-binary tree
3.24	Parse using GAP transition
3.25	Example of worst case tree of length 5 with respect to the GAP transition 54
3.26	Lexicalised discontinuous tree
4.1	Illustration of the stack-free configuration
4.2	Transitions for the set-free system
4.3	Parse using stack-free transition system
4.4	Stack-free neural parser with shared POS-tagging task architecture
5.1	Minimal example CCG lexicon
5.2	Rules for CCG 77
5.3	CCG function application
5.4	CCG composition and type raising
5.5	Coordinative use of type-raising and composition in CCG
5.6	Spurious ambiguity in CCG derivations that leads to distinct semantic represen-
	tations
5.7	Wh-movement in both DPTB and CCGrebank

5.8	Wh-word that asks for the subject from CCGrebank
5.9	Relative clause with <i>that</i> -fronting in DPTB annotation
5.10	Relative clause with <i>that</i> -fronting in CCGrebank annotation
5.11	Example for the use of <i>that</i> as a demonstrative pronoun with the corresponding
	derivation in CCGrebank
5.12	Example for the use of <i>that</i> as a conjunction in a complement clause including the
	derivation from CCGrebank
5.13	DPTB example for wh -fronting with attachment ambiguity $\ldots \ldots \ldots $ 84
5.14	CCGrebank derivation corresponding to the phrase in figure 5.13
5.15	DPTB derivation of a sentence that contains a <i>wh</i> -word that asks for an adverbial
	phrase
5.16	CCG derivation of the sentence <i>How deeply do they read it?</i> in the CCGrebank . 85
5.17	Discontinuous fronted quotation in the DPTB
5.18	Fronted quotation in CCGrebank
5.19	Projective fronted quotation with CCGrebank derivation
5.20	Extraposed subordinate clause from DPTB
5.21	Extraposed subordinate clause in CCGrebank notation 87
5.22	CCGrebank derivation where the subordinate clause that he sends is also a
0.22	modifier to the VP in the corresponding DPTB annotation 87
5.23	Extraposition together with its CCGrebank derivation where the corresponding
0.20	DPTB annotation suggests attachment of the extraposition at the NP node directly
	above anything
5.24	Circumpositioned quotation with an obscure CCCrebank analysis
5.24	Fronted extraposition with the DPTB annotation
5.26	Fronted extraposition with the CCCrebark annotation
5.20 5.27	Circumpositioned quotation and its DPTB analysis
5.28	Circumpositioned quotation and its CCCrebank analysis
5.20	DPTB analysis of an <i>it</i> -extraposition
5 30	CCCrebank analysis of an it extraposition 03
5.30	CCCrebank analysis of a projective sentence with a non-expletive use of it 03
5 32	Example sentence for subject-verb inversion in the DPTB 04
5 33	Example sentence for subject-verb inversion in the DTTD
6.1	Stack free neural parson with supertagging pipeline approach
6.2	Stack-free neural parser with supertagging pipeline approach
6.2	Design of the gated residual connection
6.4	Stack free neural parson with gated residual connections
0.4 6 5	Stack-free fieural parser with gated residual connections
0.5 6.6	Example for root and sketch subtasks in LIAG 104
0.0 6.7	Used dependence structure in the CCC shark
0.1	Deletine DOC based marking of denomber granning
0.0	Relative POS-based encoding of dependency parsing
0.9 C 10	Progression of F-score for CCC and CCC ^{are}
0.10	Progression of F-score for CCG_{600}° and $CTR_{3,600}^{\circ}$
0.11	Gold derivation of <i>wn</i> -extraction with <i>now</i> in the DPTB
0.12 6.19	Provide the produced by the PIPELINE model
0.13	Derivation of <i>wh</i> -extraction with <i>now</i> in the UUGrebank
0.14	Call designation of wheetra sting i'll the interpreter of a DDED
0.15	Gold derivation of <i>wn</i> -extraction with <i>that</i> in the DPTB
0.16	Faulty derivation of wh -extraction with that produced by the CCG ^{gaue} model . 128
0.17	UUG supertags from the UUG rebank for <i>wh</i> -extraction with <i>that</i> \dots 128

6.18	CCG supertags assigned by depccg for wh -extraction with $that$	128
6.19	Gold derivation of a fronted quotation encapsulated in parentheses in the DPTB	129
6.20	Faulty derivation of a fronted quotation encapsulated in parentheses output by the	
	$CCG^{gate} model \dots \dots$	129
6.21	DPTB derivation including a discontinuous dependency with a PP gap element	
	and an SBAR right element	131
6.22	CCGrebank supertag assignment for a sentence that includes discontinuous de-	
	pendency with a PP gap element and an SBAR right element $\hdots \hdots \hddots \hdots \hdot$	131
6.23	Alternative CCG supertag assignment for discontinuous dependency using crossed	
	composition	132
6.24	Alternative CCG supertag assignment for discontinuous dependency with a verbal	
	category that preserves the argument of its subject	132

List of Tables

4.1	Hyperparameters of the stack-free discontinuous constituent parser of Coavoux and				
	Cohen (2019)	73			
6.1	Hyperparameters of the supertag pipeline model	98			
6.2	Hyperparameters of the supertag auxiliary-task models	111			
6.3	Hyperparameters of the supertag multi-task models	111			
6.4	Results of the experiments.	113			
6.5	Results of additional experiments with a variation of auxiliary tasks $\ldots \ldots \ldots$	117			
6.6	Per-phenomenon results	120			
6.7	Error types	122			
6.8	Labelled evaluation results specific to lexical triggers of wh -extraction $\ldots \ldots \ldots$	123			
6.9	Labelled evaluation results for discontinuous dependency according to the node label				
	of the gap element	130			
6.10	Labelled evaluation results for discontinuous dependency according to the node label				
	of the right element.	130			
6.12	Comparison with previously published labelled per-phenomenon F-scores	133			
6.11	Global comparison of parsing results on the DPTB	134			

List of Algorithms

3.1	Removing useless variable separation	20
3.2	Weighted deductive CYK parsing for PLCFRS	23
3.3	Deterministic transition-based parsing algorithm	29
3.4	$\operatorname{PostOrder}(r)$ algorithm	38
3.5	SWAP oracle extraction algorithm	40
3.6	ML-Gap oracle extraction	53
4.1	Stack-free static oracle extraction	61
6.1	Proportional task sampling	101

1. Introduction

1.1. Motivation

Syntactic parsing lies at the foundation of many natural-language processing applications. The term denotes the automatic extraction of a syntactic representation from a natural language text, usually in the form of a sequence of words and punctuation marks. The representation is designed to capture information about the elements' grammatical relations and functions. This output is traditionally used as a basis for higher-level tasks like grammar checking, semantic analysis, question answering and information extraction (Jurafsky and Martin, 2009, chapter 12).

One widely used description of syntax is so-called *constituent structure*. Rooted in X-bar theory (Chomsky, 1970)¹, the idea of constituency arises from the observation that groups of words can behave as single units (Jurafsky and Martin, 2009, chapter 12). Take for instance the noun phrases in (1.1).

- (1.1) a. the memory
 - b. a new computer
 - c. the person she really likes

Despite having completely different meanings, these three terms can occur in similar syntactic environments, for example before a verb as in (1.2).

(1.2) a. the memory fades

- b. a new computer arrives...
- c. the person she really likes *waits...*

However, this is not true for all of the individual components of the phrases, as can be seen in (1.3). Therefore, one assumes the existence of an abstract category called a *constituent* to which one or more words can belong and for which rules like "noun phrases can be followed by verbs" can be postulated (Jurafsky and Martin, 2009, chapter 12).

- (1.3) a. *the fades
 - b. *new arrives...
 - c. *likes spends...

Note: The asterisk marks constructions that are not grammatical.

Traditional views of constituency demand that constituents consist of adjacent words. This is rooted in a system for modelling languages called *context-free grammar* (CFG) (Chomsky, 1956), a generative device that derives sentences by using rewrite rules, starting with an initial symbol. The derivation path marks a hierarchy of constituents that can be arranged into a *derivation/parse tree*. Figure 1.1 gives an example for such a description. Note that the tree only contains non-crossing edges which is why it is called *projective* or *continuous*.

This view of constituency was adopted by many *treebanks*. Treebanks are linguistic corpora in which the sentences have been annotated with syntactic representations (e.g. constituency trees) (Jurafsky and Martin, 2009, chapter 12). However, this leads to difficulties when analysing syntactic phenomena that are believed to exhibit non-local dependencies. While the analysis of German or other languages with varying degrees of free word order evidently questions the

 $^{^1\}mathrm{Please}$ refer to Kornai and Pullum (1990) for a critical reflection on Chomsky's ideas.



Figure 1.1: Projective derivation tree for the man took the book (Chomsky, 1956).

projective approach, the English language also exhibits a fair amount of discontinuous phenomena, see (1.4) and (1.5).

- (1.4) a. sie *packte* ihre Werkzeuge *ein* she packed her tools up 'she packed her tools'
 - b. *den Dank* möchte er nicht *annehmen* the thanks want he not accept

'he does not want to accept the thanks'

c. ohne Scham scheinen sie das komplette Buffet aufgegessen zu haben without shame seem they the complete buffet eaten to have 'they seem to have eaten the entire buffet without shame'

(1.5) a. areas of the factory were particularly dusty where the crocidolite was used (Evang, 2011)

b. a man entered who was wearing a black suit (McCawley, 1982)

Some treebanks like the English *Penn Treebank* (Marcus et al., 1993) opted to mark discontinuous phenomena through indexing and trace nodes, i.e. empty nodes in the place where a discontinuous subsection of the sentence would be expected in "standard" word order. This is motivated by the tradition of analysing discontinuities as instances of transformation or syntactic movement as introduced by Chomsky (1975). The tree in figure 1.2 features three cases of co-indexing. While such a strategy acknowledges the existence of discontinuous relationships, the long-range dependencies of the markers cannot be fully captured by traditional parsing techniques for projective grammars and often remain unused.



Figure 1.2: A tree from Penn Treebank, without arc-labels, adopted from Evang (2011).

In a number of treebanks like the German NeGra (Skut et al., 1998) and TIGER treebanks (Brants et al., 2004) or English *discontinuous Penn Treebank* (DPTB) (Evang and Kallmeyer, 2011)

long-range dependencies are represented by crossing edges and constituents with non-adjacent elements. The resulting trees are called *discontinuous constituent trees*. Figure 1.3 shows an example from the NeGra corpus.



Figure 1.3: A tree from NeGra corpus, without arc-labels (Kallmeyer, 2010, chapter 6).

In order to capture syntactic descriptions with discontinuous constituents, several formalisms have been proposed. Among these *linear context-free rewriting systems* (LCFRS) (Vijay-Shanker et al., 1987) have been shown to be a natural candidate for accurately covering the notion of discontinuous constituents employed by the aforementioned treebanks.

Being a foundation for many natural language processing tasks, parsing speed is a key concern and a driver of active research. Parsing LCFRS is non-trivial in terms of efficiency. The adaptation of traditional chart parsing for LCFRS results in a polynomial time complexity of $\mathcal{O}(n^{3\cdot dim(G)})$ where *n* is the length of a sentence and dim(G) is the maximal number of discontinuous blocks of a constituent in the grammar. Transition-based parsing approaches aim at reducing this factor by doing away with the need for an explicit grammar. Instead, an artificial neural network is trained to produce discontinuous constituent trees given only raw text input using *supervised learning* on large annotated corpora. Errors in prediction are backpropagated through the network resulting in optimisations of its parameters. In this way, it implicitly learns to identify regularities that it can effectively use to predict trees for unseen data. A recent and elegant proposal for a neural stack-free transition-based parser developed by Coavoux and Cohen (2019) successfully allows for the derivation of any discontinuous constituent tree over a sentence in worst-case quadratic time.

1.2. Thesis Aims

The purpose of this work is to explore the possibility of enhancing the accuracy of neural grammarless discontinuous constituent parsing by introducing *supertag* information. In so-called *lexicalised* grammar formalisms like *lexicalised tree adjoining grammars* (LTAG) (Schabes and Joshi, 1991) and *combinatory categorial grammar* (CCG) (Steedman, 1989, 1996, 2000) informative categories are assigned to words in a sentence that act as the sole building blocks in the composition of the sentence's overarching syntactic representation. These categories are called *supertags*. The set of rules for connecting these building blocks is kept minimal. Therefore, a supertag assignment gives strong information about the structural role of the word in the sentence and about its syntactic relationship with the surrounding items.

Figure 1.4 shows a possible supertag assignment for the word *took* in LTAG. The incomplete tree serves as a primitive structure in the formalism that directly dictates the position and number of the verb's nominal argument phrases (NP). Given this assignment, a parser would not need to reconstruct the higher-level constituent structure any more. Instead, it would simply need to link a group of elementary trees.²

 $^{^{2}}$ This illustrative description of LTAG is of course incomplete. A second operation called *adjoining* also allows the insertion of trees into the inner structure of other trees.



Figure 1.4: LTAG elementary tree for *likes*.

Information of this sort may aid in the analysis of discontinuous constituents. "Non-standard" word order, including discontinuity, can be expected to be directly encoded in the lexical category assignment. This work shall therefore explore its integration based on two approaches: by using the output of a dedicated supertagger as additional input for a neural parser (pipeline) and by jointly training a neural model to perform both parsing and supertagging (auxiliary/multi-task learning). Due to the public availability of a state-of-the-art supertagger for CCG (Yoshikawa et al., 2017), the focus will be on CCG supertags. Additionally, several other kinds of supertags (LTAG-spinal, LCFRS) and sequence labellings tasks (chunking, dependency parsing) will be compared in terms of their suitability as auxiliary tasks for neural discontinuous constituent parsing.

A secondary aim is to present the LCFRS formalism and to give an overview of the landscape of both grammar-based and grammar-less parsing approaches for discontinuous constituent trees. This is done to outline the advancements that led to the development of the stack-free transition system of Coavoux and Cohen (2019) used as the basis for the incorporation of supertags in this work.

1.3. Related Work

The potential of assigning supertags to word sequences using statistical methods as a pre-step to parsing was initially successfully explored by Bangalore and Joshi (1999) for lexicalised tree adjoining grammar (LTAG) and adapted for combinatory categorial grammar (CCG) by Clark and Curran (2010). Recently, Ruprecht and Mörbitz (2021); Ivliev (2020) have presented the first extraction algorithm for supertags based on linear context-free rewriting systems (LCFRS) and achieved state-of-the-art results in discontinuous constituent parsing.

These approaches differ from this work in that they use supertagging as an upstream task for grammar-based parsers of the respective formalism, for instance LCFRS chart-parsing (Ruprecht and Mörbitz, 2021). The extracted supertags or supertag distributions are directly utilised to reduce ambiguity and increase parsing speed by ruling out or prioritizing certain pre-terminal rule applications which effectively prunes the derivational search space. In contrast to that, I investigate the effect of supertagging in the context of a neural, grammar-less parsing approach as an input feature and as an auxiliary task. From this also follows that my choice of supertags is not structurally restricted to a certain formalism but rather to statistical correlations between supertag assignments and parsing actions.

The first thorough exploration of auxiliary/multi-task learning for natural language processing was performed by Collobert and Weston (2008) who simultaneously predict part-of-speech tags, chunk labels³, named entity tags, semantic roles, semantic similarity and the likelihood of a sentence's well-formedness from jointly trained word embeddings. Investigations into the realm of multi-task learning that followed are mainly centred around dependency parsing⁴ and in many

³Chunking separates a sentence into simple constituent types like NP or VP and is sometimes called *shallow* parsing (Collobert and Weston, 2008). Each word is only assigned one unique type. The sequence a pink car would be tagged with B-NP I-NP with B-NP marking the start of the chunk.

 $^{^4\}mathrm{D}$ ependency grammars are a class of grammatical theories centred around the notion of dependency as opposed to

cases deal with incorporating higher-level semantic tasks.

Candito (2022) extracts auxiliary tasks from semantic dependency graphs like the number of dependents of a word and a concatenation of incoming arc labels to increase the accuracy of a dependency parser by training the model to predict these features from a shared bidirectional recurrent neural network. Zhou et al. (2020a) explore jointly learning dependency and span-based constituent parsing together with semantic parsing. This approach is extended by Zhou et al. (2020b) who propose improving the quality of language models like BERT (Devlin et al., 2019) with informed linguistic knowledge by jointly training on the aforementioned tasks.

These approaches differ from this work's objective through the choice of task(s) they incorporate. They are centred around bridging the gap between syntactic and semantic representations while I try to deal with the open problem of discontinuous constituency by searching for a syntactical localised lower-level task that is informative of these phenomena.

The research on multi-task approaches pertaining to supertagging is fairly limited. Yoshikawa et al. (2017) extract dependency graphs from CCG derivations and train a neural model to predict both dependency structure and CCG supertags from a shared representation to resolve derivational ambiguity in the CCG formalism. But the actual CCG parsing is performed using an A* algorithm based on the supertag probabilities predicted by the model. Zhu and Sarkar (2019) deconstruct LTAG supertags into several components like *head*, *root* or *spine* effectively bootstrapping new tasks from an existing supertag annotated corpus. They use these tasks to improve LTAG supertagging performance. Søgaard and Goldberg (2016) show that combining CCG supertagging and POS tagging using a hierarchical approach where the easier POS tagging task is predicted from a lower level yields improvements in supertagging accuracy. This work borrows from their idea of a hierarchical auxiliary task arrangement. Bingel and Søgaard (2017) explore binary multitask relations between a range of natural language processing tasks including CCG tagging, POS tagging, chunking and semantic tasks. The effect of supertag prediction as an auxiliary task for parsing has not been examined to this date.

The work on beneficial auxiliary tasks for neural constituent parsing is also sparse. Coavoux and Crabbé (2017b) predict several word tagging tasks from a shared representation of a model for projective constituent parsing: POS tags, morphological features and functional label, i.e. the relation of a word to its head in a constituent. To the best of my knowledge, besides jointly predicted POS tags (Coavoux and Cohen, 2019), the only exploration of multi-task frameworks for grammar-less discontinuous constituent parsers to this date has been recently brought forward by Johansson and Adesam (2020) who train constituent parsing on a Swedish discontinuous treebank with a similar annotation scheme to NeGra/TIGER. They pursue parsing based on differently annotated corpora, including dependency treebanks, as an auxiliary task with predictions from a shared intermediate representation to leverage the limited availability of linguistic resources in Swedish. This differs from the work at hand in that the objectives of Johansson and Adesam (2020) are all parsing tasks predicted at the same model level while I explore the implementation of a lower-level task. Furthermore, Johansson and Adesam (2020) use a different transition system for discontinuous constituent parsing.⁵

To the best of my knowledge, the work at hand is the first exploration of supertagging in a pipeline arrangement and as an auxiliary task for grammar-less neural constituent parsers. It might therefore help to identify if a shared neural representation benefits parsing and specifically the resolvement of challenging discontinuous phenomena and if further research in this direction is a worthwile endeavour.

Note that there has been a substantial amount of research on the question of beneficial schedul-

constituency. While constituent phrase structure grammars arrange words into hierarchical constituents, dependency grammars express word-to-word relations through labelled directed arcs (Jurafsky and Martin, 2009, chapter 12.7). ⁵They utilise the *Swap* action for reordering (Versley, 2014) while this work is based on the stack-free MERGE-approach proposed by Coavoux and Cohen (2019).

ing of multi-task learning (Kiperwasser and Ballesteros, 2018; Zaremoodi and Haffari, 2019) and on metrics for the exploration of well complementing auxiliary tasks (Ben-David and Schuller, 2003; Bingel and Søgaard, 2017) in the context of natural language processing. I will leave an examination of supertagging and parsing in this regard to future work.

1.4. Structure

The following sections are structured as follows. Section 2 gives a detailed introduction to linear context-free rewriting systems (LCFRS), a natural extension of context-free grammars that accounts for discontinuous constituents. Section 3 presents two discontinuous parsing approaches: LCFRS CYK chart-parsing and grammar-less transition-based parsers. Then, in section 4 follows a discussion of the stack-free transition system proposed by Coavoux and Cohen (2019) including its implementation. Section 5 explains the formalism of combinatory categorial grammar (CCG), a major lexicalised grammar theory that gives rise to informative supertags, and examines its account of discontinuous phenomena that can be found in the discontinuous Penn Treebank (DPTB). Finally, section 6 describes the implementations and results of several experiments for the integration of CCG supertags into the stack-free transition-based neural parser. Section 7 summarises the findings and discusses them in a broader context.

2. Transgressing Context-Freeness

Discontinuous constituent trees can be seen as structural descriptions produced by derivations of linear context-free rewriting systems (LCFRS). They were introduced by Vijay-Shanker et al. (1987) and are an extension of traditional context-free grammars (CFG), which do not suffice to adequately treat discontinuous structures. This section shall be concerned with defining both CFGs and LCFRSs as well as with the derivation trees that accompany them.

2.1. Prerequisites

The formalism of context-free grammars (CFG), developed by Chomsky (1956), has a long history as the foundation for modelling the syntactic structure of natural languages. In order to define CFGs, several prerequisites are needed. I follow the definitions in Kallmeyer (2010).

Definition 2.1 (Alphabet, word, language).

Let X be a non-empty finite set of symbols.

- 1. X is called an alphabet.
- 2. A string $x_1x_2...x_n$ with $n \in \mathbb{N}$ and $x_i \in X$ for all $i \in \{1, ..., n\}$ is called a *non-empty word* over X. X^+ is defined as the set of all non-empty words over X.
- 3. I define $X^* := X^+ \cup \{\varepsilon\}$ for a new element $\varepsilon \notin X^+$ called *empty word*. ε is defined as the neutral element of concatenation on X^* , i.e. for $w \in X^* : w\varepsilon = \varepsilon w = w$. $w \in X^*$ is called a *word over* X.
- 4. A set L is called a language over X iff $L \subseteq X^*$.

Two explanatory notes on the preceding definition: firstly, for $i, j \in \mathbb{Z}$ the expression $\{i, ..., j\}$ is used as a shorthand notation for $\{n \in \mathbb{Z} \mid i \leq n \leq j\}$. Secondly, in the context of natural language an alphabet X can be a set of natural language words treated as unique symbols. For clarity, the term *word* will only refer to the formal definition given above in this work unless stated otherwise.

Definition 2.2 (Length of a word).

Let X be an alphabet and $w \in X^*$. The *length of* w is denoted by |w| and defined recursively:

$$|w| := \begin{cases} 1 + |w'|, & \text{if } w = xw' \text{ for some } x \in X \\ 0, & \text{otherwise.} \end{cases}$$

Definition 2.3 (Word repetition).

Let X be an alphabet and $w \in X^*$. Word exponentiation for $k \in \mathbb{N}_0$ is defined as word repetition:

$$w^k := \begin{cases} \varepsilon, & \text{if } k = 0, \\ ww^{k-1}, & \text{otherwise.} \end{cases}$$

Definition 2.4 (Context-free grammar).

A context-free grammar G is defined by a 4-tuple $G = \langle N, T, P, S \rangle$ where

- 1. N is an alphabet. The elements of N are called *nonterminals*.
- 2. T is an alphabet. The elements of T are called *terminals*.
- 3. it holds that $N \cap T = \emptyset$.

8

- 4. $P \subset N \times (N \cup T)^*$ is a finite set of *productions/rewriting rules*. A production $\langle A, \beta \rangle \in P$ can be written as $A \to \beta$.
- 5. $S \in N$ is the start symbol.

Context-free grammars describe languages by generating the words of the language through derivation. The following definition is given by Jurafsky and Martin (2009, chapter 12):

Definition 2.5 (Derivation).

Let $G = \langle N, T, P, S \rangle$ be a CFG.

- 1. Given $w, w' \in (N \cup T)^*$, w directly derives w' iff there exists a production $A \to \beta \in P$ and there are strings $\alpha, \gamma \in (T \cup N)^*$ such that $w = \alpha A \gamma$ and $w' = \alpha \beta \gamma$. In this case we write $w \Rightarrow w'$.
- 2. For $w_1, w_2, ..., w_n \in (N \cup T)^*, n \in \mathbb{N}$ such that $w_1 \Rightarrow w_2, w_2 \Rightarrow w_3, ..., w_{n-1} \Rightarrow w_n$ we write $w_1 \stackrel{*}{\Rightarrow} w_n$ and say that w derives w'. $\stackrel{*}{\Rightarrow}$ is the reflexive transitive closure of \Rightarrow .

For $w_1, w_2, w_3 \in (N \cup T)^*$ such that $w_1 \Rightarrow w_2$ and $w_2 \Rightarrow w_3$ the shorthand-notation $w_1 \Rightarrow w_2 \Rightarrow w_3$ is used.

Definition 2.6 (Context-free language).

For a CFG $G = \langle N, T, P, S \rangle$ the language of G is denoted by \mathscr{L}_G and defined as the set of all strings that consist only of terminal symbols and can be derived from the start symbol:

$$\mathscr{L}_G := \{ w \mid w \in T^* \land S \stackrel{*}{\Rightarrow} w \}.$$

A language L is called *context-free* iff there exists a CFG G such that $L = \mathscr{L}_G$.

Example 2.7.

Equation 2.1 gives a set of CFG rules for the language L over $\{a, b, c\}$ with $L = \{xyz \mid x, z \in \{a, b\}^+, y \in \{c\}^+\}$, numbered for easy reference. A derivation of the word $abcba \in L$ is shown in equation 2.2 with the numbers below the arrows referencing the production that was applied.

$$r_{1}: S \rightarrow ACA$$

$$r_{2}: A \rightarrow aA$$

$$r_{3}: A \rightarrow bA$$

$$r_{4}: A \rightarrow a$$

$$r_{5}: A \rightarrow b$$

$$r_{6}: C \rightarrow cC$$

$$r_{7}: C \rightarrow c$$

$$(2.1)$$

$$S \underset{r_1}{\Rightarrow} ACA \underset{r_2}{\Rightarrow} aACA \underset{r_5}{\Rightarrow} abCA \underset{r_7}{\Rightarrow} abcA \underset{r_3}{\Rightarrow} abcbA \underset{r_4}{\Rightarrow} abcba$$
(2.2)

The definition of trees is essential to our matter. I base the following definitions on Kallmeyer (2010, chapter 1).

Definition 2.8 (Directed graph).

- 1. A directed graph is a pair $\langle V, E \rangle$ where V is a finite set of nodes/vertices and $E \subseteq V \times V$ a set of edges.
- 2. For $v \in V$ the *in-degree of* v is defined as $|\{v' \in V \mid \langle v', v \rangle \in E\}|$.

- 3. For $v \in V$ the *out-degree of* v is defined as $|\{v' \in V \mid \langle v, v' \rangle \in E\}|$.
- 4. E^+ is the transitive closure of E; E^* is the reflexive transitive closure of E.

Definition 2.9 (Path).

A sequence $v_1, ..., v_n$ of vertices in V with $n \in \mathbb{N}$ such that $v_k E v_{k+1}$ for all $k \in \{1, ..., n-1\}$ and all vertices in the sequence are distinct is called a *path in* $\langle V, E \rangle$.

Definition 2.10 (Tree).

- 1. A tree is a triple $\langle V, \triangleleft, \hat{v} \rangle$ such that⁶
 - (a) $\langle V, \triangleleft \rangle$ is a directed graph and $\hat{v} \in V$ (called the *root node*),
 - (b) \triangleleft is acyclic, i.e., $\nexists v \in V : v \triangleleft^+ v$,
 - (c) \hat{v} has in-degree 0,
 - (d) all nodes $v \in V \setminus \{\hat{v}\}$ have in-degree 1 and
 - (e) every node is accessible from \hat{v} , i.e., $\forall v \in V : \hat{v} \triangleleft^* v$.
- 2. Given a tree $\langle V, \triangleleft, \hat{v} \rangle$,
 - (a) we say that v_1 directly dominates v_2 iff $v_1 \triangleleft v_2$ for $v_1, v_2 \in V$. In this case, we call v_1 the parent of v_2 and v_2 a daughter of v_1 .
 - (b) we say that v_1 dominates v_2 iff $v_1 \triangleleft^* v_2$ for $v_1, v_2 \in V$.
 - (c) we say that v_2 is a *sister* of v_3 iff $v_1 \triangleleft v_2$ and $v_1 \triangleleft v_3$ for $v_1, v_2, v_3 \in V$.
- 3. An ordered tree is a quadruple $\langle V, \triangleleft, \prec, \hat{v} \rangle$ such that $\langle V, \triangleleft, \hat{v} \rangle$ is a tree and \prec is a *linear* precedence relation with the following properties:
 - (a) \prec is irreflexive, antisymmetric and transitive,
 - (b) $\forall v_1, v_2 \in V : (v_1 \not \lhd^* v_2 \land v_2 \not \lhd^* v_1) \Rightarrow v_1 \prec v_2 \lor v_2 \prec v_1,$
 - (c) $\forall v_1, v_2 \in V : (v_1 \not A^* v_2 \wedge v_2 \not A^* v_1) \wedge ((\exists v_3 \in V : v_3 \triangleleft v_1 \wedge v_3 \prec v_2) \lor (\exists v_4 \in V : v_4 \triangleleft v_2 \wedge v_1 \prec v_4)) \Rightarrow v_1 \prec v_2,$
 - (d) nothing else is in \prec .

A node with out-degree 0 is called a *leaf*. All other nodes are called *internal*.

Definition 2.11 (Labelling).

A labelled directed graph is a triple $\langle V, E, \lambda \rangle$ such that $\langle V, E \rangle$ is a directed graph and λ is a function $\lambda : V \to A$ called the node labelling of the graph over signature A. This definition naturally extends to labelled trees $\langle V, \triangleleft, \hat{v}, \lambda \rangle$ and labelled ordered trees $\langle V, \triangleleft, \neg, \hat{v}, \lambda \rangle$.

For ordered and for labelled trees I explicitly include the precedence relation and the labelling function in the tuple defining the respective tree type since it will facilitate working with these structures in the following sections.

Definition 2.12 (Syntactic tree).

Let N and T be disjoint alphabets (of nonterminal and terminal symbols).

A syntactic tree over N and T is an ordered finite labelled tree $\langle V, \triangleleft, \prec, \hat{v}, \lambda \rangle$ such that $\lambda(v) \in N$ for each internal node v and $\lambda(u) \in (N \cup T \cup \{\varepsilon\})$ for each leaf u.

Definition 2.13 (Parse tree, derivation tree, tree language of a CFG). Let $G = \langle N, T, P, S \rangle$ be a CFG.

⁶The use of the symbol \triangleleft was adopted from Versley (2014). I naturally extend the notation to $v_1 \not \lhd v_2 :\Leftrightarrow \langle v_1, v_2 \rangle \notin \triangleleft$.

- 1. A syntactic tree $\langle V, \triangleleft, \prec, \hat{v}, \lambda \rangle$ over N and T is a parse tree in G iff
 - (a) $\lambda(v) \in (T \cup \{\varepsilon\})$ for every $v \in V$ if v is a leaf,
 - (b) for every $v_0, v_1, ..., v_n \in V, n \in \mathbb{N}$ such that $v_0 \triangleleft v_i$ for all $i \in \{1, ..., n\}$ where there is no $u \in V : v_0 \triangleleft u$ and $u \notin \{v_1, ..., v_n\}$ and where $v_j \prec v_{j+1}$ for all $j \in \{1, ..., n-1\}$, it holds that $\lambda(v_0) \rightarrow \lambda(v_1), ..., \lambda(v_n) \in P$.
- 2. A parse tree $\langle V, \triangleleft, \prec, \hat{v}, \lambda \rangle$ is a *derivation tree* in G iff $\lambda(\hat{v}) = S$.
- 3. The tree language of G is defined as $\mathscr{L}_G^{\mathcal{D}} := \{D \mid D \text{ is a derivation tree in } G\}.$

Following Chomsky (1963) the notion of generative capacity can be introduced:

Definition 2.14 (Generative capacity).

- The weak generative capacity of a grammar G is defined by all strings it can generate, i.e. by \mathscr{L}_G .
- The strong generative capacity of a grammar G is defined by all structural descriptions it produces. For a grammar where the notion of a tree language is defined this amounts to $\mathscr{L}_G^{\mathcal{D}}$.

Example 2.15.

Figure 2.1 depicts a derivation tree for the word *abcba* in the grammar given in example 2.7.



Figure 2.1: Derivation tree for the word *abcba* using the rules from example 2.7.

2.2. Insufficiency of Context-Free Grammars

Since the early 1980s, many researchers argued that CFGs are inadequate for describing natural language. Many of these arguments, however, were based on the strong generative capacity of grammars, e.g. Bresnan et al. (1982), who analysed non-local dependencies in Dutch. They relied on assumptions about the underlying structure and were therefore susceptible to critique. Primarily, this style of argument does not rule out the possibility that a different context-free grammar formalism (with different structural descriptions) might be able to generate the language in question (Kallmeyer, 2010, chapter 2).

The matter was finally resolved when Shieber (1985) showed that nesting relative clauses with crossing dependencies in Swiss German also surpass the weak generative capacity of CFGs. These findings led to the search of appropriate grammatical formalisms strong enough to generate long-distance relationships and at the same time restricted enough to allow for efficient parsing. In this context, Joshi (1985) coined the term *mild context-sensitivity* for a category of formalisms that possess certain desirable properties. Kallmeyer (2010, chapter 2) give the following definition for mild context-sensitivity:

Definition 2.16 (Mild context-sensitivity).

1. A set of languages \mathcal{L} over an alphabet X is mildly context-sensitive iff



Figure 2.2: Illustration of the domain of locality of CFG and LCFRS, inspired by Kallmeyer (2010, chapter 1).

- (a) \mathcal{L} contains all context-free languages,
- (b) \mathcal{L} can describe a limited amount of cross-serial dependencies, i.e. there is an $n \ge 2$ such that $\{w^k \mid w \in T^*\} \in \mathcal{L}$ for all $k \le n$,
- (c) the languages in \mathcal{L} can be parsed in polynomial time and
- (d) the languages in \mathcal{L} have the constant growth property.
- 2. a formalism F is mildly context-sensitive iff the set of languages it can describe $\{\mathscr{L}_G \mid G \in F\}$ is mildly context-sensitive.

The constant-growth property means, that if one orders the lengths of the words in a language, the difference in length between two consecutive elements is restricted by a constant (Kallmeyer, 2010, chapter 2). This is defined formally by Weir (1988) in the following way:

Definition 2.17 (Constant growth property).

Let L be a language over an alphabet X. L has the constant growth property iff there is a constant $c_0 > 0$ and a finite set of constants $C \subset \mathbb{N}$ such that for all $w \in L$ with $|w| > c_0$, there is a $w' \in L$ with |w| = |w'| + c for some $c \in C$.

Several formalisms in the scope of mild context-sensitivity have been proposed. Among these are tree adjoining grammar (TAG) (Joshi et al., 1975), *head grammar* (HG) (Pollard, 1984), *linear indexed grammars* (LIG) (Gazdar, 1988) and linear context-free rewriting systems (LCFRS) (Vijay-Shanker et al., 1987) as well as certain variants of combinatory categorial grammar (CCG) (Steedman, 1989, 1996, 2000).

LCFRSs were shown to naturally arise from (discontinuous) treebank representations (Kallmeyer, 2010, chapter 1). They are an adequate formalism for describing discontinuous constituents utilised in treebanks and will therefore be a subject for further inspection in the following sections.

2.3. Defining LCFRSs

Linear context-free rewriting systems (LCFRS) are a powerful type of mildly context-sensitive grammar formalism that extends the rewriting-nature of CFGs. Like context-free grammars, the formalism possesses rewriting rules with *left-hand sides* (LHS) and *right-hand sides* (RHS). Words are derived through recursive application of the rules which in turn constitute the constituent structure desirable from a linguist's point of view.

The key difference is the fact that nonterminals do not yield a string of terminals, but a tuple of strings. The strings in one tuple may be intertwined with elements of other tuples. This allows an element to have a yield of non-adjacent strings. The size of the tuples is fixed for each nonterminal A and is called its *fan-out*. Thus, the fan-out of A gives the maximum number of (discontinuous) components A dominates. The fan-out of the start symbol is set to 1. Figure 2.2 shows a tree where the nonterminal A yields three non-adjacent components $\gamma_1, \gamma_2, \gamma_3$.

LCFRSs are equivalent to simple *range concatenation grammars* (SRCG) (Boullier, 1998b) and to *multiple context-free grammars* (MCFG) (Seki et al., 1991). A formal definition for LCFRS was first outlined by Vijay-Shanker et al. (1987). In the following I will reference the characterisation given by Evang (2011) who adapts the SRCG notation from Boullier (1998b) for LCFRS.

Definition 2.18 (Linear context-free rewriting system).

A linear context-free rewriting system (LCFRS) is a 5-tuple $G = \langle N, T, V, P, S \rangle$ where

- 1. N, T and V are alphabets. The elements of N are called *nonterminals*, those of T terminals and those of V variables. T and V are disjoint.
- 2. each nonterminal $A \in N$ has an associated fan-out denoted by a function $\dim : N \to \mathbb{N}$.
- 3. the maximum fan-out is denoted by $dim(G) := \max(\{dim(A) \mid A \in N\})$. For k = dim(G), G is said to have fan-out k and is called a k-LCFRS.
- 4. $P \subset (N \times ((T \cup V)^*)^{\leq \dim(G)}) \times (N \times V^{\leq \dim(G)})^{<\mathbb{N}_0}$ is a finite set of productions/rewriting rules.⁷ A rule

$$r = \langle \langle A_0, \langle \chi_{0,1}, ..., \chi_{0,dim(A_0)} \rangle \rangle, \langle \langle A_1, \langle \chi_{1,1}, ..., \chi_{1,dim(A_1)} \rangle \rangle, ..., \langle A_m, \langle \chi_{m,1}, ..., \chi_{m,dim(A_m)} \rangle \rangle \rangle$$

can be written as

$$A_0(\langle \chi_{0,1}, ..., \chi_{0,dim(A_0)} \rangle) \to A_1(\langle \chi_{1,1}, ..., \chi_{1,dim(A_1)} \rangle) ... A_m(\langle \chi_{m,1}, ..., \chi_{m,dim(A_m)} \rangle)$$

where the part left of the arrow is called left-hand side (LHS) and denoted by LHS(r) and the part right of the arrow is called right-hand side (RHS) and denoted by RHS(r). Furthermore:

- (a) Every variable occurring in a rule occurs exactly once on the LHS and exactly once on the RHS.
- (b) There is a function

$$rank: P \to \mathbb{N}, \qquad r \mapsto |RHS(r)|,$$

associated with the grammar which assigns the number of its RHS's components to a rule. Furthermore, *rank* is also defined for the grammar as the maximum rank occurring for any rule:

$$rank(G) := \max(\{rank(r) \mid r \in P\}).$$

- (c) The strings $\chi_{0,1}, ..., \chi_{0,dim(A_0)} \in (T \cup V)^*$ are called the LHS arguments. $\langle \chi_{0,1}, ..., \chi_{0,dim(A_0)} \rangle$ is called the LHS argument tuple.
- (d) The variables $\chi_{1,1}, ..., \chi_{1,dim(A_1)}, ..., \chi_{m,1}, ..., \chi_{m,dim(A_m)} \in V$ are called the RHS arguments. The tuples $\langle \chi_{1,1}, ..., \chi_{1,dim(A_1)} \rangle, ..., \langle \chi_{m,1}, ..., \chi_{m,dim(A_m)} \rangle$ are called the RHS argument tuples.
- (e) if rank(r) = 0, the RHS is written as ε .

By convention $A_k(\langle \chi_{n,1}, ..., \chi_{n,dim(A_k)} \rangle)$ is equivalent to $A_k(\chi_{n,1}, ..., \chi_{n,dim(A_k)})$ and a tuple can be denoted by a variable in bold, e.g. $\boldsymbol{\alpha}$. $\boldsymbol{\alpha}^{[i]}$ refers to the *i*-th element of the tuple. The fan-out of a grammar gives the maximum number of non-adjecent strings in the yield of a nonterminal.

In this formalism, rewriting rules describe how the yield of the nonterminal on the LHS can be computed using the yields of the RHS nonterminals. The notion of a yield can be defined formally (Evang and Kallmeyer, 2011):

 $^{{}^{7}}X^{\leq n}$ denotes the set of all tuples with maximum number of components *n* over a set *X*. $X^{<\mathbb{N}_{0}}$ is the set of all finite tuples over *X*.

Let $G = \langle N, T, V, P, S \rangle$ be an LCFRS.

- 1. The function yield : $N \to \mathcal{P}((T^*)^{\leq \dim(G)})$ is defined as follows: for every $A \in N$:
 - (a) for every $A(\boldsymbol{\alpha}) \to \varepsilon \in P$, it holds that $\boldsymbol{\alpha} \in yield(A)$.
 - (b) for every rule

 $A_0(\chi_{0,1},...,\chi_{0,dim(A_0)}) \rightarrow A_1(\chi_{1,1},...,\chi_{1,dim(A_1)})...A_m(\chi_{m,1},...,\chi_{m,dim(A_m)}) \in P$ and for all $\boldsymbol{\alpha}_i \in yield(A_i)$ with $i \in \{1,...,m\}$ it holds that $\langle f(\chi_{0,1}),...,f(\chi_{0,dim(A)}) \rangle \in$ yield(A) where the composition function f is defined as follows:

- i. f(t) = t for all $t \in T$,
- ii. $f(\chi_{i,j}) = \boldsymbol{\alpha}_i^{[j]}$ if $\chi_{i,j} \in V$ for all $i \in \{1, ..., m\}$ and all $j \in \{1, ..., dim(A_i)\}$.
- iii. $f(\beta\gamma) = f(\beta)f(\gamma)$ for all $\beta, \gamma \in (T \cup V)^+$.
- (c) nothing else is in yield(A).
- 2. The string language of G is $\mathscr{L}_G = \{w \mid \langle w \rangle \in yield(S)\}.$

The *yield*-function maps each nonterminal A to its set of possible yields. Every element in the yield is the result of a distinct combination of available yields of the RHS nonterminals of a rule where A stands on the LHS. Each of these elements is a tuple of sequences of terminals.

The function f replaces each variable with the corresponding component of the yield of the RHS argument it also occurs in. Since for every $\chi_{0,k} \in V$ with $k \in \{1, ..., dim(A_0)\}$ there exists exactly one $\chi_{i,j}$ among the RHS arguments with $\chi_{i,j} = \chi_{0,k}$ and unique indices i, j with $i \in \{1, ..., m\}$ and $j \in \{1, ..., dim(A_i)\}$ (definition 2.18 (3.f)), it follows that $f(\chi_{0,k}) = \boldsymbol{\alpha}_i^{[j]}$. Thus, the function is well-defined.

Example 2.20.

Equation 2.3 gives a set of LCFRS production for the language L over $\{a, b, c\}$ with $L = \{wcccw \mid w \in \{a, b\}^+\}$, numbered for easy reference.

number	LHS		RHS	
r_1	S(UVWX)	\rightarrow	A(U, X)B(V, W)	
r_2	A(aU, aX)	\rightarrow	A(U, X)	
r_3	A(bU, bX)	\rightarrow	A(U, X)	
r_4	A(a,a)	\rightarrow	ε	
r_5	A(b,b)	\rightarrow	ε	
r_6	B(c,cc)	\rightarrow	ε	
r_7	B(cc,c)	\rightarrow	ε	

The production $A(b, b) \to \varepsilon$ specifies that the tuple $\langle b, b \rangle$ is in the yield of A and $A(aU, aX) \to A(U, X)$ means that it is possible to generate a new tuple in the yield of A from an already existing one by prepending a to both components.

It holds that $\langle abcccab \rangle \in yield(S)$ since:

$$\begin{array}{ll} \langle b,b\rangle & \in yield(A), & \text{using } r_5 \\ \langle ab,ab\rangle & \in yield(A), & \text{using } r_2, & \text{identifying } \langle U,X\rangle \text{ with } \langle b,b\rangle \in yield(A), \\ \langle c,cc\rangle & \in yield(B), & \text{using } r_6 \\ \langle abcccab\rangle & \in yield(S), & \text{using } r_1, & \text{identifying } \langle V,W\rangle \text{ with } \langle c,cc\rangle \in yield(B) \text{ and} \\ & \langle U,X\rangle \text{ with } \langle ab,ab\rangle \in yield(A). \end{array}$$

Similar to context-free grammars it is possible to define derivation trees. Derivation trees model the construction of a word beginning with the start symbol. Unlike phrase-structure trees of context free-grammars, LCFRS derivation trees are unordered. For two nodes it is not possible to say that one precedes the other since their children might be intertwined, leading to crossing branches. Nevertheless, LCFRS derivation trees are usually drawn so that the positions of the leaves correspond to their positions in the word. A *range* is assigned to each leaf that uniquely identifies its position in the word. For two internal nodes v_1, v_2 , usually v_1 is drawn left of v_2 if it dominates some leaf that precedes all leaves dominated by v_2 in the word.

An internal node with a nonterminal as label together with its children represents a rule application. The parent corresponds to the LHS nonterminal while the children consist of other internal nodes labeled with the RHS nonterminals and/or leaves that correspond to terminals on the LHS.

Evang (2011) gives a formal definition for derivation trees that I outline in the following. It requires defining the concepts of *range*, *range concatenation* and *rule instance*, which are taken from range concatenation grammars (Boullier, 1998b). I modified the range notation slightly to account for its use in other contexts in this work.

Definition 2.21 (Indices).

Let X be an alphabet. The set of all indices for a word w over X is denoted as $Ind(w) := \{1, ..., |w|\}$.

Definition 2.22 (Positions).

Let X be an alphabet. The set of all positions for a word w over X is denoted as $Pos(w) := Ind(w) \cup \{0\}.$

Definition 2.23 (Tuple projection).

1. Let X be an alphabet. For some tuple of positive integers $\langle i_1, ..., i_n \rangle \in \mathbb{N}^{<\mathbb{N}_0}$ the projection function $\pi_{\langle i_1,...,i_n \rangle}$ is defined on all words $w \in X^*$ where $i_1, ..., i_n \in Ind(w)$ as

$$\pi_{\langle i_1,...,i_n \rangle}(w) = w_{i_1},...,w_{i_n}.$$

 $\pi_{\langle i_1,...,i_n \rangle}(w)$ is usually written as $w_{\langle i_1,...,i_n \rangle}$. For the 0-tuple $\langle \rangle$ it naturally holds that $w_{\langle \rangle} = \varepsilon$.

2. Let X be an alphabet. For some tuple of tuples of positive integers

$$\langle\langle i_{1,1}, \dots, i_{1,n_1}\rangle, \dots, \langle i_{m,1}, \dots, i_{m,n_m}\rangle\rangle \in (\mathbb{N}^{<\mathbb{N}_0})^{<\mathbb{N}_0}$$

the second-degree tuple projection function $\pi^2_{\langle \langle i_{1,1},...,i_{1,n_1}\rangle,...,\langle i_{m,1},...,i_{m,n_m}\rangle\rangle}$ is defined on all words $w \in X^*$ where $i_{1,1},...,i_{1,n_1},...,i_{m,1},...,i_{m,n_m} \in Ind(w)$ as

$$\pi^{2}_{\langle \langle i_{1,1},...,i_{1,n_{1}}\rangle,...,\langle i_{m,1},...,i_{m,n_{m}}\rangle\rangle}(w) = \langle \pi_{\langle 1,1,...,i_{1,n_{1}}\rangle}(w),...,\pi_{\langle i_{m,1},...,i_{m,n_{m}}\rangle}(w)\rangle.$$

 $\pi^2_{\langle\langle i_{1,1},\ldots,i_{1,n_1}\rangle,\ldots,\langle i_{m,1},\ldots,i_{m,n_m}\rangle\rangle}(w)$ is usually written as $w_{\langle\langle i_{1,1},\ldots,i_{1,n_1}\rangle,\ldots,\langle i_{m,1},\ldots,i_{m,n_m}\rangle\rangle}$. For the 0-tuple $\langle \rangle$ it naturally holds that $\pi^2_{\langle \rangle}(w) = \langle \rangle$. It is not abbreviated to $w_{\langle \rangle}$ to avoid confusion with the first-order 0-tuple projection.

Definition 2.24 (Range).

- 1. A tuple $\langle a, b \rangle \in \mathbb{N}_0 \times \mathbb{N}_0$ with $a \leq b$ is called a *range*. A range $\langle a, b \rangle$ with a = b is called an *empty range*.
- 2. The set of all ranges is denoted by Ranges.
- 3. The function $\langle : \rangle : Ranges \to \mathbb{N}^{<\mathbb{N}}$ maps a range $\langle a, b \rangle$ to a tuple where all natural numbers from a + 1 to b are arranged in order, i.e.:

$$\langle a, b \rangle \mapsto \langle a+1, a+2, ..., b-1, b \rangle$$

 $\langle : \rangle (\langle a, b \rangle)$ can be written as $\langle a : b \rangle$.

- 4. $\langle : \rangle$ is extended to tuples of ranges, e.g. $\langle : \rangle (\langle \langle a_1, b_1 \rangle, \langle a_2, b_2 \rangle \rangle) = \langle \langle a_1 : b_1 \rangle, \langle a_2 : b_2 \rangle \rangle$.
- 5. The length of a range $\langle a, b \rangle$ is defined as $|\langle a, b \rangle| := b a$.
- 6. Two ranges $\langle a_1, b_1 \rangle$ and $\langle a_2, b_2 \rangle$ are called *non-overlapping* iff $b_1 \leq a_2 \lor b_2 \leq a_1$.
- 7. The precedence relation $<_{range} \subset Ranges \times Ranges$ is defined as follows:

$$\langle a_1, b_1 \rangle <_{\text{range}} \langle a_2, b_2 \rangle :\Leftrightarrow b_1 \le a_2$$

Definition 2.25 (Ranges in a word).

Let X be an alphabet and w a word over X.

- 1. A range $\langle a, b \rangle$ is called a range in w iff $a, b \in Pos(w)$.
- 2. The set of all ranges in w is denoted by Ranges(w).
- 3. For some range $\langle a, b \rangle$ in w, $w_{\langle a:b \rangle}$ is called a *substring of* w and conventionally written as $w_{a:b}$.

Definition 2.26 (Range concatenation).

- 1. The concatenation operation on two ranges $\langle a_1, b_1 \rangle$ and $\langle a_2, b_2 \rangle$ in a word w over an alphabet X where $b_1 = a_2$ is naturally defined as $\langle a_1, b_1 \rangle \circ \langle a_2, b_2 \rangle = \langle a_1, b_2 \rangle$. It is undefined if $b_1 \neq a_2$.
- 2. A string of ranges $\rho_1 \dots \rho_n \in Ranges(w)^*$ where $\rho_i \circ \rho_{i+1}$ is defined for all $i \in \{1, \dots, n-1\}$, is called *contiguous*. For such a string, range concatenation is defined as follows:

$$\bigcirc (\rho_1 \dots \rho_n) := \begin{cases} \varepsilon & \text{if } n = 0, \\ \rho_1 & \text{if } n = 1, \\ \bigcirc ((\rho_1 \circ \rho_2) \rho_3 \dots \rho_n) & \text{otherwise.} \end{cases}$$

It follows directly from this definition that for two ranges ρ_1, ρ_2 in a word w over some alphabet X where the concatenation of ρ_1 and ρ_2 is defined, the following holds: $w_{\langle : \rangle(\rho_1)}w_{\langle : \rangle(\rho_2)} = w_{\langle : \rangle(\rho_1 \circ \rho_2)}$.

Definition 2.27 (Rule instance).

Let $G = \langle N, T, V, P, S \rangle$ be an LCFRS, $r = A_0(\chi_{0,1}, ..., \chi_{0,dim(A)}) \rightarrow A_1(\chi_{1,1}, ..., \chi_{1,dim(A_1)})... A_m(\chi_{m,1}, ..., \chi_{m,dim(A_m)}) \in P$ a rule and $w \in T^*$ a word.

- 1. r can be transformed into a *rule instance of* r *with respect to* w by replacing the characters of the argument strings $\chi_{0,1}, ..., \chi_{0,dim(A_0)}, \chi_{1,1}, ..., \chi_{1,dim(A_1)}, ..., \chi_{m,1}, ..., \chi_{m,dim(A_m)}$ of p in the following fashion:
 - (a) All terminals t get replaced with some range $\langle a-1, a \rangle$ in w such that $w_{a-1:a} = t$.
 - (b) All variables get replaced with some range $\langle a, b \rangle$ in w.
 - (c) All empty arguments, i.e. if $\chi_{i,j} = \varepsilon$ for some i, j with $i \in \{0, ..., m\}$ and $j \in \{1, ..., dim(A_i)\}$, get replaced with some range $\langle a, a \rangle$ in w.

The resulting arguments are strings of ranges. Each of these strings $\rho_{i,j}$ must be contiguous and gets replaced by $\bigcirc (\rho_{i,j})$.

2. The resulting rule instance has the form $A_0(\boldsymbol{\rho}_0) \to A_1(\boldsymbol{\rho}_1), ..., A_m(\boldsymbol{\rho}_m)$ where $\boldsymbol{\rho}_k \in Ranges(w)^{dim(A_k)}$ is a tuple of ranges for all $k \in \{1, ..., m\}$.

3. The set of all rule instances of r wrt. w is called Instances(r, w).

The process described in definition 2.27 converts each of the arguments to a range in the given word where all the terminal symbol match an occurrence in the word. The range for variables can be freely chosen since there are no indications as to the substring they span. Each of the resulting sequences of ranges needs to be contiguous, i.e. the chosen ranges need to be adjacent, since a tuple component always has to match a continuous substring in the word.

Example 2.28.

Given the LCFRS from example 2.20 and the word w = abcccab, the following are some examples for rule instances for rule $r_2 = A(aU, aX) \rightarrow A(U, X)$ along with the two substrings of w their LHS spans:

$A(\langle 0:2\rangle, \langle 5:7\rangle) \to A(\langle 1:2\rangle, \langle 6:7\rangle),$	<u>ab</u> cccab	
$A(\langle 0:5\rangle, \langle 5:7\rangle) \to A(\langle 1:5\rangle, \langle 6:7\rangle),$	<u>abccc</u> ab	
$A(\langle 0:2\rangle, \langle 5:6\rangle) \to A(\langle 1:2\rangle, \langle 6:6\rangle),$	<u>ab</u> cccab	(2.5)
$A(\langle 5:7\rangle,\langle 0:2\rangle)\to A(\langle 6:7\rangle,\langle 1:2\rangle),$	<u>abcccab</u>	
$A(\langle 5:7\rangle, \langle 0:7\rangle) \to A(\langle 6:7\rangle, \langle 1:7\rangle) \in Instances(w, r_2)$	abccc <u>ab</u>	

Note that a rule instance can map a terminal to any matching occurrence in the word regardless of order. Two ranges in a range tuple can even overlap. However, as established, the start symbol must span one continuous range. Thus, all spans in a derivation must be reducible through concatenation to a single range. This is checked by determining if all of the RHS elements of a rule instance, i.e. assumptions about the span of its variables, can be found as LHS of some instantiated rule. Such a derivation naturally forms a tree.

Definition 2.29 (LCFRS derivation tree).

Let $G = \langle N, T, V_G, P, S \rangle$ be an LCFRS and $w = w_1 \dots w_n \in T^*$ a word.

- 1. Let $D = \langle V, \triangleleft, \hat{v}, \lambda \rangle$ be a labelled tree such that there are pairwise different leaves $v_1, ..., v_n$ in D with $\lambda(v_i) = \langle i - 1, i \rangle$ for $i \in \{1, ..., n\}$ and all other leaves have a label $\langle i, i \rangle$ for some $i \in \{0, ..., n\}$. The function *n*-yield : $V \to \mathcal{P}(Ranges(w)^{\leq dim(G)})$ assigns a set of tuples of ranges to each node in the tree and is defined as follows:
 - (a) For every leaf $u \in V$: n-yield $(u) = \{ \langle \lambda(u) \rangle \}$.
 - (b) For every internal node $v_0 \in V$, for every order $v_1, ..., v_m$ of all pairwise different internal nodes $v_1, ..., v_m \in V$, where $v_0 \triangleleft v_i$ and $\lambda(v_i) = A_i \in N$ for all $i \in \{1, ..., m\}$, for every rule $r = A_0(\boldsymbol{\alpha}_0) \rightarrow A_1(\boldsymbol{\alpha}_1), ..., A_m(\boldsymbol{\alpha}_m) \in P$, for every instance $A_0(\boldsymbol{\rho}_0) \rightarrow A_1(\boldsymbol{\rho}_1), ..., A_m(\boldsymbol{\rho}_m) \in Instances(r, w)$, it holds that $\boldsymbol{\rho}_0 \in n$ -yield (v_0) if:
 - i. $\boldsymbol{\rho}_i \in n\text{-yield}(v_i)$ for all $i \in \{1, ..., m\}$,
 - ii. for every terminal t occurring in $\boldsymbol{\alpha}_0$, there exists a leaf $u \in V$ such that $v_0 \triangleleft u$ and $\lambda(u)$ is the range with which t was replaced to obtain $\boldsymbol{\rho}_0$.
 - iii. for all $i \in \{1, ..., dim(A_0)\}$ such that $\boldsymbol{\alpha}_0^{[i]}$ is empty, there is a $u \in V$ such that $v_0 \triangleleft u$, u is a leaf and $\lambda(u) = \boldsymbol{\rho}_0^{[i]}$.

Nothing else is in n-yield (v_0) .

2. D is a derivation tree of w in G iff $\lambda(r) = S$ and $\langle \langle 0 : n \rangle \rangle \in n$ -yield(r). We write \mathcal{D}_G^w for the set of all derivation trees of w in G.

3. The tree language of G is defined as $\mathscr{L}_{G}^{\mathcal{D}} := \{D \mid \exists w \in T^* : D \in \mathcal{D}_{G}^w\}.$

It will prove useful to define the class of syntactic representations arising from LCFRS derivation trees independently of the formalism. It can be seen that every LCFRS derivation tree is a *range-labelled tree*, and that for any *range-labelled tree*, an LCFRS and a word exist for which it is a derivation tree.

Definition 2.30 (Partial range-labelled tree).

Let $n \in \mathbb{N}_0$ be an integer and N an alphabet (of nonterminal symbols).

- 1. A partial range-labelled tree over N in length n is an unordered finite labelled tree $\langle V, \triangleleft, \hat{v}, \lambda \rangle$ such that
 - (a) $\lambda(v) \in N$ for each internal node $v \in V$,
 - (b) $\lambda(u) = \langle i 1, i \rangle$ with $i \in \{1, ..., n\}$ or $\lambda(u) = \langle i, i \rangle$ with $i \in \{0, ..., n\}$ for every leaf node $u \in V$,
 - (c) for all leaf nodes $u_1, u_2 \in V$ if $|\lambda(u_1)| = |\lambda(u_2)| = 1$ and $u_1 \neq u_2$ it must hold that $\lambda(u_1) \neq \lambda(u_2)$.
- 2. A partial ε -free range-labelled tree over N in length n is a partial range-labelled tree over N in length n such that either
 - (a) n > 0 and no leaf is labelled with an empty range or
 - (b) n = 0 and there is only one leaf.

Definition 2.31 (Range-labelled tree).

Let $n \in \mathbb{N}_0$ be an integer and N an alphabet (of nonterminal symbols).

- 1. A range-labelled tree over N spanning length n is a partial range-labelled tree $\langle V, \triangleleft, \hat{v}, \lambda \rangle$ such that for every $i \in \{1, ..., n\}$ there exists a leaf $u \in V$ such that $\lambda(u) = \langle i-1, i \rangle$. This definition naturally extends to ε -free partial range-labelled trees.
- 2. Given an alphabet T and a word $w \in T^*$ such that |w| = n, a range-labelled tree D over N spanning length n is called a range-labelled tree for w.

In essence, a range labelled tree is an unordered tree with an ordering defined on its leaves. In the following, the term *discontinuous tree* will also be used to refer to this class of representations.

Example 2.32.

Given the LCFRS from Example 2.20 and the word *abcccab*, a possible derivation tree is shown in figure 2.3.



Figure 2.3: Derivation tree for the word *abcccab* in the grammar from example 2.20, including *n-yields*.

This tree illustrates that it would not suffice to map n-yield(v) for $v \in V$ to a single tuple of ranges since adjacent terminals like in the case of B can allow for different combinations of ranges in the components.

2.4. From Treebanks to LCFRSs

As shown in the last section, LCFRS derivation trees result in the desired tree representations of non-overlapping discontinuous constituents found in constituency treebanks with discontinuities. These trees can be directly interpreted as LCFRS derivation trees. Søgaard and Maier (2008) describe an algorithm for extracting probabilistic LCFRSs (PLCFRS) from discontinuous treebanks. Kallmeyer and Maier (2010) give the following definition for PLCFRSs.

Definition 2.33 (Probablistic LCFRS).

A probabilistic LCFRS (PLCFRS) is a 6-tuple $\langle N, T, V, P, S, q \rangle$ such that $\langle N, T, V, P, S \rangle$ is an LCFRS and $q: P \to [0,1]^8$ is a function such that for all $A \in N$:

$$\sum_{A(\boldsymbol{\alpha}) \to \Phi \in P} q(A(\boldsymbol{\alpha}) \to \Phi) = 1$$

The function q assigns a conditional probability to each rule which is based on the frequency with which instances of this rule have been found in the treebank. For more information on the extraction algorithm, please refer to Søgaard and Maier (2008).

Related research includes the work of Kuhlmann and Satta (2009) who have shown that it is possible to extract LCFRSs from dependency treebanks. Evang and Kallmeyer (2011) describe a technique to convert context-free phrase structure trees in the English language Penn Treebank into a discontinuous format using inherent additional information about non-local dependencies. This research established the discontinuous Penn Treebank (DPTB).

⁸For $i, j \in \mathbb{R}$ the expression [i, j] shall equal $\{n \in \mathbb{R} \mid i \leq n \leq j\}$.

3. Parsing LCFRS

Parsers for LCFRSs can be utilised for producing discontinuous constituent trees from text. In the following, I will outline two main parsing approaches: traditional chart parsing and transition-based parsing. Both approaches are usually *data-driven* in that they extract their syntactic knowledge from treebank descriptions. In the case of chart-parsers, this extraction is explicit. To build the parser, LCFRS rules are automatically extracted from an annotated corpus. Transition-based parsers allow the implementation of grammar-less parsing. They are usually built on a neural-network model that is directly trained on a treebank.

3.1. CYK Parser

Traditional *Cocke-Younger-Kasami* (CYK) parsers were first introduced for CFG by Sakai (1961). Their underlying idea is to assert the applicability of rules in a bottom-up fashion, starting with the terminal symbols and progressively validating derivation steps on the basis of the set of rule applications that has been recovered already. Seki et al. (1991) extended the algorithm to MCFGs. It was adapted for probabilistic LCFRS by Kallmeyer and Maier (2010). I will discuss the latter approach in more detail.

3.1.1 Prerequisites

The input of the CYK PLCFRS parser consists of a grammar and an input word. The parser requires a PLCFRS $\langle N, T, V, P, S, p \rangle$ to have the following properties:⁹

- 1. binary: For every rule $r \in P$: rank $(r) \leq 2$ (Gómez-Rodríguez et al., 2009).
- 2. terminal-restricted: Terminals occur only in terminating rules of the form $A(a) \rightarrow \varepsilon$ where a is a terminal. The LHS of all other rules contains only variables (Evang, 2011).
- 3. gap-explicit: Two different variables in a RHS element of a rule cannot appear next to each other in the same component on the LHS side. This means that a separation of two variables always entails that there is a gap between them (Kallmeyer and Maier, 2010).
- 4. *ordered*: The order of the variables within each RHS element is the same as their order of occurrence in the LHS (Villemonte de la Clergerie, 2002).
- 5. ε -free: There exists no rule with some empty LHS argument or there exists exactly one ε -rule: $S(\varepsilon) \to \varepsilon$ and S does not occur in the RHS of any rule (Boullier, 1998a).

The algorithm in Søgaard and Maier (2008) produces PLCFRSs that fulfil properties 2, 3, 4 and 5. Note that for every LCFRS there is an equivalent LCFRS (i.e. generating the same language) that is ε -free (Boullier, 1998a), ordered (Kallmeyer, 2010, chapter 7), binary (Gómez-Rodríguez et al., 2009) and terminal-restricted (trivial).

Furthermore, for every LCFRS there exists an equivalent LCFRS that is *gap-explicit*. An algorithm for converting an ordered LCFRS into this form is given below. Thus, the properties above can all be assumed without loss of generality in the following sections.

The algorithm uses the fact that the LCFRS is ordered: variables $X_1, ..., X_n$ that appear next to each other in the same component in the LHS argument tuple and as components in the same argument tuple on the RHS cannot have another component in between on the RHS since this would conflict with the order. Therefore only the case $\langle ..., X_1, ..., X_n, ... \rangle$ for the RHS is relevant.

The algorithm traverses all rules and all elements $A_k(...)$ on the rule's RHS. For each element, the occurrence of useless variables is checked. For overlapping sequences of variables adjacent

⁹For lack of established nomenclature, I introduce the terms *terminal-restricted* and *gap-explicit*.

Algorithm 3.1 Removing	useless	variable	separation
------------------------	---------	----------	------------

Input: an ordered LCFRS $\langle N, T, V, P, S \rangle$ **Output:** an equivalent LCFRS that is *gap-explicit* 1: for all rules $r = A_0(\boldsymbol{\alpha}_0) \to A_1(\boldsymbol{\alpha}_1)...A_m(\boldsymbol{\alpha}_m) \in P$ do for all $k \in \{1, ..., m\}$ do 2: $u \leftarrow \{\langle l, r \rangle \mid r-l \ge 1, \boldsymbol{\alpha}_k^{[l]} \boldsymbol{\alpha}_k^{[l+1]} \dots \boldsymbol{\alpha}_k^{[r]} \text{ is a max. adjacent substr. of some LHS argument} \}$ 3: if not $A_k^u \in N$ then 4: introduce new nonterminal A_k^u to N with $dim(A_k^u) = dim(A_k) - \sum_{(l,r) \in u} r - l$ 5:for all rules of form $A_k(\boldsymbol{\gamma}) \to \Phi \in P$ do 6: $oldsymbol{\gamma}' \leftarrow oldsymbol{\gamma}$ 7: for all $\langle l, r \rangle \in u$ do 8: merge $\boldsymbol{\gamma}^{\prime[l]}, ..., \boldsymbol{\gamma}^{\prime[r]}$ into a single component $\boldsymbol{\gamma}^{\prime[l]}...\boldsymbol{\gamma}^{\prime[r]}$ 9: end for 10:add new rule $A_k^u(\boldsymbol{\gamma}')$ to P 11: end for 12:end if 13:for all $\langle l, r \rangle \in u$ do 14: pick a new variable X15:replace components $\boldsymbol{\alpha}_{k}^{[l]},...,\boldsymbol{\alpha}_{k}^{[r]}$ with a single component X 16: replace corresponding substring on the LHS with X17:18:end for replace the nonterminal symbol A_k on the RHS with A_k^u 19: end for 20:21: end for 22:Remove useless productions

on the LHS, the sequence with the largest number of elements is considered (maximal adjacent substring). A set u is introduced to mark the start and end index in the RHS tuple for each sequence of useless components.

Now, we can replace all of these occurrences with new single variables both on the LHS and on the RHS and postpone the variable split to the productions governed by A_k .

To achieve this, a new nonterminal A_k^u marked with the information which components were merged when deriving it from A_k , is introduced. All of the rules in P where A_k stands on the LHS are duplicated for A_k^u with all of the components described by u concatenated into single components since they were already asserted to be adjacent in the rule deriving A_k . In order to avoid generating new nonterminals for the same A_k with the same merged components multiple times, no new nonterminal is introduced if A_k^u has been created already.

Through the introduction of A_k^u , the original nonterminal A_k might have become unreachable if it cannot be found in the RHS of some other rule. Useless productions can be eliminated in a final step with the algorithm given in Kallmeyer (2010, chapter 7).

Finally, the parser demands that the input already be assigned with part-of-speech (POS) tags, given as unary productions for each token. Therefore, when respecting all of the assumptions given above, every rule has to be of one of the following forms (Kallmeyer and Maier, 2013):

$$\begin{array}{ll}
A(a) & \to \varepsilon, & \text{with } A \text{ being a POS tag, } a \in T \\
A_0(\boldsymbol{\alpha}) & \to A_1(\boldsymbol{\alpha}), & \text{with } \boldsymbol{\alpha} \in V^{dim(A_0)}, \ dim(A_0) = dim(A_1) \\
A_0(\boldsymbol{\alpha}_0) & \to A_1(\boldsymbol{\alpha}_1)A_2(\boldsymbol{\alpha}_2), & \text{with } \boldsymbol{\alpha}_0 \in (V^+)^{dim(A_0)}, \boldsymbol{\alpha}_1 \in V^{dim(A_1)}, \boldsymbol{\alpha}_2 \in V^{dim(A_2)}
\end{array}$$
(3.1)

3.1.2 Deduction Rules

Parsing is formulated as a set of deduction rules which facilitates proofs of correctness and the examination of complexity. Deduction rules describe how to infer new elements from existing ones.

Figure 3.1 shows the general form of a deduction rule. It dictates that the *consequent item* is derivable from the *antecedent items* if the *side condition* is fulfilled. The side condition can be empty. Then no condition needs to be fulfilled.

 $RULE \quad \frac{antecedent_1, ..., antecedent_n}{consequent} \ side \ condition$

Figure 3.1: Deduction principle.

An axiom can be expressed as a deduction rule with no antecedent items. A parsing schema is comprised of deduction rules with one or more axiom(s) and one or several goal item(s). A successful parse is characterised by the deducibility of a goal item for a given input (Kallmeyer, 2010, chapter 3).

Following Sikkel and Nijholt (1997), I define a parsing system and the inference relation:

Definition 3.1 (Parsing system).

A parsing system \mathfrak{P} for some grammar G and a word w is a triple $\langle \mathcal{I}, H, R \rangle$ such that:

- 1. \mathcal{I} is a set of items,
- 2. H is a set of axioms/hypotheses,
- 3. $R \subseteq \mathcal{P}_{fin}(\mathcal{P}_{fin}(\mathcal{I} \cup H) \times \mathcal{I})$ is a set of deduction rules.^{10,11}

Definition 3.2 (Inference relation).

Let $\mathfrak{P} = \langle \mathcal{I}, H, R \rangle$ be a parsing system for some grammar G and a word w. The following relations are defined on $\mathcal{P}_{fin}(\mathcal{I} \cup H) \times \mathcal{I}$.

- 1. We write $Y \vdash_r c$ iff there is $Y' \subseteq Y$ such that c can be deduced from the elements in Y' using rule $r \in R$, i.e. such that $\langle Y', c \rangle \in r$.
- 2. We write $Y \vdash c$ iff $\exists r \in R : Y \vdash_r c$.
- 3. We call $Y, c_1, c_2, ..., c_n$ a deduction sequence iff $Y \cup \{c_1, ..., c_{i-1}\} \vdash c_i$ for all $i \in \{1, ..., n\}$ with $n \in \mathbb{N}$. We conventionally write $Y \vdash c_1 \vdash c_2 \vdash ... \vdash c_n$.
- 4. We write $Y \vdash^* c$ iff $c \in Y$ or $Y \vdash ... \vdash c$.
- 5. An item $c \in \mathcal{I}$ is called *valid* iff $H \vdash^* c$.

In the context of LCFRS CYK parsing, the items of the parsing system are *intermediate parsing* results:

Definition 3.3 (PLCFRS CYK intermediate parsing result).

Let $G = \langle N, T, V, P, S, q \rangle$ be a PLCFRS, $A \in N$ a nonterminal, $\boldsymbol{\rho} \in Ranges^{dim(A)}$ and $m \in \{n \in \mathbb{R} \mid n \geq 0\}$. $m : [A, \boldsymbol{\rho}]$ is called an intermediate parsing result or an item. m is called the *weight* of the item. The set of all items for G is called \mathcal{I}_{CYK} .

The tuple of ranges ρ characterises all components in the span of A. m represents a logarithmically encoded probability that is used to determine the relative probability of different parses.

A set of deduction rules constrains the set of valid intermediate parsing results for an input word w. The idea here is that a valid item should represent a parse for a part of w that is licensed by the grammar. This property is proven in section 3.1.5. The full set of deduction rules for PLCFRS CYK parsing is given in figure 3.2.

 $^{{}^{10}\}mathcal{P}_{fin}$ denotes the powerset containing only finite sets.

¹¹Note that Sikkel and Nijholt (1997) do not differentiate between different deduction rules but subsume all deductions in a single set in the definition of \mathfrak{P} .

$$\begin{array}{l} \text{SCAN} \quad \overline{0: [A, \ \langle \langle i, i+1 \rangle \rangle]} \ A \text{ is a POS tag of } w_{i+1} \\ \\ \text{UNARY} \quad \frac{in: [B, \pmb{\rho}]}{in + |\log(q(r))|: [A, \pmb{\rho}]} \ r = A(\pmb{\alpha}) \to B(\pmb{\alpha}) \in P \\ \\ \text{BINARY} \quad \frac{in_B: [B, \pmb{\rho}_B], \ in_C: [C, \pmb{\rho}_C]}{in_B + in_C + |\log(q(r))|: [A, \pmb{\rho}_A]} \ \exists r \in P: (A(\pmb{\rho}_A) \to B(\pmb{\rho}_B)C(\pmb{\rho}_C)) \in Instances(r, w) \\ \\ goal \quad m: [S, \langle \langle 0: |w| \rangle \rangle] \end{array}$$

Figure 3.2: Weighted CYK deduction system for parsing $w \in T^*$ with a PLCFRS $\langle N, T, V, P, S, q \rangle$. $R_{\text{CYK}} = \{\text{UNARY}, \text{BINARY}\}, H_{\text{CYK}} = \{0 : [A, \langle \langle i, i+1 \rangle \rangle] \mid A \text{ is a POS tag of some } w_{i+1}\}.$

The deduction system works bottom-up. First SCAN provides the axioms. These are the terminating rules that have RHS ε . Then the derivation tree is traversed bottom-up. UNARY and BINARY find ranges for rules of rank 1 or 2 respectively. Any rule instance where the pairs of nonterminals and range tuples in the RHS have been found can be applied. Therefore, its LHS is deduced. The contiguity constraint of the rule instances ensures that only elements can be chosen that can be joined in a way that is compatible with the rule to deduce, i.e. adjacent elements on the LHS are only deduced from adjacent and non-overlapping rule instances already found.

Definition 3.4 (PLCFRS-CYK parsing system).

The parsing schema **PLCFRS-CYK** is defined by the parsing system $\mathfrak{P}_{CYK} = \langle \mathcal{I}_{CYK}, H_{CYK}, R_{CYK} \rangle$ for any PLCFRS $G = \langle N, T, V, P, S, p \rangle$ and any word $w \in T^*$.

3.1.3 Chart Parsing

Natural language grammars are highly ambiguous. For a given input word w, a parsing schema may give rise to a variety of items spanning different, possibly overlapping, portions of w. There can be several unique sequences of deductions leading to a goal item for w. On the other hand, different analyses can be based on a common sub-analysis for a part of w (Kallmeyer, 2010, chapter 3).

Chart parsing is a framework rooted in dynamic programming that allows to easily store intermediate parsing results and to reuse them at any time for the deduction of new elements (Grune and Jacobs, 2010, chapter 7). For this purpose, a table - called a *chart* C - is used. For CYK parsing, where an intermediate parsing result [A, i, j] spans a projective substring of w, the table can have three dimensions with the first dimension corresponding to a unique index for each nonterminal. The second and third dimension have length |w| + 1 and correspond to the start and end index of the item's range. [A, i, j] would be represented by a mark in cell (index(A), i, j) of C(Kallmeyer, 2010, chapter 3). The cell (index(S), 0, |w|) corresponds to the goal item.

Furthermore, a list called an *agenda* \mathcal{A} is used. It consists of the items that are yet to be checked for possible deductions in combination with the asserted items in \mathcal{C} . Axioms and new items found by deduction are added to \mathcal{A} (Kallmeyer, 2010, chapter 3).

A recogniser can be converted into a parser by saving references with each entry in C that refer to the items that were used to deduce it. In this way, one can retrieve the parse tree from the goal item. The references are called *backpointers* (Kallmeyer, 2010, chapter 3).

In our context, intermediate parsing results can span more than one consecutive substring and therefore a chart with three dimensions does not suffice. The number of dimensions needed is determined by the fan-out of the grammar G. The maximum number of range borders is given by $2 \cdot dim(G)$. When representing nonterminals in a separate dimension, the chart dimensionality is $1 + 2 \cdot dim(G)$. The goal item then is (index(S), 0, |w|, n, n, ..., n, n) where n = |w| + 1 is a special index used for those nonterminals A where dim(A) < dim(G). Since this does not allow for an easy visualisation, I will represent C as a list containing intermediate parsing results.

3.1.4 Algorithm

A deductive engine can be used to deduce every valid item from the set of axioms, eventually finding the goal item with the highest weight. This naive approach would be computationally inefficient due to the large number of useless intermediate parsing results generated. Kallmeyer and Maier (2010) give a strategy for fast exploration of the best goal item based on weighted deductive parsing (Nederhof, 2003). It is reproduced in an adapted form in algorithm 3.2.

\mathbf{A}	lgorithm	3.2	Weighted	deductive	CYK	parsing	for	PLCFRS.	
--------------	----------	-----	----------	-----------	-----	---------	-----	---------	--

	Input: a parsing system $\langle I_{CYK}, H_{CYK}, R_{CYK} \rangle$ for a grammar G and a word w
	Dutput: a filled chart containing the best goal item
1:	$4 \leftarrow H_{ m CYK}$
2:	$\mathbf{vhile}\;\mathcal{A}\neq \emptyset\;\mathbf{do}$
3:	$x: I \leftarrow$ best item in \mathcal{A} \triangleright select the item with the lowest weight
4:	$\mathcal{A}.\mathrm{remove}(x:I)$
5:	$\mathcal{C}.\mathrm{add}(x:I)$
6:	$\mathbf{if} I = [S, \langle \langle 0: w \rangle \rangle] \mathbf{then}$
7:	stop
8:	else
9:	for all $y : I'$ in (all items deducible from $x : I$ and from items in \mathcal{C}) do
10:	$\mathbf{if} \nexists z \text{ with } z: I' \in \mathcal{C} \cup \mathcal{A} \mathbf{ then }$
11:	$\mathcal{A}.\mathrm{add}(y:I')$
12:	else if $\exists z \text{ with } z : I' \in \mathcal{A} \text{ then}$
13:	$\mathcal{A}. ext{remove}(z:I')$
14:	$\mathcal{A}.\mathrm{add}(\min(y,z):I')$
15:	end if
16:	end for
17:	end if
18:	end while

In each step, the best item in the agenda x : I is written to the chart and checked for deducibility of new items by combining with the items present on the chart. This amounts to retrieving all y : I' such that $\mathcal{C} \cup \{x : I\} \vdash y : I'$ since no new elements can be generated purely from items on \mathcal{C} according to the logic of the algorithm. All newly generated items y : I' that are not yet present in the agenda or on the chart are put on the agenda. If an item already exists on the agenda but the newly generated one has a lower (i.e. better) weight, the weight gets updated with the lower one.

By always retrieving the agenda item with the best score, the algorithm guarantees that the first goal item derived is the best one (Knuth, 1977).

Example 3.5.

Let G be a PLCFRS characterised by the productions and probabilities in equation 3.2 with $\mathscr{L}_G = \{aaa\}.$

number	LHS		RHS	probability	
r_1	S(XYZ)	\rightarrow	A(X,Y)B(Z)	0.1	
r_2	S(XYZ)	\rightarrow	A(X,Z)B(Y)	0.9	(3.2)
r_3	A(X,Y)	\rightarrow	B(X)B(Y)	1	
r_4	B(a)	\rightarrow	ε	1	

For an ordered grammar, the algorithm can be optimised to avoid deriving elements with ranges in their arguments that do not respect precedence. Furthermore, for trivial reasons, the algorithm does not need to compute tuples of ranges that contain overlaps. Figure 3.3 shows the sequence of parsing steps for the word *aaa* and PLCFRS G.

\mathcal{A}	С	new items
$B_0^{0:1}, B_0^{1:2}, B_0^{2:3}$	Ø	
$B_0^{1:2}, B_0^{2:3}$	$\underline{B_{0}^{0:1}}$	
$B_0^{2:3}$	$\underline{B_0^{1:2}}, B_0^{0:1}$	$A_0^{0:1,1:2}$
$A_0^{0:1,1:2}$	$\underline{B_0^{2:3}}, \underline{B_0^{1:2}}, \underline{B_0^{0:1}}$	$A_0^{1:2,2:3}, A_0^{0:1,2:3}$
$A_0^{1:2,2:3}, A_0^{0:1,2:3}$	$\underline{A_0^{0:1,1:2}}, B_0^{2:3}, B_0^{1:2}, B_0^{0:1}$	$S^{0:3}_{ log(0.1) }$
$A_0^{0:1,2:3}, S_{ log(0.1) }^{0:3}$	$\underline{A_0^{1:2,2:3}}, A_0^{0:1,1:2}, B_0^{2:3}, B_0^{1:2}, B_0^{0:1}$	
$S^{0:3}_{ log(0.1) }$	$\underline{A_0^{0:1,2:3}}, A_0^{1:2,2:3}, A_0^{0:1,1:2}, B_0^{2:3}, B_0^{1:2}, B_0^{0:1}$	$S^{0:3}_{ log(0.9) }$
$S^{0:3}_{ log(0.1) }$	$\underbrace{S^{0:3}_{ log(0.9) }}_{\uparrow}, A^{0:1,2:3}_{0}, A^{1:2,2:3}_{0}, A^{0:1,1:2}_{0}, B^{2:3}_{0}, B^{1:2}_{0}, B^{0:1}_{0}$	

Figure 3.3: CYK parse for word *aaa* and LCFRS G from equation 3.2 according to algorithm 3.2; representing intermediate parsing results $m : [A, \rho]$ as A_m^{ρ} ; the best result taken from the agenda is <u>underlined</u>, the chart item(s) it combines with are <u>underlined</u> with dots; visualising backpointers at the last step.

3.1.5 Soundness and Completeness

Up to this point, I assumed that the parser indeed does what it claims to do. But generally, parser *correctness* has to be proven by showing *soundness* and *completeness* for the parsing algorithm (Kallmeyer, 2010, chapter 3).

In the following, a recogniser algorithm A is generalised as a function on a grammar G and a word w returning *true* or *false*. A class of grammars could be, for instance, the set of all CFGs or in this case the set of all ordered, binary, gap-explicit, ε -free, terminal-restricted LCFRSs.

Definition 3.6 (Recogniser soundness, completeness, correctness).

Let A be a recogniser algorithm.

- 1. A is sound for a class of grammars \mathbb{G} , iff for every grammar $G \in \mathbb{G}$ and every input word w, it holds that $A(G, w) = true \Rightarrow w \in \mathscr{L}_G$.
- 2. A is complete for a class of grammars \mathbb{G} , iff for every grammar $G \in \mathbb{G}$ and every input word w, it holds that $w \in \mathscr{L}_G \Rightarrow A(G, w) = true$.
- 3. A is correct for a class of grammars \mathbb{G} , iff A is sound and complete for \mathbb{G} .

For a recogniser, *soundness* means that only the words of the language of the given grammar are recognised as correct while *completeness* is defined by the fact that all of the words of the language are recognised by the algorithm. If both properties hold, the system is called *correct*.

This notion can be extended to parsers. Here, A is understood as a function on a grammar G and a word $w \in \mathscr{L}_G$ returning a set of final syntactic representations (usually trees spanning the input and rooted at S).

Definition 3.7 (Parser soundness, completeness, correctness).

Let A be a parsing algorithm.

- 1. A is sound for a class of grammars \mathbb{G} , iff for every $G \in \mathbb{G}$, every input word $w, \forall D \in A(G, w) : D \in \mathcal{D}_G^w$.
- 2. A is complete for a class of grammars \mathbb{G} , iff for every grammar $G \in \mathbb{G}$ and every input word $w, \forall D \in \mathcal{D}_G^w : D \in A(G, w).$
- 3. A is *correct* for a class of grammars \mathbb{G} iff it is sound and complete for \mathbb{G} .

For the deductive chart-parsing approach, the proof of correctness is two-fold:

- 1. proving soundness and completeness for \mathbb{G} where A(G, w) tells the presence of a goal item in the set of valid items (recogniser) or gives the set of valid goal items including backpointers (parser) (Gómez-Rodríguez et al., 2008) and
- 2. showing that the deductive engine actually explores all possible deductions or the best one in case of weighted deductive parsing.

Usually, 1. is shown by proving a stronger result: soundness and completeness for the intermediate parsing results as is shown in the following.

Corollary 3.8.

- 1. Given a fixed LCFRS G with the properties from section 3.1.1 and an input sentence $w = w_1...w_n$: $[A, \rho]$ iff $w_{\langle : \rangle(\rho)} = \langle w_{\langle : \rangle \rho^{[1]}}, ..., w_{\langle : \rangle \rho^{[1\rho]}} \rangle \in yield(A)$.
- 2. The **CYK-LCFRS** parsing schema is correct for the class of all LCFRSs with the properties given in Section 3.1.1.

Proof. Part 1 is proven via induction over the sum of lengths for the range tuple $\boldsymbol{\rho}$ (i.e. the number of terminals the item spans) defined by $s = \sum_{i=1}^{|\boldsymbol{\rho}|} |\boldsymbol{\rho}^{[i]}|$. This is inspired by the proof for CFG CYK parsing given by Kallmeyer (2010, chapter 3).

- 1. Case s = 1:
 - (\Rightarrow) Given: $[A, \rho]$.
 - The item must have form $[A, \langle \langle i-1, i \rangle \rangle]$ since empty ranges do not occur.
 - i. If the item is an axiom, then by axiom definition, it holds that $w_{i-1:i} = w_i \in yield(A)$.
 - ii. If the item was derived using UNARY, the antecedent must have form $[B, \langle \langle i-1, i \rangle \rangle]$. If the induction claim holds for the antecedent, then the UNARY side-condition claims the existence of rule $A(X) \to B(X)$ for some $X \in V$. Thus, by definition of the yield, it follows that $\langle w_i \rangle \in yield(A)$. The induction claim must hold for the antecedent B (by induction over the deduction sequence) since it must have been derived either by another UNARY or be an axiom because all valid items c must satisfy $H_{\text{CYK}} \vdash^* c$ and s cannot decrease by application of deductions rules.
 - (\Leftarrow) Given: There is some range $\langle i-1,i\rangle$ such that $\langle w_i\rangle \in yield(A)$. The definition of yield allows two options:
 - i. Either there exists a rule $A(w_i) \to \varepsilon \in P$. Then by the definition of H_{CYK} it holds that $[A, \langle \langle i-1, i \rangle \rangle]$.
 - ii. Or there is some $A(X) \to B(X) \in P$ with $X \in V$ such that $w_i \in yield(B)$. In the same fashion as above, let us assume that $[B, \langle \langle i 1, i \rangle \rangle]$. Then it follows by the UNARY side-condition that $[A, \langle \langle i 1, i \rangle \rangle]$.

25

 (\Rightarrow) Given: $[A, \rho]$.

The item is not in H_{CYK} since it spans more than one symbol.

- i. If the item was derived via BINARY from two items $[B, \rho_B], [C, \rho_C]$, the induction claim must hold for both antecedents since their individual range length sum must be smaller (productions with empty ranges are not allowed). Thus, $w_{\langle : \rangle(\rho_B)} \in$ yield(B) and $w_{\langle : \rangle(\rho_C)} \in yield(C)$. Now, since by the side condition there exists an instance $A(\rho) \to B(\rho_B)C(\rho_C)$, it follows that there is a composition function such that $w_{\langle : \rangle(\rho)} \in yield(A)$.
- ii. If the item was derived via UNARY from an item $[B, \rho]$: assume that the claim holds for $[B, \rho]$. Since the side condition demanded the existence of a unary production $A(\alpha) \to B(\alpha)$, we can assert that also $w_{\langle : \rangle(\rho)} \in yield(A)$. Now, by induction over the deduction sequence, the claim must hold for $[B, \rho]$ since there must be valid items c_x, c_y, c_z such that $\{c_x, c_y\} \vdash_{\text{BINARY}} c_z$ and $\{c_z\} \vdash^* [B, \rho]$ because only BINARY can compose items with larger s than their antecedents.
- (\Leftarrow) Given: There is a range tuple ρ with $w_{\langle : \rangle(\rho)} \in yield(A)$. The definition of yield allows two options:
 - i. There is some binary production

 $A(\chi_{A,1},...,\chi_{A,dim(A)}) \to B(\chi_{B,1},...,\chi_{B,dim(B)})C(\chi_{C,1},...,\chi_{C,dim(C)})$ such that $w_{\langle : \rangle(\boldsymbol{\rho})}$ was composed from the elements of a yield $w_{\langle : \rangle(\boldsymbol{\rho}_B)} \in yield(B)$ and $w_{\langle : \rangle(\boldsymbol{\rho}_C)} \in yield(C)$ according to the production. From this composition follows that there is a rule instance $A(\boldsymbol{\rho}_A) \to B(\boldsymbol{\rho}_B)C(\boldsymbol{\rho}_C)$. Furthermore, since there are no empty nonterminal arguments allowed, both of these yields span a number of tokens smaller than s. Thus, by induction claim there must be $[B, \boldsymbol{\rho}_B]$ and $[C, \boldsymbol{\rho}_C]$. Therefore, BINARY gives $[A, \boldsymbol{\rho}]$.

ii. Or there is a unary production $A(\boldsymbol{\alpha}) \to B(\boldsymbol{\alpha})$ and also $w_{\langle : \rangle(\boldsymbol{\rho})} \in yield(B)$. Like above, we can assume that the induction claim holds for this antecedent. Therefore, $[B, \boldsymbol{\rho}]$ and by use of the side-condition also $[A, \boldsymbol{\rho}]$.

Part 2 follows from this result. The deduction schema is correct since for every LCFRS G and for every w,

- 1. if $[S, \langle \langle 0, |w| \rangle \rangle]$ it follows that $\langle w_{0:|w|} \rangle = \langle w \rangle \in yield(S)$ i.e. $w \in \mathscr{L}_G$ (sound) and
- 2. if $w \in \mathscr{L}_G$, i.e. $\langle w \rangle \in yield(S)$, then $[S, \langle \langle 0 : |w| \rangle]$ (complete).

Correctness of the parser built by saving backpointers for each item follows directly from this proof in combination with definition 2.29. The items represent nodes of the tree, an item's non-terminal is its node label and the result of the n-yield function is given by the range tuple of the item.

A proof of correctness for the weighted deductive parsing algorithm for finding the best scoring derivation (algorithm 3.2) is given by Knuth (1977) for systems where the weight is non-decreasing. This is the case for the deduction rules given in figure 3.2.

3.1.6 Complexity

The separation of parsing algorithm and schema allows for a concise examination of complexity. Given a grammar G and an input word w one has to consider the maximum number of possible deduction rule applications (Kallmeyer, 2010, chapter 3). This depends on the most complex rule, in this case: BINARY with $A(\rho_A) \rightarrow B(\rho_B)C(\rho_C)$. There is a maximum of dim(G) variables and therefore $2 \cdot dim(G)$ range boundaries for each of the antecedents. For two variables adjacent in a LHS argument, the number of independent range boundaries reduces by one since for ranges ρ_1, ρ_2 the concatenation $\rho_1 \circ \rho_2$ must be defined, i.e. the right boundary of ρ_1 is the left boundary of ρ_2 . In the worst case, dim(A) = dim(B) =dim(C) = dim(G) and each component of the LHS contains exactly two variables. More adjacent variables would lead to a greater boundary reduction. Each LHS component now gives rise to 3 independent boundaries, which means that the total maximum number of boundaries is $3 \cdot dim(G)$. For a word of length *n* this leads to $n^{3 \cdot dim(G)}$ possibilities to choose these boundaries and thus a time complexity of $\mathcal{O}(n^{3 \cdot dim(G)})$ for the algorithm (Evang, 2011).

Note that the time complexity of parsing LCFRSs is fundamentally dependent on the maximum number of discontinuous blocks for a nonterminal. Søgaard and Maier (2008) show that for a typical treebank LCFRS fan-out is around 3, which makes CYK parsing very computationally expensive.

3.1.7 Related Work

Several strategies were explored to speed up LCFRS CYK chart parsing. Kallmeyer and Maier (2013) use A^* search. A coarse-to-fine approach was presented by Van Cranenburgh (2012) while Angelov and Ljunglöf (2014) propose a new cost estimation for ranking parser items. Kallmeyer (2010, chapter 7) presents several approaches for optimisations on LCFRS like optimal binarisation and elimination of useless rules. Furthermore she explains the use of filters during parsing that reject items that cannot lead to a goal item. Kallmeyer and Maier (2009) present an incremental Earley parser for SRCG.

Maier et al. (2012) focus on optimizing the extraction from treebanks by limiting the fan-out of the resulting grammar to 2. They note that in the DPTB, which features constituents with at most three discontinuous blocks, the overwhelming majority of the cases with three blocks is caused by punctuation. They show that by changing punctuation annotation the extraction of a 2-LCFRS with minimal informational loss is possible.

3.2. Transition-Based Parsing

Parsing with transitions is based on a pseudo-deterministic approach and widely used for dependency parsing. The parser traverses the input from left to right and performs one local action from a set of possible transitions at each step to incrementally build a graph. In the case of constituent parsing, this graph is a tree. While chart parsers handle ambiguities by maintaining multiple analyses in parallel through the use of dynamic programming and narrowing down the search space with a statistical model, transition-based parsers maintain only one (incomplete) analysis of the input sequence at a time. This is called a *greedy* strategy. They usually utilise a classifier trained on treebank-data to predict the next action when faced with a choice (Nivre, 2008). While this narrows down the search space to the one best scored action at each step, it has been shown that by using neural context-aware classifiers, this approach can yield state-of-the-art results.¹²

In the following subsections, I will first provide some necessary prerequisites and define transitionbased parsing formally. Then, a short outline of the traditional projective constituent *shift-reduce* parser is given, followed by an exploration of two extensions of this approach for discontinuous constituent parsing: SwAP and GAP.

3.2.1 Transition Systems

The following definitions for *transition systems* and *transition sequences* are based on Nivre (2008). They were introduced for dependency parsing but are, with small modifications, applicable to

 $^{^{12}}$ See Section 6.3 for a comparison of results for discontinuous constituent parsing.

constituent parsing.

Definition 3.9 (Transition system).

Let T be an alphabet of terminals and N an alphabet of nonterminals. A transition system wrt. T and N is a 5-tuple $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$, where

- 1. C is a set of *configurations*,
- 2. \mathcal{T} is a set of transitions, which are (partial) functions $\tau : \mathcal{C} \to \mathcal{C}$,
- 3. σ is an *initialisation function* that maps an input word to a configuration, i.e. $\sigma: T^* \to \mathcal{C}$,
- 4. $F \subseteq C$ is a set of terminal configurations.
- 5. d is an *output function* that maps a terminal configuration to a syntactic representation labelled over the signature N.

The use of the terms *terminal* and *nonterminal* is disconnected from any formal grammar here and only occurs to allude to their function as the input sequence alphabet and as the signature of the resulting tree.

Usually, a configuration possesses a *stack* S that contains the constructed constituents and a *buffer* B of remaining words. S and B can be thought of as ordered lists of elements where access is only allowed on one end. Only the top element can be retrieved. For S, a new element can be pushed onto the structure to constitute the new top while B is read-only. Usually, S is written with its top facing to the right while B has its top on the left side. The symbol | aids in marking the top element. S|s denotes a stack where s is the top element and S is the (possibly empty) rest of the stack. b|B denotes a buffer with b as its top element. The initialisation function σ returns an initial configuration where the input sentence words are contained in the buffer and where the stack is empty. The stack and the buffer can be written down in list notation $[x_1, ..., x_n]$ which is done to specify the initial buffer configuration. [] denotes an empty structure.

The use of a stack and a buffer originates in *push-down automata* (PDA), a design of automata that is equivalent to CFGs (Hopcroft et al., 2007, chapter 6). The transitions can also be expressed as a formal deduction system. Then, $\sigma(w)$ for an input sequence w serves as an axiom. The goal of the transition system is to construct a terminal configuration by means of transitions. Since the deductions have only one antecedent item, deduction sequences are purely linear. Therefore, the following notation differs from the inference relation given in definition 3.2 in that it is not closed under addition of antecedents.

Definition 3.10 (Transition derivability relation).

Let T be an alphabet of terminals and N an alphabet of nonterminals. Let $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$ be a transition system wrt. T and N. Let $c, c' \in \mathcal{C}$ be configurations.

- The relation $c \Rightarrow c'$ holds iff $c' = \tau(c)$ for some $\tau \in \mathcal{T}$ $c \Rightarrow c'$ reads: c' is directly derivable from c using τ .
- The relation $c \Rightarrow c'$ holds iff there exists any transition $\tau \in \mathcal{T}$ such that $c \Rightarrow c'$. $c \Rightarrow c'$ reads: c' is *directly derivable from c*.
- $\stackrel{*}{\Rightarrow}$ is the reflexive and transitive closure of \Rightarrow . $c \stackrel{*}{\Rightarrow} c'$ reads: c' is *derivable from c*.

For $c_1 \Rightarrow c_2, c_2 \Rightarrow c_3$ we conventionally write $c_1 \Rightarrow c_2 \Rightarrow c_3$. This also holds for \Rightarrow and $\stackrel{*}{\Rightarrow}$.

Definition 3.11 (Transition sequence).

Let T be an alphabet of terminals and N an alphabet of nonterminals. Let $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$ be a transition system and $w \in T^*$ a word. A transition sequence for w in \mathfrak{S} is a sequence $c_{0:n} = c_1, c_2, ..., c_n$ of configurations with $n \in \mathbb{N}$, such that

- 1. $c_1 = \sigma(w)$,
- 2. $c_n \in F$,
- 3. $c_1 \Rightarrow c_2 \Rightarrow ... \Rightarrow c_n$

Definition 3.12 (Valid configuration).

Let T be an alphabet of terminals and N an alphabet of nonterminals. Let $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$ be a transition system wrt. T and N and $w \in T^*$ a word. A configuration $c \in \mathcal{C}$ is called *valid for* w iff there exists a transition sequence $c_{0:n}$ for w in \mathfrak{S} such that $c = c_i$ for some $i \in \mathbb{N}, 1 \leq i \leq n$.

Definition 3.13 (Parse).

Let T be an alphabet of terminals and N an alphabet of nonterminals. Let $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$ be a transition system wrt. T and N and $c_{0:n}$ a transition sequence for $w \in T^*$ in \mathfrak{S} . $d(c_n)$ is called the *parse* assigned to w by $c_{0:n}$

Oracles act as a system for searching and predicting the next transition given a configuration. Nivre (2008) defines oracles in the following way:

Definition 3.14 (Oracle).

Let T be an alphabet of terminals and N an alphabet of nonterminals. Let $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$ be a transition system wrt. T and N.

- 1. An oracle o for \mathfrak{S} is a function $o: \mathcal{C} \to \mathcal{T}$.
- 2. A transition sequence $c_{0:n}$ for some word $w \in T^*$ in \mathfrak{S} is *licensed by o* iff for every $i \in \{1, ..., n-1\}$ it holds that $c_{i+1} = (o(c_i))(c_i)$.

In data-driven parsing, the oracle has access to the gold trees of a given corpus and can base its predictions on them. A *scorer* is trained with the goal of approximating the oracle. It can take the form of a neural classifier and receives input features at each step that are associated with certain elements in the configuration.¹³

Given a transition system $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$ wrt. an alphabet of terminals T and an alphabet of nonterminals N and an oracle o for \mathfrak{S} , the following simple algorithm for deterministic parsing is defined (Nivre, 2008):

Algorithm 3.3 $Parse(w)$				
Input: a sentence $w \in T^*$				
Output: a parse assigned to w				
1: $c \leftarrow \sigma(w)$				
2: while $c \notin F$ do				
3: $c \leftarrow (o(c))(c)$				
4: end while				
5: return $d(c)$				

The benefit of this formal characterisation is the separation of oracles, transitions and configurations which facilitates the analysis of the complexity and other formal properties of the system. According to Nivre (2008) one can usually assume that o(c) and $\tau(c)$ have constant time complexity for all transitions and configurations. Therefore, the complexity of the transition system is given by the worst-case maximum length of transition sequences for an input sentence. The space complexity is characterised by the maximum size of a configuration since the system only needs to store one configuration at a time.

The examination of correctness differs from that of grammar-based parsers. Correctness for a transition system is characterised not by a set of syntactic representations for a specific grammar

 $^{^{13}\}mathrm{A}$ more detailed account of scorers is given in section 3.2.7.
(i.e. its tree language) but by a class of syntactic representations (e.g. the set of all possible labelled trees). Nivre (2008) gives the following definition of soundness and completeness which I slightly modified to account for the tree labelling.

Definition 3.15 (Soundness, completeness, correctness).

Let T be an alphabet of terminals and N an alphabet of nonterminals. Let $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$ be a transition system wrt. T and N.

- 1. \mathfrak{S} is sound for a class \mathbb{D} of syntactic representations, iff for every word $w \in T^*$ and every transition sequence $c_{0:n}$ for w in \mathfrak{S} , it holds that $d(c_n) \in \mathbb{D}$.
- 2. \mathfrak{S} is *complete* for a class \mathbb{D} of syntactic representations iff for every syntactic representation $D_w \in \mathbb{D}$ for some word w, there is a transition sequence $c_{0:n}$ for w in \mathfrak{S} such that $d(c_n) = D_w$.
- 3. \mathfrak{S} is *correct* for a class \mathbb{D} of syntactic representations iff it is sound and complete for \mathbb{D} .

For the class of range-labelled trees correctness of the transition system ensures that every tree can be built using the system and that every syntactic representation built is indeed formally a range-labelled tree. Additionally, to match the notion of parser correctness given in definition 3.7, it needs to be proven that the optimal steps predicted by the oracle lead indeed to the desired tree for an input.

Following Goldberg and Nivre (2012a) one can define oracle correctness as follows:

Definition 3.16 (Oracle correctness).

Let T be an alphabet of terminals, N an alphabet of nonterminals and $\mathfrak{S} = \langle \mathcal{C}, \mathcal{T}, \sigma, F, d \rangle$ be a transition system wrt. T and N that is correct for a class \mathbb{D} of syntactic representations and $o: \mathcal{C} \to \mathcal{T}$ and oracle for \mathfrak{S} .

o is correct iff for every word $w \in T^*$ and a syntactic representation $D_w \in \mathbb{D}$ for w the transition sequence $c_{0:n}$ for w licensed by o assigns exactly the parse $d(c_n) = D_w$.

Note that a proof of correctness can only be done for the oracle and not for the final model trained to approximate the oracle. Its properties rely crucially on the quality of the neural model and its learning algorithm.

3.2.2 Comparing Chart Parsing and Transition-Based Parsing

In this section I will point out several differences between chart parsers and transition-based parsers. In chart parsers the items are intermediate parsing results which represent a partial analysis. One or more items can be used to deduce a new analysis for some part of the input. In a transitionbased system the items are configurations that in turn contain all available partial analyses for the input. They can be seen as a data structure that represents the state of analysis at a certain time-step, comparable to a view of chart and agenda. But unlike in chart-parsing, analyses cannot conflict with each other.

A deduction can be used to alter the inner structure of the item that restricts which elements are available for derivation (e.g. shift an element from the buffer to the stack) or to derive a new partial analysis from certain elements in the item. When deriving a new partial analysis from elements using a transition, they are "used up" and no longer available for future derivation steps. Thus, by design there is no derivational ambiguity in the retrieved analysis. This reduces the formalism's complexity. The parsing process works bottom-up and the transition system can be seen as a framework to keep track of active partial analyses which have not been used in a derivation yet.

While both approaches are usually data-driven, the weighted CYK chart parser works with an explicitly formalised grammar that was extracted from a treebank in a preceding step. Weights are

assigned to productions based on their observed frequency of occurrence. Transition-based parsers are generally built on a neural network framework. They are trained on a treebank and learn to replicate the patterns found in it. The transition system specifies the order and manner in which this replication takes place (e.g. bottom-up retrieval of derivation steps, left-to-right processing). Thus, one can view the model as implicitly learning a weighted grammar that licenses the treebank. Adopting this view, the model can be said to assign the highest score to the chain of actions that leads to the analysis with the best overall weight.

Note however that while a chart-parser based on an extracted grammar can only apply derivations that were observed during the extraction process on a corpus, a neural approach yields probabilities for all derivations allowed by the transition system including those that it did not observe during training. This has two consequences: on the one hand, it greatly expands the field of potential parsing errors far beyond the scope of derivational ambiguity inherent to grammar-based approaches. On the other hand, it enables the parser to generalise over constituents that exhibit a similar behaviour and to predict the correct action if faced with a context that is only attested for a portion of these constituents in the training data.

In contrast to the production weights used in weighted chart parsing, the likelihood of a production in a transition-based neural model is not independent of surrounding items. A prediction is computed using a complex neural function based on features like items present on the stack and the buffer or previously derived items. This mitigates the disadvantage that arises from pursuing a greedy 1-best strategy. The parameters of this function are adjusted during training to produce higher scores for derivations found in the gold derivation trees. Improbable derivations are penalised by the cost-function of the learning algorithm.

3.2.3 Standard Shift-Reduce Projective Parsing

Shift-reduce parsing is a bottom-up left-to-right parsing method for CFGs. It serves as the basis for most transition-based parsers. Here, a configuration is a tuple $\langle S, B \rangle$ with a stack S and a buffer B. The configuration is visualised in figure 3.5.

The stack consists of derived constituents usually characterised by a nonterminal and a tuple of pointers to its daughters. The goal is to construct a single item via transitions from which an ordered derivation tree can be recovered. This mirrors the use of backpointers in charts.

Definition 3.17 (Recursive labelled ordered tree).

Let X be an alphabet of labels.

- 1. A symbol $A \in X$ is a recursive ordered tree labelled over signature X, more specifically a leaf.
- 2. A tuple $\langle A_1, \mathbf{D}_1 \rangle$ is a recursive ordered tree labelled over signature X if
 - (a) $A_1 \in X$,
 - (b) \mathbf{D}_1 is a tuple of recursive ordered trees labelled over signature X and
 - (c) there is no cyclic sequence of recursive ordered trees $\langle A_1, \mathbf{D}_1 \rangle, \langle A_2, \mathbf{D}_2 \rangle, ..., \langle A_n, \mathbf{D}_n \rangle$ where $\mathbf{D}_i^{[j]} = \langle A_{i+1}, \mathbf{D}_{i+1} \rangle$ for some $j \in \{1, ..., |\mathbf{D}_i|\}$ for all $i \in \{1, ..., n-1\}$ and $\langle A_n, \mathbf{D}_n \rangle = \langle A_1, \mathbf{D}_1 \rangle.$
- 3. Nothing else is a recursive ordered tree labelled over signature X.

The notation in definition 3.17 is an alternative to the traditional graph notation $\langle V, \triangleleft, \prec, \hat{v}, \lambda \rangle$ of an ordered tree with a node labelling over signature X introduced in section 2.1. It is the common way to implement trees as a data structure. A recursive ordered tree $\langle A, \mathbf{D} \rangle$ can be likewise treated as an ordered tree with a root node that is labelled with A and has daughters defined by the subtrees in D for which precedence is determined by their position in **D**. In the following, I will use the two notations interchangeably.

The transition system of Sagae and Lavie (2005) is used as the foundation for many transitionbased constituent parsers. They work with binarised trees with headedness information.¹⁴ Coavoux and Crabbé (2017a) give the following reduced set of transitions for unlexicalised parsing:

- 1. SHIFT moves the first element from the buffer onto the stack.
- 2. REDUCEUNARY-X creates a new constituent, removes the top element from the stack, assigns it as the single daughter of the new constituent and puts the new constituent onto the stack.
- 3. REDUCE-X creates a new constituent, removes the two topmost elements from the stack, assigns them as the daughters of the new constituent and puts the new constituent onto the stack.

The formalisations as deductions are presented in figure 3.4. Cross and Huang (2016b) use a similar set of transitions that allows non-binary trees.

$$\begin{array}{l} axiom \quad \overline{\langle [], [w_1, ..., w_n] \rangle} \\ \text{SHIFT} \quad \frac{\langle S, w_i | B \rangle}{\langle S | w_i, B \rangle} \\ \text{Reduce-X} \quad \frac{\langle S | s_1 | s_0, B \rangle}{\langle S | \langle X, \langle s_1, s_0 \rangle \rangle, B \rangle} \\ \text{ReduceUnary-X} \quad \frac{\langle S | s_0, B \rangle}{\langle S | \langle X, \langle s_0 \rangle \rangle, B \rangle} \\ goal \quad \langle [ROOT], [] \rangle \end{array}$$

Figure 3.4: Shift-reduce as deductions, $\mathcal{T}^{SR} = \{SHIFT, REDUCE-X, REDUCEUNARY-X\}.$

Definition 3.18 (Shift-reduce transition system).

A SHIFT-REDUCE transition system for an alphabet of terminals T and an alphabet of nonterminals N is a 5-tuple $\mathfrak{S}^{SR} = \langle \mathcal{C}^{SR}, \mathcal{T}^{SR}, \sigma, F, d \rangle$ where

- 1. $C^{SR} = \{ \langle S, B \rangle \mid \forall c \in S \cup B : c \text{ is a binary ordered tree with leaf labels from } T \text{ and internal labels from } N \},^{15}$
- 2. \mathcal{T}^{SR} is the set of transitions defined for \mathcal{C}^{SR} as given in figure 3.4,
- 3. $\sigma(w_{0:n}) = \langle [], [w_1, ..., w_n] \rangle$ for every $w_{0:n} \in T^+$,
- 4. $F = \{ \langle [\text{ROOT}], [] \rangle \mid \text{ROOT} \text{ is a binary orderd tree with leaf labels from } T \text{ and internal labels from } N \}$ and
- 5. d gives the sole element in S for every $\langle S, B \rangle \in F$.

Example 3.19.

Figure 3.6 shows an example transition sequence for the sentence *the man took the book* with the derivation tree depicted in figure 1.1.

 $^{^{14}}$ A discussion on the matter of head-information and lexicalisation is given in section 3.2.6.

 $^{^{15}}$ In a slight abuse of notation, set operators are used here to refer to all elements in the two sequences.



Figure 3.5: Illustration of a configuration in C^{SR} . Transitions involve changes to the local part of the configuration (dashed line). Illustration reproduced from Coavoux and Cohen (2019).

Step	Stack	Buffer	Action
0		the, man, took, the, book	Shift
1	the	man, took, the, book	Shift
2	the, man $\overset{\text{the, man}}{\sim}$	took, the, book	REDUCE-NP
3	$\langle \mathrm{NP}, \langle ullet, ullet angle angle$	took, the, book	Shift
4	$\langle NP, \langle \bullet, \bullet \rangle \rangle, took$	the, book	REDUCEUNARY-Verb
5	$\langle NP, \langle \bullet, \bullet \rangle \rangle, \langle Verb, \langle \bullet \rangle \rangle$	the, book	Shift
6	$\langle NP, \langle \bullet, \bullet \rangle \rangle, \langle Verb, \langle \bullet \rangle \rangle, the$	book	Shift
7	$\langle NP, \langle \bullet, \bullet \rangle \rangle, \langle Verb, \langle \bullet \rangle \rangle, the, book$		REDUCE-NP
8	$\langle NP, \langle \bullet, \bullet \rangle \rangle, \langle Verb_{\epsilon} \langle \bullet \rangle \rangle, \langle NP, \langle \bullet, \bullet \rangle \rangle$		REDUCE-VP
9	$\langle \mathrm{NP}_{\mathbf{z}} \langle \mathbf{\bullet}, \mathbf{\bullet} \rangle \rangle, \langle \mathrm{VP}_{\mathbf{z}}, \langle \mathbf{\bullet}, \mathbf{\bullet} \rangle \rangle$		REDUCE-Sentence
10	$\langle \text{Sentence}, \langle ullet, ullet \rangle \rangle$		

Figure 3.6: Shift-Reduce parse for the man took the book producing the tree from Figure 1.1.

For an input and a parse tree to derive, there exists only one transition sequence. If the topmost element's root v_0 is the unary daughter of another constituent or if v_0 and the second topmost element's root v_1 represent a gold constituent, a reduction must take place. If not, SHIFT must be performed. Shifting when elements are reducible would move a new item onto the stack making v_0 inaccessible. From this step on, the top of the stack can only be occupied by elements succeeding v_0 in the precedence relation. This minimal ambiguity makes deriving an oracle a trivial task.

The elements in a valid configuration for a word w represent a forest of derivation trees that are subtrees of some possible tree D_w for w. The roots of the forest elements are the surface elements on the stack and in the buffer. They represent a *tree-cut* of D_w . A REDUCE transition is a transition from a tree-cut $v_1, ..., v_i, v_{i+1}, ..., v_m$ to another tree-cut $v_1, ..., v_p, ..., v_m$ where v_p is the parent of v_i, v_{i+1} . This illustrates why no two conflicting analyses of constituents can cooccur in this framework. Items dominated by an element in the current tree-cut are not present at surface-level in the configuration and cannot be used to compose alternate analyses.

The following definition of a tree-cut is based on Versley (2014).

Definition 3.20 (Tree-cut).

Let $D = \langle V, \triangleleft, \hat{v} \rangle$ be a tree. A *tree-cut in* D is a sequence $v_1, ..., v_n$ with $v_k \in V$ for all $k \in \{1, ..., n\}$ for some $n \in \mathbb{N}$ such that for all paths from \hat{v} to a leaf, exactly one node from the path is contained in the sequence.

Definition 3.21 (Partial parse).

Let T be an alphabet of terminals and $w \in T^*$ a word. A *partial parse* for w is an ordered forest $D_{0:n}$ where there exists some tree D_w for w such that the roots of $D_1, ..., D_n$ are a tree-cut in D_w and such that $D_1, ..., D_n$ are subtrees of D_w .

Example 3.22.

Figure 3.7 shows two possible tree-cuts in a tree. These demonstrate the reduction of two nodes in a tree-cut: Verb and NP in the left tree and their parent VP in the right tree.



Figure 3.7: Two possible tree-cuts (<u>underlined</u>) in a projective tree. The tree is taken from Chomsky (1956).

Note that this set of transitions cannot predict ε -productions. There is no deduction rule to predict empty leaves between elements or to reduce them. Therefore, the initialisation function also does not accept the empty word. One could add a configuration with empty stack and buffer to the set of terminal configurations to enable the latter but this shall not be of any concern here as the corpora at hand do not feature empty leaves.

What is significant, however, is this system's inability to deduce crossing edges. It can only reduce adjacent elements into constituents. Several strategies were developed to solve this problem: new transitions like SWAP and GAP reorder items and make discontinuous parts of a constituent adjacent in order to reduce them. In the following, I will discuss these approaches in more detail.

3.2.4 Reordering with the Swap Action

Versley (2014) was the first to adapt the SWAP reordering action from dependency parsing to constituent parsing. The underlying idea is that for every non-projective tree, there is a projective tree that can be constructed by reordering the input words. This projective tree can be parsed with traditional methods. SWAP can be used to change the order of two elements and thus allows for out-of-order processing. In the following, this approach will be referred to as SHIFT-REDUCE-SWAP or SR-SWAP.

Versley (2014) uses an *easy-first* approach inspired by Goldberg and Elhadad (2010) where a parsing action or a SWAP action can occur anywhere in an input string based on the prediction of a scorer. Maier (2015) adapted the SWAP action to the shift-reduce transition system. I will outline this approach in more detail.

Discontinuous trees are not ordered (cf. definition 2.29). Only the leaves have the range precedence relation $<_{\rm range}$ defined on them that describes the surface order of the input tokens. Therefore, we need to modify the definition of a transition constituent for discontinuous constituent parsing. The list of daughters for a constituent no longer needs to be ordered and can therefore be defined as a set.

Definition 3.23 (Recursive unordered tree).

Let X be a set of labels.

1. A symbol $A \in X$ is a recursive unordered tree labelled over signature X, more specifically a leaf,

- 2. A tuple $\langle A_1, \mathcal{D}_1 \rangle$ is a recursive unordered tree labelled over signature X if
 - (a) $A_1 \in N$,
 - (b) \mathcal{D}_1 is a set of recursive unordered trees labelled over signature X and
 - (c) there is no cyclic sequence of recursive unordered trees $\langle A_1, \mathcal{D}_1 \rangle, \langle A_2, \mathcal{D}_2 \rangle, ..., \langle A_n, \mathcal{D}_n \rangle$ such that $\langle A_{i+1}, \mathcal{D}_{i+1} \rangle \in \mathcal{D}_i$ for $i \in \{1, ..., n-1\}$ and $\langle A_n, \mathcal{D}_n \rangle = \langle A_1, \mathcal{D}_1 \rangle$.
- 3. Nothing else is a recursive unordered tree labelled over signature X.

This notation is equivalent to the graph-based definition $\langle V, \triangleleft, \hat{v}, \lambda \rangle$ of an unordered labelled tree and the two notations will be used interchangeably within the following paragraphs. The REDUCE-X and REDUCEUNARY-X transitions are modified accordingly to reflect the new notation. Furthermore, the parser introduced in Maier (2015) also marks the lexical head of a constituent. I omit this aspect for reasons of brevity.

Figure 3.8: SWAP transition.

The central innovation is the introduction of the SWAP transition shown in figure 3.8. Using this transition, it is possible to derive discontinuous constituents by reordering items such that for two elements that have to be reduced, every element in between that blocks this reduction by not being a sister node is moved to a different position. The SR-SWAP transition system is formally defined as follows:

Definition 3.24 (SR-Swap transition system).

A SHIFT-REDUCE-SWAP transition system for an alphabet of terminals T and an alphabet of nonterminals N is a 5-tuple $\mathfrak{S}^{SRS} = \langle \mathcal{C}^{SRS}, \mathcal{T}^{SRS}, \sigma, F, d \rangle$ where

- 1. C^{SRS} is the set of all configurations $\langle S, B \rangle$ composed of partial ε -free binary range-labelled trees over N,
- 2. $\mathcal{T}^{SRS} = \{SHIFT, REDUCE-X, REDUCEUNARY-X, SWAP\}$ the set of transitions defined on \mathcal{C}^{SRS} .
- 3. $\sigma(w_{0:n}) = \langle [], [\langle 0, 1 \rangle, ..., \langle n-1, n \rangle] \rangle$ for every $w_{0:n} \in T^+$,
- 4. $F = \{ \langle [\text{ROOT}], [] \rangle \mid \text{ROOT} \text{ is an } \varepsilon \text{-free binary range-labelled tree over } N \}$ and
- 5. d retrieves the tree from the sole element in S for every $\langle S, B \rangle \in F$.

Example 3.25.

Figure 3.9 shows an example transition sequence for the sentence $Dar\ddot{u}ber mu\beta$ nachgedacht werden and the derivation tree depicted in figure 1.3.

It is straightforward to see that one can apply different transitions to retrieve the same result. More specifically, in figure 3.9 a SWAP at step 2 would have already brought the configuration into a form where reducing the tree-cut would be possible with the standard set of transitions. Indeed, the possibility to put items from the stack back onto the buffer leads to significant ambiguity and even allows for looping behaviour by alternating between SHIFT and SWAP transitions. This poses a challenge for defining oracles that lead to an optimally short transition sequence.

Step	Stack	Buffer	Action
0		$Dar"uber_1, mu \beta_2, nachgedacht_3, werden_4$	Shift
1	$Dar"uber_1,$	$mu\beta_2$, nachgedacht ₃ , werden ₄	Shift
2	$Dar"uber_1, mu\beta_2,$	$nachgedacht_3, werden_4$	Shift
3	$Dar"uber_1, mu \beta_2, nachgedacht_3,$	$werden_4$	SWAP
4	$Dar"upper_1$, nachgedacht ₃	$mu\beta_2$, werden ₄	REDUCE-VP
5	$\langle VP, \{\bullet, \bullet\} \rangle$	$mu\beta_2$, werden ₄	Shift
6	$\langle VP, \{\bullet, \bullet\} \rangle, mu\beta_2$	$werden_4$	Shift
7	$\langle VP, \{\bullet, \bullet\} \rangle, mu\beta_2, werden_4$		SWAP
8	$\langle V\underline{P}, \{\bullet, \bullet\} \rangle$, werden ₄	$mu\beta_2$	REDUCE-VP
9	$\langle VP, \{\bullet, \bullet\} \rangle$	$mu\beta_2$	Shift
10	$\langle V\underline{P}, \{\bullet, \bullet\} \rangle, mu\beta_2$		Reduce-S
11	$\langle \mathrm{S}, \{ullet, , ullet \} angle$		

Figure 3.9: Example parse using SWAP transition for $Dar\ddot{u}ber mu\beta$ nachgedacht werden with tree shown in figure 1.3, POS tags omitted, indices given alongside tokens instead of ranges.

Swap Loops Several suggestions have been made for restricting the application of transitions to reduce ambiguity. A straightforward restriction is to prohibit SWAP loops by ensuring that two elements cannot be swapped twice. For this, Maier and Lichte (2016) define the following order.

Definition 3.26 (Indices of ranges).

- 1. $Ind_{range} : Ranges \to \mathcal{P}(\mathbb{N})$ is defined as $\langle a, b \rangle \mapsto \{a+1, ..., b\}$.
- 2. Let N be an alphabet. Let $D = \langle V, \triangleleft, \hat{v}, \lambda \rangle$ be a range-labelled tree over N. $Ind_D : V \to \mathcal{P}(\mathbb{N})$ gives the set of all indices dominated by a node:

$$v\mapsto \{i\mid \exists u\in V: v\triangleleft^* u, u \text{ is a leaf}, i\in Ind_{\mathrm{range}}(\lambda(u))\}.$$

Definition 3.27 (Leftmost index ordering).

Let $D_{0:n}$ be a partial parse for w with $D_i = \langle V_i, \triangleleft, \hat{v}_i, \lambda \rangle$ for every $i \in \{1, ..., n\}$. $<_{\text{ind}}$ is defined as a strict total order on the elements of $D_{0:n}$:

$$D_i <_{\text{ind}} D_j :\Leftrightarrow \min(Ind_{D_i}(\hat{v}_i)) < \min(Ind_{D_j}(\hat{v}_j))$$

Now, for two trees on the stack s_1, s_0 SwAP is only allowed if they satisfy $s_1 <_{\text{ind}} s_0$ as shown in the modified deduction rule in figure 3.10. This is the case for all elements in the initial order. If SwAP is performed, this changes the order of s_1, s_0 in the configuration. Since $s_0 <_{\text{ind}} s_1$ cannot be fulfilled, SwAP cannot be performed a second time for these two elements.

SWAP
$$\frac{\langle S|s_1|s_0, B \rangle}{\langle S|s_0, s_1|B \rangle} s_1 <_{\text{ind}} s_0$$

Figure 3.10: SWAP transition standard condition.

Eager Swap Nivre (2009) and Nivre et al. (2009) outline two strategies for reducing transition sequence lengths for SWAP in dependency parsing: *eager swap* and *lazy swap*. They were adapted for constituent parsing by Versley (2014). These are conditions concerning oracle extraction. The oracle has access to the gold tree and is constructed to yield only partial analyses of the gold tree.

Therefore, in the following, trees in the deduction rules are simply treated as their roots, i.e. as nodes in the gold tree.

The idea of the SwAP transition is to convert the input sequence into a projective order, or in other words, to an order where the target derivation tree can be recovered by reducing only adjacent elements. Therefore eager swap restricts the SwAP transition to cases where the two elements in consideration are not in a projective order. It prohibits swapping when a SwAP would be unnecessary to retrieve the tree. The quest of Versley (2014) is to define a desirable order for every tree-cut that allows to parse the elements projectively. For this, he defines the following property.

Definition 3.28 (<-compatibility).

Let $D = \langle V, \triangleleft, \hat{v} \rangle$ be a tree. A node ordering $\prec \subseteq V \times V$ defined on all pairs of nodes that co-occur in some tree-cut of D is called \triangleleft -compatible iff for $v, v' \in V$ with $v \triangleleft^* v'$ and $u, u' \in V$ with $u \triangleleft^* u'$ it holds that $u \prec v \Rightarrow u' \prec v'$.

Note that this is equivalent to conditions 3.(b) and 3.(c) for the linear precedence relation of an ordered tree given in definition 2.10.

Versley (2014) shows that sorting the leaves of a tree in a \triangleleft -compatible order \prec makes it possible to find a derivation using standard transition sets for projective parsing. This stems from the observation that reducing a \prec -ordered tree-cut yields another \prec -ordered tree-cut. Thus, no crossing edges are encountered and no reductions of non-adjacent elements are necessary.

Corollary 3.29.

Let $\langle V, \triangleleft, \hat{v} \rangle$ be a tree, \prec a \triangleleft -compatible order and $v_1, ..., v_n$ a \prec -ordered tree-cut. Reducing some elements v_i, v_{i+1} with $i \in \{1, ..., n-1\}$ to $p \in V$ with $p \triangleleft v_i$ and $p \triangleleft v_{i+1}$ results in another \prec -ordered tree cut $v_1, ..., v_{i-1}, p, v_{i+2}, ..., v_n$.

Proof. Assumption: $v_1, ..., v_{i-1}, v_i, v_{i+1}, v_{i+2}, ..., v_n$ is \prec -ordered and there is $p \in V$ such that $p \triangleleft v_i, p \triangleleft v_{i+1}$. In particular this means that $v_{i-1} \prec v_i \prec v_{i+1} \prec v_{i+2}$. If $p \prec v_{i-1}$ then \triangleleft -compatibility of \prec demands that also $v_i \prec v_{i-1}$ which contradicts the assumption. Therefore, it must hold that $v_{i-1} \prec v_i$. Likewise, if $v_{i+2} \prec p$ then per \triangleleft -compatibility of \prec it would also hold that $v_{i+2} \prec v_{i+1}$ which contradicts the assumption. Therefore, $v_{i+2} \prec v_{i+1}$ which contradicts the assumption. Therefore $p \prec v_{i+2}$. Thus, $v_1, ..., v_{i-1}, p, v_{i+2}, ..., v_n$ is \prec -ordered.

A generalisation to non-binary trees is straightforward. The precedence order $<_{\text{range}}$ on leaves can be naturally extended to a partial \triangleleft -compatible order for internal nodes in the following way (Versley, 2014):

$$v_1 <_{\mathsf{N}} v_2 \Leftrightarrow \forall u_1, u_2 \in V : (v_1 \triangleleft^* u_1 \land v_2 \triangleleft^* u_2 \land u_1 \text{ and } u_2 \text{ are leaves}) \Rightarrow u_1 <_{\mathrm{range}} u_2$$
(3.3)

Note however that this does not suffice for nonterminals v_1, v_2 with intertwined leaves since in such a case neither $v_1 <_N v_2$ nor $v_2 <_N v_1$. To solve this issue, one can instead extend $<_{\text{range}}$ to a <-compatible order \prec by defining a standard descendent for internal sister nodes to base the order on. This way, $<_N$ is respected locally, i.e. for two sister nodes v_1, v_2 it holds that $v_1 <_N v_2$ entails $v_1 \prec v_2$.

Versley (2014) works with a lexicalised model. His trees provide a function $h: V \to V$ that gives the head daughter for each node. Therefore he suggests recursively assigning the head leaf to an internal node:

$$h^*(v) := \begin{cases} v & \text{if } v \text{ is a leaf,} \\ h^*(h(v)) & \text{otherwise.} \end{cases}$$
(3.4)

Now, the order $<_h$ defined on all tree-cuts can be characterised as follows:

$$v_{1} <_{h} v_{2} \quad :\Leftrightarrow \begin{cases} \text{if } \exists p \in V : p \triangleleft v_{1}, p \triangleleft v_{2} \text{ and } h^{*}(v_{1}) <_{\text{range}} h^{*}(v_{2}) \text{ or} \\ \text{if } \exists v_{1}', v_{2}' \in V : v_{1}' \triangleleft^{*} v_{1}, v_{2}' \triangleleft^{*} v_{2} \text{ and } v_{1}' <_{h} v_{2}'. \end{cases}$$
(3.5)

The actual choice of leaf is arbitrary as Versley (2014) notes. One could likewise use leftmost index ordering as the base:

$$v_{1} <_{\mathbf{G}} v_{2} \quad :\Leftrightarrow \begin{cases} \text{if } \exists p : p \triangleleft v_{1}, p \triangleleft v_{2} \text{ and } v_{1} <_{\text{ind}} v_{2} \text{ or} \\ \text{if } \exists v_{1}', v_{2}' : v_{1}' \triangleleft^{*} v_{1}, v_{2}' \triangleleft^{*} v_{2} \text{ and } v_{1}' <_{\mathbf{G}} v_{2}'. \end{cases}$$
(3.6)

Maier (2015) extracts a projective ordering by use of *post-order traversal* on the tree. Post-order traversal on an ordered binary tree retrieves a sequence of all nodes by starting with the root node and for every node recursively traversing 1) through the left daughter, 2) through the right daughter and 3) appending the node itself to the sequence. In an ordered tree, the local order of node children is provided with the tree. Here, leftmost index ordering $<_{ind}$ is used instead.

Algorithm 3.4 shows how to extract a post-order sequence from a tree. $\langle_{\rm G}$ is then defined by precedence in the extracted sequence. It can be seen that $\langle_{\rm G}$ as defined by equation 3.6 entails the relation extracted from algorithm 3.4. Note however that algorithm 3.4 also defines an order for nodes that stand in the dominance relation. This does not influence parsing since two such nodes never co-occur in a valid parsing configuration.

Algorithm 3.4 $\operatorname{PostOrder}(r)$				
Input: a binary tree identified by its root node r				
Output: a sequence of nodes				
1: if r is leaf then				
2: return $[r]$				
3: end if				
4: $v_1, v_2 \leftarrow \text{children}(r)$				
5: if $v_1 <_{\text{ind}} v_2$ then				
6: return PostOrder $(v_1) \circ$ PostOrder $(v_2) \circ [r]$				
7: else				
8: return PostOrder $(v_2) \circ$ PostOrder $(v_1) \circ [r]$				
9: end if				

Maier (2015) uses a strict eager strategy based on binary trees that demands that reductions take place according to the elements' post-ordering. For the topmost element s_0 of the stack, the oracle checks if the first element of the buffer is the direct projective successor. If not, SHIFT gets predicted until the direct successor b_i is at the top of the stack. Then, all the elements that were shifted onto the stack located between s_0 and b_i get swapped back to the buffer. This forces a left-to-right completion of the projective tree. Since this is the only allowed case for a SWAP transition, it follows that the buffer consists only of leaves.

Example 3.30.

Figure 3.11 visualises the ordering retrieved by a post-order traversal on a discontinuous tree. Swapping w_1 and w_2 would be permissible by condition $<_{\text{ind}}$ but not by $<_{\text{G}}$. Figure 3.12 shows part of a transition sequence for the tree using the strict eager strategy of Maier (2015).

Although this approach is straightforward, one can see that it is not optimal. It predicts SWAP transitions in cases where completing constituents first would result in shorter derivations. Here, reducing the intermediary subtree rooted at B would decrease the transition count.

A more flexible approach is taken by Stanojević and Alhama (2017) who perform online reordering. This is permissible since the \leq_{G} -relationships of nodes are preserved under REDUCE



Figure 3.11: Example for projective ordering by post-order traversal.

Step	Stack	Buffer	Action
0		w_1, w_2, w_3, w_4, w_5	Shift
1	w_1	w_2, w_3, w_4, w_5	Shift
2	w_1, w_2	w_3, w_4, w_5	Shift
3	w_1, w_2, w_3	w_4, w_5	Shift
4	w_1, w_2, w_3, w_4	w_5	Shift
5	w_1, w_2, w_3, w_4, w_5		SWAP
6	w_1, w_2, w_3, w_5	w_4	SWAP
7	w_1, w_2, w_5	w_3, w_4	SWAP
8	w_1, w_5	w_2, w_3, w_4	Reduce-C
9	$\langle \mathrm{C}, \{ ullet, ullet \} angle$	w_2, w_3, w_4	

Figure 3.12: Example parse using the eager swap strategy outlined in Maier (2015) for the tree in figure 3.11.

and UNARAYREDUCE transitions due to \triangleleft -compatibility. Algorithm 3.5 extracts the oracle.¹⁶ It demands that nonterminals be allowed to be swapped onto the buffer.

Now, the oracle function returns the immediate successor of a configuration c in O. This oracle prioritises the completion of substructures that are already in projective order. The swapAllowed condition is given in equation 3.7.

$$swapAllowed(\langle S|s_1|s_0, B\rangle) :\Leftrightarrow s_0 <_{\mathbf{G}} s_1 \tag{3.7}$$

Note however that the condition does not benefit parsing of figure 3.13. In this case, $\langle_{\rm G}$ blocks the more efficient SWAP of Darüber₁ and muß₂ since Darüber₁ $\langle_{\rm G}$ muß₂. This pinpoints at the fact that the projective ordering that leads to a minimal number of transitions is dependent on the form of the individual tree. Here, a rightmost index ordering would lead to a smaller transition count. Generally, ordering by median leaf index or by start index of the largest consecutive span of the constituent could be used to make the parser complete constituents in an optimal place to reduce the number of SWAP transitions. To my knowledge, this has not been explored yet.

Lazy Swap Eager swap can lead to suboptimal transition sequence lengths when it swaps items into groups of nodes that could first be reduced projectively without the need for further swapping. This leads to chains of SWAP applications. Take for instance the tree in figure 3.14. Equation 3.8

 $^{^{16}}$ I adapted the algorithm to the unlexicalised transition set defined for binary trees in section 3.2.3.

Algorithm 3.5 Oracle extraction

Input: a word $w_1, ..., w_n$, a tree D over w **Output:** an oracle O1: $c \leftarrow \langle [], [w_1, ..., w_n] \rangle$ 2: $O \leftarrow List[]$ 3: while $c \neq goal$ do if REDUCE- $X(c) \in D$ then 4: $c \leftarrow \text{REDUCE-X}(c)$ 5: O.append(REDUCE-X) 6: else if REDUCEUNARY- $X(c) \in D$ then 7: $c \leftarrow \text{ReduceUnary-X}(c)$ 8: *O*.append(REDUCEUNARY-X) 9: else if swapAllowed(c) = True then 10: $c \leftarrow \text{SWAP}(c)$ 11: O.append(SWAP)12:13:else 14: $c \leftarrow \text{SHIFT}(c)$ O.append(SHIFT) 15:end if 16: 17: end while 18: return O



Figure 3.13: Projective ordering for the tree over Darüber muß nachgedacht werden.

shows the transition sequence that would be induced by algorithm 3.5.¹⁷

$$\begin{aligned} ||w_{1}, w_{2}, w_{3}, w_{4} \underset{\text{SHIFT}}{\Rightarrow} w_{1}||w_{2}, w_{3}, w_{4} \underset{\text{SHIFT}}{\Rightarrow} w_{1}, w_{2}||w_{3}, w_{4} \underset{\text{SHIFT}}{\Rightarrow} \\ w_{1}, w_{2}, w_{3}||w_{4} \underset{\text{Swap}}{\Rightarrow} w_{1}, w_{3}||w_{2}, w_{4} \underset{\text{SHIFT}}{\Rightarrow} w_{1}, w_{3}, w_{2}||w_{4} \underset{\text{SHIFT}}{\Rightarrow} \\ w_{1}, w_{3}, w_{2}, w_{4}|| \underset{\text{Swap}}{\Rightarrow} w_{1}, w_{3}, w_{4}||w_{2} \underset{\text{RedCUCE-A}}{\Rightarrow} w_{1}, A||w_{2} \end{aligned}$$

$$(3.8)$$

 w_2 and w_3 get swapped since $w_3 <_G w_2$. But this blocks the reduction of w_3 and w_4 to A so that a second SWAP is necessary changing the order of w_2 and w_4 . Then, w_3 and w_4 are adjacent again and can be reduced. Postponing the SWAP of w_2 and w_3 and instead shifting, reducing A and then swapping w_2 and A would lower the transition count and reduce the number of SWAPs by one. Equation 3.9 shows this shorter transition sequence leading to the same configuration.

$$||w_{1}, w_{2}, w_{3}, w_{4} \underset{\text{SHIFT}}{\Rightarrow} w_{1}||w_{2}, w_{3}, w_{4} \underset{\text{SHIFT}}{\Rightarrow} w_{1}, w_{2}||w_{3}, w_{4} \underset{\text{SHIFT}}{\Rightarrow} w_{1}, w_{2}, w_{3}||w_{4} \underset{\text{SHIFT}}{\Rightarrow} w_{1}, w_{2}, w_{3}, w_{4}|| \underset{\text{Reducce-A}}{\Rightarrow} w_{1}, w_{2}, A|| \underset{\text{SWAP}}{\Rightarrow} w_{1}, A||w_{2}$$

$$(3.9)$$

Motivated by this observation, *lazy swap* adds a second condition to the SWAP transition: the two topmost stack items are not swapped if the adjacent element to the right (i.e. the first position in the buffer) has the same closest maximal fully projective constituent as the stack top. This prevents the parser from swapping elements into groups of nodes that can be reduced to a

 $^{^{17}}$ || represents the border between stack and buffer in this shorthand notation.



Figure 3.14: Minimal example for the motivation of lazy swap.

projective constituent. To formally define this notion, one needs the following (Stanojević and Alhama, 2017):

Definition 3.31 (Projective Constituent).

Let $w = w_1, ..., w_n$ be a word and $D = \langle V, \triangleleft, \hat{v} \rangle$ a tree for w. Let $v \in V$ be a node.

- 1. v is called a *projective constituent* iff v dominates a span of leafs without a gap.
- 2. v is called a *fully projective constituent* iff v is a projective constituent and if all descendants of v are projective constituents.
- 3. v is called a *maximal fully projective constituent* iff v is a fully projective constituent and the parent of v is not a fully projective constituent.

Example 3.32.

Figure 3.15 shows a tree where all projective, fully projective and maximal fully projective constituents are marked. Note that a projective constituent can have children that are not projective constituents, here for instance S. The property does not tell anything about the order of the nodes dominated by the constituent.



Figure 3.15: Tree with maximal fully projective node A. Projective constituents are circled in solid, fully projective constituents dotted and maximal fully projective constituents dashed.

Now, one can define a function that returns the closest ancestor that is a maximally projective constituent for a given constituent:

Definition 3.33 (Closest maximal projective constituent).

Let w be a word, $D = \langle V, \triangleleft, \hat{v} \rangle$ a tree over w. $MPC : V \to V$ is defined as follows:

$$v \mapsto \begin{cases} \arg\min_{m \in M} |Ind_D(m)| & \text{if } M \neq \emptyset, \\ v & \text{otherwise,} \end{cases}$$

where M denotes the set of all maximally fully projective constituents that are accendants of v defined as follows:

 $M = \{ p \in V \mid p \text{ is a maximally fully projective constituent and } p \triangleleft^+ v \}$

Figure 3.10 shows the modified swapAllowed condition for lazy swap. Versley (2014) found that a lazy approach outperforms an eager one, presumably because a large number of SWAP transitions is harder to predict.

$$swapAllowed_{lazv}(\langle S|s_1|s_0, b_0|B\rangle) :\Leftrightarrow s_0 <_{\mathcal{G}} s_1 \land MPC(s_0) \neq MPC(s_0)$$
(3.10)

Lazier Swap Stanojević and Alhama (2017) took the idea of lazy swap and introduced an even *lazier swap* strategy. They noticed that lazy swap does not help when the substructure that would be more efficient to complete first does not exhibit a maximal fully projective constituent but could be reduced to a projective constituent nonetheless.

Example 3.34.

Figure 3.16 shows a tree where only the terminals are maximally fully projective constituents. Therefore a lazy oracle would predict a SWAP for w_2 and w_3 . w_2 would be swapped into the subtree rooted at B which would make two additional SWAP transitions necessary to allow for the treatment of the substructure. First completing the subtree rooted at B by resolving its lower-level discontinuity would lead to a smaller step count.



Figure 3.16: Tree without maximal fully projective nodes (except for terminals). Projective constituents are circled in solid, fully projective constituents dotted and maximal fully projective constituents dashed.

To achieve this, Stanojević and Alhama (2017) introduce a function that returns the lowest projective acendant of a given node:

Definition 3.35 (Closest Projective Constituent).

Let w be a word and $D = \langle V, \triangleleft, \hat{v} \rangle$ a tree over w. The function $CPC : V \to V$ is defined as follows:

$$v \mapsto \underset{m \in M}{\operatorname{arg\,min}} |Ind_D(m)|$$

where M denotes the set of all maximal fully projective constituents that are ascendants of v as established in definition 3.33.

$$\operatorname{swapAllowed}_{\operatorname{lazier}}(\langle S|v_1|s_0, s_0|B\rangle) :\Leftrightarrow s_0 <_{\operatorname{G}} s_1 \wedge CPC(s_1) = CPC(s_0)$$
(3.11)

A SwAP is only allowed if the closest projective constituent of the topmost stack element and the second topmost stack element are the same. If they are not the same, one of the elements belongs to a substructure that can be reduced first (potentially by bringing it first internally into projective order). If they are the same there is no projective subtree for which the analysis would be blocked by swapping the second topmost element. Then, a SwAP is necessary to proceed with the derivation. This version of the transition condition is given in equation 3.11. Stanojević and Alhama (2017) show that the lazier swap condition entails lazy swap:

Corollary 3.36.

Let $D = \langle V, \triangleleft, \hat{v} \rangle$ be a tree and $s_i, ..., s_1, s_0, b_0, ..., b_j$ a tree-cut of D where $s_0 <_G s_1$. It holds that $CPC(s_0) = CPC(s_1) \Rightarrow MPC(s_0) \neq MPC(b_0)$.

Proof. Assumption: $CPC(s_0) = CPC(s_1)$. Either 1. there is a non-empty sequence $v_1, ..., v_k$ of non-projective constituents such that $CPC(s_0) \triangleleft v_1, v_1 \triangleleft v_2, ..., v_k \triangleleft s_0$ or 2. $CPC(s_0) \triangleleft s_0$.

- 1. Since the parent of s_0 is non-projective, it follows that s_0 is a maximal projective constituent. By definition 3.33 it holds that $MPC(s_0) = s_0$. Since s_0 per definition of a tree-cut does not dominate b_0 , it holds that $MPC(s_0) \neq MPC(b_0)$.
- 2. Either (a) $CPC(s_0)$ is fully projective or (b) $CPC(s_0)$ is not fully projective.
 - (a) This contradicts $s_0 <_G s_1$. The relation demands that the two sister nodes p_0 with $p_0 \triangleleft^* s_0$ and p_1 with $p_1 \triangleleft^* s_1$ fulfil $p_0 <_{ind} p_1$. It must hold that $p_0 = s_0$ since $CPC(s_0) \triangleleft s_0$ and $CPC(s_0) \triangleleft^* s_1$. Therefore $s_0 <_{ind} p_1$. Since s_0 is fully projective, it follows that $s_0 <_N p_1$. But this means that s_1 must have been swapped from a position right of s_0 to the position left of s_0 which would not have been allowed by $<_G$ since the two elements already respected projective ordering. Therefore, $CPC(s_0)$ cannot be fully projective.
 - (b) It follows that $MPC(s_0) = s_0$ and therefore $MPC(s_0) \neq MPC(b_0)$ (see 1.).

Example 3.37.

Figure 3.17 shows the transition sequence predicted by the oracle derived using algorithm 3.5 with the lazier swap condition given in figure 3.11 for the tree in equation 3.16. The lazy swap strategy would have predicted a SWAP at step 3 since $MPC(w_3) = w_3 \neq MPC(w_4) = w_4$. This would move w_2 into the subtree rooted at B, blocking its derivation.

Step	Stack	Buffer	Action	check of $SwapAllowed_{lazier}$
0		w_1, w_2, w_3, w_4, w_5	Shift	
1	w_1	w_2, w_3, w_4, w_5	Shift	
2	w_1, w_2	w_3,w_4,w_5	Shift	$w_2 \not<_G w_1$
3	w_1, w_2, w_3	w_4,w_5	Shift	$CPC(w_2) = S \neq B = CPC(w_3)$
4	w_1, w_2, w_3, w_4	w_5	Shift	$w_4 \not<_G w_3$
5	w_1, w_2, w_3, w_4, w_5		SWAP	$w_5 <_G w_4 \land CPC(w_4) = B = CPC(w_5)$
6	w_1, w_2, w_3, w_5	w_4	Reduce-A	
7	w_1, w_2, A	w_4	Shift	$CPC(w_2) = S \neq B = CPC(A)$
8	w_1, w_2, A, w_4		Reduce-B	
9	w_1, w_2, B		SWAP	$\mathbf{B} <_G w_2 \land CPC(w_2) = \mathbf{S} = CPC(\mathbf{B})$
10	w_1, B	w_2	Reduce-C	
11	С	w_2	Shift	
12	C, w_2		Reduce-S	
13	S			

Figure 3.17: Example parse using lazier swap strategy for tree in figure 3.16. Pointers are omitted since the nodes are uniquely labelled.

Complexity For an input of length n the number of REDUCE-X transitions to recover a binary tree is n-1 (one for every internal node). The number of REDUCEUNARY-X transitions is potentially unbounded. Let m denote the maximum unary chain length encountered during training used to hard-code a limit into the parser. Then the maximum number of REDUCEUNARY-X transitions is m(2n-1) (one chain for every leaf and for every internal node).

Every leaf must be shifted once. The number of SWAP transitions and further SHIFT transitions is dependent on the choice of oracle. The worst case (not considering loops) is the strict eager oracle used by Maier (2015) since it features the maximum SWAP number before a reduction. When establishing a discontinuous constituents, no intermediate elements are merged to reduce the number of elements to swap.

Figure 3.18 shows the worst case tree for length 5 given by Coavoux and Crabbé (2017a). With 1 and 2 being on top of the stack, the parser needs to perform 3 intermediate SHIFT transitions and 3 SWAP transitions to achieve projective order for 1 and 5. After REDUCE-A, the same procedure follows with 2 intermediate SHIFTs and 2 SWAPs. For a sentence with length n, where k = n - 2is the number of intermediate elements, this leads to

$$2(k + (k - 1) + (k - 2) + \dots + 1) = 2\left(\frac{k(k + 1)}{2}\right) = (n - 2)(n - 1) = n^2 - 3n + 2$$
(3.12)

transitions for projective reordering. This results in a total worst-case transition count of $n^2 - n + 1 + m(2n-1)$.



Figure 3.18: Example for worst case tree of length 5 with respect to SWAP transition, adapted from Coavoux and Crabbé (2017a).

Correctness To show the correctness of the transition system for the set of all ε -free rangelabelled binary trees, completeness and soundness must be shown.

For completeness, it has to be shown that every tree can be retrieved from some transition sequence. Any two nodes in a tree-cut found in S and B can be moved into an adjacent position via SHIFT and SWAP and assigned a parent via REDUCE-X. Furthermore, any single node can be assigned a unary parent using REDUCEUNARY-X after shifting or composing it via REDUCE-X. Thus, from the initial configuration (tree-cut with leaves $\langle 0, 1 \rangle, ..., \langle |w| - 1, |w| \rangle$ for some word w), any ε -free range-labelled binary tree for w can be built.

For soundness, it must be shown that any transition sequence produces a tree. First, it can be seen that a node cannot have several parents. This follows from the fact that the daughter(s) are removed from S when applying REDUCE-X or REDUCEUNARY-X. Furthermore, by definition, a final configuration consists of an empty buffer and only one constituent on the stack. Since the number of constituents can only be reduced via REDUCE-X, it follows that this one constituent spans all input elements. Lastly, by definition of the initialisation function, the leaves are labelled with one-length ranges that cover all indices of the input word. Thus, an ε -free range-labelled binary tree can be retrieved.

The design of the transitions guarantees that at every step at least one transition is possible. When prohibiting swap-loops via the $<_{ind}$ -condition and setting a maximum number of unary chains, the maximum number of transitions is final. This follows directly from the fact that if all possible SWAPs were performed, the end of the buffer was reached and the maximum number of unary chains were assigned to the topmost element of the stack only REDUCE-X could be

performed, reducing the number of elements. Eventually, only one root element is left. Therefore, with these restrictions, the algorithm always reaches a final configuration.

3.2.5 Reordering with a Split Stack and the Gap Action

Coavoux and Crabbé (2017a) introduce a new transition called GAP. Instead of altering the order of two elements as the SWAP transition does, they give up the restriction that only the two topmost elements of the stack can combine. Instead, any element of the stack is allowed to combine with the topmost element.

They achieve this by splitting up the stack into two parts: the stack S and a deque Δ .¹⁸ A dequeue is a sequence that can be accessed from both ends. Reductions can take place between the tops of S and Δ . The GAP transition takes an element from the top of S and adds it to the bottom of Δ . The SHIFT transition is modified so that it pushes all elements from Δ onto S and then pops an element from the buffer onto Δ .

In their original paper, Coavoux and Crabbé (2017a) use a lexicalised model that assigns heads to constituents. In later work they present an unlexicalised approach that will be the basis for this section (Coavoux et al., 2019).

The traditional REDUCE transition is split into a LABEL-X and a MERGE transition. Thus, they call their approach ML-GAP (MERGE-LABEL-GAP). Furthermore, by introducing a NO-LABEL transition, they can drop the requirement that the trees must be binary. Several nodes can be grouped together via MERGE before assigning a common parent via LABEL.

Inspired by span-based parsing for projective trees (Cross and Huang, 2016b), ML-GAP represents nodes by referencing the indices they dominate. In order to account for discontinuities, a set of indices is used instead of a single range. The labelling of the nodes is stored in a separate set K. The final elements of K can be arranged into a hierarchy from which the parse tree can be restored. The following definitions extend the formalisation of Coavoux et al. (2019).

Definition 3.38 (Constituent candidate).

- 1. A set of integers $s \subseteq \{1, ..., n\}$ with $n \in \mathbb{N}$ is called a *constituent candidate (in length n)*.
- 2. The set of all constituent candidates in length $n \in \mathbb{N}$ is referred to by $ConstCand(n) := \mathcal{P}(\{1, ..., n\}).$
- 3. The set of all constituent candidates is referred to by $ConstCand := \mathcal{P}_{fin}(\mathbb{N})$.
- 4. Two constituent candidates s_1, s_2 are called *compatible* iff $(s_1 \subset s_2) \lor (s_2 \subset s_1) \lor (s_1 \cap s_2 = \emptyset)$

Definition 3.39 (Instantiated constituent).

Let N be an alphabet of nonterminal labels.

- 1. A tuple $\langle A, s \rangle$ is called an *instantiated constituent wrt*. N (in length n) iff $A \in N$ and s is a constituent candidate in length n.
- 2. The set of all instantiated constituents wrt. N in length $n \in \mathbb{N}$ is called $Const(N, n) := N \times ConstCand(n)$.
- 3. The set of all instantiated constituents wrt. N is called $Const := N \times ConstCand$.
- 4. $\langle A_1, s_1 \rangle, \langle A_2, s_2 \rangle \in Const(N, n)$ are called *compatible* iff s_1 and s_2 are compatible.

Definition 3.40 (Sets of compatible constituents).

Let N be an alphabet of nonterminals and $n \in \mathbb{N}$. Let $S \subseteq ConstCand(n)$ be a set of constituent candidates in length n.

¹⁸Coavoux and Crabbé (2017a) use D to refer to the deque which I changed to Δ to prevent confusion since I frequently use D to refer to a tree.

- 1. S is called *consistent* iff for all $s_1, s_2 \in S$, s_1 and s_2 are compatible.
- 2. S is called *rooted* iff there exists $s \in S$ such that $s' \subset s$ for all $s' \in S$ with $s \neq s'$. s is called the *root of* S and denoted by root(S) = s.
- 3. S is said to span length n iff for all $i \in \{1, ..., n\}$ there exists some element $s \in S$ such that $i \in s$.
- 4. S is called *complete* for length n iff it is consistent, rooted and spans length n.
- 5. The terms consistent, rooted, spanning length n and complete naturally extend to sets of instantiated constituents.

Definition 3.41 (Maximal subset).

Let $S \subseteq ConstCand(n)$ be a consistent set of constituent candidates for some $n \in \mathbb{N}$. The relation $\underset{\max}{\subset} \subset S \times S$ is defined as follows:

$$s_1 \subset s_2 :\Leftrightarrow s_1 \subset s_2 \land \nexists s_3 \in S : s_1 \subset s_3 \subset s_2, \quad \text{for all } s_1, s_2 \in S$$

The relation is naturally extended to consistent sets of constituents.

Corollary 3.42.

Let $n \in \mathbb{N}$ and $S \subseteq ConstCand(n)$ be a consistent set of constituent candidates. For every $s_1 \in S$ there exists at most one element s_2 such that $s_1 \subset s_2$.

Proof. Let us assume, there are $s_1, s_2, s_3 \in S$ such that $s_1 \underset{\max}{\subset} s_2$ and $s_1 \underset{\max}{\subset} s_3$. Since S is a compatible constituent candidate set, it must hold that either $s_2 \subset s_3$ or $s_3 \subset s_2$ or $s_2 \cap s_3 = \emptyset$. Take some index $i \in s_1$. By assumption also $i \in s_2$ and $i \in s_3$. Now, it cannot be that $s_2 \cap s_3 = \emptyset$. Assume that $s_2 \subset s_3$. This contradicts the assumption that $s_1 \underset{\max}{\subset} s_3$. The reverse also holds true.

Definition 3.43 (Unary-free tree).

Let $D = \langle V, \triangleleft, \hat{v} \rangle$ be a tree. D is called *unary-free* iff for all internal nodes $v \in V$ either v has a daughter that is a leaf or v has at least two daughters.

It can be seen that in a unary-free tree, there are no two internal nodes (i.e. nonterminals) that dominate exactly the same set of leaves. The parent of an internal node must always enlarge the indices covered by the yield and a daughter must always narrow them down. Since the leaves are labelled with ranges, a unary relationship is allowed for pre-terminals to grant the possibility of assigning a nonterminal label spanning a single symbol.

Corollary 3.44 (Equivalence of trees and constituent sets).

1. Let $n \in \mathbb{N}$ and $D = \langle V, \triangleleft, \hat{v}, \lambda \rangle$ a unary-free ε -free range-labelled tree spanning length n. One can define a function ψ converting nodes to instantiated constituents as follows:

$$\psi: V \to Const(N, n), \qquad v \mapsto \langle \lambda(v), Ind_D(v) \rangle,$$

where it holds that:

- (a) $\{\psi(v) \mid v \in V \text{ and } v \text{ is internal}\}$ is complete for length n.
- (b) $v_1 \triangleleft v_2 \Leftrightarrow \psi(v_2) \underset{\max}{\subset} \psi(v_1)$ for all $v_1, v_2 \in V$ where v_1, v_2 are internal.
- 2. Let $C \subseteq Const(N, n)$ be complete for length n. One can construct a unary-free ε -free rangelabelled tree $D = \langle V, \triangleleft, \hat{v}, \lambda \rangle$ spanning length n in the following fashion:

(a)
$$V := root(K) \cup K$$
,

(b) $\hat{v} := root(K),$

$$(c) \triangleleft := \underset{\max}{\subset} {}^{T} \cup \{ \langle c, i \rangle \mid i \in \hat{v}, \begin{pmatrix} \{i\} \in K, c = \{i\} \text{ or} \\ \{i\} \notin K, \{i\} \subset c, c \in K, \nexists c' \in K : (c \neq c' \land \{i\} \subset c' \subset c) \end{pmatrix} \}^{19}$$

(d)
$$\lambda(\langle A, s \rangle) := A$$
 for all $\langle A, s \rangle \in K$
 $\lambda(i) := \langle i - 1, i \rangle$ for all $i \in \hat{v}$.

Now it holds that $c_1 \subset c_2 \Leftrightarrow c_2 \triangleleft c_2$.

It can be seen that the two notations are equivalent and that the conversions preserve the dominance relation. Set notation is useful for transition-based parsers since they do not rely on the range tuples that arise from LCFRS rules. Note that set notation does not allow for two constituents to span the same indices since dominance is expressed via the subset relation. Therefore, trees are not allowed to exhibit chains of unary relations. It is possible to convert any tree to a unary-free tree by collapsing every path of internal unaries $A_0, A_1, ..., A_n$ to a single node $A_0@A_1@...@A_n$ and to convert it back by expanding the chain.²⁰

Differentiating between constituent candidates and instantiated constituents makes it possible for a transition system to construct constituents with more than two daughters using multiple MERGE actions. Every MERGE enlarges the constituent candidate by one additional daughter. A full constituent is only asserted when performing LABEL-X. This is comparable to *implicit binarisation* as described by Gómez-Rodríguez (2014) for *Earley* and *Left-Corner* parsers since the sequence of MERGE transitions implicitly gives rise to a binary derivation. Constituent candidates can be viewed as unordered *dotted items* and instantiated constituents as asserted productions. This is visualised in figure 3.19.



Figure 3.19: Implicit binarisation. The tree on the left exhibits a node with four children; on the right a derivation via MERGE and LABEL-X is shown.

A configuration in ML-GAP comprises of a stack, a deque and a set of label assignments. Instead of a buffer as in SR-SWAP, a simple index is used to keep track of the next element to shift. This is possible since — without a SWAP transition — no element can be moved onto the buffer and thus its elements are guaranteed to respect linear precedence. The stack and the deque hold a partition of the set of indices shifted up to the current point which is equivalent to the tree-cut in SR-SWAP (when concatenating the unread nodes). Figure 3.20 visualises a ML-GAP configuration.

Definition 3.45 (ML-Gap configuration).

Let N be an alphabet of nonterminal labels and Q a set of states. \mathcal{C}^{MLG} is the set of all elements $\langle S, \Delta, i, j, K \rangle : q$ such that

1. $j \in \mathbb{N}$ marks the maximum index,²¹

 $^{{}^{19}}R^T$ for a relation R denotes the converse relation.

 $^{^{20}\}mathrm{Coavoux}$ and Cohen (2019) perform this conversion in the implementation of their parser.

 $^{^{21}}$ Contrary to Coavoux and Crabbé (2017a) I include the right border index as a component of the configuration since I define the set of configurations for all possible input sequences. Therefore, a configuration must contain information as to the number of elements left to shift.

- 2. S and Δ are sequences of constituent candidates in length j,
- 3. $i \in \{1, ..., j\}$ marks the current index,
- 4. $K \subseteq Const(N, j)$ is a set of instantiated constituents²² and
- 5. $q \in Q$ is a state.



Figure 3.20: Illustration of the ML-GAP configuration (without the set of constituents K). Illustration inspired by Coavoux and Cohen (2019).

Coavoux et al. (2019) adopt the strategy of differentiating between *structural* and *labelling* transitions from Cross and Huang (2016b). Depending on the previous action(s), only structural or labelling transitions are possible which narrows down the search space for the neural scorer making predictions easier. The model of Cross and Huang (2016b) simply alternates between structural and labelling actions. To account for this, one can include a step index into the configuration that counts up the total number of transitions applied and represents an even or an odd step.

Here, the GAP transition is utilised to enable merging the top element with an element anywhere on the stack. Therefore, multiple consecutive applications of GAP must be possible. Furthermore, a sequence of GAP transitions should be followed by a MERGE to limit its application to useful cases. Coavoux et al. (2019) define which transitions can follow each other using a *finite state automaton*. For clarity, I integrate this restriction into the transitions and treat the state as part of the configuration. The following definition of *finite state automata* is taken from Hopcroft et al. (2007, chapter 2).

Definition 3.46 (Finite state automaton).

A finite state automaton (FSA) is a 5-tuple $\langle \Sigma, Q, q_0, \delta, F \rangle$ where:

- 1. Σ is an alphabet,
- 2. Q is a finite non-empty set of states,
- 3. $q_0 \in Q$ is called the *initial state*,
- 4. $\delta: Q \times \Sigma \to Q$ is called the *transition function*,
- 5. $F \subseteq Q$ is the set of final states.

Definition 3.47 (MLG FSA).

The FSA for ML-GAP action sequences is defined as $\mathfrak{A}^{MLG} = \langle \mathcal{T}^{MLG}, Q, Struct, \delta, \{Struct\} \rangle$ where

- 1. $\mathcal{T}^{\text{MLG}} = \mathcal{T}^{\text{MLG}}_{\text{struct}} \cup \mathcal{T}^{\text{MLG}}_{\text{label}}$ is the set of ML-GAP transitions with
 - (a) $\mathcal{T}_{struct}^{MLG} = \{SHIFT, GAP, Merge\}$ and
 - (b) $\mathcal{T}_{label}^{MLG} = \{ No-LABEL, LABEL-X \},\$

 $^{^{22}}$ Coavoux et al. (2019) use C to denote the set of instantiated constituents. I changed the symbol to prevent confusion with variables used for configurations.

- 2. $Q = \{$ Struct, Struct', Label $\},\$
- 3. δ is the transition function defined as follows:

$$\begin{split} \delta(\text{Struct}, \text{GAP}) &= \text{Struct'} \\ \delta(\text{Struct}, \text{SHIFT}) &= \text{Label} \\ \delta(\text{Struct}, \text{MERGE}) &= \text{Label} \\ \delta(\text{Struct'}, \text{GAP}) &= \text{Struct'} \\ \delta(\text{Struct'}, \text{MERGE}) &= \text{Label} \\ \delta(\text{Label}, \text{LABEL-X}) &= \text{Struct} \\ \delta(\text{Label}, \text{NO-LABEL}) &= \text{Struct} \end{split}$$

The state diagram for \mathfrak{A}^{MLG} is given in figure 3.21. The current state of the automaton is part of the configuration and a new configuration can only be deduced from a transition if the FSA accepts the transition as input at the given state.



Figure 3.21: Illustration of the FSA for the action sequences allowed in ML-GAP; reproduction of a figure in Coavoux et al. (2019).

The full set of transitions for ML-GAP is given as deduction rules in figure 3.22. The formalisation of the transition system is given in definition 3.48.

Definition 3.48 (Merge-Label-Gap transition system).

A MERGE-LABEL-GAP transition system for an alphabet T and an alphabet of nonterminals N is a 5-tuple $\mathfrak{S}^{\mathrm{MLG}} = \langle \mathcal{C}^{\mathrm{MLG}}, \mathcal{T}^{\mathrm{MLG}}, \sigma, F, d \rangle$ where

- 1. \mathcal{T}^{MLG} is defined for \mathcal{C}^{MLG} as described in figure 3.22,
- 2. $\sigma(w) = \sigma'(|w|) = \langle [], [], 0, |w|, \emptyset \rangle$: Struct for every $w \in T^+$,

3.
$$F = \{ \langle S, \Delta, i, j, K \rangle : q \in \mathcal{C}^{\mathrm{MLG}} \mid i = j \land \exists A \in N : \langle A, \{1, ..., j\} \rangle \in K \}$$
 and

4. d retrieves the tree from K for every $\langle S, \Delta, i, j, K \rangle : q \in F$.

The definition of constituent sets allows to abstract away from symbols of an alphabet of terminals when examining the properties of the transition system. Structurally, all words with the same length are equivalent. It is only for the action scorer that indices must be used as references to individual input items. In the following, when assuming a valid configuration $c = \langle S, \Delta, i, j, K \rangle$:

Corollary 3.49.

For every valid configuration $c = \langle S, \Delta | d_0, i, j, K \rangle : q$ it holds that

$$\forall s \in S \cup \Delta : \max(s) \le \max(d_0).$$

Proof. Let $c = \langle S, \Delta | d_0, i, j, K \rangle$: q be a valid configuration. S and Δ can only contain those elements that were added to the memory by SHIFT. Therefore, all elements are smaller than i or equal. i must have been moved as a single-element set $\{i\}$ to the top of the deque. GAP, LABEL-X and NO-LABEL do not alter the top of the deque. MERGE performs set union on the top of S and the top of Δ but since all those elements are smaller than i, i remains the largest element. Only SHIFT can move a larger element onto the deque top but it also enlarges the buffer index. Therefore $\forall s \in S \cup \Delta : \max(s) \leq i = \max(d_0)$.

Corollary 3.50.

Let N be an alphabet of nonterminals. Every valid configuration $c \in F$ has the form $c = \langle [], [\{1, ..., n\}], n, n, K \rangle$: Struct where K is a rooted set of instantiated constituents with $root(K) = \langle A, \{1, ..., n\} \rangle$ for some $A \in N$.

Proof. Let us assume that $c = \langle S, \Delta, i, j, K \rangle$: $q \in F$ is a valid configuration. From the definition of *F* already follows that i = j and $\exists A \in N$: $\langle A, \{1, ..., j\} \rangle \in K$. Since *j* does not change for any transition, it must hold that $\sigma'(j) = \langle [], [], 0, j, \emptyset \rangle$: Struct $\stackrel{*}{\Rightarrow} \langle S, \Delta, j, j, K \rangle$: *q* and therefore all instantiated constituents in *K* must be composed from elements in the range from 1 to *j* i.e. it holds that $K \subseteq Const(N, j)$. There must exist $c' = \langle S', [\{1, ..., n\}], i', j, K' \rangle$: Label $\in C^{MLG}$ such that $\sigma'(j) \stackrel{*}{\Rightarrow} c' \underset{\text{LABEL-X}}{\Rightarrow} \langle S', [\{1, ..., n\}], i', j, K' \cup \{\langle A, \{1, ..., n\} \rangle\} \rangle$: Struct $\stackrel{*}{\Rightarrow} c$. Since the top of Δ always contains the maximum index of *S* and Δ (cf. corollary 3.49), it must hold that S' = []and i' = j. Therefore, the result of the labelling is already the goal element. Hence, *q* = Struct and there cannot be another instantiated constituent in *K* with indices $\{1, ..., n\}$. Now it holds that for all $\langle A', s' \rangle \in (K \setminus \{\langle A, \{1, ..., n\} \rangle\})$: $s' \subset \{1, ..., n\}$ and thus $root(K) = \langle A, \{1, ..., n\} \rangle$. □

Example 3.51.

Figure 3.24 shows an example parse for the tree in figure 3.23.

Oracle Nodes are necessarily created according to a rightmost index ordering. As Coavoux et al. (2019) note, this stems from the property described by corollary 3.49.²³ From the deque top element, new nodes are established via LABEL-X. For two elements with different rightmost indices, the one with the lower rightmost index is created first. For two nodes in the tree that have the same rightmost index, one necessarily contains less indices than the other and is thus labelled first before applying MERGE once or more and then labelling the other. The following definition is adopted from Coavoux and Cohen (2019):

Definition 3.52 (Rightmost index ordering).

The order \leq_{right} is defined for constituent candidates as follows:

$$s_1 \leq_{\text{right}} s_2 :\Leftrightarrow \begin{cases} \max(s_1) < \max(s_2) \text{ or} \\ \max(s_1) = \max(s_2), s_1 \subseteq s_2. \end{cases}$$

²³A more thorough proof of this property is given for the stack-free successor formalism in corollary 4.8.

$$\begin{array}{ll} axiom & \\ \hline & \\ goal & \langle \ S, \Delta, j, j, K \ \rangle : \text{Struct} & \exists A \in N : \langle A, \{1, ..., j\} \rangle \in K \end{array}$$

Structural actions $\mathcal{T}_{struct}^{MLG}$

$$\begin{array}{l} \text{SHIFT} \quad \frac{\langle S, \Delta, i, j, K \rangle : q}{\langle S | \Delta, [\{i+1\}], i+1, j, K \rangle : \delta(q, \text{SHIFT})} & \delta(q, \text{SHIFT}) \text{ is defined} \\ \\ \text{MERGE} \quad \frac{\langle S | I_{s_0}, \Delta | I_{d_0}, i, j, K \rangle : q}{\langle S | \Delta, [I_{s_0} \cup I_{d_0}], i, j, K \rangle : \delta(q, \text{MERGE})} & \delta(q, \text{MERGE}) \text{ is defined} \\ \\ \text{GAP} \quad \frac{\langle S | I_{s_0}, \Delta, i, j, K \rangle : q}{\langle S, I_{s_0} | \Delta, i, j, K \rangle : \delta(q, \text{GAP})} & \delta(q, \text{GAP}) \text{ is defined} \end{array}$$

Labelling actions $\mathcal{T}^{\rm MLG}_{\rm label}$

$$\begin{array}{ll} \text{LABEL-X} & \frac{\langle \ S, [I_{d_0}], i, j, K \ \rangle : q}{\langle \ S, [I_{d_0}], i, j, K \cup \{\langle X, I_{d_0} \rangle\} \ \rangle : \delta(q, \text{LABEL-X})} & \delta(q, \text{LABEL-X}) \text{ is defined} \\ \text{NO-LABEL} & \frac{\langle \ S, [I_{d_0}], i, j, K \ \rangle : q}{\langle \ S, [I_{d_0}], i, j, K \ \rangle : \delta(q, \text{NO-LABEL})} & \delta(q, \text{NO-LABEL}) \text{ is defined} \end{array}$$





Figure 3.23: Example non-binary tree.

The order extends naturally to instantiated constituents:

$$\langle A_1, s_1 \rangle \leq_{\text{right}} \langle A_2, s_2 \rangle :\Leftrightarrow s_1 \leq_{\text{right}} s_2$$

For binary trees, there exists a bijection between derivations and trees (Coavoux and Crabbé, 2017a). For nodes with more than two daughters in non-binary trees, there exists more than one implicit binarisation. See for instance steps 4 to 9 in figure 3.24 where instead indices 3 and 4 could have been merged first. Therefore, a standard binarisation must be fixed to define an oracle. Coavoux and Crabbé (2017a) define an eager oracle that performs a leftmost binarisation by prioritizing MERGE in all possible cases. Algorithm 3.6 shows their oracle extraction in full detail.

At each step, there is at most one element on the stack that can be retrieved by use of GAP to merge with the top element. Therefore, the index used in the algorithm is unique.

Corollary 3.53.

Let T be an alphabet and $w \in T^+$ a word. Let K_g be a set of instantiated constituents complete for length |w| and $c = \langle S, \Delta | d_0, i, |w|, K \rangle$: Label a configuration in a transition sequence for w licensed by the oracle o extracted by algorithm 3.6 for K_g . Let $p \in K_g$ with $d_0 \underset{\max}{\subset} p$ be the parent of d_0 . For any two indices $m, n \in \{1, ..., |S|\}$ such that $S^{[m]} \underset{\max}{\subset} p$ and $S^{[n]} \underset{\max}{\subset} p$ it holds

step	Stack	Dequeue	index	current state	new constituent	Action
0			0	Struct		Shift
1		$\{1\}$	1	Label		No-Label
2	$\{1\}$	$\{2\}$	2	Struct		Shift
3	$\{1\}$	$\{2\}$	2	Label		No-Label
4	$\{1\}, \{2\}$	$\{3\}$	3	Struct		Merge
5	$\{1\}$	$\{2, 3\}$	3	Label		No-Label
6	$\{1\}$	$\{2, 3\}$	3	Struct		Shift
7	$\{1\}, \{2, 3\}$	$\{4\}$	4	Label		No-Label
8	$\{1\}, \{2, 3\}$	$\{4\}$	4	Struct		Merge
9	$\{1\}$	$\{2, 3, 4\}$	4	Label		Label-A
10	$\{1\}$	$\{2, 3, 4\}$	4	Struct	$\langle \mathbf{A}, \{2,3,4\}\rangle$	Shift
11	$\{1\},\{2,3,4\}$	$\{5\}$	5	Struct		Gap
12	$\{1\}$	$\{2,3,4\},\{5\}$	5	Struct'		Merge
13	$\{2, 3, 4\}$	$\{1, 5\}$	5	Label		LABEL-B
14	$\{2, 3, 4\}$	$\{1, 5\}$	5	Struct	$\langle \mathbf{B}, \{1,5\}\rangle$	Merge
15		$\{1,2,3,4,5\}$	5	Label		LABEL-S
16		$\{1,2,3,4,5\}$	5	Struct	$\langle \mathbf{S}, \{1,2,3,4,5\}\rangle$	

Figure 3.24: Example parse using the GAP transition for the tree in figure 3.14. Pointers are omitted since the nodes are uniquely labelled.

that m = n.

Proof. Let us assume, without loss of generality, that $m \leq n$. There must be some configuration $c' = \langle S', \Delta' | S^{[m]}, i', |w|, K' \rangle$: q' such that $\sigma(w) \stackrel{*}{\Rightarrow} c' \stackrel{*}{\Rightarrow} c$. If $m \neq n$, then $\exists t \in \{1, ..., |S|\}$: $S^{[n]} = S'^{[t]}$. But since $S^{[m]}$ and $S^{[n]}$ have a common parent in K_g , o(c') predicts to merge $S^{[n]}$ and $S^{[m]}$ via t - 1 GAPs and one MERGE transition. Then, neither $S^{[n]}$ nor $S^{[m]}$ can be in S. Therefore, it must hold that m = n.

Complexity As Coavoux and Crabbé (2017a) note, for an input of length n the number of SHIFT transitions is n and the number of MERGE operations n-1. Furthermore, the number of labelling actions is 2n-1 (one for every SHIFT and one after each MERGE transition). This results in a minimum number of 4n-2 transitions.

The number of GAP transitions is variable. In the worst case, the first and the last token have a common parent and the nodes in between incrementally form higher-level constituents as visualised in figure 3.25 (Coavoux and Crabbé, 2017a). Let k = n-2 be the number of intermediate elements, then the number of GAP actions needed is

$$k + (k-1) + (k-2) + \dots + 1 = \frac{k(k+1)}{2} = \frac{(n-2)(n-1)}{2} = \frac{n^2 - 3n + 2}{2}$$
(3.13)

Therefore, the total maximum number of transitions is $\frac{n^2+5n-2}{2}$.

Coavoux and Crabbé (2017a) suggest using COMPOUNDGAP transitions that perform i consecutive GAPs inspired by the COMPOUNDSWAP transition introduced by Maier (2015). Every MERGE transition must be preceded by a COMPOUNDGAP. COMPOUNDGAP₀ has no effect.

When following this proposal, the number of COMPOUNDGAPS is n-1. Then the total number of actions is exactly 5n-3. Ensuring that the number of transitions is equal for every input with length n is a way to counteract the bias of the scorer towards long transition sequences as observed by Crabbé (2014).

Algorithm 3.6 ML-Gap oracle extraction

Input: a sequence w, a gold set of instantiated constituents K_q complete for length |w|**Output:** an oracle O 1: $c \leftarrow \sigma(w)$ 2: $O \leftarrow List[]$ 3: while $c \notin F$ do if c.state = Struct then 4: if $c.\operatorname{stack.top} \subset p$ with $c.\operatorname{dequeue.top} \subset p \in K_g$ then 5: $c \leftarrow \text{MERGE}(c)$ 6: 7: O.append(MERGE)else if $\exists i \in \{1, ..., | c.stack|\} : c.stack^{[i]} \underset{\max}{\subset} p$ with $c.dequeue.top \underset{\max}{\subset} p \in K_g$ then 8: while $i \neq 1$ do 9: $c \leftarrow \text{GAP}(c)$ 10: O.append(GAP)11: $i \leftarrow i - 1$ 12:end while 13:14: else $c \leftarrow \text{SHIFT}(c)$ 15:O.append(SHIFT)16: end if 17:18: else if $\exists \langle A, s \rangle \in K_g : s = c.$ dequeue.top then 19: $c \leftarrow \text{LABEL-A}(c)$ 20:O.append(LABEL-A) 21: else 22: $c \leftarrow \text{NO-LABEL}(c)$ 23: O.append(NO-LABEL) 24:25:end if end if 26:27: end while 28: return O

Correctness Coavoux and Crabbé (2017a) proof the correctness of the transition system for the set of discontinuous binary trees. This result can be generalised for all discontinuous trees. Completeness follows from the correctness of the oracle as outlined above. For soundness, it must be shown that any transition sequence produces a discontinuous tree. First, it can be seen that a node cannot have several parents. This follows directly from the fact that a LABEL-X transition must be followed by a structural action which either enlarges the element or shifts it onto the stack from which it can only be retrieved to unify with another non-empty constituent candidate. Therefore, for any valid configuration, K is a consistent constituent set.

Per design of the transition system, at least one action can be performed at any step. The number of transitions has an upper bound. Thus, the algorithm always reaches a final configuration. The set of instantiated constituents in the final configuration spans the length of the input by definition. Furthermore, from corollary 3.50 it follows that K is rooted and thus K is a complete constituent set wrt. the input.

Comparing with Swap The main benefit of ML-GAP lies in the extended local region in comparison to shift-reduce parsing with the SWAP transition (compare figures 3.5 and 3.20). The model uses three data structures instead of two and places items onto the deque when reordering. In a SWAP model items are pushed back onto the buffer which is why Coavoux et al. (2019) assume that a SWAP parser may have difficulties differentiating between swapped elements and tokens that have not been shifted yet.

Coavoux and Crabbé (2017a) assume SWAP or GAP transitions are the hardest to predict for the respective system and that they are the driving factor for derivation length. The SWAP transition



Figure 3.25: Example of worst case tree of length 5 with respect to the GAP transition as given by Coavoux and Crabbé (2017a).

doubles the number of necessary transitions associated with reordering because it comes with the necessity to shift the swapped element(s) back from the buffer onto the stack in some future step. Shorter derivations are desirable since this reduces the overall uncertainty of the system.

Coavoux and Crabbé (2017a) show that their lexicalised ML-GAP model for binary trees approaches derivation lengths half as long as those of the SR-SWAP model for the worst case longest derivation. This result is confirmed empirically: for the TIGER corpus the average derivation length with respect to the sentence length n is 3.09n for SR-SWAP and 2.03n for ML-GAP which amounts to a 50 % derivation length reduction.

Surprisingly the average length of GAP derivation sequences roughly equals that of a standard shift-reduce parser (2n - 1) (Coavoux and Crabbé, 2017a).

3.2.6 Lexicalisation

Lexicalised models are defined by Coavoux et al. (2019) as models that (a) assign a lexical head to each constituent and (b) use heads of constituents as features to score parsing transitions. They are motivated by lexicalised probabilistic context-free grammars where a nonterminal is annotated with a terminal element representing its lexical head as shown in equation 3.14 (Coavoux et al., 2019).

$$r = VP[saw] \rightarrow VP[saw] PP[telescope]$$
 (3.14)

q(r) then gives the probability of *telescope* modifying *saw*. The general assumption is that information about a constituent's head might help the parser to resolve ambiguities. Following this idea, the original transition-based SHIFT-REDUCE constituent parser of Sagae and Lavie (2005) features two binary REDUCE transitions instead of one:

- REDUCELEFT-X: the head of the left element is assigned as head of the new constituent
- REDUCERIGHT-X: the head of the right element is assigned as head of the new constituent

While this was a long-time standard approach for transition-based constituent parsers, Coavoux et al. (2019) empirically show that unlexicalised models consistently outperform their lexicalised counterparts. Unlexicalised ML-GAP shows an improvement of 0.1 absolute points on the DPTB and as much as 1.1 and 1.3 absolute points on NeGra and TIGER when compared to ML-GAP with lexicalised MERGE transitions. The improvement on discontinuous constituents alone is even greater. Providing the scorer with the predicted heads of the constituents in its domain as a feature has varying effects and in some cases harms the score.

Coavoux et al. (2019) suggest that part of the reason for this result may lie in the head-driven oracle of lexicalised models (cf. the $<_h$ -order for SR-SWAP in equation 3.5). They hypothesise that derivations given by eager oracles are easier to learn. In particular, they observe that lexicalised models require more complicated chains of transitions when constructing constituents with more than two daughters. Unlexicalised models can use implicit left-branching-binarisation and directly start merging elements as soon as the two topmost items have the same parent while lexicalised models wait until the head of a constituent is shifted and then construct the constituent leading to longer chains of transitions. Additionally, Coavoux et al. (2019) show that this leads to a larger average size of stack and deque. They suggest that a smaller stack size is easer to represent as a feature for the scorer and thus benefits the results of the unlexicalised model.

They also find that the overall number of GAP actions, which they assume to be most difficult to predict, reduces for unlexicalised models. This is due to the fact that the lexicalised model waits for a lexical head if it is discontinuous with respect to its k sisters. Then, it needs to perform k chains of GAP transitions to move its sisters to an adjacent position while the unlexicalised model can immediately merge these k elements and then only performs one chain of GAPs. The number of consecutive GAP actions also reduces since intermediate elements might also belong to a constituent with a discontinuous head. The unlexicalised model can reduce them first to a constituent candidate which lowers the length of the GAP chains necessary for the first constituent. Figure 3.26 illustrates this case.



Figure 3.26: Lexicalised discontinuous tree; a head-driven GAP oracle would predict 4 GAP actions to construct A while in an eager unlexicalised approach only 1 GAP action would be needed.

Finally, unlexicalised models are simpler to evaluate for the scorer since they feature fewer transition types. This is in line with the findings of Cross and Huang (2016b) regarding alternating structural and labelling steps to reduce the number of actions for the scorer to choose from at a given step. Additionally, lexicalisation might increase the issue of error propagation. A wrong head assignment that is used as a feature in subsequent steps might encourage further false classifications (Coavoux et al., 2019).

Coavoux et al. (2019) point out that most recent unlexicalised transition-based parsing approaches (Coavoux and Crabbé, 2017b; Cross and Huang, 2016b; Coavoux et al., 2019, among others) are based on a *long short-term memory network (LSTM)*. LSTMs create context-aware representations that enrich individual token representations with information of their surroundings. Coavoux et al. (2019) hypothesise that the LSTM transducer might implicitly learn lexicalised information, which coincides with the result of Kuncoro et al. (2017) that suggests that LSTMs learn head rules when faced with the training objective of jointly learning surface sequences and syntactic trees.

3.2.7 Scorers

Transition-based parsers apply actions based on the predictions of a scorer. The scorer receives features associated with certain elements in the configuration. It usually takes the form of a machine learning model that is trained to predict the optimal sequence of transitions given by the oracle.

Versley (2014) use a linear classifier with a set of features that include (among others) POS tags, word forms, lemmas and morphological tags in a two-token window around the two nodes in

consideration, as well as the category and POS tag of the leftmost and rightmost dependent of the nodes themselves. Coavoux et al. (2019) successfully use the constituents in the domain of locality of their transition system (stack top, deque top, buffer top) as well as the second topmost element of both stack and deque.

Recent proposals employ a concatenation of word embeddings and character-aware embeddings to represent leaves (Coavoux et al., 2019). The vectors are then enriched with contextual information over the whole input sequence prior to parsing using a recurrent neural network. Constituents are represented using their leftmost and their rightmost leaf representation. Stanojević and Alhama (2017) pioneered using neural networks as a classifier. Most importantly, they utilise the whole configuration as a feature to a recurrent neural network for deciding the next parsing action instead of limiting themselves on the elements in the local region of the shift-reduce framework.

4. Parsing with a Stack-Free Transition System

Coavoux and Cohen (2019) present a novel approach to transition-based parsing by replacing the configuration stack with an unordered random-access set. The parser is able to consider every constituent in this memory regardless of its position in the sentence with a single action to construct a new constituent. This omits the use of a SWAP or a GAP transition for reordering and allows for the derivation of a discontinuous tree over an input sequence of length n in 4n - 2 steps.

The role of the stack top in the traditional shift-reduce approach is performed by a separate *focus item*. As in the GAP approach, a buffer of unread tokens is implemented in the form of an index that is counted up when a new token gets retrieved.

The model is a direct advancement of the ML-GAP model of Coavoux et al. (2019) and likewise driven by the idea to enable merging a focus element with any item in the memory without the need to move intermediate items back onto the buffer. It continues the trend of enlarging the domain of locality a transition has access to.

4.1. Configurations

Definition 4.1 (Stack-free configuration).

Let N be an alphabet of nonterminal labels. C^{SC} denotes the set of all elements of the form $\langle S, s_f, i, j, K \rangle : n$ such that

- 1. $j \in \mathbb{N}$ with $j > 1,^{24}$
- 2. $S \subseteq ConstCand(j-1)$ is called the memory of c,
- 3. $s_f \in ConstCand(j-1)$ is called the focus item of c,
- 4. $i \in \{1, ..., j\}$, marks the next index,²⁵
- 5. $K \subseteq Const(N, j 1)$ is a set of instantiated constituents and
- 6. $n \in \mathbb{N}_0$ marks the step count.

S is the set that replaces the traditional stack while *i* represents the index of the next token in the buffer. It differs from ML-GAP where it represented the most recent index shifted onto the memory. s_f is a separate focus element. K is the set of constituents labelled so far. A counter *n* encodes the current step count. Figure 3.5 visualises the configuration. Note the enlarged local region when compared to the SR-SHIFT and the ML-GAP configurations in figures 3.5 and 3.20.

4.2. Transitions

A set of deduction rules dictates the construction of new items. SHIFT takes the next item off the buffer and makes the one-item set containing it the new focus item. The old focus item is added to the memory set. COMBINE-s removes s from the set and forms a union with the focus item creating a new focus item. The LABEL-X transition is used to declare the focus item s_f an instantiated constituent $\langle X, s_f \rangle$ with some label X. NO-LABEL has no effect. The full list of deductions is given in figure 4.2.

The unordered set allows for the focus item to be combined with any possibly non-adjacent constituent candidate present in the memory. Therefore, there is no need for a reordering transition.

 $^{^{24}}$ Just like in definition 3.45 I include the desired maximal index as a component of the configuration since I define the set of configurations for all possible input sequences. Therefore, a configuration must contain information as to the number of elements left to shift.

 $^{^{25}}$ Coavoux and Cohen (2019) start indexing sequences with 0. I adjusted the definition to the convention of 1 in this work.



Figure 4.1: Illustration of the stack-free configuration (without the set of initiated constituents). Reproduction of an illustration in Coavoux and Cohen (2019).

$$\begin{array}{ll} axiom & \overline{\langle \ \emptyset, \emptyset, 1, |w| + 1, \emptyset \ \rangle : 0} \\ goal & \langle \ S, s_f, j, j, K \ \rangle : n & \exists A \in N : \langle A, \{1, ..., j\} \rangle \in K \end{array}$$

Structural actions $\mathcal{T}_{\text{struct}}^{\text{SC}}$

$$\begin{array}{ll} \text{SHIFT} & \frac{\langle \ S, s_f, i, j, K \ \rangle : n}{\langle \ S \cup \{s_f\}, \{i\}, i+1, j, K \ \rangle : n+1} & i < j, \ n \text{ is even} \\ \text{COMBINE-s} & \frac{\langle \ S, s_f, i, j, K \ \rangle : n}{\langle \ S \setminus \{s\}, s_f \cup s, i, j, K \ \rangle : n+1} & s \in S, \ n \text{ is even} \end{array}$$

Labelling actions \mathcal{T}_{label}^{SC}

$$\begin{array}{ll} \text{LABEL-X} & \frac{\langle \ S, s_f, i, j, K \ \rangle : n}{\langle \ S, s_f, i, j, K \ \rangle : n + 1} & n \text{ is odd} \\ \text{No-LABEL} & \frac{\langle \ S, s_f, i, j, K \ \rangle : n}{\langle \ S, s_f, i, j, K \ \rangle : n + 1} & (i \neq j \lor S \neq \emptyset), n \text{ is odd} \end{array}$$

Figure 4.2: Transitions for the set-free system, $\mathcal{T}^{SC} = \mathcal{T}^{SC}_{struct} \cup \mathcal{T}^{SC}_{label}$

The need for a NO-LABEL-transition arises from the fact that Coavoux and Cohen (2019) divide the transitions into *structural* (SHIFT, COMBINE-s) and *labelling* (LABEL-X, NO-LABEL) actions. Contrary to the GAP proposal described in section 3.2.5 and true to the original proposal by Cross and Huang (2016b), a structural action is always followed by a labelling action and vice versa. Therefore, a configuration carries a step counter that, if it is even, allows for a structural transition to follow, or if it is odd, allows for a labelling transition.

Definition 4.2 (Stack-free transition system).

A stack-free or SHIFT-COMBINE transition system for an alphabet T and an alphabet of nonterminals N is a 5-tuple $\langle \mathcal{C}^{SC}, \mathcal{T}^{SC}, \sigma, F, d \rangle$ where

1. \mathcal{T}^{SC} is defined on \mathcal{C}^{SC} as shown in figure 4.2,

2.
$$\sigma(w) = \sigma'(|w|) = \langle \emptyset, \emptyset, 1, |w| + 1, \emptyset \rangle : 0 \text{ for every } w \in T^+,$$

3.
$$F = \{ \langle S, s_f, i, j, K \rangle : n \in \mathcal{C}^{\mathrm{SC}} \mid i = j \land \exists A \in N : \langle A, \{1, ..., i\} \rangle \in K \}^{26} \text{ and }$$

4. *d* retrieves the tree from *K* for every $\langle S, s_f, i, j, K \rangle : n \in F$.

Corollary 4.3.

Let $c = \langle S, s_f, i, j, K \rangle$: *n* be a valid configuration.

²⁶Coavoux and Cohen (2019) do not require that K contain an instantiated constituent spanning $\{1, ..., i\}$ but instead that j = 4n - 2. One follows from the other as can be seen in section 4.4.

- 1. It holds that $\max(s_f) = i 1.^{27}$
- 2. Let $c' = \langle S', s'_f, i', j', K' \rangle : n' \in \mathcal{C}^{SC}$ be a configuration such that $c \Rightarrow c'$. It holds that $\max(s_f) \leq \max(s'_f)$.

Proof. Part 1. follows from the fact that the focus item cannot be swapped with a set item. It can only be enlarged with items from the set or shifted. Therefore, given the first buffer item i, i - 1 is always a member of the focus item, and therefore $\max(s_f) = i - 1$. Now, if SHIFT is performed, the new maximum focus element is larger than the preceding one. If another action is performed, it remains equal. Hence, 2. is true.

Corollary 4.4.

For every valid configuration $c = \langle S, s_f, i, j, K \rangle : n$ it holds that $\forall s \in S : \max(s) < s_f$.

Proof. For every $s \in S$ there must exists a configuration $c' = \langle S', s, i', j, K' \rangle : n' \in \mathcal{C}^{SC}$ such that $c' \underset{\text{SHIFT}}{\Rightarrow} \dots \underset{\text{C}}{*} c$. Therefore, following corollary 4.3, $\max(s) \leq \max(s_f)$. Furthermore, since SHIFT was applied, i' < i and thus $\max(s) < \max(s_f)$.

Corollary 4.5.

Given an alphabet of nonterminals N and valid configurations $c_1 = \langle S_1, s_f, i_1, j_1, K_1 \rangle : n_1, c'_1 = \langle S_1, s_f, i_1, j, K_1 \cup \{\langle A, s_f \rangle\} \rangle : n_1 + 1$ with $A \in N$ such that $c_1 \underset{\text{LABEL-A}}{\Rightarrow} c'_1$, there are no configurations $c_2 = \langle S_2, s_f, i_2, j, K_2 \rangle : n_2, c'_2 = \langle S_2, s_f, i_2, j, K_2 \cup \{\langle A', s_f \rangle\} \rangle : n_2 + 1$ with $A' \in N$ such that $c'_1 \underset{\text{LABEL-A}}{\Rightarrow} c_2 \underset{\text{LABEL-A}}{\Rightarrow} c'_2$, i.e. no set can be assigned a label twice.

Proof. Let us assume that $c \xrightarrow[LABEL-A]{} c' \stackrel{*}{\Rightarrow} c_2 \xrightarrow[LABEL-A]{} c'_2$. LABEL-A must be followed by a structural transition. If MERGE-s follows, then s_f gets merged with some non-empty s and LABEL-A' cannot be applied to s_f any more. If SHIFT follows, the maximum focus index gets increased by one and for any following configuration, it does not decrease to $\max(s_f)$ (corollary 4.3). Therefore, c_2 is not derivable from c'.

Example 4.6.

Figure 4.3 shows an example parse. Note the ambiguity: the constituent A could have also been constructed starting with {3,4}, i.e. by performing the sequence SHIFT, NO-LABEL, COMBINE-{3}, NO-LABEL, COMBINE-{2} starting at step 6.

4.3. Oracle

Coavoux and Cohen (2019) present a static oracle as well as a dynamic oracle. The latter represents the first dynamic oracle suggested for transition-based discontinuous constituent parsing.

4.3.1 Static Oracle

Coavoux and Cohen (2019) design an eager static oracle with a priority on applying COMBINE-s to keep the number of elements in the set as small as possible. A smaller memory means fewer choices for COMBINE-X actions in the future which facilitates decisions and makes scoring faster.

Algorithm 4.1 describes the static oracle extraction for the stack-free-transition system. It is a direct successor of the ML-GAP oracle given in algorithm 3.6. Instead of a chain of GAP transitions followed by a MERGE it simply predicts one COMBINE-s transition. Corollary 3.53 translates to the stack-free approach: there is at most one sister to combine with the focus item present in the set at any step.

 $^{^{27}\}mathrm{Coavoux}$ and Cohen (2019) make this a condition on all stack-free configurations.

step	set	focus	index	new constituent	action
0			1		Shift
1		1	2		No-Label
2		1	2		Shift
3	$\{1\}$	2	3		No-Label
4	$\{1\}$	2	3		Shift
5	$\{1\}, \{2\}$	3	4		No-Label
6	$\{1\}, \{2\}$	3	4		Combine- $\{2\}$
7	$\{1\}$	2, 3	4		No-Label
8	$\{1\}$	2,3	4		Shift
9	$\{1\}, \{2,3\}$	4	5		No-Label
10	$\{1\}, \{2, 3\}$	4	5		Combine- $\{2, 3\}$
11	$\{1\}$	2, 3, 4	5		LABEL-A
12	$\{1\}$	2, 3, 4	5	$\langle \mathbf{A}, \{2,3,4\}\rangle$	Shift
13	$\{1\},\{2,3,4\}$	5	6		No-Label
14	$\{1\},\{2,3,4\}$	5	6		Combine- $\{1\}$
15	$\{2, 3, 4\}$	1, 5	6		LABEL-B
16	$\{2, 3, 4\}$	1, 5	6	$\langle B, \{1,5\} \rangle$	Combine- $\{2, 3, 4\}$
17		1, 2, 3, 4, 5	6		LABEL-S
18		1, 2, 3, 4, 5	6	$\langle S, \{1,2,3,4,5\}\rangle$	

Figure 4.3: Example parse using the stack-free transition system for the tree in figure 3.14. The columns *set* and *focus* represent sets with the outer brackets being are omitted.

4.3.2 Dynamic Oracle

A dynamic oracle predicts the best actions to perform for any valid configuration — not only the configurations part of the gold path to the desired tree. This is motivated by the fact that training a neural classifier only on the gold scenario limits the quality of information provided at training. When taking a wrong turn by choosing a bad transition the parser should still be able to construct the best possible parse tree even if it is no longer on the path to the perfect prediction.

This directly addresses the issue of *error propagation*. The term refers to a common problem with incremental models where predictions are dependent on previous decisions. It describes the tendency of an initial erroneous assignment to evoke more errors in subsequent steps. McDonald and Nivre (2007) first observed this tendency in transition-based parsers.

The idea of a dynamic oracle was first brought forward by Goldberg and Nivre (2012b) for dependency parsing. Their oracle outputs a set of best actions to perform and is therefore nondeterministic. Dynamic oracles for projective constituent parsing were proposed by a number of authors, for instance Cross and Huang (2016b). The dynamic oracle for discontinuous constituent parsing proposed by Coavoux and Cohen (2019) is an extension of this work. The following prerequisites are needed.

Definition 4.7 (Reachability).

Let N be an alphabet of nonterminal labels.

1. Let $c = \langle S, s_f, i, j, K \rangle : n \in C^{SC}$ be a configuration. An instantiated constituent $\langle A, s \rangle \in Const(N)$ such that $\langle A, s \rangle \notin K$ is said to be *reachable* from c iff there exists a configuration $c' = \langle S', s'_f, i', j, K' \rangle$ such that $\langle A, s \rangle \in K'$ and $c \stackrel{*}{\Rightarrow} c'.^{28}$

²⁸In contrast to Coavoux and Cohen (2019) I explicitly require that $c \notin K$. This facilitates the following definitions.

Algorithm 4.1 Stack-free static oracle extraction

Input: a word w, a gold set of instantiated constituents K_q complete for length |w|**Output:** an oracle O 1: $c \leftarrow \sigma(w)$ 2: $O \leftarrow List[]$ 3: while $c \notin F$ do if *c*.step is even then 4: if $\exists s \in c.$ memory : $s \underset{\max}{\subset} p$ with c.focus $\underset{\max}{\subset} p \in K_g$ then 5: $c \leftarrow \text{COMBINE-}s(c)$ 6: 7: O.append(COMBINE-s)8: else $c \leftarrow \text{Shift}(c)$ 9: O.append(SHIFT) 10: end if 11: 12: else if $\exists \langle \mathbf{A}, s \rangle \in K_g : c.$ focus = s then 13: $c \leftarrow \text{LABEL-}A(c)$ 14:O.append(LABEL-A)15:else 16: $c \leftarrow \text{NO-LABEL}(c)$ 17:18: O.append(No-LABEL) end if 19:20:end if 21: end while 22: return O

2. The function reach is defined as follows:

 $reach: \mathcal{C}^{\mathrm{SC}} \times \mathcal{P}(Const(N)) \to \mathcal{P}(Const(N)),$ $\langle c, K \rangle \mapsto \{k \in K \mid k \text{ is reachable from } c\}.$

In other words, for a set of instantiated constituents K wrt. N in any length, *reach* gives all of the constituents in K that are reachable from a configuration $\langle S, s_f, i, j, K \rangle : n$. It is straightforward to see that only constituents in length j-1 are reachable.

Corollary 4.8.

Let N be an alphabet of nonterminal labels, $\langle A, s \rangle, \langle A', s' \rangle \in Const(N, j - 1)$ for some $j \in \mathbb{N}$ instantiated constituents with $\langle A, s \rangle \leq_{\text{right}} \langle A', s' \rangle$ and $s \neq s'$ and $c_b = \langle S_b, s_b, i_b, j, K_b \rangle : n_b \in \mathcal{C}^{\text{SC}}$ a valid configuration with $\langle A, s \rangle, \langle A', s' \rangle \in K_b$. Then it follows that there exist valid configurations $\langle S, s, i, j, K \rangle : n, \langle S', s', i', j, K' \rangle : n' \in \mathcal{C}^{\text{SC}}$ such that $\langle S, s, i, j + 1, K \rangle : n \underset{\text{LABEL-A}}{\Rightarrow} \langle S, s, i, j + 1, K' \cup \{\langle A', s' \rangle\} \rangle :$ $n' + 1 \stackrel{*}{\Rightarrow} c_b$. In other words: $\langle A, s \rangle$ must be constructed before $\langle A', s' \rangle$.

Proof. Let us assume the contrary: there is a transition sequence that contains c_b , that does not fulfil this property. By definition of the transitions in figure 4.2: if $\langle A, s \rangle, \langle A', s' \rangle \in K_b$, there must exist $c = \langle S, s, i, j+1, K \rangle : n, c_l = \langle S, s, i, j+1, K \cup \{\langle A, s \rangle\}\rangle : n+1, c' = \langle S', s', i', j+1, K' \rangle : n'$ and $c'_l = \langle S', s', i', j, K' \cup \{\langle A's' \rangle\} \rangle : n' + 1$ such that $\sigma'(j-1) \stackrel{*}{\Rightarrow} c \stackrel{\Rightarrow}{\underset{\text{LABEL-A}} c_l \stackrel{*}{\Rightarrow} c_b$ and $\sigma'(j-1) \stackrel{*}{\Rightarrow} c' \stackrel{\Rightarrow}{\underset{\text{LABEL-A}}, c'_l \stackrel{*}{\Rightarrow} c_b$. By assumption, it cannot be that $c_l \stackrel{*}{\Rightarrow} c'$, so either 1. the variables denote the same transition or 2. $\sigma'(j-1) \stackrel{*}{\Rightarrow} c' \stackrel{\Rightarrow}{\underset{\text{LABEL-A}} c_l \stackrel{*}{\Rightarrow} c_b$.

- 1. This contradicts the corollary assumption that $s \neq s'$.
- 2. Following corollary 4.3 it must hold that $\max(s') \le \max(s)$. Now, there are two cases: (a) $\max(s') < \max(s)$ or (b) $\max(s') = \max(s)$

- (a) This contradicts the assumption that $\langle A, s \rangle \leq_{\text{right}} \langle A', s' \rangle$.
- (b) First, i = i' (corollary 4.3). By definition 3.52 it must hold that $s \subseteq s'$. Since $s \neq s'$ by assumption it follows that $s \subset s'$. Thus, there exists $x \in s'$ such that $x \notin s$. This contradicts with $c' \stackrel{*}{\Rightarrow} c$ since there is no structural action to achieve this. Thus, this also contradicts the assumptions and the corollary is true.

In their statement about the property in corollary 4.8, Coavoux and Cohen (2019) do not require that $s \neq s'$. But s = s' would trivially mean that the variables denote the same nonterminal and the same configuration since no set can be assigned a label twice (corollary 4.5).

Coavoux and Cohen (2019) note the following:

Corollary 4.9.

Given an alphabet of nonterminals N, a valid configuration $c = \langle S, s_f, i, j, K \rangle : n \in \mathcal{C}^{SC}$ where n is odd and an instantiated constituent $\langle A, s_g \rangle \in Const(N, j - 1)$ with $\langle A, s_g \rangle \notin K$. $\langle A, s_g \rangle$ is reachable from c iff:

- 1. $\max(s_f) \leq \max(s_g)$ and
- 2. $\forall s \in S \cup \{s_f\} : (s \subseteq s_g) \text{ or } (s \cap s_g = \emptyset).$

Proof.

- (⇒) From the reachability of $\langle A, s_g \rangle$ follows that there exists a configuration $c' = \langle S', s'_f, i', j', K' \rangle$: n' such that $\langle A, s_g \rangle \in K'$ and $c \stackrel{*}{\Rightarrow} c'$. Since $\langle A, s_g \rangle \notin K$, it follows that there exist configurations $c_g = \langle S_g, s_g, i_g, j, K_g \rangle$: n_g and $c_{g'} = \langle S_g, s_g, i_g, j, K_g \cup \{s_g\} \rangle$: $n_g + 1$ such that $c \stackrel{*}{\Rightarrow} c_g \stackrel{\Rightarrow}{\underset{\text{LABEL-A}}{\Rightarrow}} c_{g'}$. Condition 1. now follows directly from corollary 4.3. Furthermore, if there was an element $s \in S \cup \{s_f\}$ such that there was an index $x \in s \cap s_g$ and an index $y \in s \setminus s_g, \langle A, s_g \rangle$ would not be reachable from c, since MERGE-s would be necessary to build s_g but MERGE-s would also make y an element in the focus set. Therefore condition 2 must also hold.
- (⇐) Either (a) $\max(s_f) = \max(s_g)$ or (b) $\max(s_f) < \max(s_g)$.
 - (a) By condition 2 it must hold that $s_f \subseteq s_g$. Let $\{y_1, ..., y_m\} = s_g \setminus s_f$. No element in $s_g \setminus s_f$ was introduced by a SHIFT following c since $\max(s_f) = \max(s_g)$. Therefore it holds that $\exists Y_z \in S : y_z \in Y_z$ for all $z \in \{1, ..., m\}$. Due to condition 2, it must hold that $Y_z \subseteq s_g$ for every $z \in \{1, ..., m\}$. It holds that $s_f \cup Y_1 \cup ... \cup Y_m = s_g$. One can construct a path: $c \rightleftharpoons_{\lambda_1} ... \rightleftharpoons_{COMBINE-Y_1} \langle S \setminus \{Y_1\}, s_f \cup Y_1, i, j, K' \rangle : n + 2 \rightleftharpoons_{\lambda_2} ... \rightleftharpoons_{COMBINE-Y_m} \langle S \setminus \{Y_1, ..., Y_m\}, s_g, i, j, K'' \rangle : n + 2m \rightleftharpoons_{LABEL-A} \langle S \setminus \{Y_1, ..., Y_m\}, s_g, i, j, K \cup \langle A, s_g \rangle \rangle : n + 2m + 1$ with $\lambda_1, ..., \lambda_m \in \mathcal{T}_{label}^{SC}$. Therefore $\langle A, s_g \rangle$ is reachable from c. If $s_f = s_g$, then simply m = 0.
 - (b) For all $s \in S$ it must hold that $\max(s) < \max(s_f)$ (corollary 4.4). Therefore, all elements $i, i+1, ..., \max(s_g)$ must be shifted from the buffer to construct s_g . Now construct a path: $c \rightleftharpoons_{\lambda_1} c' \rightleftharpoons_{\text{SHIFT}} \langle S \cup \{s_f\}, \{i\}, i+1, j, K' \rangle : n+2 \rightleftharpoons_{\lambda_2} ... \rightleftharpoons_{\text{SHIFT}} \langle S \cup \{s_f, \{i\}, ..., \{\max(s_g) 1\}\}, \{\max(s_g)\}, \max(s_g) + 1, j, K'' \rangle : n+2(\max(s_g) \max(s_f)) = c' \text{ with } \lambda_1, ..., \lambda_{\max(s_g) \max(s_f)} \in \mathcal{T}_{\text{label}}^{\text{SC}}$. It holds that $\forall s \in \{\{i\}, ..., \{\max(s_g) 1\}\} : (s \subseteq s_g) \text{ or } s \cap s_g = \emptyset$ since s has only one element. Now, for c' it follows by case (a), that $\langle A, s_g \rangle$ is reachable from c'. Therefore $\langle A, s_g \rangle$ is also reachable from c.

Corollary 4.10.

Let N be an alphabet of nonterminals, and $c = \langle S, s_f, i, j, K \rangle : n \in C^{SC}$ a valid configuration where n is even. Let $\langle A, s_g \rangle \in Const(N, j - 1)$ be an instantiated constituent with $\langle A, s_g \rangle \notin K$. $\langle A, s_g \rangle$ is reachable from c iff:

- 1. $\max(s_f) \le \max(s_g),$
- 2. $\forall s \in S \cup \{s_f\} : (s \subseteq s_q) \text{ or } (s \cap s_q = \emptyset) \text{ and }$
- 3. $s_f \neq s_g$.

Proof. This is straightforward to see following corollary 4.9 in combination with the fact that the current focus item cannot be labelled since only SHIFT and COMBINE are allowed. \Box

Following Cross and Huang (2016b), Coavoux and Cohen (2019) define a function to retrieve the smallest (with respect to \leq_{right}) reachable gold constituent. I adapted the function to the notation used here:

Definition 4.11 (Next function).

Let N be an alphabet of nonterminal labels. The function next is defined as follows:

$$\begin{aligned} \{\langle c, K_g \rangle \mid \exists m \in \mathbb{N} : \sigma'(m) \xrightarrow{\sim} c, c \notin F, K_g \subseteq Const(N, m), K_g \text{ is complete for } m\} \to Const(N) \\ \langle c, K_g \rangle \mapsto \underset{\leq_{\text{right}}}{\operatorname{arg\,min}} reach(c, K_g) \end{aligned}$$

Restricting the function to valid but not final configurations for an input length m and to sets of constituents complete for length m guarantees that at least $\{1, ..., m\} \in reach(c, K_g)$. Furthermore, since K_g is consistent, the elements in $reach(c, K_g)$ are pairwise-compatible and therefore a minimal element exists. Thus, next is well-defined.

Now, the oracle is defined separately for odd and for even steps:

Definition 4.12 (Dynamic oracle odd).

Let N be an alphabet of nonterminals, $m \in \mathbb{N}$ an integer, K_g a complete set of instantiated constituents wrt. N for length m and $c = \langle S, s_f, i, m+1, K \rangle$: n a configuration such that n is odd and $\sigma'(m) \stackrel{*}{\Rightarrow} c$.

$$o_{odd}(c, K_g) = \begin{cases} \{\text{LABEL-}A\}, & \text{if } \exists \langle A, s_f \rangle \in K_g, \\ \{\text{NO-LABEL}\}, & \text{otherwise.} \end{cases}$$

Definition 4.13 (Dynamic oracle even).

Let N be an alphabet of nonterminals, $m \in \mathbb{N}$ an integer, K_g a complete set of instantiated constituents wrt. N for length m and $c = \langle S, s_f, i, m+1, K \rangle$: n a configuration such that n is even and $\sigma'(m) \stackrel{*}{\Rightarrow} c$.

$$o_{even}(c, K_g) = \begin{cases} \{\text{COMBINE-}s \mid (s_f \cup s) \subseteq s'_f\}, & \text{if } \max(s'_f) = \max(s_f), \\ \{\text{COMBINE-}s \mid (s_f \cup s) \subseteq s'_f\} \cup \{\text{SHIFT}\}, & \text{if } \max(s'_f) > \max(s_f), \end{cases}$$

where $next(c, K_g) = \langle A, s'_f \rangle$.

For training, Coavoux and Cohen (2019) make the oracle deterministic by choosing COMBINE at an even step if the oracle allows for it. This is done to keep the stack as small as possible which reduces the number of elements future COMBINE actions need to evaluate. If several COMBINE actions are possible, the element with the highest right index is merged. This is done to reduce the number of non-adjacent parts in the new set which could potentially be impossible to represent uniquely by the four index constituent embedding representation (see section 4.5.2).

4.3.3 Oracle Correctness

Coavoux and Cohen (2019) proof that the oracle described in definitions 4.12 and 4.13 always leads to the reachable tree with the highest *F*-score with respect to a gold tree for any configuration. They give the following definitions for *precision*, *recall* and *F*-score:

Definition 4.14 (F, precision, recall).

Let T be an alphabet of terminals, N an alphabet of nonterminals, $w \in T^+$ a word and K_g, \hat{K} sets of instantiated constituents wrt. N complete for length |w|. K_g is the gold tree and \hat{K} the predicted tree.

$$\operatorname{precision}(\hat{K}, K_g) = \frac{|K \cap K_g|}{|\hat{K}|}$$
$$\operatorname{recall}(\hat{K}, K_g) = \frac{|\hat{K} \cap K_g|}{|K_g|}$$
$$\operatorname{F}_1(\hat{K}, K_g) = \frac{2 \cdot \operatorname{precision}(\hat{K}, K_g) \cdot \operatorname{recall}(\hat{K}, K_g)}{\operatorname{precision}(\hat{K}, K_g) + \operatorname{recall}(\hat{K}, K_g)}$$

Precision, recall and F_1 -score are common evaluation metrics for information retrieval. Precision measures the percentage of identified constituents that are actually correct (i.e. that are gold constituents). In other words: it rises when the number of *false positives* goes down. Recall measures the percentage of gold constituents that were identified by the parser. It increases when the number of *false negatives* reduces. F_1 -score is a means to combine the two metrics into one measure (Jurafsky and Martin, 2009, chapter 13.5.3).

These concepts are related to the notions of soundness, completeness and correctness introduced in section 3.1.5. Proving that the dynamic oracle is optimal for precision and recall also entails that the oracle is correct in a static scenario. A proof can be performed by showing optimality both for odd and even steps.

Coavoux and Cohen (2019) give the following proof: By definition, for a configuration $c = \langle S, s_f, i, j, K \rangle$: n and a gold tree K_g , $o_{odd}(c, K_g)$ is optimal for precision because it only predicts a LABEL-A-action if the constituent $\langle A, s_f \rangle$ is actually in the gold tree K_g . It does not add to the number of false positives. Furthermore, it is optimal for recall since, if the current focus is a gold constituent, it will assign a label to it. It does not add to the number of false negatives by omitting a label.

 $o_{even}(c, K_g)$ is optimal for precision, since it does not label new constituents at all, so it cannot add to the number of false positives. For its relationship with recall, the question stands whether its prediction can have the effect of prohibiting reaching a gold constituent. Coavoux and Cohen (2019) show the following:

Corollary 4.15.

Given an alphabet of nonterminals N, a valid configuration $c = \langle S, s_f, i, j, K \rangle$: n such that n is even and a set of instantiated constituents complete for length j - 1 called K_g , it holds that $\forall k_g \in K_g : k_g \in \operatorname{reach}(c, K_g) \Rightarrow (\forall c' \in o_{even}(c, K_g) : k_g \in \operatorname{reach}(c', K_g))$, i.e. after any transition in the oracle's even prediction, k_g will still be reachable.

Proof. Assume that $next(c, K_g) = \langle A', s_h \rangle$ and that $\langle A, s_g \rangle \in K_g$ is a constituent reachable from c. There are two cases to explore, based on the definition of o_{even} : 1. SHIFT or 2. COMBINE-s.

1. If SHIFT $\in o_{even}(c, K_g)$, then $\max(s_h) > \max(s_f)$. Following corollary 4.10, it holds that $\max(s_f) \le \max(s_g)$ as well as $\forall s \in S \cup \{s_f\} : (s \subseteq s_g)$ or $(s \cap s_g = \emptyset)$. Furthermore $s_g \neq s_f$. Let us assume that $\langle A, s_g \rangle$ is not reachable from SHIFT $(c) = \langle S \cup \{s_f\}, \{i\}, i+1, j, K \rangle : n+1$. Then either (a) $\max(\{i\}) > \max(s_g)$ or (b) $\exists s \in S \cup \{s_f, \{i\}\} : (s \notin s_g)$ and $(s \cap s_g \neq \emptyset)$ (corollary 4.9).

- (a) This entails that $\max(s_g) = \max(s_f) < \max(s_h)$. But then $s_g \leq_{\text{right}} s_h$ which contradicts the assumption that $\operatorname{next}(c, K_g) = \langle A', s_h \rangle$.
- (b) Per assumption this can only be true for $\{i\}$. From $\{i\} \cap s_g \neq \emptyset$ follows $\{i\} \subseteq s_g$. This contradicts $\{i\} \not\subseteq s_g$. Therefore, $\langle A, s_g \rangle$ must be reachable from SHIFT(c).
- 2. For any COMBINE- $s_b \in o_{even}(c, K_g)$: per oracle definition it holds that $(s_f \cup s_b) \subseteq s_h$. Since $\langle A', s_h \rangle$ is also reachable from c, it must hold that s_g and s_h are compatible or $s_g = s_h$. Four cases:
 - (a) $s_q = s_h$. Then $\langle A, s_q \rangle$ is trivially also reachable from COMBINE- $s_b(c)$.
 - (b) $s_h \subset s_g$. Therefore, also $\forall s \in (S \setminus \{s_b\}) \cup \{s_f \cup s_b\} : (s \subseteq s_g) \text{ or } (s \cap s_g = \emptyset)$. Therefore, by corollary 4.9, $\langle A, s_g \rangle$ is still reachable from COMBINE- $s_b(c)$.
 - (c) $s_g \subset s_h$. It follows that $s_g \leq_{\text{right}} s_h$. But then, $next(c, K_g) \neq \langle A', s_h \rangle$ since s_g is smaller than s_h which contradicts the definition of s_h .
 - (d) $s_g \cap s_h = \emptyset$. Then also $s_f \cap s_g = \emptyset$ and $s \cap s_g = \emptyset$. Therefore $\forall s \in (S \setminus \{s_b\}) \cup \{s_f \cup s_b\} :$ $(s \subseteq s_g) \lor (s \cap s_g = \emptyset)$ and thus, by corollary 4.9, $\langle A, s_g \rangle$ is still reachable from COMBINE- $s_b(c)$.

4.4. Complexity

Given a word $w = w_0...w_n$ and a transition sequence, for every element in $\{1, ..., n\}$, four actions are applied: It is shifted to become the focus item, then a labelling transition is performed. Furthermore, each SHIFT puts the previous focus item into the memory from which it is retrieved at some future point by MERGE. Then, another labelling step follows. Only for the focus element identified by $\max(s_f) = |w|$, the second pair of steps is not present, since it is never put into the memory. When all these actions have been performed, the focus element consists of $\{1, ..., n\}$ and a labelling action has established the root. Therefore one can construct a tree for a sentence of length n in 4n - 2 steps.

As Coavoux and Cohen (2019) note, for structural steps, |S|+1 different transitions are possible. In the worst case, S contains n-1 items (i.e. when no MERGE has been performed and all items in $\{1, ..., n\}$ have been shifted). For labelling steps, |N|+1 different actions can be predicted. Thus, the time complexity is $\mathcal{O}(n(n-1) + n(|N|+1)) = \mathcal{O}(n(|N|+n))$ (Coavoux and Cohen, 2019). This is a slight downgrade in comparison to the time complexity of ML-GAP (cf. section 3.2.5). In practice, fast matrix-multiplication for one-step evaluation of all COMBINE actions with elements in the memory mitigates this disadvantage.

4.5. Implementation

Coavoux and Cohen (2019) implement their parsing algorithm using the Python programming language. The action scorer, trained to predict the optimal action according to the oracle, is a neural network model built with the Pytorch library (Paszke et al., 2017). From this arises the question of how to represent tokens and sets of tokens in a form usable by the neural model. Furthermore, an objective function must be set. Additionally, Coavoux and Cohen (2019) include a POS-tagging component.

4.5.1 Token Representations

Coavoux and Cohen (2019) follow previous proposals for constituency and dependency parsing like Cross and Huang (2016a) and Kiperwasser and Goldberg (2016) that suggest using a recurrent
neural network (RNN) to construct context-aware representations for the input tokens. In the following, I will deviate slightly from Coavoux and Cohen (2019) in favour of the notation used in Goldberg (2022).

Given a sequence of input tokens $w_{1:n} = w_1, ..., w_n$, Coavoux and Cohen (2019) first compute a representation for each token using a concatenation of a character-aware embedding and a word embedding.

Word Embedding When encoding a set of k distinct categorial features, it is common to associate each of these features with a dense vector representation called *embedding*. These vectors are trained and optimised as parameters of the model to achieve the optimal representation for each symbol (Goldberg, 2022, chapter 4.8).

Definition 4.16 (Alphabet index).

Let V be an alphabet. $idx_V : V \to \{i \in \mathbb{N} \mid i \leq |V|\}$ denotes a bijective function from symbols in V to unique indices.

Definition 4.17 (Embedding matrix).

For an alphabet V and an embedding dimension $d, E \in \mathbb{R}^{|V| \times d}$ is called an *embedding matrix*. For a symbol $s \in V$, the idx_V(s)-th row of E is defined as the *embedding of s*.

In many natural language processing tasks symbols represent natural language words. Thus, the term *word embedding* was coined. Traditionally, features of an alphabet V are encoded as vectors with dimensionality |V| filled with 0s except for one position with value 1 that corresponds to a distinct symbol. This is called a *one-hot encoding*. Dense feature representations allow it to represent a vocabulary in a much smaller dimensionality than |V|.²⁹

For the vocabulary of observed natural language words V, an embedding matrix \mathbf{E}_{word} for V with dimension $d_{word\text{-}emb}$ and an input sequence $w = w_1...w_n$, Coavoux and Cohen (2019) use $\mathbf{e}(w_i)$ to denote the embedding for w_i with $i \in \{1, ..., n\}$.

$$\mathbf{e}(w_i) := \mathbf{E}_{\mathbf{word}}^{[\mathrm{id}_{\mathbf{X}_V}(w_i)]} \tag{4.1}$$

 $\mathbf{E}_{\mathbf{word}} \in \mathbb{R}^{|V| \times d_{word\text{-}emb}}, \ \mathbf{e}(w_i) \in \mathbb{R}^{d_{word\text{-}emb}}$

When training on the DPTB corpus, Coavoux and Cohen (2019) extract 44530 unique words, including special <PAD> and <UNK> symbols. They embed them into a 32-dimensional space.

Character-Aware Embedding While word embeddings are a very capable way to represent discrete categories, they disregard the internal structure of words. Prefixes and suffixes may give information about the word's role in a sentence. For instance, the suffix *-ing* for English verbs indicates an infinitive. In our context, it usually means that it is dominated by another (main) verb in the constituent tree.

An embedding matrix that assigns distinct embeddings to different word surface forms can be assumed to capture the similarities and differences that are related to suffixes or prefixes from their distribution. However, it has been shown that combining word embeddings with a layer based on sub-word information improves POS-tagging results (Plank et al., 2016). This has two reasons. First, treating every word discretely generally requires more training data. Secondly, looking deeper allows to compute representations for words not seen at all during training (Gaddy et al., 2018).

Coavoux and Cohen (2019) use a character-aware-embedding as the second foundation for their token representations. It is generated using a *bidirectional long short-term memory (biLSTM)*

 $^{^{29} {\}rm Please}$ refer to Goldberg (2022, chapter 8.4) for more information on the relation between one-hot and dense encoding.

encoder. Viewed abstractly, an LSTM (Hochreiter and Schmidhuber, 1997) is a function that receives as input an arbitrarily long sequence of vectors $\mathbf{x}_{0:n} = \mathbf{x}_1, ..., \mathbf{x}_n$ with $\mathbf{x}_i \in \mathbb{R}^{d_{in}}$ for all $i \in \{1, ..., n\}$ and returns as output a vector $\mathbf{y}_n \in \mathbb{R}^{d_{out}}$ with d_{in} and d_{out} being pre-defined input and output dimensionalities (Goldberg, 2022).

When used as a *transducer*, each input vector \mathbf{x}_i is mapped to the output generated for the sequence up until \mathbf{x}_i , i.e. for $\mathbf{x}_{0:i}$. The internal computation function of the LSTM makes it possible for information from the previous entries in the sequence to influence the output for \mathbf{x}_i . When used as an *encoder*, only the final output is used. It represents a context-aware embedding for the whole input sequence.

The following full definition of an LSTM is based on Goldberg (2022, chapter 15). \odot denotes element-wise vector multiplication.

Definition 4.18 (LSTM).

A long short-term memory (LSTM) model is defined as a function with the following properties:

$$LSTM^{*}(\mathbf{x}_{0:n}) = \mathbf{y}_{0:n}$$
$$\mathbf{y}_{i} = LSTM(\mathbf{x}_{1:i})$$
$$LSTM(\mathbf{x}_{0:t}) = \mathbf{h}_{t}$$
$$\mathbf{h}_{t} = \mathbf{o} \odot \tanh(\mathbf{c}_{t})$$
$$\mathbf{c}_{t} = \mathbf{f} \odot \mathbf{c}_{t-1} + \mathbf{i} \odot \mathbf{g}$$
$$\mathbf{i} = \operatorname{sigm}(\mathbf{h}_{t-1}\mathbf{U}_{\mathbf{i}} + \mathbf{x}_{t}\mathbf{W}_{\mathbf{i}})$$
$$\mathbf{f} = \operatorname{sigm}(\mathbf{h}_{t-1}\mathbf{U}_{\mathbf{f}} + \mathbf{x}_{t}\mathbf{W}_{\mathbf{f}})$$
$$\mathbf{o} = \operatorname{sigm}(\mathbf{h}_{t-1}\mathbf{U}_{\mathbf{g}} + \mathbf{x}_{t}\mathbf{W}_{\mathbf{g}})$$
$$\mathbf{g} = \tanh(\mathbf{h}_{t-1}\mathbf{U}_{\mathbf{g}} + \mathbf{x}_{t}\mathbf{W}_{\mathbf{g}})$$
$$\mathbf{x}_{i} \in \mathbb{R}^{d_{in}}, \ \mathbf{y}_{i} \in \mathbb{R}^{d_{out}}, \ \mathbf{U}_{\mathbf{i}}, \mathbf{U}_{\mathbf{f}}, \mathbf{U}_{\mathbf{o}}, \mathbf{U}_{\mathbf{g}} \in \mathbb{R}^{d_{out} \times d_{out}}$$
$$\mathbf{W}_{\mathbf{i}}, \mathbf{W}_{\mathbf{f}}, \mathbf{W}_{\mathbf{o}}, \mathbf{W}_{\mathbf{g}} \in \mathbb{R}^{d_{in} \times d_{out}}, \ \mathbf{i}, \mathbf{f}, \mathbf{o}, \mathbf{g} \in \mathbb{R}^{d_{out}}$$

A bidirectional LSTM model (Graves and Schmidhuber, 2005) consists of two separate LSTMs: one for the forward and one for the backward direction. When considering an input sequence $\mathbf{x}_{0:n}$ the desired output for word $i \in \mathbb{N}$ with $i \in \{1, ..., n\}$ is constructed by using the standard forward sequence $\mathbf{x}_{0:i} = \mathbf{x}_1, \mathbf{x}_2, ..., \mathbf{x}_i$ as the input for the first LSTM and the backward sequence $\mathbf{x}_{n:i-1} = \mathbf{x}_n, \mathbf{x}_{n-1}, ..., \mathbf{x}_i$ for the second LSTM. The output is a concatenation of both results. This enables contextual influence from elements preceding and elements following \mathbf{x}_i .

Definition 4.19 (bi-LSTM notation).

A bi-LSTM is defined as a function where $LSTM_f$ and $LSTM_b$ denote LSTMs of input dimension d_{in} and output dimension d_{out} :

$$biLSTM^*(\mathbf{x}_{0:n}) = LSTM(\mathbf{x}_{0:n}, 1), ..., LSTM^*(\mathbf{x}_{0:n}, n)$$
$$biLSTM(\mathbf{x}_{0:n}, i) = LSTM_f(\mathbf{x}_{0:i}) \circ LSTM_b(\mathbf{x}_{n:i-1})$$
$$\mathbf{x}_i \in \mathbb{R}^{d_{in}}, \ \mathbf{y}_i \in \mathbb{R}^{2 \cdot d_{out}}$$

To construct character-aware embeddings, Coavoux and Cohen (2019) deconstruct tokens into sequences of individual characters. L denotes the alphabet of all character symbols. Every symbol gets assigned a character-embeddings:

$$\mathbf{c}_{\mathbf{e}}(l) := \mathbf{E}_{\mathbf{char}}^{[\mathrm{idx}_{L}(l)]} \tag{4.2}$$

$$\mathbf{E_{char}} \in \mathbb{R}^{|L| \times d_{char-emb}}, \ \mathbf{c_e}(l) \in \mathbb{R}^{d_{char-emb}}$$

Then, for each token w_i comprised of character sequence $w_{i,0:m}$, a biLSTM is used to encode a context-aware embedding. The backward network focuses on prefixes while the forward network is more sensitive to suffixes.

$$\mathbf{c}(w_i) := biLSTM_l(\mathbf{c}_{\mathbf{e}}(w_{i,1}), ..., \mathbf{c}_{\mathbf{e}}(w_{i,m}))$$

$$\mathbf{c}(w_i) \in \mathbb{R}^{d_{char}}, \ \mathbf{c}_{\mathbf{e}}(w_{i,j}) \in \mathbb{R}^{d_{char-emb}}$$

$$(4.3)$$

Token Representation Coavoux and Cohen (2019) construct a token representation \mathbf{w}_i as the concatenation of the character-aware embedding and the word embedding:

$$\mathbf{w}_i = \mathbf{c}(w_i) \circ \mathbf{e}(w_i) \tag{4.4}$$

$$\mathbf{w}_i \in \mathbb{R}^{a_{token}}, \ d_{token} = d_{char} + d_{word\text{-}emb}$$

Coavoux and Cohen (2019) build context-aware token embeddings by using a 2-layer biLSTM as a transducer. In this way, the network can capture contextual information regarding preceding and following tokens in each token vector. The first biLSTM has an input dimension denoted by d_{token} and an output dimension denoted by d_{hid} . The second layer biLSTM has input dimension d_{hid} and output dimension d_{hid} . The representations for $w_{0:n}$ are computed in the following way:³⁰

$$\mathbf{m}_{1,0:n} = biLSTM_1^*(\mathbf{w}_{0:n})$$

$$\mathbf{m}_{2,0:n} = biLSTM_2^*(\mathbf{m}_{1,0:n}) + \mathbf{m}_{1,0:n}$$

$$\mathbf{w}_i \in \mathbb{R}^{d_{token}}, \ \mathbf{m}_{1,i}, \mathbf{m}_{2,i} \in \mathbb{R}^{d_{hid}}$$

$$(4.5)$$

The first layer output is added to the second layer output as a *residual connection*. Residual mappings facilitate the learning of the identity function. Furthermore, they counteract the problem of exploding and vanishing gradients encountered in deep neural networks (He et al., 2016).³¹

As shall be seen, the output of the first layer biLSTM is used for the prediction of part-of-speech tags while the second layer is used to predict parsing actions. The architecture described so far is depicted in figure 4.4.

4.5.2 Set Representations

There are different possibilities of constructing constituent representations using token vectors. In projective constituency parsing it is common to use the leftmost leaf and the rightmost leaf of a constituent (Hall et al., 2014; Crabbé, 2015; Durrett and Klein, 2015). Using the notation established above such a constituent representation could be defined as follows for a constituent candidate s:

$$\mathbf{r}_{\mathbf{proj}}(s) = \mathbf{m}_{2,\min(s)} \circ \mathbf{m}_{2,\max(s)}$$

$$\mathbf{m}_{2,t} \in \mathbb{R}^{d_{hid}}, \mathbf{r}_{\mathbf{proj}}(s) \in \mathbb{R}^{2 \cdot d_{hid}}$$

$$(4.6)$$

In the view of overlapping constituents in discontinuous parsing, Coavoux and Cohen (2019) suggest enriching the constituent representation with information about gaps in the constituent. First, they define the following notation for a constituent candidate s, which yields all indices not in s located between the minimum and maximum of s.

Definition 4.20 (Gap set).

Let s be a constituent candidate.

$$\overline{s} := \{i \mid \min(s) < i < \max(s), i \notin s\}$$

 $^{^{30}}$ Coavoux and Cohen (2019) do not mention the use of a residual connections in their report, however it is included in the released code. Therefore I introduce an explicit variable $\mathbf{m}_{l,t}$ for each layer output (LSTM plus optional residual connection). $^{31}{\rm A}$ more thorough exploration of residual connections is given in section 6.2.2.



Figure 4.4: Stack-free neural parser with shared POS-tagging task architecture. For clarity the token representations are subscripted with words instead of indices. Figure inspired by Goldberg (2022, chapter 20.2).

The constituent is now represented using the set boundaries as well as the minimum gap element and the maximum gap element. A special vector $\mathbf{m_{nil}}$ that is randomly initialised and learned as part of the model is used if the set does not contain a gap.

$$\mathbf{r}(s) = \begin{cases} \mathbf{m}_{2,\min(s)} \circ \mathbf{m}_{2,\max(s)} \circ \mathbf{m}_{2,\min(\overline{s})} \circ \mathbf{m}_{2,\max(\overline{s})} & \text{if } \overline{s} \neq \emptyset, \\ \mathbf{m}_{2,\min(s)} \circ \mathbf{m}_{2,\max(s)} \circ \mathbf{m}_{\mathbf{nil}} \circ \mathbf{m}_{\mathbf{nil}} & \text{else} \end{cases}$$

$$\mathbf{m}_{\mathbf{nil}}, \mathbf{m}_{2,t} \in \mathbb{R}^{d_{hid}}, \mathbf{r}(s) \in \mathbb{R}^{4 \cdot d_{hid}}$$

$$(4.7)$$

This modification enables the model to differentiate between constituents with maximally two consecutive parts. It is not enough to distinguish constituents with more than one gap, i.e. with three or more consecutive parts. See the following two equations. The 4-index formalisation does not consider any intermediate continuous blocks lying between the two outermost gap indices.

$$s_1 = \{1, 3, 5\}$$

$$\overline{s_1} = \{2, 4\}$$

$$\langle \min(s_1), \max(\overline{s_1}), \max(\overline{s_1}) \rangle = \langle 1, 5, 2, 4 \rangle$$

$$(4.8)$$

$$s_2 = \{1, 5\}$$

$$\overline{s_2} = \{2, 3, 4\}$$

$$\langle \min(s_2), \max(s_2), \min(\overline{s_2}), \max(\overline{s_2}) \rangle = \langle 1, 5, 2, 4 \rangle$$
(4.9)

Despite this limitation, one could argue that this notation should be capable of providing the scorer with adequate information about the form of a constituent since in the DPTB, which features at most 3 discontinuous blocks, the overwhelming majority of 3 block occurrences is caused by punctuation (Maier et al., 2012).³²

 $^{^{32}}$ cf. section 3.1.7

4.5.3 Action Scorer

Differentiating between structural and labelling actions as odd and even steps allows using two separate feed-forward networks for action prediction.

Structural actions At a structural step for a configuration $c = \langle S, s_f, i, j, K \rangle : n \in C^{SC}$, the model has to decide between a COMBINE action with some element in S and SHIFT. Coavoux and Cohen (2019) compute the score of a COMBINE-s action using only the representation of s_f and s as input. The score of a SHIFT action is computed using s_f and $\{i\}$, i.e. the element to shift.

These |S| + 1 scores can be computed in parallel using the following matrix. On the right side I note the action the row corresponds to.

$$\mathbf{M} = \begin{pmatrix} \mathbf{r}(s_1) & \circ & \mathbf{r}(s_f) \\ \dots & \vdots & \dots \\ \mathbf{r}(s_n) & \circ & \mathbf{r}(s_f) \\ \mathbf{r}(\{i\}) & \circ & \mathbf{r}(s_f) \end{pmatrix}$$
COMBINE- s_n
(4.10)
$$\mathbf{M} \in \mathbb{R}^{(|S|+1) \times 8 \cdot d_{hid}}, \mathbf{r}(s) \in \mathbb{R}^{4 \cdot d_{hid}}$$

The score of an action is predicted using a feed-forward network with initial dropout.³³ For a pair of constituent representations, it computes a single scalar. tanh is used as the *activation function*. The *weight matrices* $W_{1struct}, W_{2struct}W_{3struct}$ and the *bias vectors* $b_{1struct}, b_{2struct}$ are trained by the model.

$$FF_{struct}(\mathbf{a}) = \mathbf{h_2W_{3struct}}$$

$$\mathbf{h_2} = \tanh(\mathbf{h_1W_{2struct}} + \mathbf{b_{2struct}})$$

$$\mathbf{h_1} = \tanh(\mathbf{a'W_{1struct}} + \mathbf{b_{1struct}})$$

$$\mathbf{a'} = \mathbf{d} \odot \mathbf{a}$$

$$\mathbf{d} \sim \text{Bernoulli}(drop_{action})$$

$$\mathbf{a'} = \mathbf{d} \odot \mathbf{a}$$

$$\mathbf{a}, \mathbf{a}', \mathbf{d} \in \mathbb{R}^{8 \cdot d_{hid}}, \ \mathbf{W}_{1\mathsf{struct}} \in \mathbb{R}^{8 \cdot d_{hid} \times d_{action}}, \mathbf{W}_{2\mathsf{struct}} \in \mathbb{R}^{d_{action} \times d_{action}}, \\ \mathbf{W}_{3\mathsf{struct}} \in \mathbb{R}^{d_{action} \times 1}, \ \mathbf{b}_{1\mathsf{struct}}, \mathbf{b}_{2\mathsf{struct}} \in \mathbb{R}^{d_{action}}, \ drop_{action} \in [0, 1]$$

A probability distribution over the possible structural actions is generated by applying the Softmax function on the |S|+1 dimensional output of FF_{struct} on **M**. Finally, the model performs the action corresponding to the highest score.

$$P(\cdot|c) = \text{Softmax}(FF_{struct}(\mathbf{M}))$$

$$P(\cdot|c) \in \mathbb{R}^{|S|+1}, \ \mathbf{M} \in \mathbb{R}^{(|S|+1) \times 8 \cdot d_{hid}}$$

$$(4.12)$$

Labelling Action At a labelling step with configuration $c = \langle S, s_f, i, j, K \rangle : n \in C^{SC}, |N| + 1$ different actions can be performed, where N is the alphabet of nonterminals. s_f can be labelled with some nonterminal $A \in N$ or No-LABEL can be performed.

An action is predicted using only s_f as input for a feed-forward network with an architecture that corresponds to that of FF_{struct} with the exception of the input and the output dimensionality.

 $^{^{33}}$ A detailed description of the dropout technique is given in Goldberg (2022, chapter 4.6).

Here, the output is a vector of length |N| + 1.

$$FF_{label}(\mathbf{a}) = \mathbf{h_2W_{3label}}$$
$$\mathbf{h_2} = \tanh(\mathbf{h_1W_{2label}} + \mathbf{b_{2label}})$$
$$\mathbf{h_1} = \tanh(\mathbf{s'W_{1label}} + \mathbf{b_{1label}})$$
$$\mathbf{a'} = \mathbf{d} \odot \mathbf{a}$$
$$\mathbf{d} \sim \operatorname{Bernoulli}(drop_{action})$$
(4.13)

$$\mathbf{a}, \mathbf{a}', \mathbf{d} \in \mathbb{R}^{4 \cdot d_{hid}}, \ \mathbf{W_{1label}} \in \mathbb{R}^{4 \cdot d_{hid} \times d_{action}}, \ \mathbf{W_{2label}} \in \mathbb{R}^{d_{action} \times d_{action}},$$
$$\mathbf{W_{3label}} \in \mathbb{R}^{d_{action} \times (|N|+1)}, \ \mathbf{b_{1label}}, \mathbf{b_{a2label}} \in \mathbb{R}^{d_{action}}, \ drop_{action} \in [0, 1]$$

A probability distribution over the possible labelling actions is generated by applying the Softmax function on the output of FF_{label} on $\mathbf{r}(s_f)$ Finally, the model performs the action corresponding to the highest score.

$$P(\cdot|c) = \text{Softmax}(FF_{label}(\mathbf{r}(s_f)))$$

$$P(\cdot|c) \in \mathbb{R}^{|N|+1}, \ \mathbf{r}(s_f) \in \mathbb{R}^{4 \cdot d_{hid}}$$

$$(4.14)$$

4.5.4 POS Tagger

Following Søgaard and Goldberg (2016), POS tagging is performed as an auxiliary task using the output of the first layer LSTM transducer. A linear transformation is used to predict the POS tag probabilities for a token. Let A be the alphabet of all POS-tags.

$$FF_{POS}(\mathbf{x}) = \mathbf{x}' \mathbf{W}_{POS}$$
$$\mathbf{x}' = \mathbf{d} \odot \mathbf{x}$$
$$\mathbf{d} \sim \text{Bernoulli}(drop_{POS})$$
(4.15)

$$\mathbf{W}_{\mathbf{POS}} \in \mathbb{R}^{dim_{hid} \times |A|}, \ drop_{POS} \in [0, 1]$$

For an input sequence $w_{0:n}$ the probability of a sequence of POS tags $t_{0:n} = t_1, ..., t_n$ is then computed as follows:

$$P(t_{0:n}|w_{0:n}) = \prod_{i=1}^{n} \operatorname{Softmax}(FF_{POS}(\mathbf{m}_{1,i}))^{[\operatorname{idx}_{A}(t_{i})]}$$

$$\mathbf{m}_{1,i} \in \mathbb{R}^{d_{hid}}, \ FF_{POS}(\mathbf{m}_{1,i}) \in \mathbb{R}^{|A|}$$

$$(4.16)$$

4.5.5 Objective Function

The word and character embeddings, the LSTM parameters and the parameters of the feed-forward networks introduced above are optimised during training. The following objective functions determine the loss that the model is trained to minimise.

Static Oracle For the static oracle given an input sequence $w_{0:n}$ with gold POS-tags $t_{0:n}$ and gold tree K_g , the sum of the log-likelihood of $t_{0:n}$ and the log-likelihood of gold transition actions $a_{0:4n-2}$ given by the oracle with respect to K_g is optimised:

$$\mathcal{L} = \mathcal{L}_t + \mathcal{L}_p$$

$$\mathcal{L}_t = -\sum_{i=1}^n \log P(t_i | w_{0:n})$$

$$\mathcal{L}_p = -\sum_{i=1}^{4n-2} \log P(a_i | (a_{i-1}(...(a_1(\sigma(w_{1:n}))))))$$
(4.17)

In practice, Coavoux and Cohen (2019) alternate between optimizing parsing and POS-tagging. This has been shown by Caruana (1997) to be effective for multitask-learning. At each step, they provide a sentence from the train corpus and optimise either for parsing or for POS-tagging.

Dynamic Oracle Instead of optimizing the likelihood of the gold actions, in the dynamic setting Coavoux and Cohen (2019) sample a configuration from the set of all valid configurations. Then the likelihood of the best action with respect to this configuration given by the dynamic oracle is optimised.

Before a training epoch, each sentence from the training set is sampled with probability p. The current parameters are used to parse and retrieve a sequence of configurations. Inspired by (Ballesteros et al., 2016) at each step an action is chosen randomly based on the softmax distribution instead of taking the best-scoring action.

Let *Choice* sample an action randomly based on a tuple of likelihoods. Then the sequence of (suboptimal) configurations is denoted by $d_{0:4n-2}$ and generated for an input $w_{0:n}$ as follows.

$$d_{1} = Choice(P(\cdot | \sigma(w_{0:n})))$$

$$d_{i} = Choice(P(\cdot | d_{i-1})) \quad \text{for } i \in \{2, ..., 4n-2\}$$
(4.18)

Finally, the likelihood of the best action determined by the oracle for each step is optimised.

$$\mathcal{L}_p = -\sum_{i=1}^{4n-2} \log P(o(d_i)|d_i)$$
(4.19)

When the oracle predicts more than one best action, one is chosen deterministically as discribed in section 4.3.2.

4.5.6 Dealing with Unknown Words

Coavoux and Cohen (2019) assign a special UNK word embedding to words that were not observed during training. UNK is integrated as part of the vocabulary V. To learn a suitable representation for unknown words, before each training step they replace embeddings that belong to the $\frac{2}{3}$ least frequent words with a probability of 0.3 by UNK.

4.5.7 Training

Coavoux and Cohen (2019) train the model with the averaged stochastic gradient descent (ASGD) algorithm (Polyak and Juditsky, 1992) for 100 epochs. Every 4 epochs, the model is evaluated using the development set and saved if the validation F-score surpasses the results of previous evaluations. This way, degradation by overfitting does not affect the final model.

Coavoux and Cohen (2019) train the model on the DPTB³⁴ using sections 2-21 for training (39832 sentences), 22 for development (1700 sentences) and 23 for testing (2416 sentences). Furthermore, they report results for TIGER and NeGra. Table 4.1 gives the total list of hyperparameters.

4.5.8 Evaluation

The model is evaluated by Coavoux and Cohen (2019) using a dedicated module of disco-dop³⁷ (van Cranenburgh et al., 2016). disco-dop (*Discontinuous Data-Oriented Parsing*) is an LCFRS

 $^{^{34}}_{}\mathrm{DPTB7}$ which sections 5.1 and 5.2 in Evang and Kallmeyer (2011) are based on.

 $^{^{35}}$ Xavier refers to a weight initialisation scheme proposed by Glorot and Bengio (2010) that has shown to allow faster convergence in deep neural networks.

 $^{^{36}}$ Gradient clipping is a common technique to prevent the problem of exploding gradients in deep (recurrent) networks (Pascanu et al., 2013).

³⁷https://github.com/andreasvc/disco-dop

Architecture h	yperparamet	ters
Dimension of word embeddings	$d_{word\text{-}emb}$	32
Dimension of character embeddings	$d_{char-emb}$	100
Dimension of character biLSTM state	d_{char}	100 (50 per direction)
Dimension of sentence biLSTM	d_{hid}	400 (200 per direction)
Dimension of hidden layers for action scorer	d_{action}	200
Number of action scorer FF layers		3
Number of tagger FF layers		1
Final bias in action scorer FF networks		False
Final bias in tagger FF network		False
Optimisation h	nyperparame	ters
Initial learning rate	l_0	0.01
Learning rate decay	l_t	$\frac{l_0}{1+t\cdot 10^{-7}}$, (t is step number)
Dropout for tagger input	$drop_{POS}$	0.5
Dropout for parser input	$drop_{action}$	0.2
Training epochs		100
Batch size		1 sentence
Optimisation algorithm		ASGD
Word and character embedding initialisation		$\mathcal{U}([-0.1,0.1])$
Other parameters initialisation (incl. LSTMs)		$Xavier^{35}$
Gradient clipping $(norm)^{36}$		100
Dynamic oracle	p	0.15

Table 4.1: Hyperparameters of the model given in Coavoux and Cohen (2019).

parser that comes with tools that aid in the evaluation of parsers for discontinuous treebanks.

Coavoux and Cohen (2019) ignore punctuation and root symbols for evaluation (as set in the standard evaluation parameters in proper.prm). The evaluation metrics are F-score (F) and F-score limited to only the discontinuous constituents (Disc. F). A detailed comparison of results is given in section 6.3.

5. Explaining Supertags

In the most successful transition-based discontinuous constituent parsing approaches, POS-tags are jointly learned as an auxiliary task (Coavoux and Crabbé, 2017b; Coavoux et al., 2019). This strategy was first explored by Søgaard and Goldberg (2016) as an auxiliary task for chunking³⁸ and CCG supertagging³⁹ and adopted for parsing by Coavoux and Crabbé (2017b).

POS-tags very coarsely indicate the relations their carriers can stand in (Jurafsky and Martin, 2009, chapter 5) (e.g. *bark* marked as a verb gives rise to a different constituent structure than *bark* marked as a noun). A hierarchical LSTM network seems to be able to learn useful common representations for performing both POS tagging and parsing decisions which suggests a strong relationship between these tasks. I assume that training to predict POS tags from an intermediate layer leads the model to associate similar vectors to tokens that bear the same POS tag at that level. This implicitly evokes a distribution over constituent structures. The LSTM networks serve as a means of lexicalizing this distribution, giving more weight to those structures that occur more frequently given the surroundings.

Coavoux et al. (2019) note that the unlexicalised GAP-transition parser resolves discontinuities especially well when a lexical trigger like why in wh-extractions is present. Most unrecognised discontinuities occur where no token in itself is indicative of a discontinuous structure. In such cases, the POS-tag assignments are usually not helpful either. See for instance (5.2) where a. is a discontinuous extraposition taken from the DPTB (*evidence* and *that commissions were paid* belong to one constituent; example given by Coavoux et al. (2019)) and b. a projective sentence using the same words. The POS tag assignments for each word are the same.

(5.1)	a.	In	April	1987	,	evidence	surfa	ced	that	con	missic	ons	we	re	paid	•
		IN	NNP	CD	,	NN	VBD)	IN	NN	S		VE	BD	VBN	•
	b.	In	April	1987	,	evidence	that	\cos	nmissi	ions	were	pai	id	sur	faced	
		IN	NNP	CD	,	NN	IN	NN	\mathbf{S}		VBD	VE	ΒN	VB	D	

Nudging the parser into building more distinct vector representations in such cases by training to predict fine-grained tag assignments that differentiate the projective and the discontinuous use of certain words might be beneficial for the ability of the parser to correctly resolve discontinuities. Supertags could fulfil this role - especially since they were shown to be compatible with POS-tags in an auxiliary-scenario (Søgaard and Goldberg, 2016).

Furthermore, they could help the parser to recognise types of discontinuous constituents which occur with a very low frequency. Maier (2015) observe that grammar-based parsers perform well on low frequency discontinuities while transition-based neural parsers perform poorly. This is not surprising considering that a multi-layer neural approach needs a lot of data to learn successfully. The resources available for phenomena that are only sparsely attested in the DPTB simply do not suffice. Thus, the statistical model, having no restrictions beyond the transition system design, resorts to erroneous actions. Meanwhile, an exact grammar extraction algorithm builds rules for all cases that occur in a given corpus. However low their frequency may be, the parser eventually will apply them if no other more frequent rules yield a successful parse.

Using subparts of structural supertag information to learn representations for certain aspects of discontinuity may aid in dealing with rare constituent types much akin to the effect that characteraware word embeddings have on parsing. The idea is to increase the similarity of the vector representation of items whose syntactic impact is alike in some aspect and reduce the similarity where it is different. This could help to improve representations for rare combinations of phenomena and thereby reduce the problem of data availability for rare discontinuities.

 $^{^{38}}$ Please refer to Jurafsky and Martin (2009, chapter 13.5) for an introduction to chunking.

 $^{^{39}\}mathrm{See}$ section 5.2 for an account of CCG supertags.

In the following, I will first introduce the concept of supertags in the context of combinatory categorial grammar (CCG). Then I will examine several categories of discontinuous phenomena in more detail and analyse how CCG supertags may help a discontinuous constituent parser to correctly recognise them. Finally, I will test my hypothesis by integrating supertags both as an input feature and as an auxiliary task into the transition-based parser of Coavoux and Cohen (2019) and explore ways to give a hierarchical model more control over the use of auxiliary information for the top level task. A thorough analysis of the results will be provided as well.

5.1. Lexicalised Grammar Formalisms

Broadly speaking, grammar formalisms fall into two groups. In the traditional approach, a set of very simple primitive components is used as a foundation. Operations of varying complexity and usually large in quantity are then introduced to construct complex structures out of the primitives. CFG and LCFRS fall into this category where productions serve as the combining operations and where the set of lexical category types (POS-tags) is very small. Contrasting that, strongly lexicalised formalisms like lexicalised tree adjoining grammar (LTAG) (Bangalore and Joshi, 2010a) and combinatory categorial grammar (CCG) (Steedman, 1989, 1996, 2000) follow an approach called *complicate locally, simplify globally* (CLSG) (Bangalore and Joshi, 2010a). They associate the lexical items (i.e. words in a sentence) with informative primitive structures from which a complex syntactical description of the phrase can be retrieved using a minimal and language-independent set of combining operations (Bangalore and Joshi, 2010a).

The primitive structures associated with lexical items are called *supertags*. A supertag localises information about (a) the number of arguments (b) (type) constraints for the arguments and/or (c) the positions of the arguments relative to the position of the item bearing the supertag. There may be several possible supertags associated with one lexical item. In CFGs, this information is distributed over more than one primitive structure (Bangalore and Joshi, 2010a)

As a result, non-local syntactic dependencies are localised into the primitive structure in CLSG formalisms. Thus, knowledge of supertags may be beneficial for parsing discontinuous constituents. In the following, I will give an overview of the CCG formalism and the resulting supertags.

5.2. Combinatory Categorial Grammar

Combinatory categorial grammar (CCG), developed by Steedman (1989, 1996, 2000), is an extension to *categorial grammars* (Adjukiewicz, 1935; Bar-Hillel, 1953). The central assumption of categorial grammars is that constituents are composed using function application. They associate each word in a sentence with a (complex) function that can receive a word to its left or to its right as an argument.

CCG extends this formalism by introducing directed versions of *combinators* originally found in combinatory logic (Curry, 1930). This approach can be described as a *deduction-based logical framework* (Kallmeyer, 2010, chapter 1). It produces constituent-like descriptions and features a novel strategy to deal with discontinuous phenomena. Its efficient parsibility and its straightforward compositional semantic interpretation makes the formalism of computational interest for covering natural language (Bozsahin, 2013). The following overview of the CCG formalism is based on Jurafsky and Martin (2023).

In the CCG formalism categories can be *atomic elements* or *functional categories*:

Definition 5.1 (Categories).

Let $\mathscr A$ be a set of atomic elements. The set of categories $\mathscr C$ is defined as:

• $\mathscr{A} \subseteq \mathscr{C},$

- $(X/Y), (X \setminus Y) \in \mathcal{C}, \text{ if } X, Y \in \mathcal{C},$
- nothing else is in \mathscr{C} .

Functional categories represent functions of a single argument category to a desired category. In (X/Y), the symbol Y is called the *argument* and X the *functor*. It means that an element of type Y is expected to the right of category (X/Y). On the other hand $(X\setminus Y)$ expects an element of type Y to its left. X is the result of application. The outermost brackets of functional categories are conventionally omitted.

Usually, the set of atomic elements is small (e.g. sentences and noun-phrases). Verb phrases, conjunctions and other complex types are realised though complex functional categories (Jurafsky and Martin, 2023). A lexicon contains assignments from words to sets of possible categories. Due to ambiguity, the assignments often contain several choices.

Example 5.2.

A minimal example lexicon is given in figure 5.1. Note that *lie* has two possible categories: N marks its use as a noun while $S \ P$ allows for a predicative use like in *You lie*. $S \ P$ is a function that expects an argument of type NP to its left and returns a category of type S.

Figure 5.1: Minimal example CCG lexicon.

The rules in figure 5.2 define the standard combining operations of most forms of CCG. (>) and (<) are the rules for forward and backward function application respectively. Forward and backward composition (>B, <B) allow combining two functions where one asks for an argument of type Y and the second gives Y as a result. The resulting function has the return type of the first function and the argument of the second. Type raising (>T, <T) converts a category into a function that receives as argument a function that seeks said category as its argument. Finally the conjunction operator (< Φ >) allows combining the CONJ category and two elements of the same type on both sides of the conjunction into a single element of this type.

Rule application for sequences of lexical items is performed on their associated categories in the style of natural deduction with the restriction that only adjacent elements can be used in deductions. This way, the deduction process adheres to the *principle of adjacency* (Steedman, 2000, chapter 4.1) implicitly yielding an ordered tree. The goal is to use up every input element and to retrieve a single category at the end. The resulting category is the type of the whole sequence.

A strong argument in favour of the CCG formalism is the compositional semantics that naturally emerges from it. Interpretations of constituents can be formalised as terms of the λ -calculus (Church, 1933).⁴⁰ Every lexical entry is assigned with a λ -expression and each combinator is given an interpretation within the λ -calculus, e.g. function application for forward and backward application. Through the replacement of bound variables, the terms of a sentence are brought into an interpretable logical form and order.

Example 5.3.

Figure 5.3 shows an example CCG derivation for the sentence that is a lie using the lexicon given

 $^{^{40}\}mathrm{An}$ interested reader is referred to Rojas (2015) for an introduction to the $\lambda\text{-calculus}$



Figure 5.2: Rules for CCG. X, Y, Z and T are category variables.

in figure 5.1. To the right, a tree visualisation of the constituents in the CCG derivation is given. As can be seen, this analysis amounts to the canonical CFG derivation tree for this sentence.

Function application allows to fill in the argument slot of a category. Take for instance *is* and its category $(S\NP)/NP$. It requires an NP category to the right. Applying (>) on $(S\NP)/NP$ and on category NP to the right yields the functor $S\NP$.



Figure 5.3: Example for CCG function application.

Figure 5.4 shows an example for forward composition (>B) and forward type-raising (>T) using the same supertag assignments for the sentence *that is a lie*. The category $(S\NP)/NP$ is combined with NP/N via composition. $(S\NP)/NP$ requires an argument of type NP while NP/N provides category NP if provided with N. Composition allows it to combine both items before filling missing arguments. The argument requirement of the category on the right is inherited by the resulting category $(S\NP)/N$.

NP is type-raised to a complex function yielding S and expecting a category that looks for an NP to its left with functor S. This way, an argument takes the syntactic role of a functor (Koller and Kuhlmann, 2009). In this way, forward composition can be used to fill the left argument slot of a VP before the right one. Here, *that* is combined before *lie*.

Note the equivalent tree structure to the right in figure 5.4. Grouping together a verb and a determiner or attaching the subject at a lower level than the object contradicts the canonical constituent analysis found in S-bar-inspired constituency trees. Both CCG tree descriptions are licensed by the same CCG lexicon in combination with the minimal set of combinatory rules. It becomes apparent that the formalism produces constituents that are not comparable with our traditional notion of constituency.



Figure 5.4: Example for CCG composition and type raising.

Example 5.4.

Composition and type-raising are not necessary to analyse the sentence in example 5.3. In other cases, like in coordination of non-canonical constituents they are argued to be required tools (Steedman, 2000, chapter 3). Figure 5.5 shows an example. Here, *Laura* and *a gift* as well as *Tommy* and *an invoice*, in both cases two elements with type NP, combine via type-raising and composition. *Laura* is type-raised to expect a ditransitive VP to its left and to return a transitive VP. *a gift* is type-raised to expect a transitive VP to its left and to return an intransitive VP. Then, the two type-raised terms are combined via backward composition. Doing the same for *Tommy* and *an invoice* allows using the conjunction operator ($\langle \Phi \rangle$) for these non-constituents.

The two combinators play an important role in dealing with long-range dependencies which I will analyse in the following subsection.

Ι	sent	Laura	a	gift	and	Tommy	an	invoice
NP	$\overline{((S \setminus NP)/NP)/NP}$	NP	NP/N	Ν	CONJ	NP	NP/N	Ν
		$\overline{(((S\backslash NP)/NP)\backslash(((S\backslash NP)/NP)/NP)} \leq T$		NP >				
			(S\N	P)∖((S\NP)/NP)				
		(S\NP)\(((S\NP)/M	NP)/NP)	< <u> </u>		(S\NP)\(((S\N	NP)/NP)/NP)
			$(S \setminus NP) \setminus$	(((S\NP)/NP)/NI	^{>})			<Φ>
			$S \setminus NP$					<
_			S					<

Figure 5.5: Coordinative use of type-raising and composition in CCG.

As became apparent, type raising and coordination allow any substring in a sentence to form a constituent (Houtman, 1994, 63–85).⁴¹ This enables strict left-to-right derivation of sequences which is argued by Jurafsky and Martin (2023) to be a desirable feature from a psycholinguistic point of view since it resembles the way humans process language. Cases of derivational ambiguity where the same input sequence with the same supertag assignments can be reduced to the same category are claimed by (Lewis and Steedman, 2014) to license an identical same semantic composition in many cases.

A key argument of the proponents of CCG as a theory of grammar is that there is no necessity to model several layers of syntactic representation to account for a transformation from a deep form grouping together the words in a sentence that belong together in terms of our intuition of meaning, and from which a semantic representation can be extracted, to a surface form cluttered with discontinuities, scrambling or other phenomena inert to human utterance while adhering to the reflection of a narrow definition of constituency in the syntactic representation. Instead, syntax should be treated as a transparent interface between spoken language and logical form (Steedman, 2000, chapters 1 and 2). CCG is a *monostratal* grammar formalism in the sense that it features

 $^{^{41}}$ This only holds if type-raising is not restricted to some limited set of operations as explained by Steedman (2000, chapter 4).

only one syntactic level of representation with the goal of assigning every syntactic rule with a semantic rule applied in parallel (Steedman, 2000, chapter 1). Since it aims at describing the competent composition of a semantic form, over-generation of spuriously ambiguous derivational structures is not considered problematic.

While in contrast the discontinuous constituent structure found in the DPTB is very strict and designed to exhibit one clear derivational analysis per phenomenon, its motivation for analysing constituents as discontinuous and not as instances of a separate movement process as employed by Chomsky (1975) and grammatical traditions alike is similar to that of CCG: constructing only one syntactic level of representation. Discontinuous tree descriptions achieve this by loosening the adjacency requirement of constituency while maintaining a narrow requirement for what can form a constituent based on semantic intuition. This precisely leads to the necessity of mild context-sensitive generative devices like LCFRS. In CCG the idea is executed by binding arguments into predicate-argument structure while combining adjacent words into syntactic constituents resulting in a semantic expression that does not necessarily correspond with the sentence word order (Steedman, 2000, Chapter 3).

My hope is that the strong correspondence of syntactic rules and semantic function application inherent to CCG categories as well as the fact that word order is projected entirely from the lexicon might enable a neural model to retrieve information about the predicate-argument structure from lexical category assignments that motivated the notion of constituency found in discontinuous treebanks.

5.2.1 CCG Treebanks

The *CCGbank* (Hockenmaier and Steedman, 2007) is a version of the Penn Treebank annotated with CCG supertags and derivations. It was automatically extracted from the Penn Treebank and contains 48,934 fully annotated sentences as well as a lexicon of 44,000 words. Sections 2-21 feature 1,286 distinct lexical category types.

CCGbank differs from the pure CCG presented above. It uses features to distinguish different types of primitive categories such as S[dcl] for declarative sentences and S[q] for yes-no questions. Furthermore, it introduces a range of special combinatory rules, some of which deal with punctuation, others allow unary category changes, e.g. from N to NP. Without further specification, I will use the notation (<r>) for all punctuation rules and (<u>) for all unary rules.

Honnibal et al. (2010) revised the original CCGbank by introducing a more precise analysis of compound nouns, verb argument structure, verb-particles and noun-arguments. Furthermore, they reintroduced quotation marks which were previously removed by Hockenmaier and Steedman (2007). The resulting treebank is known as *CCGrebank* and features 1,574 distinct lexical category types in sections 2-21. I will base my analysis of discontinuous phenomena in section 5.3 on a comparison of derivation examples in DPTB and in CCGrebank.

5.2.2 Supertaggers and Parsers

The lexicalised nature of combinatory categorial grammar allows for a unique approach to parsing for the formalism. Usually, a word is assigned more than one category in a CCG lexicon. This fact and the non-standard notion of constituency have the effect that a sentence can yield a large number of different parses. This problem can be approached by assigning lexical categories in a pre-parsing step using a statistical *supertagger*. Bangalore and Joshi (1999) call this *an approach to almost parsing*. Since the set of combinators is kept minimal and the syntactic information inherent in supertags is so rich most ambiguities are resolved through the assignment of adequate supertags. The idea of using a dedicated supertagger before parsing to narrow down the space of possible supertag assignments for an input sequence was first explored by Bangalore and Joshi (1999) for LTAG and adapted by Clark and Curran (2010) for parsing CCG. Note, however, that some ambiguity remains that is not resolved through supertagging (cf. example 5.3).

Lewis and Steedman (2014) propose a new CCG parsing strategy that allows for an efficient A^* search by assuming that the optimal CCG parse for a sentence is determined by the highest probability supertag assignment. Then, they use a heuristic to eliminate spurious derivational ambiguity. They use 511 unique supertags and a total of 23 combinatory rules. Their model is trained on sections 02-21 of CCGbank.⁴²

Yoshikawa et al. (2017) propose an extension to this parser called $depccg^{43}$ that resolves spurious attachment ambiguity by jointly modelling a dependency structure. Their model successfully approaches cases where spurious ambiguity indeed leads to a difference in semantic representation. They illustrate this problem by help of the Japanese example reproduced in figure 5.6.

Yesterday	buy-PAST	$\operatorname{curry}\!-\!ACC$	eat-PAST		Yesterday	buy-PAST	$\operatorname{curry}\!-\!ACC$	eat-PAST
Kinoo	kat-ta	karee-wo	tabe-ta		Kinoo	kat-ta	karee-wo	tabe-ta
S/S	S	NP	S\NP	-	S/S	S	NP	S\NP
	S >					NP/NP		
NF	P/NP <u></u>					1	NP >	
	NP	>					S	<
		S	<				S	>

Figure 5.6: Spurious ambiguity in CCG derivations that leads to distinct semantic representations; adapted from Yoshikawa et al. (2017). The English translation is *I ate the curry I bought yesterday. yesterday* can refer to either *bought* or *ate*.

Alongside the standard parser trained on the original CCGbank which they describe in their paper, Yoshikawa et al. (2017) also release a model trained on the CCGrebank, which I will use as a supertagger in the pipeline approach presented in section 6.1.

5.3. Discontinuous Constituents and CCG

Due to its monostratal nature, CCG deals differently with phenomena that are traditionally analysed to exhibit discontinuous constituents. Being a lexicalised approach, the recipe for dealing with discontinuities is largely (with the exception of the combining rules) encoded in the assigned supertags. In the following, I will take a closer look at the way CCG resolves sentences in nonstandard word order and explore the relationship with common discontinuous constituent analyses from the DPTB.⁴⁴

Several suggestions were made to account for discontinuous constituents using the CCG framework. One example is the introduction of infixation points and *wrap* and *infix* operations, see Warstadt (2015) for a recent proposal. It introduces the notion of discontinuity by weakening the formalism's *principle of adjacency* (Steedman, 2000, chapter 4.1). Since annotated corpora for nonstandard proposals are generally unavailable, I will only focus on the handling of discontinuities by CCG in the form described above.

Evang (2011) enumerates six classes of discontinuities found in the DPTB. Coavoux et al. (2019) revise these categories resulting in the following list and example phrases:

⁴²https://github.com/mikelewis0/easyccg

⁴³https://github.com/masashi-y/depccg

 $^{^{44}}$ In the following derivation examples pre-terminal nodes are regularly omitted for reasons of space. In most cases only extracts of the full trees are shown.

1. wh-extraction:	<u>What</u> should I \underline{do} ?
2. fronted quotations:	Absolutely, he <u>said</u> .
3. extraposed dependent:	In April 1987, <u>evidence</u> surfaced <u>that commissions were paid</u> .
4. circumpositioned quotations:	In general, they say, avoid takeover stocks.
5. <i>it</i> -extrapositions	$\underline{\text{It}}$'s better to wait.
6. subject-verb inversion	Said the spokeswoman: "The whole structure has changed."

Note that terms like *movement* and *extraction* are not meant to allude to a specific grammatical theory that uses the notion of movement as a syntactic operation but only to refer to syntactic phenomena metaphorically.

5.3.1 Wh-movement

Wh-movement denotes the non-canonical word-order correlated with the use of interrogative words like what and why. The wh-word is placed at the front of the sentence or clause instead of the position it would be expected in with respect to its grammatical function. Take for example (5.2b) where this, the direct object of do, is found in the standard position, following the predicate. Compare this with (5.2a) where What, corresponding to do, is positioned at the beginning of the sentence.

- (5.2) a. <u>What</u> should I \underline{do} ?
 - b. I should do this.

Figure 5.7 shows the analysis for the sentence *What should I do?* in the DPTB as well as in CCGrebank. In the constituent tree, the fronted *wh*-word *What* is attached at the level of the VP that directly dominates the verb *do*. In the CCG analysis, the subpart *should I do* is derived first via forward application and composition. The resulting category S/NP expects an object NP in standard position to its right. The discontinuity is dealt with by assigning *what* a higher-order function: S/(S/NP). It fills the missing argument slot by simply "consuming" the incomplete sentence.

This supertag assignment could give a discontinuous constituent parser a hint: It could learn to identify the fact that the right NP argument slot of do is not removed by combination with the two items on its left. It could then conclude from the supertag assignment of the first word, S/(S/NP), that it should be used as a filler of this argument slot promoting a MERGE operation between *What* and *do* before merging with *should* or *I*.



Figure 5.7: Wh-movement. Derivation in the DPTB (on the left) and in CCGrebank (on the right).

Sentences, where the wh-word asks for the subject, are usually projective. The first position corresponds with the subject position and so the supertag assignment differs compared to the previous case. This can be seen in figure 5.8. Here, What expects a sentence with a missing NP in left position. One can clearly differentiate the projective and the unprojective case on the basis of supertag assignments.



Figure 5.8: Wh-word that asks for the subject from CCGrebank.

Wh-movement also occurs in relative clauses. Figure 5.9 shows a DPTB derivation for the nominal phrase *federal statutes that the Supreme Court invalidated*. Here *that* acts as an object of the verb. The subject *the Supreme Court* blocks a projective analysis.

The CCGrebank analysis, given in figure 5.10, is similar to figure 5.7: The subject and the verb are first composed into a non-standard constituent by use of type-raising and composition that then serves as an argument for *that*. Interrogative words like *what* and *which* are also assigned the category $(N\setminus NP)/(S[dcl]/NP)$ when they occur in this kind of construction. It differs from the assignment in questions in that it does not return a sentence type but type $N\setminus N$ that makes it function as a modifier of a preceding noun.



Figure 5.9: Relative clause with that-fronting in DPTB annotation.

federal	statutes	that	the	Supreme	Court	invalidated
N/N	N	$\overline{(N \setminus N)/(S[dcl]/NP)}$	$\overline{NP/N}$	N/N	N	$\overline{(S[dcl]\backslashNP)/NP}$
	N >			N	>	
				NP	>	
				$S/(S \setminus NP)$))	
				9	6[dcl]/N	NP>₿
				N\N		>
			Ν			<

Figure 5.10: Relative clause with that-fronting in CCGrebank annotation.

that is given a different supertag when it does not accompany a relative clause but is used as a demonstrative pronoun (see figure 5.11) or as a conjunction in a complement clause (see figure 5.12). Coavoux et al. (2019) point out that the ambiguity of *that*-clauses is a notable source of error concerning *wh*-movement. Supertag information may enable the parser to better differentiate between the structurally distinct uses of *that*.

That	\mathbf{s}	good	news	for marketers of walking shoes	
Ν	$\overline{(S[dcl]\backslashNP)/NP}$	$\overline{N/N}$	Ν	$(S \setminus NP) \setminus (S \setminus NP))$	•
Ň₽			N >		
		Ν	<mark>∎</mark> <">		
	S[dcl]\I	NP	>		
			S[d	dcl]\NP	
			S[d	cl] <	
			S	[dcl]	r>

Figure 5.11: Example for the use of that as a demonstrative pronoun with the corresponding derivation in CCGrebank.

Ι	know	that	I 'm shouting into the breeze here as far as what we 're doing now
NP	$\overline{(S[dcl] \setminus NP)/S[em]}$	$\overline{S[em]/S[dcl]}$	S[dcl]
			S[em]
			S[dcl]\NP
			S[dcl]

Figure 5.12: Example for the use of that as a conjunction in a complement clause including the derivation from CCGrebank.

Another import reason for parsing-errors in *wh*-extracted phrases, according to Coavoux et al. (2019), is ambiguity on the extraction site, i.e. the node the fronted question word should attach to to form a discontinuous constituent. As an example they mention the phrase *which many clients didn't know about* for which the DPTB analysis and its CCGrebank counterpart are given in figures 5.13 and 5.14 respectively.⁴⁵ Their parser commits the error of treating *which* as an argument of *know*. Given the correct CCG analysis, it would be immediately clear that the fronted phrase must attach to *about* since *know* does not posses an NP argument. Direct attachment of *What* to *know* would be licensed by a different supertag, namely $(S\NP)/NP$. Supertag information could therefore help to resolve extraction site ambiguities of this kind.

⁴⁵The CCGrebank analysis uses the backward crossed composition combinator $\langle \mathbf{B}_{\times} \rangle$ here. Please refer to Steedman (2000, chapter 3) for more information on it.



Figure 5.13: DPTB example for wh-fronting with attachment ambiguity (PP vs. VP)

which	many	clients	did	n't	know	about
$\overline{(NP \setminus NP)/(S[dcl]/NP)}$	N/N	N	$\overline{(S[dcl]\backslashNP)/(S[b]\backslashNP)}$	$\overline{(S \setminus NP) \setminus (S \setminus NP)}$	$\overline{(S[b] \setminus NP)/PP}$	PP/NP
		N >	(S[dcl]\NP)/(S	5[b]\NP)	(S[b]\NP)	/NP
		NP<		$(S[dcl] \setminus NP)/NP$		——>B
	S/(5	S∖NP)				_
			S[d	cl]/NP		>₿
			NP\NP			>

Figure 5.14: CCGrebank derivation corresponding to the phrase in figure 5.13.

Unfortunately, supertags are not always that informative for retrieving the exact discontinuous constituent structure. Figure 5.15 depicts a DPTB analysis containing the fronted adverbial phrase *How deeply*. It is, just like in the previous example, analysed as a daughter of the VP node. Compare this with the CCGrebank analysis in figure 5.16. Here, *How deeply* is simply treated as a sentential modifier for the rest of the phrase. Neither is it associated with the subject-less verbal phrase S[b] NP nor is there an indicator for the fact that it is to be placed at a lower level since the phrase on its right does not show any missing arguments.

This also happens for other extractions of features realised as adjuncts (prepositional and adverbial phrases) instead of arguments of right-side components. I assume that the parser will be able to infer from the appearance of *do* in a sentence that the preceding sentential modifier most likely refers to a constituent to the right of *do*. Nonetheless, if there was more than one valid option for attachment beyond *do*, supertags would not help in resolving such an ambiguity.



Figure 5.15: DPTB derivation of a sentence that contains a wh-word that asks for an adverbial phrase.

How	deeply	do	they	read	it	?
$\overline{(S[wq]/S[q])/(S[adj]\backslashNP)}$	$\overline{S[adj] \backslash NP}$	$\overline{(S[q]/(S[b]\backslashNP))/NP}$	NP	$\overline{(S[b]\backslashNP)/NP}$	NP	•
S[wq]/S[q]	>	S[q]/(S[b] NP)	>	$S[b] \setminus NP$	—>	
			S[q]		_>	
		S[wq]			—>	
		S[wq]			— <r< td=""><td>.></td></r<>	.>

Figure 5.16: CCG derivation of the sentence How deeply do they read it? in the CCGrebank.

5.3.2 Fronted Quotations

Since the Penn Treebank is based on newspaper articles, it features a lot of quotations (Evang, 2011) - some of which result in discontinuous structures. Figure 5.17 shows an example of a fronted quotation from the DPTB. The phrase framed by quotation marks is analysed as a complement of *said*. The two elements are separated by a gap containing a comma and the closing quotation marks as well as the subject of *said* which are all attached one level higher in the tree.



Figure 5.17: Discontinuous fronted quotation in the DPTB.

Figure 5.18 shows the CCGrebank analysis of the sentence. In the CCGrebank, opening and closing quotation marks are assigned special categories: LQU and RQU respectively. They are removed by use of $(\langle \mathbf{r} \rangle)$ and thus cannot serve as an explicit indicator of the fronted quotation. The non-standard word-order is resolved through the category assignment of *said*. It expects as an inner argument a sentential category to its left. The first argument is the subject in standard position. This necessarily means that an intermediate constituent must be constructed by providing its subject *he* before merging with the quotation which is opposite to the order of the DPTB analysis.

Nevertheless, the supertag assignment can be an indicator for the constituent parser since it differs from the projective case where the quotation is positioned to the right as can be seen in the partial derivation in figure 5.19. Here the order of derivation still does not reflect the canonical analysis but the supertag assignment conforms to the fact that the quotation is expected directly to the left without intermediate arguments. Thus, supertag information could help to resolve discontinuities caused by fronted quotations.



Figure 5.18: Fronted quotation in CCGrebank.



Figure 5.19: Projective fronted quotation with CCGrebank derivation.

5.3.3 Extraposed Dependents

The category of extraposed dependents mainly contains discontinuities that arise from appended clauses that refer to some preceding word. Contrary to wh-extractions and fronted quotations, no lexical trigger is present (Coavoux et al., 2019). Figure 5.20 shows an example where a subordinate clause is appended after the predicate of the main clause but relates to its subject.



Figure 5.20: Extraposed subordinate clause from DPTB.

The CCGrebank analysis is shown in figure 5.21. Here, the subordinate clause *that commissions* were paid functions as an adjunct to the predicate *surfaced*. The lexical category of *that* expects a sentence to its right and as inner argument a VP (i.e. S[dcl]\NP) to its left, returning a VP. Then, the NP evidence is realised as a regular argument.

The attachment of the relative clause at the level of the NP cannot be recovered from this analysis. In fact, the same supertag assignment is given when the extraposed clause actually attaches at VP level in the DPTB analysis, see figure 5.22. Thus, the parser could be tempted to attach the extraposed clause incorrectly in one of the cases purely based on distributional properties of the corpus.

In	April	1987	,	evidence	surfaced	that	commissions	were	paid
$\overline{(S/S)/NP}$	Ν	$\overline{N \setminus N}$,	N	$\overline{S[dcl] \setminus NP}$	$\overline{((S\backslashNP)\backslash(S\backslashNP))/S[dcl]}$	N	$\overline{(S[dcl]\backslashNP)/(S[pss]\backslashNP)}$	$\overline{S[pss] \setminus NF}$
		N_<		— <u> NP</u>			< <u>u></u>	S[dcl]\NP	>
	N	- <u> 1P</u>						S[dcl]	<
S	s/S	>					(S\NP)\	(S\NP)	>
	S/S	<i< td=""><td>r></td><td></td><td></td><td></td><td>$S[dcl] \setminus NP$</td><td></td><td><</td></i<>	r>				$S[dcl] \setminus NP$		<
							S[dcl]		<
						S[dcl]			>

Figure 5.21: Extraposed subordinate clause in CCGrebank notation.



Figure 5.22: CCGrebank derivation where the subordinate clause *that he sends* ... is also a modifier to the VP in the corresponding DPTB annotation.

This unhelpful analysis is not limited to the conjunctive use of *that*. The same supertag is assigned to *than* in (5.3). The parser may be able to infer from the distribution of such cases that the extraposed clause should not form a direct constituent with the verb since *than* rarely modifies a VP. With that being said, the supertags are neither aiding in such cases nor do they help in resolving attachment ambiguity past the verb boundary. See for instance the CCG derivation in figure 5.23 that would correspond to three distinct DPTB analyses. The extraposed clause could be attached to *anything*, *the fundamentals* or *our business*.

(5.3) In any case, the firms are clearly moving <u>faster</u> to create new ads than they did in the fall.

Note that this issue does not arise from derivation ambiguity as explored in example 5.3 and in figure 5.6. It is either that the annotations in CCGrebank simply do not treat such phenomena properly or that the CCG formalism itself cannot appropriately account for discontinuities of this kind. While there is, by formulation of its objective, no necessity for CCG to make discontinuous constituents deducible from lexical category assignments; nonetheless, the observed behaviour does not live up to the paradigm of the formalism. Steedman (2000, chapter 2.1) postulates that every syntactic rule shall have a semantic representation. In our case, several different semantic representations arise from exactly the same syntactic operation of backward application. Applying $((S\NP)\(S\NP))/(S[dcl]\NP)$ would prescribe that the meaning of its right argument be associated with the object of the verbal phrase on its left in figure 5.23, with its missing left NP argument in figure 5.21 and with the verbal phrase as a whole in figure 5.22. Alternatively one could assume that in each of these cases, *that* refers to a distinct lexicon entry with a distinct semantic representation. While this is reasonable in the case of the VP modifier in figure 5.22, it is at least questionable for figures 5.21 and 5.23 that only differ in regards to the function the NP, modified by the extraposed clause, serves (i.e. which argument slot of the VP it fills).



Figure 5.23: Extraposition together with its CCGrebank derivation where the corresponding DPTB annotation suggests attachment of the extraposition at the NP node directly above anything.





There is also a rare case of fronted extrapositions without a lexical indicator in the form of a wh-word. This happens when an argument or an adjunct appears before the subject and the verb and is attached at VP level instead of S (Evang, 2011). Figure 5.25 shows an example. Here, CCGrebank treats the extraposed part as a sentential modifier in the same way as And is treated. Since cases like this are very infrequent, I predict this shortcoming to have a negligible effect on the score of a supertag-enriched parser.



Figure 5.25: Fronted extraposition with the DPTB annotation.



Figure 5.26: Fronted extraposition with the CCGrebank annotation.

5.3.4 Circumpositioned Quotations

Figure 5.27 shows an example of a circumpositioned quotation from the DPTB. The two parts of the quotation *Working with lawyers* and *I need it* are dominated by a common parent S which is assigned as a complement to the intermediary verb *says* to form a VP. The structure is similar to the fronted quotation with the addition of a tail expression.



Figure 5.27: Circumpositioned quotation and its DPTB analysis.



Figure 5.28: Circumpositioned quotation and its CCGrebank analysis.

The corresponding CCGrebank analysis shown in figure 5.28 is problematic. The substring *she* says, "I need it." is analysed like a regular quotation in standard position yielding an S category. Thus, the prepended participle construction takes as argument not only the right sub-quote but also the quotative verb and its subject. No supertags give any indication of a circumpositioned arrangement.

The structure is, with the exception of an additional pair of quotation marks, indistinguishable from a sentence like *Reluctantly he said*, "I want to go." where *Reluctantly* would receive category S/S. Therefore, I suspect that supertag information will not help the parser to resolve circumpositioned quotations. In the worst case, it might even mislead it.

The presence of quotation marks will probably counteract such an analysis. However, as Coavoux et al. (2019) points out, there are cases of circumpositioned quotations without quotation marks that are already hard to resolve by an unmodified neural model. Here, supertag information should be least beneficial.

In other cases, the CCG analysis is even more questionable. Figure 5.24 shows a case where the quotative verb and its subject are treated as a modifier of a preceding NP inside the quotation. This is done by converting the S[dcl]/S[dcl] category of Mr. Jones says into NP\NP by some unary rule. The supertag assignments alone suggest that Mr. Jones says should be followed by

an S category. This could mislead the parser since the predicate can actually be found to the left. Overall, I assume supertag information on the basis of the CCGrebank relating to circumpositioned quotations to be of little informative value for the parser.

5.3.5 It-Extrapositions

It-extrapositions denote cases where an expletive, i.e. semantically empty, it is found in the standard-place of a clausal argument that is instead positioned at the end of the sentence (Evang, 2011). See (5.4) for an example. The extraposed clause could be regarded as specifying it semantically and is therefore analysed as one common discontinuous NP in the DPTB.

- (5.4) a. <u>It</u> is easy to forget something.
 - b. To forget something is easy.

Figure 5.29 shows the DPTB analysis of part of the direct quotation in (5.5a) to know what to expect at this point and It are dominated by a common NP node.

- (5.5) a. "<u>It</u>'s hard to know what to expect at this point," said Peter Rogers, an analyst at Robertson Stephens & Co.
 - b. It is also likely to bolster fears that the Japanese will use their foothold in U.S. biotechnology to gain certain trade competitive advantages.

Figure 5.30 shows the CCGrebank counterpart. Here, to know what to expect has category S[to] NP and thus expects an NP to its left. This is in line with the canonical analysis of *it* and the extraposed clause forming an NP constituent. However, due to the discontinuity, this argument slot cannot be filled. Instead, the lexical category assigned to 's deals with both elements. To account for to know what ... with category S NP, 's has a second, inner right argument.

While the subject *it* functions as a regular, innermost NP argument of the predicate and while the non-finite clause is also realised as an argument of 's, the specialised supertag assignment in itself could still give the parser an indication to merge the preceding NP *it* when to know what ... is the focus item before merging 's. This clue is even stronger when including category features. The NP category of *It* is marked with the feature [expl], as is the left argument slot of 's. Thus, the parser could learn to associate this supertag with the necessity to find a discontinuous appended clause. The supertag assignments could further specify the boundaries between the projective parts and the extraposed clause in such a scenario.

This assumption is solidified by the fact that the projective use of infinite clauses as an adjectival modifier in combination with *It is* is tagged differently in the CCGrebank. Take for instance (5.5b) for which figure 5.31 shows part of the CCGrebank derivation. Here, the adjective *likely* receives a different supertag than *hard* in the last example. It expects a non-finite *to*-clause to its right as an additional argument. Consequently, *is* does not require a non-finite clause as a second right argument here. When one considers categorial feature the difference is even more pronounced since the NP and its corresponding argument slot lack the [expl] feature. Therefore, I expect the parser to be able to infer useful information for the treatment of *it*-extrapositions from supertag information.

5.3.6 Subject-Verb Inversion

Subject-verb inversion is an infrequent class of discontinuities where the subject can be found between the predicate on the left and one of the verb dependents on the right Evang (2011). It occurs mainly in combination with quotations and thus can be considered as the reverse case of



Figure 5.29: DPTB analysis of an *it*-extraposition.

fronted quotations. Figure 5.32 shows an example where the quotative verb *Said* takes the initial position in the sentence.

The corresponding CCGrebank analysis is shown in figure 5.33. Here, the lexical category assignment of *Said* accounts for the reversed argument structure. A manual analysis showed that the category (S[dcl]/S[dcl])/NP is only assigned in cases of subject-verb inversion with quotations. It should therefore be a useful hint for the parser.

When disregarding the category feature [dcl], (S/S)/NP is also used for prepositions in front position sentential modifiers. The following sentence gives an example where In is assigned the category (S/S)/NP in the CCGrebank.

(5.6) In any case, supplies to patients won't be interrupted, the company added.

Following the category, *In* first accepts *any case* as an NP-argument and after that the Sargument *supplies to patients won't be interrupted*. This derivation order corresponds to the projective analysis found in DPTB. While the neural model might learn a correlation between the supertag and verbs of speech, I do not assume that supertag information will benefit the analysis of subject-verb inversion when the feature [dcl] is dropped. In the worst case, it could produce noise that pushes the parser towards analysing the quotative verb and the subject as a modifier of the quotation.

expect (NP) (S[b]/NP)/N	NP)/NP												
to [] (S[to]\NP)/(S[b]	(S[to]) NP	5[b]\NP					tter fears that	P)/NP NP >		^	^		V
$\frac{\rm what}{\rm NP/((S[to]\backslash NP)/N}$		0,	S[to]\NP			position.	to bols)/(S[b]\NP) (S[b]\N	S[to]/NP	[adj]\NP			
know \NP) (S[b]\NP)/NF				lcl]∖NP[expl]	lcl]	alysis of an <i>it</i> -extra		[to]\NP) (S[to]\NP		S	S\NP	lcl]\NP	dcl]
to (S[to]\NP)/(S[b]				Slo	S): CCGrebank an	likely	P) (S[adj]\NP)/(9				Slo	S
hard s[adj]\NP) S[adj]\NF	[to]\NP)					Figure 5.30	also]\NP) (S\NP)/(S\N					
`s S[dcl]\NP[expl])/(S[to]\NP))/(S	(S[dcl]\NP[e×pl])/(S]						It is	NP (S[dcl]\NP)/(S[adj					
It JP[expl] ((:	I			I									





Figure 5.32: Example sentence for subject-verb inversion in the DPTB.



Figure 5.33: Example sentence for subject-verb inversion in the CCGrebank.

5.3.7 Summary

In the preceding paragraphs, I provided a detailed analysis of the treatment of discontinuous phenomena in the DPTB and in the CCGrebank. Furthermore, I explored the relationship between the two approaches concerning the information supertags may give a discontinuous constituent parser to construct correct parses for long-range phenomena. The analysis has shown that CCG supertags may be beneficial for resolving some types of discontinuities, namely *wh*-movement, fronted quotations and *it*-extrapositions. On the other hand, the analysis did not show a transparent correlation in the case of circumpositioned quotations, extraposed dependents and subject-verb inversion and in some cases even suggested a negative effect. Noteworthy is the result that quotations seem to be particularly difficult to resolve for the CCG formalism and that they are inconsistently analysed in the CCGrebank.

In their analysis of the ML-GAP parser, Coavoux et al. (2019) found that *wh*-extractions and *it*extrapositions are, in absolute figures, the second and third largest sources of false negatives among the discontinuous phenomena. Due to the transition system and the neural network implementation being similar to the stack-free approach (Coavoux and Cohen, 2019) serving as the base for the following experiments, I expect the integration of supertag information to reduce these numbers.

Coavoux et al. (2019) report that extrapositions are the third largest class of discontinuities in the DPTB development split while they account for half of the false negatives their model produces. Unfortunately, deviating from my initial assumption, in many cases the supertags assigned in CCGrebank can not accurately account for attachment ambiguities with extraposed dependents. This class therefore most likely will remain to pose a challenge and as Coavoux et al. (2019) note, in many cases world-knowledge would be needed to resolve such cases.

It is important to point out that this prognosis is based on the assumption that the system is provided with gold supertags on the basis of CCGrebank. The actual performance will be dependent on the accuracy and style of implementation of the supertags and the variety of CCG, for instance whether categorial features are included.

On a general note not limited to discontinuities, supertag assignments alone can not be expected to help a constituent parser where they lead to spurious attachment ambiguity in their own right. Take for instance the case of figure 5.6 which was briefly discussed in the previous section. A constituent parser could not infer at what depth an attachment would be preferable based on the lexical category assignment. To deal with such cases, it might be beneficial to include information about gold parsing attachments in CCGrebank as a feature by enriching the supertags or training an additional auxiliary task.

A related area for exploration would be the integration of some notion of dependency structures into the model. While the depccg parser presented in section 5.2.2 has shown that a dependency factored approach can successfully treat attachment ambiguities inert to the CCG formalism itself, it would be worth to investigate if it could also complement a CCG-enriched constituent parser in those cases where CCG does not model discontinuous attachment ambiguities at all as in the case of extraposed dependents explored in section 5.3.3.

6. Integrating Supertags

In the following, two approaches for the integration of supertag information into discontinuous constituent parsing are explored. The *pipeline* model uses a pre-trained supertagger to assign the input sequences with lexical categories both at training and deployment. This assignment is then used as an additional input feature for the parser. In the *auxiliary* approach, the neural model is trained to predict both parsing actions and supertag assignments from a shared representation, exploiting statistical correlations between the two tasks.

6.1. Pipeline Approach

In natural language processing, it is common to process several tasks that build on each other in a sequential order. Using the outputs of one dedicated system as input for another model where the models have parameters that are independent of each other is called a *pipeline* approach (Goldberg, 2022, chapter 20).

The implementation of the stack-free discontinuous transition-based parser from Coavoux and Cohen (2019), presented in section 4, will serve as the basis for a supertagging pipeline approach. I use the supertag output of the depccg model trained on the CCGrebank (Yoshikawa et al., 2017, described in section 5.2.2) as an additional input to the parser. One option would be to supertag the input sentences and use a trainable embedding for the one-best supertag for each token as an input. Instead, I decided to use the probability distribution over the 511 lexical categories that depccg outputs for each token as an input feature. This should give the parser more useful information in cases where depccg is unsure of the correct assignment and assigns high probabilities to more than one choice.

Such an approach is called *cascading* (Goldberg, 2022, chapter 20). Instead of feeding the first model's final output to the next task, an intermediate representation that is trained to be informative for the lower-task prediction is used. Here, the last layer before the application of arg max is passed forward.

A common strategy is to *fine-tune* an upstream model when training the main task by including all or some of its parameters in the error gradient backpropagation (Jiang et al., 2020). I will not pursue such a strategy here since this would effectively enlarge the parsing model and skew a direct comparison with the baseline. Furthermore, re-optimizing the supertag predictor on the parsing task would contradict the goal of identifying what effect supertag features have on the parser.

6.1.1 Supertag Representation

On an abstract level, the depccg supertagger is a function depccg that takes a sequence of tokens $w_{0:n}$ and outputs a sequence of probability distributions over 511 categories. These include categorial features as in S[dcl] and NP[expl].

$$depccg(w_{0:n}) = \mathbf{d}_1, ..., \mathbf{d}_n \tag{6.1}$$
$$\mathbf{d}_i \in \mathbb{R}^{511}$$

 \mathbf{d}_i contain log probabilities. Thus, to obtain probabilities in the unit interval, one has to apply the exponential function to the entries. Prior to enriching the token representation, I pre-process the distribution using a two-layer feed-forward network with dropout before the second layer.

$$FF_{sup}(\mathbf{a}) = \mathbf{h}' \mathbf{W}_{2sup} + \mathbf{b}_{2sup}$$

$$\mathbf{h}' = \mathbf{d} \odot \mathbf{h}$$

$$\mathbf{d} \sim \text{Bernoulli}(drop_{sup})$$

$$\mathbf{h} = \tanh(\mathbf{a}\mathbf{W}_{1sup} + \mathbf{b}_{1sup})$$

$$\mathbf{a} \in \mathbb{R}^{511}, \ \mathbf{W}_{1sup} \in \mathbb{R}^{511 \times d_{sup}}, \mathbf{W}_{2sup} \in \mathbb{R}^{d_{sup} \times d_{sup}}, \ \mathbf{b}_{1sup}, \mathbf{b}_{2sup} \in \mathbb{R}^{d_{sup}}$$
(6.2)

Then I enrich the token representation \mathbf{w}_i by concatenating the output of the feed-forward network to the word embedding and the character-aware encoding:

$$\mathbf{w}_{i} = \mathbf{c}(w_{i}) \circ \mathbf{e}(w_{i}) \circ \mathbf{s}(w_{i})$$

$$\mathbf{s}(w_{i}) = FF_{sup}(\exp(\mathbf{d}_{i}))$$

$$\mathbf{d}_{i} \in \mathbb{R}^{511}, \ \mathbf{s}(w_{i}) \in \mathbb{R}^{d_{sup}}, \ \mathbf{w}_{i} \in \mathbb{R}^{d_{token}}, \ d_{token} = d_{char} + d_{word-emb} + d_{sup}$$
(6.3)

The 2-layer bi-LSTM transducer then operates over the enriched token representations constructing context-aware representations that encode character and word information as well as information about the supertag structure in the sentence for each token. The architecture of the model is visualised in figure 6.1.



Figure 6.1: Stack-free neural parser with supertagging pipeline approach. For clarity the token representations and the supertag prediction vectors are subscripted with words instead of indices. Figure inspired by Goldberg (2022, chapter 20.2).

6.1.2 Hyperparameters

The rest of the model resembles the implementation of Coavoux and Cohen (2019). Their choice of hyperparameters given in table 4.1 remains unchanged. Approach-specific hyperparameters are listed in table 6.1. The dimensionality of the supertag feed-forward network is set to the same value as the size of the probability distribution it receives as input in order to rule out any loss of

Architecture hyperparameters		
Dimension of supertag input representation	d_{sup}	511
Number of layers for supertag FF network		2
Activation function for supertag FF network		anh
Optimisation hyperparameters		
Dropout for supertag FF network	$drop_{sup}$	0.1

information that could impede model performance. In the comparison of results given in section 6.3 the model will be called PIPE_{rebank}.

Table 6.1: Hyperparameters of the supertag pipeline model. The other architecture and optimisation parameters match those of the baseline listed in table 4.1.

6.2. Auxiliary Approach

A common technique related to fine-tuning in pipeline-systems is *auxiliary* or *multi-task learning* which was first explored by Caruana (1997). Instead of using the predictions of a specialised system as additional input for the main task model, the idea is to jointly train individual networks for both tasks that share some part of their structures (Goldberg, 2022, chapter 20). Such an architecture learns a shared core representation that is thought to leverage synergies between the tasks.

In deep neural network models, it is common practice to arrange easier tasks at lower levels, stipulating a hierarchy between tasks. Instead of sharing the entire deep neural network, some task-specific predictors are provided with vectors resulting from lower layers of the network. This strategy is known as *collective sharing* or *stack propagation* (Goldberg, 2022, chapter 20) and was first explored for natural language processing by Søgaard and Goldberg (2016).

Auxiliary tasks based on the main task dataset effectively allow to enlarge the available training data without the need to acquire additional corpora. Often it is possible to break down the main objective into lower-level auxiliary tasks (Zhu and Sarkar, 2019). Furthermore, when training in parallel, multi-task learning can have a regulatory effect on large models that tend to overfit (Goldberg, 2022, chapter 20). Introducing auxiliary objectives forces the shared substructures to be more general which in turn helps with the treatment of data not seen during training.

The following sections present several architectures for the inclusion of CCG supertagging as an auxiliary task in the discontinuous constituent transition-based parser of Coavoux and Cohen (2019).

6.2.1 Simple Model

The original stack-free discontinuous parser of Coavoux and Cohen (2019) features two biLSTM layers where POS tagging is supervised at the first layer and parsing actions are supervised at the second layer. Thus, the first biLSTM network is shared by both POS tag prediction and parsing while the second biLSTM network is only included in the computation of parsing transitions. I propose integrating CCG supertagging as an auxiliary task by enlarging the LSTM stack through the insertion of an intermediate layer. The output of this biLSTM module is then used as input for a task-specific feed-forward network for CCG supertagging and as input for the top biLSTM network that is exclusive to parsing action prediction.

This arrangement is motivated by two factors. Firstly, Søgaard and Goldberg (2016) show that using POS-tagging as an auxiliary lower-level task for CCG supertag prediction in a threelayer recurrent neural network setting yields improvements in supertagging accuracy suggesting a natural hierarchy between the two tasks. Secondly, supervising CCG supertagging at a lower level than parsing is in line with the traditional view on supertags as "almost parsing" (Bangalore and Joshi, 1999). Furthermore, the findings of the manual analysis in section 5.3 suggested that a constituent parser would need to be able to contextualise supertag assignments to benefit from them. This motivates maintaining a final biLSTM layer that is only included in parsing action prediction and operates over the vectors that are also provided to the supertagging feed-forward network.

The proposed architecture can be formalised as follows. Each layer l consists of a biLSTM module receiving the last layer's output as input.

$$\mathbf{m}_{l,0:n} = biLSTM_l^*(\mathbf{m}_{l-1,0:n}) \tag{6.4}$$

For the first layer the token representations are used as input:

$$\mathbf{m}_{1:0:n} = biLSTM_1^*(\mathbf{w}_{0:n}) \tag{6.5}$$

In the baseline model, the output of the first LSTM layer is added to the output of the second LSTM layer (residual connection). The combined output is directed to the feed-forward networks for parsing action prediction. I opted to replicate the simple residual connection from the second LSTM's input to its output for the additional layer. The first LSTM does not feature a residual connection since its input and output dimensionalities do not match.

$$\mathbf{m}_{l,0:n} = biLSTM_l^*(\mathbf{m}_{l-1,0:n}) + \mathbf{m}_{l-1,0:n} \quad \text{for } l \in \{2,3\}$$
(6.6)

Now, the feed-forward network for POS-tagging FF_{POS} still receives \mathbf{m}_1 as input while the parsing action scorers FF_{label} and FF_{struct} are provided with the last layer \mathbf{m}_3 . For supertagging, a specialised feed-forward network with the same architecture as FF_{POS} is introduced that receives \mathbf{m}_2 as input. Let B be the alphabet of supertags.

$$FF_{sup}(\mathbf{x}) = \mathbf{x}' \mathbf{W}_{sup}$$
$$\mathbf{x}' = \mathbf{d} \odot \mathbf{x}$$
$$\mathbf{d} \sim \text{Bernoulli}(drop_{sup})$$
(6.7)

$$\mathbf{W}_{\mathbf{sup}} \in \mathbb{R}^{dim_{hid} \times |B|}, \ drop_{sup} \in [0, 1]$$

For an input sequence $w_{0:n}$ the probability of a sequence of supertages $s_{0:n} = s_1, ..., s_n$ is then computed as follows:

$$P(s_{0:n}|w_{0:n}) = \prod_{i=1}^{n} \operatorname{Softmax}(FF_{sup}(\mathbf{m}_{2,i}))^{[\operatorname{idx}_{B}(s_{i})]}$$

$$\mathbf{m}_{2,i} \in \mathbb{R}^{d_{hid}}, \ FF_{sup}(\mathbf{m}_{2,i}) \in \mathbb{R}^{|B|}$$
(6.8)

In the train split of the CCGrebank, 1574 distinct lexical category assignments occur, thus |B| = 1574. Only 511 supertags appear 10 or more times (Honnibal et al., 2010). The large number of supertags and the small amount of training data for two thirds of the categories may make it hard to achieve useful generalisations for rare phenomena.

An overview of the modified architecture of the model is given in figure 6.2.

Objective Function I perform supervised learning using the lexical category annotations provided by the CCGrebank. For training, sections 2-21 are used so that the auxiliary task does not extend the domain of the main task DPTB dataset.

The neural network parameters are optimised during training. The objective function introduced in section 4.5 now includes the supertagging loss.

$$\mathcal{L} = \mathcal{L}_t + \mathcal{L}_p + \mathcal{L}_{sup}$$
$$\mathcal{L}_{sup} = -\sum_{i=1}^n \log P(s_i | w_{0:n})$$
(6.9)



Figure 6.2: Stack-free neural parser with a three-layer biLSTM network and supertagging as an auxiliary task. For clarity the token representations and the supertag prediction vectors are subscripted with words instead of indices. Figure inspired by Goldberg (2022, chapter 20.2).

In practice, instead of strictly alternating between the three tasks, a strategy called *uniform* sampling (Sanh et al., 2019), I iterate through the randomly shuffled list of all available pairs of task and annotated sentence. At each step, the network performs an optimisation for the chosen pair. This is known as proportional sampling where the probability of a task is proportional to its corpus size and was first proposed by Sanh et al. (2019). Each epoch iterates through the full list. The pseudo-code for this training procedure is given in algorithm 6.1.

This modification to the model was necessary since aligning the CCGrebank with the DPTB is not possible due to a few missing sentences in the CCGrebank (0.56%, i.e. 274 of the 49.208 sentences in the Penn Treebank). These sentences were already left out in the original CCGbank since they could not be converted to CCG by the translation algorithm utilised by Hockenmaier and Steedman (2007). They note that 107 of these cases include sentential gapping and 66 non-sentential gapping while only a minor number of the remaining sentences was excluded because of an erroneous treatment of long-range dependencies. According to Hockenmaier and Steedman (2007) the translation algorithm incorrectly identified a complement as an adjunct. Despite their small number and without investigating these exclusions any closer, it could of course be the case that they would provide important information for the treatment of certain discontinuous phenomena that already occur very infrequently in the corpus (cf. section 5.3). However, manually reintegrating these sentences would be very labour-intensive.

Algorithm 6.1 Proportional task sampling. Inspired by (Goldberg, 2022, chapter 5).

Input: a list of task-dataset pairs M, a number of training epochs n1: $D \leftarrow []$ 2: for all $\langle task, dataset \rangle$ in M do 3: for all training example $\langle x, y \rangle$ in *dataset* do $D.append(\langle task, x, y \rangle)$ 4: end for 5:6: end for 7: for all epoch = 1 to n do for all training example $\langle task, x, y \rangle$ in D do 8: loss_node \leftarrow build_computation_graph_{task}(x, y) 9: 10: loss node.forward() gradients \leftarrow loss_node.backward() 11: 12:update_parameters_{task}(gradients) 13: end for 14: $\operatorname{shuffle}(D)$ 15: end for

6.2.2 Residual Connections

In deep multi-layer neural network models trained with gradient descent the effective backpropagation of errors can be difficult to achieve. A central challenge stems from the problem of vanishing gradients where the gradient being backpropagated through the network decays with each additional layer, hardly affecting early parameters of the system, or grows exponentially due to repeated scaling (Bengio et al., 1994; Hochreiter et al., 2001; Glorot and Bengio, 2010). Thus, with increasing depth, a model tends to be harder to train effectively.

Residual, skip or *shortcut connections* are used to mitigate this effect by allowing for the direct flow of gradients from the final loss computation to lower parts of the system through the establishment of connections that leave out certain components or layers of the network. Motivated by the *ResNet* model of He et al. (2016), this is typically. This strategy has the additional benefit of enabling the model to easily learn the identity function which tends to be difficult for models with non-linear activation functions (He et al., 2016).

Stacked recurrent neural networks are connected both horizontally, with the hidden units in the same layer representing a time-step in the sequence, and vertically, as a feed-forward process through the layers at the same time-step. The architecture of LSTMs is designed to prevent the problem of vanishing gradients for repeated horizontal application by enforcing constant error flow along the sequence using a special cell unit controlled by gates that perform element-wise multiplication to *forget* information in the cell or add new (transformed) *input* (Hochreiter and Schmidhuber, 1997; Sundermeyer et al., 2012) (cf. definition 4.18). However, when stacking several LSTM models vertically, the problem reoccurs and leads to slow and difficult training (Wu et al., 2016b).

Therefore, it is common practice to include residual connections in stacked LSTM models. They were first proposed for stacked recurrent neural networks by Raiko et al. (2012) and explored for LSTM stacks specifically by Wu et al. (2016b). As explained in section 4.5, Coavoux and Cohen (2019) use a simple skip connection that adds the output vector of the first LSTM layer to the output of the second layer element-wise. This allows information to bypass the second LSTM network.

With larger vertical LSTM stacks, the question of where to add residual connections becomes non-trivial. While simply using an additive residual skip connection for every LSTM layer or blocks of LSTM layers appears to be the standard approach, several alternatives have been suggested, one of which motivates a modification to the three-layer auxiliary approach at hand. Inspired by the gate design of the LSTM, Wu et al. (2016a) propose integrating a multiplicative gate for the
residual connection directly into the LSTM hidden unit. Their best scoring approach is to connect the output of layer l - 2 to the cell output of layer l. This entails the following change to the standard LSTM equation given in definition 4.18.

$$\mathbf{h}_{l,t} = \mathbf{o}_{l,t} \odot \tanh(\mathbf{c}_{l,t-1}) + \mathbf{r}_{l,t} \odot \mathbf{h}_{l-2,t}$$

$$\mathbf{h}_{l-2,t}, \mathbf{o}_{l,t}, \mathbf{c}_{l,t-1}, \mathbf{r}_{l,t}, \mathbf{h}_{l-1,t} \in \mathbb{R}^{d_{out}}$$
(6.10)

where t represents the time step in the sequence and $\mathbf{r}_{l,t}$ is a gate computed alongside the other gates in the LSTM:

$$\mathbf{r}_{l,t} = \operatorname{sigm}(\mathbf{h}_{l,t-1}\mathbf{W}_{\mathbf{r},l} + \mathbf{h}_{l-1,t}\mathbf{U}_{\mathbf{r},l})$$

$$\mathbf{r}_{l,t} \in \mathbb{R}^{d_{out}}, \ \mathbf{W}_{\mathbf{r},l} \in \mathbb{R}^{d_{out} \times d_{out}}, \ \mathbf{U}_{\mathbf{r},l} \in \mathbb{R}^{d_{in} \times dim_{out}}$$
(6.11)

When $\mathbf{r}_{l,t}$ equals 1 at some position, the output of $\mathbf{h}_{l-,t}$ at that position gets passed through unobstructed. When it equals 0, the network behaves like a traditional stacked feed-forward network (Wu et al., 2016a). This strategy has the benefit of allowing the network to choose the scales with which the individual entries in the tensor should be directed forward to improve the result depending on the sequential context encoded in the internal states. It should lend itself well to a multitask approach that tries to build a general feature representation benefitting several objectives. An LSTM layer can fully specialise on retrieving context-dependent information for a certain task while being able to directly retrieve information already encoded by a lower level task depending on input and context. Wu et al. (2016a) report that their proposal improves results for CCG supertagging over a model simply adding the penultimate layer output to $\mathbf{h}_{l,t}$.

I would like to propose a complementary perspective on skip connections in the context of hierarchical multi-task models. Direct additive connections from early layers to the final layer allow for the unobstructed backpropagation of errors from the main task's feed-forward network to lower levels from which auxiliary tasks are supervised. In a fully connected architecture, with shortcut connections in each layer, I assume that for layer l this significantly works towards the construction of mutually beneficial representations with respect to tasks supervised on levels $\geq l$. This, however, presupposes closely related tasks which is a key condition for effective multi-task learning as explained above. When auxiliary tasks are expected in some systematic way to be synergistic in a certain number of classes but noisy and unhelpful for the main task in others, using gates to enable the model to dampen or to increase the intensity of values in skip connections dynamically and individually depending on context may allow the model to benefit from less closely related auxiliary tasks, as in the case of CCG supertagging for discontinuous constituent parsing. This might be a balanced compromise between an ungated residual LSTM stack based on element-wise summation and a computationally expensive but more informative residual connection realised through concatenation with free access to all preceding outputs.

Unfortunately, using a custom LSTM cell architecture in pytorch would mean a significant increase in computational cost. The baseline model of Coavoux and Cohen (2019) uses the fast native LSTM implementation written in the C++ programming language. To maintain this efficiency and to better compare the effect of a gated residual connection to the baseline implementation of Coavoux and Cohen (2019) that skips only a single LSTM layer at a time instead of two, I explore simply using the current and the last layer output as inputs to the gate. The residual connection is established from the output of the preceding layer and does not alter the hidden unit of the LSTM but is connected thereafter. The exploration of Wu et al. (2016a)'s approach is left for future work. I expect the LSTM hidden state of the current layer to be able to encode the contextual information provided by the preceding time step hidden state $\mathbf{h}_{l,t-1}$ used in Wu et al. (2016a). My approach can also be seen as a variation of the concept of *highway connections* (Srivastava et al., 2015).

Definition 6.1 (Gated Residual).

A gated residual module can be defined as a function that takes two vectors $\mathbf{x}, \mathbf{y} \in \mathbb{R}^{dim}$ and has learnable parameters $\mathbf{b} \in \mathbb{R}^{dim}, \mathbf{W} \in \mathbb{R}^{2dim \times dim}$ for some dimensionality dim:

$$Res(\mathbf{x}, \mathbf{y}) := \mathbf{x} + \mathbf{r} \odot \mathbf{y}$$
$$\mathbf{r} = \text{sigm}((\mathbf{x} \circ \mathbf{y})\mathbf{W} + \mathbf{b})$$

A visualisation of the gated residual module is given in figure 6.3.



Input

Figure 6.3: Design of the gated residual connection.

For each LSTM layer, the output and the previous layer output are combined via a layer-specific gate. The result of this computation is then forwarded to the specific task feed-forward network(s) supervised at this level and to the next LSTM recurrent layer as well as the following gated residual module.

$$\mathbf{m}_{l,t} = Res_l(biLSTM_l^*(\mathbf{m}_{l-1,0:n})^{[t]}, \mathbf{m}_{l-1,t})$$
(6.12)

Furthermore, I suggest adding a shortcut connection from the LSTM stack input, i.e. from the token representations, by performing an initial linear transformation that converts this tensor into the LSTM hidden layer dimensionality d_{hid} . This way, the first LSTM layer can also carry a residual module. The full modified architecture is visualised in figure 6.4.

6.2.3 Increasing Model Width

To allow for a fair comparison with the original model of Coavoux and Cohen (2019), the simple CCG approach maintains the LSTM hidden state dimension of 400. However, against the backdrop of the manual analysis in section 5.3 which suggested that CCG lexical category assignments might statistically correlate with DPTB descriptions for some phenomena while being unhelpful for resolving others, one can expect a multi-task model optimised for both tasks to require a higher dimensionality to establish a representation that can encode the additional information in uncorrelated cases.

At the same time, Goldberg (2022, chapter 20.2.6) points out that if an auxiliary model with k tasks only started to show improvements if the dimensionality was simply enlarged k-fold, the network likely did not established shared knowledge for several tasks and the improvement was only built on increased neural capacity. Therefore, in order to identify if the model demands increased capacity to support the additional task in a shared representation, I also perform training with the LSTM hidden dimension d_{hid} enlarged by factor 1.5 to 600.



Figure 6.4: Stack-free neural parser with a three-layer biLSTM network, supertagging as an auxiliary task and gated residual connections to cell outputs. For clarity the token representations and the supertag prediction vectors are subscripted with words instead of indices. Figure inspired by Goldberg (2022, chapter 20.2).

6.2.4 Feature Bootstrapping

Zhu and Sarkar (2019) perform multi-task learning for lexicalised tree adjoining grammar (LTAG) supertagging by deconstructing complex TAG categories into classes of substructures like *root*, which naturally refers to the label of the root node of the lexical tree, or *sketch*, which captures only the structure of the lexical category without labels. Six objectives, including supertag prediction, are trained using separate biLSTM models. After training, the supertagger is used to generate a certain number of best supertags. The auxiliary models then compute the probability of these assignments' substructures from which the best supertag among the predictions is determined. Zhu and Sarkar (2019) report consistent improvements in a range of settings compared to a single-task network for supertagging. By reusing the training set for the supertagging task the authors effectively generate more data instances without enlarging the dataset.



Figure 6.5: Example for root and sketch subtasks for an LTAG lexical tree as proposed by Zhu and Sarkar (2019).

Figure 6.5 shows an example for the decomposition of lexical trees in LTAG as proposed by Zhu and Sarkar (2019). In the following, I borrow from this idea and present the extraction of subtasks for CCG supertags.

Argument For each supertag the rightmost/outermost argument as well as its directionality symbolised by "-" for left and "+" for right are concatenated. For categories without arguments, "+" is used as a dedicated label. (6.1) gives an example of this schema. 148 distinct arguments are found in the training split.

(6.1) a. I like this train NP $(S[dcl] \setminus NP)/NP NP/N N$ +NP++N+b. You sing beautifully $S[dcl] \setminus NP (S \setminus NP) \setminus (S \setminus NP)$ NΡ +-NP $-(S \setminus NP)$

Head The *head* subtask predicts the innermost return type of a CCG category. It is also known as the *range* of a category (Steedman, 2000, chapter 3). There are 26 distinct head tags.

Sketch The *sketch* ignores atomic categories in function category descriptions and only represents the overall argument structure of the supertag. It is inspired by the sketch task of Zhu and Sarkar (2019) who found that it helps with disambiguation for TAG. To extract the sketch feature from a CCG supertag, all atomic categories are replaced by an X. The number of distinct sketch labels is 212.

Given the hierarchical framework of the approach at hand, subtasks will be supervised at a dedicated additional layer between POS-tagging and supertagging. The three subtasks are supervised from the same layer since there is no obvious hierarchy regarding their difficulty and inter-relationship. An illustration of the full modified architecture is given in figure 6.6.

Each of the three subtasks has a dedicated feed-forward network that outputs a distribution over the distinct labels found in the training dataset. It is optimised in parallel according to algorithm 6.1. The inclusion of these three tasks leads to twice the amount of training instances per epoch compared with the simple auxiliary model (237,940 sentences in total). In case the approach shows to be successful, the extraction of other subtasks like functor or full argument structure could be performed.



Figure 6.6: Stack-free neural parser with a multi-task four-layer biLSTM network, including gated residual connections. Supertag-component tagging (sketch, head, arg) and supertagging (CCG) are added as auxiliary tasks, with supertagging being supervised one layer higher than the subcomponents. For clarity the token representations and the supertag prediction vectors are subscripted with words instead of indices. Figure inspired by Goldberg (2022, chapter 20.2)

6.2.5 Head-Dependency Structure

The CCGbank and its derivative, the CCGrebank, provide head-dependency structures that encode the saturation of category arguments. The dependencies hold between a lexical functor and the heads of the constituents that fill the functor's argument slots (Hockenmaier and Steedman, 2007).

Figure 6.7 shows an example CCG derivation from the CCG rebank as well as its head-dependency structure depicted using directed labelled edges. The edge labels refer to the respective argument slot that is filled in the category the edge originates in. For transparency, I added subscripts to the supertag assignments that make explicit the argument number. As can be seen in the case of *does*, a word can have two incoming edges. This stems from the fact that adjuncts are not treated as arguments of their governor, but instead take it as their argument, as prescribed by combinatory categorial grammar.



Figure 6.7: Derivation from the CCGrebank alongside its head-dependency structure visualised as labelled directed edges above the tokens.

The head-argument information used here closely correlates with the CCGrebank gold derivation and is less akin to conventional dependency descriptions. This can be seen in figure 6.7 where *what* is not encoded as an argument of *believe* but as taking a constituent with *does* as the lexical head as an argument. Therefore, while this information might be useful to solve projective attachment ambiguities (cf. figure 5.6) its effect on difficult discontinuous cases that arise from extraposed dependents where the CCGrebank unhelpfully analyses constituents as adjuncts of higher-level clauses (cf. figure 5.23) remains to be seen.

To test the usefulness of head-dependency structure prediction as an additional auxiliary task, I encode this information into two distinct sequence labelling tasks encompassing left dependencies and right dependencies respectively. For each position i, the relative position of the words right of i connected with an outgoing edge as well as the corresponding argument number is predicted (*right action*). The same is done for left dependents (*left action*). (6.4) shows the tags resulting from the annotation in figure 6.7 where each outgoing edge takes the form <argument_number>:<relative_position>. 0 represents the absence of outgoing edges. In cases of multiple arguments they are separated by an underscore symbol. To keep the number of distinct categories manageable, relative distances above absolute value 5 are all set to 5 or -5 respectively.

(6.4)		\mathbf{So}	what	does	George	Bush	really	believe	?
	right:	1:1	1:1	$2:2_1:4$	1:1	0	0	0	0
	left:	0	0	0	0	0	1:-3	0	0

In the train split, 351 distinct left action labels and 955 right action labels are predicted. This corresponds with the right-branching tendency of the English language (Kiparsky, 1996). The two sequence tagging tasks are treated as higher-level tasks compared to CCG supertagging since they encode relationships between different supertag assignments in the sentence, and therefore, in a four-biLSTM-layer network the tasks are supervised in the following order: POS-tags, CCG supertags, action labels, parsing (from bottom to top).

6.2.6 Other Auxiliary Tasks

In addition to auxiliary tasks based on the CCGrebank, I perform a number of experiments to assess the suitability of other syntactically motivated sequence labelling tasks for discontinuous constituent parsing. Due to space limitations, the representations and formalisms which the auxiliary tasks are derived from will not be introduced in detail and an interested reader will be referred to more comprehensive introductions and formal accounts.

LCFRS Ruprecht and Mörbitz (2021); Ivliev (2020) present an algorithm for the extraction of supertags in the LCFRS formalism from discontinuous constituent treebanks. For LCFRSs have shown to be able to adequately capture the syntactic descriptions in the DPTB (Evang, 2011), LCFRS supertagging immediately suggests itself as an auxiliary task for neural parsing. Using the **prepare_data** script of the LCFRS Supertag Parser released by Ruprecht and Mörbitz (2021)⁴⁶, I converted the DPTB into a format annotated with LCFRS supertags. The model is trained to predict these from an intermediate layer between POS tagging and parsing in the same fashion as suggested for CCG supertags in section 6.2.1.

Unfortunately, the extraction process results in a large number of distinct supertags. In total 4504 lexical categories are assigned in sections 2–21. Since this number is three times larger than the 1574 distinct CCG categories found in the CCGrebank train split, LCFRS-supertag prediction is likely to be more difficult to learn for the neural model. This is an important thing to keep in mind when comparing the results of different auxiliary tasks to draw conclusions about their informativeness for discontinuous constituent parsing.

Chunking Chunking separates a sentence into simple constituent types like NP or VP and is sometimes called *shallow parsing* (Collobert and Weston, 2008) or *partial parsing* (Jurafsky and Martin, 2009, chapter 13.5). Each word is only assigned to a single constituent. Using the so-called IBO notation (Jurafsky and Martin, 2009, chapter 13.5), chunking can be treated as sequence labelling by introducing special symbols where the beginning word of a constituent is marked with B and internal parts of a chunk are marked with I. Furthermore, if elements are outside of any chunk they are simply marked with O. Constituent type assignment and I, B and O features are treated as a single task, which doubles the total number of tags. (6.5) shows an example sentence chunked with brackets as well as the equivalent IBO-encoding taken from Jurafsky and Martin (2009, chapter 13.5).

 $^{^{46}}$ https://github.com/truprecht/lcfrs-supertagger

In order to integrate chunking as an auxiliary task, I use the dataset made available as part of the CoNLL-2000 shared chunking task (Tjong Kim Sang and Buchholz, 2000). It is based on sections 15–18 (train split, 8936 sentences) and section 20 (test split, 2012 sentences) of the Penn Treebank. Therefore, the inclusion of this task does not enlarge the domain beyond the sentences used to train the main parsing task (PTB sections 2–21). Since no development split is defined, I use the last 399 sentences of the train split for development and remove them from the training set.

The model is trained to predict the chunking labels from an intermediate biLSTM layer between POS tagging and parsing in the same fashion as suggested for CCG supertagging in section 6.2.1. There are 22 distinct labels in the test split. Since task sampling is proportional to the dataset size, chunking only accounts for 9.7% of total training instances.⁴⁷ For all other models explored in this work, the individual task datasets are roughly equal in size.

Dependency Parsing Dependency grammars are a class of grammatical theories centred around the notion of dependency as opposed to constituency. While constituent phrase structure grammars arrange words into hierarchical constituents, dependency grammars exhibit word-to-word syntactic relations expressed through labelled directed edges (Jurafsky and Martin, 2009, chapter 12.7).

A dependency descriptions can be formalised as a tree with labelled edges over the tokens in a sentence. Each token has one incoming edge and zero or more outgoing edges. A single token is connected via an incoming edge with a special *root* symbol. For each token, the source of its incoming edge is called its *head*. This notion closely correlates with the concept of a constituent head discussed in section 3.2.6 (Jurafsky and Martin, 2009, chapter 12.7.1).

Several suggestions for the joint learning of constituent and dependency parsing have been made. Strzyz et al. (2019a) perform sequence labelling-based projective constituent and dependency parsing in a multi-task framework that yields improvements over comparable single-task models for both tasks. Zhou et al. (2020a) explore dependency parsing and span-based constituent parsing together with semantic parsing. To the best of my knowledge, no exploration of synergistic parsing of dependency relations and discontinuous constituents has been performed to this date.

Strzyz et al. (2019b) compare several approaches for the conversion of dependency parsing into a sequence labelling task. Among these approaches, so-called *relative POS-based encoding*, which was first used by Spoustová and Spousta (2010), leads to the best parsing score for single task dependency parsing. A dependency tree for a sentence $w_1, ..., w_n$ is linearised by assigning each token w_i the label of its incoming edge l_i , the POS tag t_j of its head w_j and an integer o_i such that if i < j, w_j is the o_i th closest of the words to the right of w_i that bear POS tag t_j and if $j < i, w_j$ is the $-o_i$ th closest of the words to the left of w_i that have POS tag t_j . This results in a triple (t_j, o_i, l_i) that is treated as a distinct tag. As Strzyz et al. (2019b) note, such a scheme is closer to the notion of valency, i.e. argument structure, when compared to strictly encoding relative position.

Figure 6.8 shows an example dependency description together with the corresponding relative POS-based encoding. For instance, the label VB_1_aux of the token *does* encodes an edge with label aux that connects the first token to the right of *does* that has POS tag VB with *does*. In case of a root relation, the distinct label root_-1_root is used as a standard.

I convert the dependency PTB (de Marneffe et al., 2006) into a dataset tagged with relative POS-based labels. The PTB POS tags are taken as the base. Subtypes for syntactic relations like :poss in nmod:poss are removed to reduce the number of distinct labels. Furthermore, as suggested by Spoustová and Spousta (2010), the maximum relative position is capped at -3 and 3. The conversion results in 1634 distinct labels for the train split of sections 2–22 of the PTB.

⁴⁷Parsing and POS tagging amount to 39,832 instances each.



VB_1_advmod VB_1_dobj VB_1_aux NNP_1_compound VB_1_nsubj VB_1_advmod root_-1_root VB_-1_punct

Figure 6.8: Dependency description using directed arcs from the dependency PTB as well as the corresponding relative POS-based encoding underneath.

In the experiment, dependency parsing is treated as a parallel task to constituent parsing and also supervised from the final layer. In order to be able to compare the model with the other auxiliary task experiments, the network maintains a 3-layer biLSTM stack. This amounts to a similar architecture as in Strzyz et al. (2019a) with the notable difference that they deconstruct the parsing tasks into 3 constituent labelling tasks and 2 dependency labelling tasks whereas I employ only one task per paradigm. Furthermore, the transition-based approach used here allows for discontinuous constituents while the sequence labelling encoding of constituent trees used by Strzyz et al. (2019a) does not.

LTAG Tree adjoining grammar (TAG) is a mildly context-sensitive formalism that was first introduced by Joshi et al. (1975). A lexicalised variant of TAG was proposed by Schabes and Joshi (1991) which led to the formalisation of lexicalised tree adjoining grammar (LTAG). An illustrative example of an LTAG elementary tree has been given in figure 6.5. Much of the work on supertags like the extraction of supertagged corpora from treebanks and the development of parsers that use supertaggers was initially conducted on LTAG. Bangalore and Joshi (2010b) give a comprehensive overview of important research in this regard.

The TAG extracted PTB of Chen et al. (2006) is commonly used as a basis for training LTAG supertaggers (Zhu and Sarkar, 2019; Kasai et al., 2018). Unfortunately, I was not able to acquire this corpus. Instead, I used the publicly available⁴⁸ LTAG-spinal treebank developed by Shen et al. (2007). LTAG-spinal omits subcategorisation information and the argument-adjunct distinction. This reduces the overall number of distinct supertags and the average number of elementary trees associated with a lexical item.

I train a discontinuous constituent parser with LTAG-spinal supertag prediction as an intermediate auxiliary task using the same 3-layer architecture as in the CCG model. The LTAG-spinal treebank is based on the PTB and split into the same training, development and testing datasets. 1202 distinct tag symbols occur in the training set.

6.2.7 Hyperparameters

The hyperparameters for the multi-task models introduced in the preceding section are given in tables 6.2 and 6.3. All remaining parameters are equal to those in the baseline as listed in table 4.1. The task specific feed-forward networks have the same architecture and are assigned the same input dropout as the POS tag feed-forward network.

⁴⁸Available at: https://www.cis.upenn.edu/~xtag/spinal

Architecture hyperparameters		CCG	$\mathrm{CCG}^{\mathrm{gate}}$	$\mathrm{CCG}_{600}^{\mathrm{gate}}$	$\mathrm{LCFRS}^{\mathrm{gate}}$	Chunkgate	$\mathrm{Dep}^{\mathrm{gate}}$	$LTAG^{gate}$
Dimension of sentence biLSTM	d_{hid}	400	400	600	400	400	400	400
Number of biLSTM layers		3	3	3	3	3	3	3
Number of auxiliary tasks labels		1574	1574	1574	4504	22	1634	1202
Type of residual connection		add	gate	gate	gate	gate	gate	gate

Table 6.2: Hyperparameters of the supertag models with one auxiliary task. The other architecture and optimisation parameters match those of the baseline listed in table 4.1.

Architecture hyperparameters	$\mathrm{CCG}^{\mathrm{gate}}_{\mathrm{multi}}$	$\mathrm{CCG}_{\mathrm{dep}}^{\mathrm{gate}}$
Dimension of sentence biLSTM d_{hid}	400	400
Number of biLSTM layers	4	4
Number of auxiliary tasks	4	3
Number of auxiliary task labels	148, 26, 212; 1574	1574;351,955
Type of residual connection	gate	gate

Table 6.3: Hyperparameters of the supertag models with more than one auxiliary task. The task label numbers are listed according to supervision depth from bottom to top. A semicolon indicates the next layer. The other architecture and optimisation parameters match those of the baseline listed in table 4.1.

Due to the promising results of the gated residual model CCG^{gate} on the development dataset compared to the ungated residual model CCG (cf. section 6.3), the other experiments adhere to the gated residual approach.

Since all auxiliary models enlarge the baseline approach in some way, it is important to isolate the contribution of this change in architecture on the model performance. Therefore, for every combination of biLSTM stack depth, biLSTM hidden dimensionality and type of residual connection used in the experiments, I also train a control model that does not include auxiliary tasks at the intermediate biLSTMs. These are called CTR_3 , CCG_2^{gate} , CTR_3^{gate} , $CTR_{3,600}^{gate}$ and CTR_4^{gate} where the subscript indicates the number of biLSTM stacks. The auxiliary models will be compared with the respective control model to reliably assess the effect of training with auxiliary tasks.

6.3. Experiments

I carried out experiments that investigate the effect of integrating supertags into discontinuous constituent parsing by training the systems proposed in the preceding sections. In section 6.3.1, I present the experimental protocol, then I discuss the results of the experiments in section 6.3.2. Section 6.3.3 gives a per-phenomenon analysis of the pipeline model and the best scoring auxiliary-task model on discontinuous structures and section 6.3.4 evaluates these models against several error types. Section 6.3.5 investigates noticeable effects of supertag integration in more detail via a sample comparison of gold and predicted trees. Finally, I give a comparison with previously published discontinuous constituent parsing results in section 6.3.6.

All approaches are implemented as modifications of the discontinuous stack-free transitionbased parser created by Coavoux and Cohen (2019) which I have presented in section 4. The code including all trained models and results will be made available at https://github.com/ filemon11/discoparset-supertag.

The modified codebase includes a dynamic multi-task framework. Via a hyperparameter one can choose the number of layers for the biLSTM stack and what tasks should be shared up to which layer. Combining several tasks based on different corpora is possible without manual code changes. This allows for a straightforward exploration of a multitude of architectures. Furthermore, I included the optional evaluation of auxiliary task accuracy on the respective development and test corpora in the main evaluation function. This way, a multi-task-model's capabilities on an auxiliary task can be easily compared with previously reported scores of other sequence labelling systems.

6.3.1 Experimental Protocol

In order to use the dynamic oracle model of Coavoux and Cohen (2019) as a baseline and to allow a reliable comparison with their results the protocol remains unchanged. The models here also use the dynamic oracle and are trained with the ASGD algorithm (Polyak and Juditsky, 1992) for 100 epochs. Every 4 epochs the model is evaluated on the development set and selected if it surpasses the previous best scoring instance in parsing F-score.

Training took approximately two days for the PIPE_{rebank} model and three days for the simple auxiliary CCG model using an NVIDIA GeForce RTX 2060 SUPER 8GB GPU. For the model with the highest number of parameters, CCG_{multi}^{gate} , training took roughly a week. Evaluation is performed using the standard evaluator **discodop** (van Cranenburgh et al., 2016) and follows the standard procedure of ignoring punctuation and root symbols. **discodop** retrieves the number of matching brackets in the parsing prediction compared to the gold annotation using precision, recall and F-score which has been agreed upon as the common metric for constituency parser evaluation (Black et al., 1991).

6.3.2 General Effects of Supertag Integration

Table 6.4 shows the results of the experiments on the development set of the DPTB corpus (section 22) as well as the baseline score as reported by Coavoux and Cohen (2019) and a rerun with the baseline model architecture and parameters using the modified codebase. The evaluation includes precision, recall and F-score as well as precision, recall and F-score on discontinuous constituents only. Furthermore, accuracy is reported for all auxiliary tasks. All models achieve convergence within the 100 training epochs.

Baseline The differences between Coavoux and Cohen (2019) and the rerun with the modified codebase can be attributed to the proportional task sampling used here that does not strictly alternate between tasks but chooses them at random (cf. section 6.2) and to other non-deterministic factors when training the network (e.g. dropout). Since the experiments here are all performed using the modified code, the rerun is taken as a baseline in the comparisons that follow.

Pipeline The PIPE_{rebank} model is unable to make improvements over the baseline parser. Slight decreases in general precision (-0.35 in absolute score) and discontinuous precision (-5.01) result in degradations of F-score. Providing supertag distributions generated by the pre-trained rebank **depccg** model to the parser seems to decrease the model's syntactic capabilities questioning the helpfulness of CCG supertags for resolving discontinuous constituency. Interestingly, recall slightly increases in the general (+0.08) and in the discontinuous case (+1.79).

A possible cause for the poor results of the PIPE_{rebank} model might be the quality of supertag distributions generated by the upstream supertagger. False lexical category assignments used as input to the parser could lead the transition system astray, especially in cases of only sparsely attested phenomena where the amount of training data cannot teach the parser successfully to counteract wrong information at input level. I test this hypothesis by evaluating the depccg rebank model on the CCGrebank development split which yields an accuracy of 81.61 when taking the 1-best supertag — much lower than the CCG auxiliary model's supertagging score, which amounts to 92.64. For 89% of tokens the best supertag is amongst the 3 highest scoring tags.

depccg rebank was trained on the 511 most frequent supertags in the CCGrebank training dataset. 1% of the tokens in the development split are not covered by this set. Strikingly, for those

Model	Р	R	F	DP	DR	DF	POS	SUP	HEA	ARG	SKE	LEF	RIG
original	91.5	91.3	91.4	76.1	66.4	70.9	97.2	_	_	_	_	_	_
baseline	91.45	91.15	91.30	79.95	65.25	71.85	97.25	_	_	-	-	_	_
PIPE _{rebank}	91.1	91.23	91.16	74.94	67.04	70.77	97.25	_	_	_	_	_	_
$\operatorname{PiPE_{bank}}$	91.52	91.55	91.54	75.65	65.47	70.19	97.30	_	_	_	_	_	_
CCG	91.32	91.31	91.32	78.92	62.11	69.51	97.24	92.64	_	_	_	_	_
Ctr_3	91.34	91.14	91.24	72.25	61.88	66.67	97.18	_	_	_	_	_	-
$\operatorname{Ctr}_2^{\operatorname{gate}}$	91.67	91.2	91.43	81.23	65.02	72.23	97.31	_	_	-	_	_	_
$\mathrm{CCG}^{\mathrm{gate}}$	91.86	91.67	91.77	82.48	68.61	74.91	97.22	92.85	_	_	_	_	-
$\mathrm{Ctr}_3^{\mathrm{gate}}$	91.82	91.56	91.69	78.2	69.96	73.85	97.23	_	_	_	_	_	-
CCG_{600}^{gate}	91.85	91.57	91.71	78.36	66.59	72.0	97.21	92.66	_	_	_	_	_
$\operatorname{Ctr}_{3,600}^{\operatorname{gate}}$	91.99	91.76	91.88	80.63	69.06	74.4	97.26	_	_	_	_	_	-
$\mathrm{CCG}^{\mathrm{gate}}_{\mathrm{multi}}$	91.61	91.4	91.51	79.63	68.39	73.58	97.18	92.49	95.31	95.51	93.05	_	-
$\mathrm{CCG}^{\mathrm{gate}}_{\mathrm{dep}}$	91.57	91.24	91.4	83.33	65.02	73.05	97.20	92.67	_	-	_	95.64	92.76
$\operatorname{Ctr}_{4}^{\operatorname{gate}}$	91.82	91.71	91.77	79.53	67.94	73.28	97.16	_	_	_	_	_	_

Table 6.4: Results of the experiments on the DPTB development set. P, R and F are precision, recall and F-score. DP, DR and DF represent discontinuous precision, recall and F-score. POS and SUP are POS-tagging and supertagging accuracy respectively. HEA, ARG and SKE refer to the supertag subcomponents head, argument and sketch while LEF and RIG are left and right action labels. The best score for each metric across all models is set in bold if the metric is reported for more than one model. original refers to the dynamic oracle results of Coavoux and Cohen (2019) while baseline refers to the rerun using a modified codebase and proportional task sampling.

tags that can be predicted, the gold tag has an average rank of 19.3 in the distributions which the model outputs. Thus, there are some strong outliers with very low score in the prediction. A more detailed analysis showed that in 67.7% of the cases where the gold supertag is not amongst the best scoring 20 labels, the supertag is "," while the 1-best prediction is "conj". Perhaps there was a mix-up of datasets when the depccg rebank model was trained.

In order to check whether this problem is specific to this supertagger variant or if the pipeline approach is simply unsuitable, I decided to train an additional model (PIPE_{bank}) with the standard **depccg** trained on the original CCGbank as the supertagger. This model performs better for standard F-score (+0.24 compared to the baseline) but unfortunately even lower for discontinuities (-1.66). Using this model, for 98.4% of tokens in the DPTB development split the gold supertag is among the 3-best predictions which is why I conclude that there is indeed some kind of inconsistency in the rebank supertagger. For this reason further comparisons of the experiments in regards to discontinuous constituent analysis in sections 6.3.3 and 6.3.4 will be based on the results of PIPE_{bank}. Unfortunately, when comparing PIPE_{bank} and the CCG auxiliary approach differences in results might be caused by changes in the analysis of some phenomena that CCGrebank introduced (Honnibal et al., 2010) and not only by the style of implementation (pipeline or auxiliary task) which I tried to evaluate. Furthermore, the results of the manual analysis of correlations between the DPTB and CCGrebank in section 5.3 may not always be applicable to PIPE_{bank}.

Simple Auxiliary It can be seen that the simple 3-layer control model with summative residual connections CTR_3 produces worse results than the baseline (-0.06 F, -5.18 DF) despite the fact that enlarging the LSTM stack depth gives the model more computational capacity. The simple CCG model is able to improve these results with slight increases in F-score (+0.08) and discontinuous F-score (+2.84) compared to the control model. However, it only leads to a marginal improvement in F-score and even a decrease in discontinuous F-score compared to the baseline. Both results

suggest that the architecture is suboptimal for harnessing the increase in computational capacity provided by the additional LSTM transducer.

Gated Residual Connection Using gated residual connections in the 3-layer control model CTR_3^{gate} leads to solid improvements in all metrics compared to the baseline (+0.39 F, +2.0 DF), with the notable exception of discontinuous precision (-1.75). The increase in discontinuous F-score therefore stems from better recall which is why I assume that the gate-controlled residual connections make it easier for the model to recognise long-range dependencies by having more control over the composition of the relevant contextual features across the vertical pass through the network. Improvements are not limited to networks with 3 biLSTM layers. They also show when training the 2-layer baseline-model in a gated residual setting (CTR_2^{gate} ; +0.13 F, +0.38 DF).

The inclusion of supertags as an auxiliary task at the intermediate LSTM (CCG^{gate}) results in solid improvements both for general F-score and for discontinuous F-score which confirms the compatibility of gated residual connections with a hierarchical multi-task approach. Figure 6.9 depicts the progression of F-score over the epochs for the simple CCG model and for CCG^{gate}. The results for both the development split and a random selection of 425 sentences from the train split are included. It shows that the gain in F-score for the gated model is not simply a result of increased regularisation since both training and development results improve.



Figure 6.9: Progression of F-score for CCG and CCG^{gate} on the development split and a random subset of the training split.

F-score for CCG^{gate} increases by 0.08 absolute points and discontinuous F-score by 1.06 against the control model CTR_3^{gate} as well by 0.47 for F-score and 3.06 for discontinuous F-score compared to the baseline. The former isolates the contribution resulting from CCG supertagging as an auxiliary task. CCG^{gate} strongly corrects discontinuous precision from 78.2 for the control model to 82.48. This suggests that the network is able to acquire increased syntactic knowledge and competence for the analysis of long-range dependencies.

This finding confirms that CCG supertags can be helpful in discontinuous constituent parsing despite the seemingly incompatible nature of DPTB trees and CCG categories. The multi-layer neural auxiliary-task model is able to generate beneficial representations based on complex correlations which the pipeline model is apparently unable to acquire from the supertag distributions it is provided with at input level.

The increase in discontinuous precision comes with the cost of a slight decrease in discontinuous recall compared to the control model (-1.35). A large gap between precision (higher) and recall (lower) for discontinuities is often observed in discontinuous constituent parsers (Coavoux, 2021). The baseline model is no exception to this rule (79.95 precision, 65.25 recall). The fact that the pipeline approach does not add to this issue is coherent with the observations of Coavoux (2021) who enrich the input features of their parser with BERT-contextual embeddings (Devlin et al., 2019) which close the gap between recall and precision.

Learning a joint representation for both CCG supertagging and parsing apparently makes it harder for the parser to recognise long-range dependencies. One possible cause is that conditioning an intermediate vector representation both on supertagging and parsing as opposed to a pipeline approach impedes differentiating between informative supertag assignments and incompatible cases where a supertag corresponds to both projective and unprojective structures. This can be expected since the shared architecture up to the second biLSTM output is trained to provide useful features for CCG supertag prediction independent of supertag or context. As mentioned above, I suggest that this effect is mitigated through the adaptivity of the gated residual connection.

A second possibility may lie in the sparseness of training data for certain types of discontinuities (including their distinct CCG categories) coupled with the fact that auxiliary task training is performed on the same dataset. Therefore, the model might overfit for triggers of certain phenomena. This results in worse ability to recognise discontinuities not seen during training but when they are recognised, the syntactic capabilities induced by CCG supertag prediction allow the parser to competently reconstruct them. Enriching the input features using a pipeline approach naturally does not lead to this phenomenon since the number of training instances remains the same.

Increasing Width Contrary to the expectations, enlarging the LSTM hidden dimensionality from 400 to 600 for CCG_{600}^{gate} does not increase F-score (-0.06) or discontinuous F-score (-2.91) compared to CCG^{gate} . It also does not result in increased supertagging accuracy (-0.19). This indicates that the model started to overfit.

When comparing with the very well performing single-task control model $CTR_{3,600}^{gate}$ one could assume that the decrease in F-score (-0.17) and discontinuous F-score (-2.4) is caused by the integration of supertagging and be tempted to write off the auxiliary approach as ineffective. However, given the fact that CCG supertagging showed to increase F-score in the other settings (CCG and CCG^{gate}) I argue that this observation should be taken as another indicator for overfitting. This is supported by figure 6.10 which shows that the F-score on training instances significantly increases while the development score is slightly lower than $CTR_{3,600}^{gate}$ from epoch 20 onwards. Apparently the model adapts too closely to the training data due to a combination of increased network width and the fact that the number of training instances on the same dataset is enlarged by factor 1.5.

As usual in cases of overfitting, one could either enlarge the training dataset, which is not an option here, or experiment with forms of regularisation like increased dropout (Goldberg, 2022, chapter 4.6) or the introduction of variational dropout for recurrent neural networks (Gal and Ghahramani, 2016). An exploration of beneficial dropout application is left for future work.

Feature Bootstrapping The CCG_{multi}^{gate} multi-task model with four biLSTM-layers and subcomponent supertag features performs worse than the three-layer model CCG_{gate}^{gate} without bootstrapped sequence labelling tasks. General F-score is lower by 0.26 absolute points and discontinuous F-score by 1.33 points. When comparing with the 4-layer control model CTR_4^{gate} one can assess that while the approach leads to a degradation of default precision, recall and F-score, these metrics on only discontinuous constituents actually slightly increase. This is especially surprising for discontinuous recall (+0.45), which distinguishes this approach and the simple CCG model as



Figure 6.10: Progression of F-score for CCG_{600}^{gate} and $CTR_{3,600}^{gate}$ on the development split and a random subset of the training split.

the only two multi-task models that do not lead to a decline in discontinuous recall relative to their control model. Deconstructing supertags into distinct sequence labelling tasks may be a way to counteract the previously observed tendency of neural multi-task approaches to worsen in the model's ability to recognise discontinuities. Given the fact that CCG_{multi}^{gate} still underperforms compared to the single-auxiliary task model CCG_{gate}^{gate} , further research regarding viable subcomponent tasks and suitable architectures is necessary.

Head-Dependency As can be seen in table 6.4, using head-dependency information provided by the CCGrebank in a four-layer multi-task approach alongside CCG supertagging does not lead to improvements. The strongest drop in score occurs for discontinuous constituents, especially recall: -3.59 compared to CCG_{gate}^{gate} . This observation cannot be explained by the choice of architecture since the control model CTR_4^{gate} outperforms CCG_{dep}^{gate} in most metrics.

Therefore, one can assess that head-dependency relations do not provide useful information for constituent parsing. My assumption is that the inclusion of this auxiliary task induces a representation for relationships between tokens that is close to the gold combinatory categorial grammar derivations found in the CCGrebank but incompatible with the DPTB constituent analysis. For instance, in figure 6.7 the *wh*-word *what* would receive the same right action label 1:1 as *George*, which indicates that its right argument is filled by a constituent that has its lexical head one position to the right in the sentence. In the case of *George*, this corresponds to the DPTB analysis of *George* and *Bush* as a nominal constituent. *what*, however, does not produce a constituent with its successor *does* but instead forms a discontinuity with *believe* in the DPTB. Clearly, the model should be able to distinguish these two cases in its representation.

On the basis of supertags only, which constitute discrete categories (here: S[wq]/(S[q]/NP) and N/N), the parser appears to learn distinct contextual token representations that lead to more competence in the prediction of constituent parsing transitions, as the results of CCG^{gate} suggest. The neural model has no information as to the workings of the CCG formalism or the internal patterns of categories and only benefits from the fact that the category distinctions rooted in differences in argument and return structure used in the CCGrebank seem to correspond statisti-

cally in some capacity with distinctions in constituent structure analysis. But when supervising head-dependency information at a higher level than CCG supertagging, these useful distinctions between supertag assigned tokens seem to be partially overridden harming the model's competence. Naturally, this particularly affects discontinuous constituents and the parser's ability to recognise them, as indicated by the decline in discontinuous recall.

Other Auxiliary Tasks Table 6.5 shows the results of the experiments with other auxiliary tasks as well as CCG^{gate}, the baseline and the 3-layer gated control model for comparison. The scores may serve as a first examination of the potential synergies between discontinuous constituent parsing and different kinds of syntactic sequence labelling tasks.

Model	Р	R	F	DP	DR	DF	POS	AUX	Auxiliary label count
original	91.5	91.3	91.4	76.1	66.4	70.9	97.2	_	_
baseline	91.45	91.15	91.30	79.95	65.25	71.85	97.25	-	-
$\mathrm{CCG}^{\mathrm{gate}}$	91.86	91.67	91.77	82.48	68.61	74.91	97.22	92.85	1574
$\mathrm{LCFRS}^{\mathrm{gate}}$	91.5	91.38	91.44	78.42	66.82	72.15	97.30	87.0	4504
$\mathrm{CHUNK}^{\mathrm{gate}}$	91.94	91.66	91.8	82.22	66.37	73.45	97.25	99.29	22
$\mathrm{Dep}^{\mathrm{gate}}$	91.73	91.39	91.56	80.34	63.23	70.77	97.25	87.67	1634
$\mathrm{LTAG}^{\mathrm{gate}}$	91.55	91.35	91.45	80.38	67.04	73.11	97.13	94.72	1202
$\mathrm{Ctr}_3^{\mathrm{gate}}$	91.82	91.56	91.69	78.2	69.96	73.85	97.23	_	_

Table 6.5: Results of additional experiments with a variation of auxiliary tasks on the DPTB development set. P, R and F are precision, recall and F-score. DP, DR and DF are discontinuous precision, recall and F-score. POS is POS-tagging accuracy while AUX refers to the respective auxiliary task accuracy. The best score for each metric across all models is set in bold, except for AUX which is incomparable.

Despite the success of Ruprecht and Mörbitz (2021) with grammar-based discontinuous constituent parsing using LCFRS-based supertags, the LCFRS^{gate} model leads to a decline in both F-score (-0.25) and discontinuous F-score (-1.7) compared to the control model. Similar to the CCG models, discontinuous precision slightly increases while recall decreases which suggests that this phenomenon is not rooted in the kind of supertags used but has more to do with overfitting and data sparseness for distinct rare labels assigned to discontinuous structures.

The generally poor results of LCFRS supertagging as an auxiliary task might be explained with the large number of distinct labels (4504) which aggrevates the problem of data sparseness and makes it more difficult to build shared common representations in a multi-task environment.

Chunking as an auxiliary task improves general precision (+0.12), recall (+0.1) and F-score (+0.11) when compared to CTR_3^{gate} . For discontinuous constituents only, precision is improved significantly (+4.02) while recall worsens (-3.59) leading to a slight decline in F-scoree (-0.4). It is not surprising that the integration of chunking as an auxiliary task makes the recognition of long-range dependencies more difficult since the task does not correlate discontinuous chunks with each other.

Performing joint dependency parsing and constituent parsing leads to lower F-score (-0.13) and discontinuous F-score (-3.08) compared to the control model CTR_3^{gate} . Furthermore, the model experiences the lowest discontinuous recall amongst the auxiliary models compared in table 6.5. This is surprising since I expected long-distance syntactic dependencies to be a good indicator of discontinuous constituency. In section 3.2.6 I discussed the results of Coavoux et al. (2019) who show that lexicalisation as a component of transition systems decreases parsing score due to duplicating some transition types and leading to more complicated oracles. The results of DEP^{gate} show that softly reintroducing the notion of constituent heads as a dependency prediction objective from the underlying contextualised token representation does not have a favourable effect either.

This may be an indicator that dependency parsing, or at least the relative POS-based encoding used here, conflict with the lexicalised information and the head rules that Coavoux et al. (2019) assume LSTMs to learn implicitly.

Using LTAG-spinal supertags decreases F-score (-0.24) and discontinuous F-score (-0.74) relative to the control model. LTAG^{gate} performs only slightly better than the LCFRS^{gate} model despite not being hampered by an unsuitably large number of distinct labels. I think that this result can in part be explained by the fact that Shen et al. (2007) remove subcategorisation information from their supertags. For instance, elementary trees of transitive and ditransitive verbs are the same in LTAG-spinal (Shen et al., 2007). But these distinctions would benefit parsing if they were induced in the network's contextualised token representations. Predicting subcategorisation structure from a number of possible choices requires a model to learn to retrieve contextual information in the input sequence effectively. Furthermore, it is a form of pre-disambiguation of parsing actions. Predicting LTAG-spinal supertags retains too much ambiguity in terms of the constituent structure patterns that can match a lexical category assignment.

The experimental results show that the CCG model is the best-performing auxiliary model in terms of discontinuous F-score among the five types of syntactic sequence labelling tasks explored. The CHUNK^{gate} model is unique in that the auxiliary task only covers 10% of the total number of training instances. It shows that auxiliary objectives can be benefitial even in cases of significant imbalance in terms of dataset sizes. The success of the chunking task might also be attributed to the small set of distinct labels (22). It may be a coincidence but is striking nonetheless that general F-score tends to behave inverse proportionally to the number of auxiliary task labels in table 6.5, an exception being LTAG-spinal which performs poorly despite being the supertag task with the smallest number of labels. For further research it might therefore be worthwhile to investigate whether models that dispense of supertag training entirely and only learn to predict deconstructed subcomponent tasks with smaller label set sizes can have a positive effect on parser performance.

6.3.3 Per-Phenomenon Evaluation

Using only F-score, precision and recall as evaluation metrics for discontinuous parsing does not allow for a detailed analysis and is hard to interpret. While the metrics of discontinuous Fscore, discontinuous precision and discontinuous recall provided by disco-dop allow for a general evaluation of the performance on long-range dependencies, they do not give information about structurally distinct discontinuous phenomena like *wh*-extraction or circumpositioned quotations which leaves the syntactic capabilities of a parser unclear.

In order to achieve a finer-grained analysis several strategies have been explored. A comparison of the approaches suggested so far is presented by Coavoux (2021) and will be reproduced briefly in the following. Evang (2011) performed a manual analysis classifying sentences from the development set of the DPTB according to the types of discontinuity that appear in them and manually checked the resulting representations constructed by a PLCFRS parser. Coavoux et al. (2019) follow this strategy in the evaluation of their ML-GAP parser. As Coavoux (2021) note, manual evaluation is very time consuming and has to be repeated in full length for every newly developed parser.

Maier et al. (2014) created a testsuite for German called **discosuite** which allows the automatic evaluation of different types of discontinuous structures found in the TIGER treebank. To achieve this, they annotated sentences from the corpus with the long-range phenomena found in them and by this means makes a per-phenomenon evaluation possible.

Inspired by this idea Coavoux (2021) released a phenomenon-driven test suite for English.⁴⁹ They reuse the categories previously employed for manual error analysis by Evang (2011) and

⁴⁹Found as supplementary material here: https://aclanthology.org/2021.findings-acl.288

Coavoux et al. (2019) and, ignoring unprojectivity resulting only from punctuation, categorise all discontinuous trees from the DPTB delevopment set according to one or several discontinuity types resulting in 266 tree instances (16% of the dataset). The per-phenomenon evaluation is achieved by retrieving the number of gold, correct and incorrect discontinuous constituents for each tree individually using disco-dop and then aggregating the results for each type. Two separate scores with and without labelling are included.

In the following, I use the testsuite provided by Coavoux (2021) to retrieve detailed and linguistically meaningful information about the syntactic capabilities of the supertag-enriched approaches presented in this work.

Effects of the Pipeline Approach Table 6.6 reports the per-phenomenon results of the PIPE_{bank} model as well as the best-scoring auxiliary task model CCG^{gate} . For each metric, the absolute improvement/decline relative to the baseline is included.

For PIPE_{bank} relevant increases in F-score can be observed for *it*-extraposition (+7.9), discontinuous dependency (+3.5) and circumpositioned quotation (+2.9). Improvements for *It*extrapositions have been predicted by the manual comparison of the treatment of discontinuous phenomena in section 5.3. The increase in F-score for the other two categories is rather unexpected. Furthermore, contrary to the analysis in section 5.3 that suggested a beneficial correlation for *wh*-movement and fronted quotation, the F-score of these types actually decreases (-3.2 and -2.1 respectively). While both recall and precision drop, the decline in precision is larger (-5,6 extraction; -2.8 fronted quotation). These two phenomena are the most significant classes of discontinuity so that their relatively small decrease in F-score dominates the overall results.

The increase in F-score for *it*-extrapositions is caused by better recall (+8.3). I assume that the unique category NP[expl] assigned to expletive uses of *it* by the supertagger makes it easier for the parser to recognise discontinuities of this type.

For discontinuous dependency the increase in F-score is driven by a significant improvement in precision (+7.1). This effect is surprising since the manual analysis of phenomena did not suggest clear synergies in these cases due to extraposed dependents not encoding their correct attachment point in their category. Perhaps supertags helped to distinguish the boundaries of continuous subsections which allowed for a more precise analysis. I will investigate this result closer in section 6.3.5.

No change can be observed in the small class of subject inversion. The largest improvement in F-score occurs for sentences with two extractions (extraction+extraction; +15.3). While this class with only five instances is too small to draw statistically relevant conclusions, the result might indicate that lexical category assignments helped the scorer to differentiate between the two extractions through CCG category distinctions.

The unlabelled results are close to the results in the labelled case. A notable exception are all metrics for circumpositioned quotations which a noticeably higher when not including labels. This is coherent with the baseline and with previous results of Coavoux (2021). He traces the fall in labelled scoring back to cases where the discontinuous quotation exhibits an infrequent label like FRAG or SINV. Compared with the baseline, the unlabelled F-score for PIPE_{bank} decreases slightly (-0.3) while the labelled score increases (+2.9), driven by recall (+8.0), which suggests that supertag information may have helped to recognise labels for such infrequent cases.

Overall, it can be seen that recall generally increases or only decreases slightly for the pipeline approach with the exception of combined discontinuous dependency and *wh*-extraction (-11.8). The network does not experience the drop in discontinuous recall inherent to the auxiliary models. Precision fluctuates more reflecting the mixed informativeness of CCG supertags for constituent parsing. CCG supertags as an input feature make analysing some types of discontinuity (*wh*extraction, fronted quotation, circumpositioned quotation) correctly more difficult for the parser.

-												
a	aseline	_			$\mathrm{PIPE}_{\mathrm{bank}}$					$\rm CCG^{gate}$		
M PM	പ	Б	Exact match	Partial match	Precision	Recall	Ъ	Exact match	Partial match	Precision	Recall	Ъ
.9 85.7	90.0	78.8 84	1.0 73.6 (-3.3)	84.6 (-1.1)	84.4 (-5.6)	77.5 (-1.3)	80.8 (-3.2)	78.0(+1.1)	87.9 (+2.2)	89.9 (-0.1)	83.1 (+4.3)	86.4(+2.4)
0.03.0	95.7	$93.0 9_{4}$	$1.3 90.1 \ (-2.9)$	91.5(-1.5)	92.9 (-2.8)	91.5(-1.5)	92.2(-2.1)	93.0(+0.0)	93.0(+0.0)	94.3(-1.4)	93.0 (+0.0)	93.6 (-0.7)
3 29.7	78.6	25.0^{-3}	$7.9 29.7 \ (+5.4)$	32.4 (+2.7)	85.7 (+7.1)	27.3 (+2.3)	41.4(+3.5)	27.0 (+2.7)	29.7 (+0.0)	92.3(+13.7)	27.3 (+2.3)	42.1 (+4.2)
0 81.2	54.5	36.0 4;	$3.4 0.0 \ (+0.0)$	87.5 (+6.3)	48.9(-5.6)	44.0(+8.0)	46.3(+2.9)	(0.0 (+0.0)	75.0 (-6.2)	46.7 (-7.8)	28.0 (-8.0)	35.0(-8.4)
.7 41.7	100	41.7 58	3.8 50.0 (+8.3)	50.0(+8.3)	100(+0.0)	50.0(+8.3)	66.7 (+7.9)	50.0(+8.3)	50.0(+8.3)	100(+0.0)	50.0(+8.3)	(6.7 (+7.9)
.4 100	84.2	84.2 8	1.2 71.4 $(+0.0)$	100 (+0.0)	84.2 (+0.0)	84.2 (+0.0)	84.2(+0.0)	100(+28.6)	100 (+0.0)	100 (+15.8)	100 (+15.8)	100 (+15.8)
0.08 0.0	61.1	64.7 6:	2.9 0.0 (-20.0)	80.0(+0.0)	75.0(+13.9)	52.9 (-11.8)	62.1 (-0.8)	0.0 (-20.0)	100(+20.0)	73.3(+12.2)	64.7 (+0.0)	68.8 (+5.9)
0.0 100	81.2	76.5 78	$3.8 \ 60.0 \ (+20.0)$	100(+0.0)	94.1 (+12.9)	94.1 (+17.6)	$94.1\ (+15.3)$	40.0(+0.0)	100 (+0.0)	93.8(+12.6)	88.2 (+11.7)	90.9(+12.1)
0.0 60.0	100	60.0 7!	(0.0+)0.00	60.0(+0.0)	100 (+0.0)	60.0(+0.0)	75.0(+0.0)	(0.0+) (0.0)	60.0(+0.0)	100(+0.0)	60.0 (+0.0)	75.0(+0.0)
						Unlab	belled					
q	aseline				$\mathrm{PIPE}_{\mathrm{bank}}$					$\rm CCG^{gate}$		
M PM	പ	Ч	Exact match	Partial match	Precision	m Recall	Ъ	Exact match	Partial match	Precision	m Recall	ц
.9 85.7	89.6	80.0 84	1.5 74.7 (-2.2)	84.6 (-1.1)	84.4 (-5.2)	79.3 (-0.7)	81.8 (-2.7)	78.0 (+1.1)	87.9 (+2.2)	90.0(+0.4)	84.0 (+4.0)	86.9(+2.4)
0.093.0	95.7	$93.0 9_{4}$	1.3 90.1 (-2.9)	91.5(-1.5)	92.9 (-2.8)	91.5(-1.5)	92.2(-2.1)	93.0(+0.0)	93.0(+0.0)	94.3(-1.4)	93.0(+0.0)	93.6 (-0.7)
.3 29.7	78.6	25.6 3	$3.6 32.4 \ (+8.1)$	35.1 (+5.4)	92.9(+14.3)	30.2 (+4.6)	45.6(+7.0)	27.0 (+2.7)	29.7 (+0.0)	$91.7\ (+13.1)$	25.6(+0.0)	40.0(+1.4)
0 87.5	93.9	81.6 8′	7.3 81.2 (+6.2)	87.5 (+0.0)	96.8(+2.9)	78.9 (-2.7)	87.0 (-0.3)	75.0 (+0.0)	87.5 (+0.0)	96.7 (+2.8)	76.3 (-5.3)	85.3 (-2.0)
.7 41.7	100	41.7 58	$3.8 50.0 \ (+8.3)$	50.0(+8.3)	100 (+0.0)	50.0(+8.3)	66.7 (+7.9)	50.0(+8.3)	50.0(+8.3)	100 (+0.0)	50.0(+8.3)	66.7 (+7.9)
.4 100	88.9	88.9 8	$3.9 71.4 \; (+0.0)$	100 (+0.0)	88.9(+0.0)	(0.0+) $(+0.0)$	(0.0+) $(+0.0)$	100(+28.6)	100 (+0.0)	100 (+11.1)	100 (+11.1)	100 (+11.1)
0.08 0.0	60.0	60.0 60	0.0 0.0 (-20.0)	80.0(+0.0)	80.0(+20.0)	53.3 (-6.7)	64.0(+4.0)	0.0 (-20.0)	100(+20.0)	$76.9\ (+16.9)$	(7.6.7)	71.4 (+11.4)
0.0 100	80.0	75.0 77	7.4 60.0 (+20.0)	100(+0.0)	93.8(+13.8)	93.8 (+18.8)	93.8(+16.4)	40.0(+0.0)	100 (+0.0)	93.3(+13.3)	87.5 (+12.5)	90.3(+12.9)
0.09 0.0	100	60.0 7!	5.0 60.0 (+0.0)	(0.0+) (0.09)	100(+0.0)	(0.0+) (0.09)	75.0(+0.0)	60.0(+0.0)	60.0(+0.0)	100(+0.0)	60.0(+0.0)	75.0(+0.0)
n result	ts on 1	the tes	t suite from C	oavoux (202)	l) for the ba	seline (rerui	n of Coavoux	and Coher	(2019) with	ı updated cc	odebase), P	PE _{bank} and
	 8.0 93.0 8.1.3 29.7 0 81.2 0 81.2 1.7 41.7 1.1 100 1.0 80.0 100 80.0 100 60.0 100 87.5 10 87.5 10 87.5 10 87.5 10 80.0 10 80.0 10 60.0 10 60.0 	 8.0 93.0 95.7 8.1.2 54.5 8.1.2 54.5 8.1.2 54.5 8.1.2 54.5 9.1.7 100 8.1.2 9.0 60.0 100 9.1.2 9.1.1 100 9.1.2 9.1.1 100 9.1.2 93.0 95.7 9.1.1 100 9.1.2 100 9.1.2 100 9.1.1 100 9.1.1 100 9.1.2 100 9.1.2 100 9.1.1 100 9.1.1 100 9.1.2 100 9.1.1 100 9.1.2 100 9.1.1 100 9.1 100 	 (a) 93.0 95.7 93.0 9 (b) 93.0 95.7 78.6 25.0 3 (c) 81.2 54.5 36.0 4 (c) 81.2 54.5 36.0 4 (c) 81.2 54.5 36.0 4 (c) 81.2 84.2 8 (c) 80.0 61.1 64.7 6 (c) 80.0 61.1 64.7 6 (c) 100 81.2 76.5 7 (c) 33.0 95.7 93.0 9 (c) 33.0 95.7 93.0 9 (c) 33.0 95.7 93.0 8 (c) 33.0 95.7 93.0 8 (c) 33.0 95.7 93.0 9 (c) 33.0 95.7 9 (c) 33.0 95.7 9 (c) 33.0 9 (c	$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	10 93.0 95.7 93.0 94.3 90.1 (-2.9) 91.5 (-1.5) 1.3 29.7 78.6 25.0 37.9 29.7 $(+5.4)$ 32.4 $(+2.7)$ 0 81.2 54.5 36.0 43.4 0.0 $(+0.0)$ 87.5 $(+6.3)$ 7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 50.0 $(+8.3)$ 7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 50.0 $(+8.3)$ 7 41.7 100 81.2 76.5 71.4 $(+0.0)$ 100 $(+0.0)$ 0.0 60.1 64.1 64.7 62.9 0.0 (-20.0) 80.0 $(+0.0)$ 0.0 80.0 81.2 78.8 60.0 $(+0.0)$ 100 $(+0.0)$ 0.0 100 81.2 75.4 81.6 (-1.1) 100 $(+0.0)$ 0.0 93.0 95.7 93.0 84.5 74.7 (-2.9) 84.6 (-1.1) 0.0	10 93.0 95.7 93.0 94.3 90.1 (-2.9) 91.5 (-1.5) 92.9 (-7.1) 1.3 29.7 78.6 25.0 37.9 29.7 $(+5.4)$ 32.4 $(+7.1)$ 0 81.2 54.5 36.0 43.4 0.0 $(+0.0)$ 87.5 $(+6.3)$ 48.9 (-5.6) 7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 100 $(+0.0)$ 7 41.7 100 84.2 84.2 71.4 $(+0.0)$ 84.2 $(+0.0)$ 9 80.0 61.1 64.7 62.9 0.0 (-20.0) 80.0 $(+0.0)$ 94.1 $(+12.9)$ 0 80.0 61.1 64.7 62.9 0.0 (-20.0) 94.1 $(+12.9)$ 0 80.0 61.1 64.7 62.9 0.0 (-20.0) 94.1 $(+12.9)$ 0 80.0 61.0 60.0 $(+0.0)$ 94.1 $(+12.9)$ 0 80.0 80.0 $(+0.0)$ <td>030 95.7 93.0 94.3 90.1 (2.9) 91.5 (1.5) 13 29.7 78.6 25.0 37.9 29.7 $(+5.4)$ 32.4 $(+2.7)$ 85.7 $(+7.1)$ 27.3 $(+2.3)$ 0 81.2 54.5 36.0 43.4 0.0 $(+0.0)$ 87.5 $(+6.3)$ 48.9 (-5.6) 44.0 $(+8.0)$.7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 50.0 $(+8.3)$ 50.0 $(+8.3)$.7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 50.0 $(+8.3)$ 50.0 $(+8.3)$.7 41.7 100 41.7 58.8 50.0 $(+2.0)$ 80.0 $(+1.12.9)$ 94.1 $(+17.6)$.0 60.0 (-0.0) 60.0 $(+0.0)$ 90.0 $(+0.0)$ (-0.0) (-0.0) (-0.0) $(-1.18.9)$ $(-1.18.9)$ $(-1.16.9)$.0 80.0 81.1 (-1.7) 84.4 (-5.2) 84.1 (-7.5) $(-1.5$</td> <td>03.0 95.7 93.0 94.3 90.1 (-2.9) 91.5 (-1.5) 92.9 (-2.3) 91.5 (-1.5) 92.2 (-2.1) 13. 29.7 78.6 25.0 37.9 29.7 $(+5.4)$ 32.4 $(+2.7)$ 85.7 $(+7.1)$ 27.3 $(+2.3)$ 41.4 $(+3.5)$ 0 81.2 54.5 36.0 $(+3.3)$ 50.0 $(+8.3)$ 50.0 $(+7.9)$.7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 50.0 $(+7.9)$.6 61.1 64.7 62.9 0.0 (-20.0) 80.0 $(+0.0)$ 84.2 $(+0.0)$ 84.2 $(+0.0)$ 0.0 81.0 61.1 64.7 (-10.0) 60.0 $(+0.0)$ 84.2 $(+0.0)$ 0.0 81.2 75.6 60.0 $(+0.0)$ 84.2 $(+1.15.3)$ 0.0 60.0 100 60.0 $(+0.0)$ 60.0 $(+0.0)$ 75.0 $(+0.0)$ 0.0 81.2 74.5 94.1 74.4</td> <td>$\begin{array}{ c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{ c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{ c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{ c c c c c c c c c c c c c c c c c c c$</td>	030 95.7 93.0 94.3 90.1 (2.9) 91.5 (1.5) 13 29.7 78.6 25.0 37.9 29.7 $(+5.4)$ 32.4 $(+2.7)$ 85.7 $(+7.1)$ 27.3 $(+2.3)$ 0 81.2 54.5 36.0 43.4 0.0 $(+0.0)$ 87.5 $(+6.3)$ 48.9 (-5.6) 44.0 $(+8.0)$.7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 50.0 $(+8.3)$ 50.0 $(+8.3)$.7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 50.0 $(+8.3)$ 50.0 $(+8.3)$.7 41.7 100 41.7 58.8 50.0 $(+2.0)$ 80.0 $(+1.12.9)$ 94.1 $(+17.6)$.0 60.0 (-0.0) 60.0 $(+0.0)$ 90.0 $(+0.0)$ (-0.0) (-0.0) (-0.0) $(-1.18.9)$ $(-1.18.9)$ $(-1.16.9)$.0 80.0 81.1 (-1.7) 84.4 (-5.2) 84.1 (-7.5) $(-1.5$	03.0 95.7 93.0 94.3 90.1 (-2.9) 91.5 (-1.5) 92.9 (-2.3) 91.5 (-1.5) 92.2 (-2.1) 13. 29.7 78.6 25.0 37.9 29.7 $(+5.4)$ 32.4 $(+2.7)$ 85.7 $(+7.1)$ 27.3 $(+2.3)$ 41.4 $(+3.5)$ 0 81.2 54.5 36.0 $(+3.3)$ 50.0 $(+8.3)$ 50.0 $(+7.9)$.7 41.7 100 41.7 58.8 50.0 $(+8.3)$ 50.0 $(+7.9)$.6 61.1 64.7 62.9 0.0 (-20.0) 80.0 $(+0.0)$ 84.2 $(+0.0)$ 84.2 $(+0.0)$ 0.0 81.0 61.1 64.7 (-10.0) 60.0 $(+0.0)$ 84.2 $(+0.0)$ 0.0 81.2 75.6 60.0 $(+0.0)$ 84.2 $(+1.15.3)$ 0.0 60.0 100 60.0 $(+0.0)$ 60.0 $(+0.0)$ 75.0 $(+0.0)$ 0.0 81.2 74.5 94.1 74.4	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$

CCG^{gaue} models. C, EM and PM are count, exact match and partial match. Sentences where several discontinuous phenomena occur are treated separately. Only combinations with at least 5 occurrences in the DPTB development split are included. Absolute changes relative to the baseline are included in parentheses.

It could be that the parser learned to identify functors, arguments and argument directions in CCG supertags. For instance it could have internalised the fact that supertags assigned to fronted wh-words like S[wq]/(S[q]/NP) predict an argument to the right and thus wrongfully merges the right neighbour constituent leading to less precision. A deeper investigation of this issue is provided in section 6.3.5.

Effects of the Auxiliary Approach For the auxiliary task model, improvements in the labelled case can be observed for *wh*-extractions (+2.4) and *it*-extrapositions (+7.9) which is true to the assumptions. However, like in the pipeline model, albeit to a smaller extent, F-score for fronted quotations decreases (-0.7).

For discontinuous dependency, circumpositioned quotations and subject inversion the manual analysis in section 5.3 did not suggest transparent correlations when including supertags in the parser. Indeed a significant drop for circumpositioned quotations can be observed (-8.4). On the other hand, discontinuous dependency F-score increases (+4.2) similar to the pipeline model. Precision even exhibits twice the amount of improvement as in CCG_{bank} (+13.7). The integration of supertagging as an auxiliary task has no effect on subject inversion.

The decrease in F-score pertaining to circumpositioned quotations is caused by a degradation of both recall (-8.0) and precision (-7.8). This differs from the pipeline result where recall increased (+8.0) and may be due to the fact that circumpositioned quotations feature some rare categories that the auxiliary model is trained to predict while depccg rebank is only trained on the 511 most frequent supertags. Thus, this type of discontinuity might suffer from training data sparseness in the auxiliary approach.

As section 5.3 showed, the analysis of circumpositioned quotations generally seems to be inconsistent in the CCGrebank. Thus, conditioning a shared representation to predict supertags for such cases likely introduces a lot of noise. It may also suggest that the pipeline-model was able to identify uninformative assignments that it learned to ignore while the auxiliary model was conditioned to learn a shared representation for conflicting analyses.

The trends are roughly mirrored in the unlabelled case. Overall the improvements of the auxiliary approach compared to the pipeline model occur across most of the types of phenomena with the exception of circumpositioned quotiations. The phenomenon-specific analysis confirms that joint learning is more effective in building representations that can benefit from correlations between CCG lexical category assignments and discontinuous constituent parsing.

6.3.4 Error Analysis

In addition to the per-phenomenon analysis, the test suite released by Coavoux (2021) automatically classifies errors according to structural patterns. For this, the author builds on the standard Berkeley Parser Analyser (Kübler et al., 2009) and extends it to discontinuous trees. Errors in the predictions are classified according to manually defined patterns by analysing transformations that convert the predicted tree into the gold tree. Coavoux (2021) reports the following error types specifically for discontinuous nodes:

- 1. PP attachment
- 2. NP internal structure
- 3. Modifier Attachment
- 4. Unary constituent
- 5. Different label
- 6. Clause attachement

- 7. Coordination
- 8. NP attachment
- 9. VP attachment

After classification, for each category the number of error occurrences as well as the number of node errors involved is returned.

Table 6.7 contains the results of the error analysis on the models of this work. Changes relative to the baseline are included in parentheses. I will focus on comparing errors on discontinuous nodes specifically. The largest contributor to discontinuous errors in the baseline model is erroneous attachment of constituents. Both the pipeline and the auxiliary model significantly reduce the number of discontinuous PP attachment errors (-25.7%, -22.9%) and modifier attachment errors (-25.0%, -33.3%) as well as the number of affected nodes. This shows that CCG supertag information can be very helpful for a parser to resolve PP and modifier attachment ambiguities. I assume that this is a significant factor for the improvements in extraposed dependency score where appended discontinuous PP constituents are frequent.

	base	eline	Pipi	E _{bank}	CC	G^{gate}
Error type	Count	Nodes	Count	Nodes	Count	Nodes
PP attachment	513	1076	446 (-13.1%)	1045 (-2.9%)	438 (-14.6%)	952 (-11.5%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	35	58	26 (-25.7%)	37 (-36.2%)	27 (-22.9%)	41 (-29.3%)
Unclassified	452	585	400 (-11.5%)	518 (-11.5%)	397 (-12.2%)	541 (-7.5%)
Single word phrase	386	453	353 (-8.5%)	422 (-6.8%)	352 (-8.8%)	409 (-9.7%)
Unary	325	325	339 (+4.3%)	339 (+4.3%)	321 (-1.2%)	321 (-1.2%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	0	0	$2 (+\infty\%)$	$2 (+\infty\%)$	0 (+0.0%)	0 (+0.0%)
Different label	268	538	214 (-20.1%)	442 (-17.8%)	255 (-4.9%)	513 (-4.6%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	17	36	19 (+11.8%)	52 (+44.4%)	19 (+11.8%)	41 (+13.9%)
Modifier attachment	261	449	258 (-1.1%)	482 (+7.3%)	229 (-12.3%)	447 (-0.4%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	12	17	9 (-25.0%)	$14 \ (-17.6\%)$	8 (-33.3%)	14 (-17.6%)
NP internal structure	258	318	262 (+1.6%)	339~(+6.6%)	253 (-1.9%)	317~(-0.3%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	0	0	0 (+0.0%)	0 (+0.0%)	$1 (+\infty\%)$	$1 (+\infty\%)$
Clause attachment	192	457	187 (-2.6%)	462 (+1.1%)	192 (+0.0%)	496 (+8.5%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	26	44	27 (+3.8%)	50 (+13.6%)	27 (+3.8%)	38 (-13.6%)
Co-ordination	158	452	150 (-5.1%)	456 (+0.9%)	146 (-7.6%)	413 (-8.6%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	6	11	3 (-50.0%)	8 (-27.3%)	3(-50.0%)	8 (-27.3%)
NP attachment	105	316	119 (+13.3%)	363 (+14.9%)	108 (+2.9%)	354 (+12.0%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	17	13	$20 \ (+17.6\%)$	29 (+123.1%)	16 (-5.9%)	18 (+38.5%)
VP attachment	51	245	44 (-13.7%)	189 (-22.9%)	44 (-13.7%)	186 (-24.1%)
${}^{\scriptscriptstyle {\sf L}}$ discontinuous	5	8	4 (-20.0%)	7 (-12.5%)	5 (+0.0%)	8 (+0.0%)
XoverX Unary	11	11	10 (-9.1%)	10 (-9.1%)	10 (-9.1%)	10 (-9.1%)

Table 6.7: Error types for the models including the baseline on the development set. Discontinuous counts are included in the respective main category. Changes relative to the baseline in percent are included in parentheses.

Discontinuous clausal attachment errors increase for PIPE_{bank} (+3.8%) and for CCG^{gate} (+3.8%). Discontinuous NP attachment errors only increase in count for the pipeline model (+17.6%) while they drop for the auxiliary model (-5.9%). Yet, the number of nodes affected increases for both models and the metrics also significantly increase for the general (including projective errors) case. The supertag-enriched models appear to experience some difficulties in the treatment of discontinuous NPs.

Errors due to wrong label assignments on discontinuous constituents increase for both models

by 11.8% relative to the baseline. This is striking considering the fact that the general error count for different labels decreases both for the pipeline model (-20.1%) and for the auxiliary model (-4.9%). Recall that CCG supertags are designed as structural descriptions operating on a small number of atomic categories (NP, S, ...). Thus, some discontinuous phenomena that are assigned different labels in the DPTB due to linguistic motivation may be structurally similar in terms of their CCG supertag analysis which makes assigning the correct node label more difficult for the parser.

Overall, the effect of the inclusion of supertags on different error types does not follow a uniform trend. While the results of $PIPE_{bank}$ and CCG^{gate} share some similar patterns (e.g. discontinuous PP attachment, discontinuous different label) some types of errors (e.g. general unary, general NP internal structure, discontinuous clause attachment) seem to be dependent on the kind of supertag integration. This may either be a hint at differences between the CCGbank and the CCGrebank analysis and/or further support for the ability of the auxiliary approach to leverage synergies that single-task models are not able to profit from.

6.3.5 Sample Analysis

In sections 6.3.3 and 6.3.4 several observations regarding the models' performance on *wh*-movement, fronted quotations and extraposed dependents were made that deserve closer investigation. In the following, I will conduct a more detailed analysis using samples from the models' predictions on the DPTB development split.

Comparing Auxiliary and Pipeline: *wh*-movement The per-phenomenon analysis in section 6.3.3 showed that F-score for *wh*-extraction is reduced relative to the baseline in the pipeline approach while it increases in the CCG^{gate} model. To investigate the performance on *wh*-extraction in more detail, I computed separate scores for each trigger of frontation in the DPTB development split, listed in table 6.8. Sentences are assigned to the categories using annotations found in the supplementary material of Coavoux (2021).

				PIPEbank					CCG ^{gate}		
Trigger	Count	Exact match	Partial match	Precision	n Recall	F	Exact match	Partial match	Precision	n Recall	F
when	35	85.7 (+0.0)	94.3 (+2.9)	90.5 (+0.5)	90.5 (+4.8)	90.5 (+2.7)	88.6 (+2.9)	94.3 (+2.9)	95.0 (+5.0)	90.5 (+4.8)	92.7 (+4.9)
where	14	85.7 (+0.0)	85.7 (+0.0)	95.5 (+0.0)	91.3 (+0.0)	93.3 (+0.0)	85.7 (+0.0)	85.7 (+0.0)	95.5 (+0.0)	91.3 (+0.0)	93.3 (+0.0)
which	12	58.3 (-16.7)	75.0 (-16.7)	85.7 (-10.5)	64.3 (-25.0)	73.5 (-19.1)	75.0 (+0.0)	91.7 (+0.0)	81.2 (-15.0)	92.9 (+3.6)	86.7 (-5.9)
how	9	66.7 (+0.0)	77.8 (+0.0)	52.4 (-16.4)	57.9 (+0.0)	55.0 (-7.9)	77.8 (+11.1)	88.9 (+11.1)	80.0 +11.2	84.2 (+26.3)	82.1 (+19.2)
that	9	44.4 (+0.0)	$66.7 \\ (+0.0)$	94.7 (+0.9)	75.0 (+12.5)	83.7 (+8.7)	33.3 (-11.1)	55.6 (-11.1)	92.3 (-1.5)	50.0 (-12.5)	64.9 (-10.1)
what	9	66.7 (-11.1)	77.8 (+0.0)	66.7 (-13.3)	61.5 (+0.0)	64.0 (-5.6)	77.8 (+0.0)	88.9 (+11.1)	$90.9 \ (+10.9)$	76.9 (+15.4)	83.3 (+13.7)

Table 6.8: Labelled evaluation results specific to lexical triggers of wh-extraction. Absolute improvements over the baseline are included in parentheses. Only cases with at least five occurrences are reported.

Section 5.3 suggested that the CCG analysis for wh-extraction is dependent on the syntactic function of the fronted word. More specifically: in cases where the wh-word represents an argument required by the predicate of the sentence it would be easier to resolve for the parser since the verb's supertag would provide it with the information where to attach the question word while adjunctive frontation where the wh-word does not match an argument slot in the verb would be more difficult for the parser, e.g. in the case of how.

The evaluation results in table 6.8 are not that clear-cut. F-score for *which*, thought to be encoded as a core-argument, drops for both models (pipeline: -19.1, auxiliary -5.9) while the score for *when*, being purely adjunctive, increases (+2.7, +4.9). The differences in F-score between PIPE_{bank} and CCG_{gate} result from extrapositions of *how* and *what* on which the pipeline model performs worse than the baseline while the auxiliary model significantly increases F-score. Likewise noteworthy is the improvement on *that*-extraposition in the pipeline model (+8.7) caused by better recall (+12.5) while recall (-12.5) causes a drop in F-score for this type in the auxiliary task model (-10.1). Interestingly, it is the only *wh*-word where the auxiliary model exhibits a degradation of recall.

Let us look at an example for the *wh*-word how. Figure 6.11 shows part of the gold DPTB derivation for the sentence in (6.6a) which is retrieved correctly by CCG^{gate} . The constituent how long is extraposed to the front and forms a discontinuous verbal phrase together with the VB take, its NP argument and the S constituent to make a contribution. The PIPE_{bank} model returns the faulty analysis reproduced in figure 6.12. Here, the final S-constituent is not analysed as part of the VP but instead a discontinuous nominal phrase with *it* as an additional member is predicted, suggesting an occurrence of *it*-extraposition, and attached at the top-level S node.

- (6.6) a. "The question is <u>how long</u> it 's going to <u>take Barry Wright to make a contribution</u>," says F. John Mirek, an analyst at Blunt Ellis Loewi in Milwaukee.
 - b. While reaching blockbuster proportions yesterday, the volume was still well within the 600 million-share capacity <u>that</u> the exchange has said it can <u>handle daily since beefing</u> up its computers after the October 1987 crash.

Figure 6.13 shows the CCGrebank supertag assignments which were used to train the auxiliary model as well as the corresponding derivation. The long-range dependency is realised through an additional NP argument for *take*. The *wh*-word *how* expects an N argument to its right, which is filled by *long*, and as a second argument a declarative sentence category with an unfilled NP slot. This way, it "consumes" the rest of the sentence. *how* is actually not realised as an adjunctive category but conforms with the way core *wh*-extractions were analysed in section 5.3. Furthermore, this coincides exactly with the orginal CCGbank assignments. The fact that *wh*-words that I thought to be purely adjunctive indeed match argument slots with their discontinuous counterpart in the CCG analysis may in part explain why the differences in performance between the triggers for *wh*-extraction in table 6.8 are not as clear-cut as predicted.

Figure 6.14 contains the 1-best supertags generated by standard depccg for the sentence in (6.6a). They deviate from the CCGbank which was used to train the supertagger in that *take* is only assigned one right NP argument. Furthermore, *how* does not expect a function with range S[dcl] and a missing NP argument to the right any more but only a complete S[dcl] type.

As figure 6.12 shows, the parser was still able to recognise the discontinuous relationship of how long and take. The error occurs thereafter. I assume that the parser learned that the tag $((S[b]\NP)/(S[to]\NP))/NP$ has only two arguments and is misled by this assignment provided by depccg. When the transition system combines how long and take it implicitly recognises one argument slot as filled and then fills the second slot with Barry Wright. The scorer notices that the following S constituent surpasses the arity of take so it predicts an alternative attachment point being led by the fact that to make a contribution can be combined to a category that looks for an NP to its left. Therefore, it merges with the NP *it* — a construction not uncommon for *it*-extrapositions.

The baseline model also commits the error of merging to make a contribution with it. These observations can be taken as an example for the help CCG predicate-argument structure can provide to the constituent parser and showcases how the auxiliary model is able to profit from



Figure 6.11: Gold derviation wh-extraction using how in the DPTB.

synergies between DPTB constituents and CCG category assignments.

However, the picture is not always that clear. (6.6b) gives an example for *wh*-extraction caused by the relative pronoun *that*. Figure 6.15 shows the DPTB analysis of this sentence which the pipeline model is able to construct correctly. However, the auxiliary model attaches *that* at a higher level predicting an additional intermediary S-node that constitutes a discontinuity between *that* and *it can handle daily...* (figure 6.16). Thus, the fronted *wh*-word is analysed as a modifier at sentential level instead of as a verbal argument. This leads to lower recall since *that* is missing as part of the lower-level unprojective constituents SBAR, S, VP and VP.

Figure 6.17 depicts the CCG category assignments provided by the CCGrebank while figure 6.18 contains the 1-best predictions of the depccg supertagger which exactly coincide with the original CCGbank. Both supertag sequences are the same except for minor differences: *that* being an N-modifier in CCGrebank and an NP-modifier in CCGbank, the categorial type NP[nb] occurring only in the CCGbank and the particle *up* being an argument of *beefing* in the rebank but an adjunct in the original treebank. Differences in analysis can therefore not be attributed to deviating lexical category assignments.

It seems as if the model had analysed *that* and *it can handle daily* ... as a circumpositioned quotation, being triggered by the occurrence of the word *said*. The per-phenomenon analysis in section 6.3.3 has already shown that the auxiliary model performs worse on circumpositioned quotations than the baseline while PIPE_{bank} improves F-score for this category. Apparently the inconsistent CCGrebank gold categories introduce so much noise into the network the analysis of other types of discontinuities is negatively affected as well.

Another factor may be the fact that with increasing sentence length it may become harder for the parser to identify whether a missing argument should be filled with some successor item or with an extraposed constituent. In this case, when having reached *handle* the discontinuous constituent parser must decide whether to perform a MERGE or to SHIFT to fill the NP argument slot with some right neighbour. Therefore, it needs to know if an adjacent subsection of the remaining sequence can be reduced to an NP. Through category assignment alone this cannot be ruled out since NP appears twice as a functor in the remainder of the sentence (*its* and *the*). Therefore, the model would need to perform a shallow form of CCG parsing. This information must be construed by the multi-layer biLSTM network. I suppose that the neural model has difficulties performing such an implicit derivation in cases of long input sequences.



Figure 6.12: Faulty derivation of wh-extraction with how produced by the PIPE_{bank} model. Instead of attaching to make a contribution to the VP dominating how long, take and Barry Wright, it predicts another discontinuity merging with it.

Unexpected Decrease: Fronted quotations Contrary to the analysis in section 5.3 which suggested a transparent correlation between the analysis of fronted quotations in the DPTB and the supertag assignments in the CCGrebank, where both the fronted quotation and the subject of the quotative verb are encoded positionally as arguments, the per-phenomenon analysis in section 6.3.3 showed that F-score for this type of discontinuity decreases in the PIPE_{bank} model (-2.1) and in the CCG^{gate} model. A drop in precision is the primary cause. In (6.7) I have included a sentence that features a fronted quotation which the baseline model analyses correctly (cf. figure 6.19). However, the auxiliary model produces erroneous output by including the predicate phrase After the game into the discontinuous VP as part of the quotation, as can be seen in figure 6.20.

(6.7) After the game ("<u>Bluefield lost</u>, 9-8, stranding three runners in ... the ninth," he <u>noted</u>), trouble began.

The sentence in (6.7) is an unusual example since the quotative clause is embedded into parentheses. Assigning a CCG analysis to them apparently was problematic which is why the sentence was left out from the CCGbank and its rebanked successor. For this reason it is also not present in the CCG training dataset used for CCG^{gate} so that it, and possibly similar sentences, were underrepresented during training. Therefore, the model might have not learned to treat clauses encapsulated by parentheses as single units. At the same time, most fronted quotations start at the beginning of the sentence. Thus, the network is misled by the usual analysis of a PP like *After* the game as a modifier of the following S category and attaches it projectively to the quotation.

On a similar note, I suspect that the reintegration of quotation marks in the CCGrebank is a factor for the better score of the auxiliary model compared to $PIPE_{bank}$. When providing the depccg standard model with quotations surrounded by quotation marks, it simply analyses them as sentential modifiers like S/S which may conflict with traditional sentential modifiers that behave differently while the CCGrebank assigns special categories LQU and RQU. Training the model to predict these labels may have induced knowledge about the restrictions for functional application across quotation mark boundaries.

to make a contribution	S[to]\NP (S\NP)/((S\NP)/(S[to]\NP))			3					t to make a contribution	S[to]\NP		^	^			Ĭ	
Barry Wright	//NP NP >>	(S\NP)/NP	NP)/NP						Barry Wrigh	N/NP NP	(S[to]\NP)	S[b]\NP	J\NP				
take	(((S[b]\NP)/(S[to]\NP))/NP ((S[b]\NP)/(S[to])		(S[to])	(S[ng]/NP)/NP	[dcl]\NP)/NP	/NP		in the CCGrebank.	take	$\frac{[b] \ NP}{((S[b] \ NP)/(S[to] \ NP))}$	(S[b]\NP)		S[tc	S[ng]\NP	S[dcl]\NP	S[dcl]	
to	IP) (S[to]\NP)/(S[b]\NP)				5)	S[dcl]	S[qem]	wh-extraction with how	1g to	$(S[to] \setminus NP) \overline{(S[to] \setminus NP)/(S)}$							S[qem]
going	5[ng]\NP)/(S[to]\N							3: Derivation of	goir	VP) (S[ng]\NP)/							
S,	(S[dcl]\NP)/(S[ng]\NP) (S							Figure 6.1.	ıg it 's	VP NP (S[dcl]/NP)/(S[ng]	Â						
it	NP <u>S/(S∖N</u> Pً []]								lon	VP) S[adj]	[dcl])						
how long	em]/(S[dcl]/NP))/N N S[qem]/(S[dcl]/NP) >								how	(S[qem]/S[dcl])/(S[adj	S[qem]/S]						



Figure 6.18: CCG supertags assigned by depccg for wh-extraction of the word that.

(NP\NP)/(S[dci]/NP) NP[nb]

128



Figure 6.19: Gold derivation of a fronted quotation encapsulated in parentheses in the DPTB.



Figure 6.20: Faulty derivation of a fronted quotation encapsulated in parentheses output by the CCG^{gate} model.

Unexpected Increase: Discontinuous Dependencies The per-phenomenon analysis in section 6.3.3 has shown consistent improvements in precision and recall for discontinuous dependencies for both PIPE_{bank} and CCG^{gate}. F-score for the former increases by +3.5 absolute points compared to the baseline while the score for the latter rises by +4.2 points. This result is surprising since the manual analysis of this type of discontinuity in section 5.3 suggested that the CCG formalism does not feature a transparent and consistent way to encode such relationships. In the examples that were examined, the right part of the discontinuous constituent was assigned a functional category that suggests analysing it as a modifier of the middle constituent surrounded by the discontinuity (cf. figure 5.21).

In his supplementary material Coavoux (2021) includes information about the structure of extraposed dependents in the DPTB development dataset in form of the label(s) of the node(s) that are surrounded by the long-range dependency, called gap here, and the node label(s) of the appended subsection of the discontinuous constituent, abbreviated to *right*. I use these assignments to compute separate scores which shed light on the parsers' increased competence regarding discontinuous dependencies.

				PIPEbank					$\mathrm{CCG}^{\mathrm{gate}}$		
Gap	Count	Exact match	Partial match	Precision	n Recall	F_1	Exact match	Partial match	Precision	n Recall	F_1
PP	19	42.1 (+10.5)	42.1 (+10.5)	88.9 (+22.2)	36.4 (+9.1)	51.6 (+12.9)	47.4 (+15.8)	47.4 (+15.8)	90.9 (+24.2)	45.5 (+18.2)	60.6 (+21.9)
VP	10	0.0 (-10.0)	0.0 (-10.0)	0.0 (-100)	0.0 (-10.0)	0.0 (-18.2)	0.0 (-10.0)	0.0 (-10.0)	0.0 (-100)	0.0 (-10.0)	0.0 (-18.2)
NP	5	60.0 (+20.0)	60.0 (+0.0)	$100 \\ (+0.0)$	50.0 (+0.0)	66.7 (+0.0)	20.0 (-20.0)	20.0 (-40.0)	100 (+0.0)	16.7 (-33.3)	28.6 (-38.1)

Table 6.9: Labelled evaluation results for discontinuous dependency according to the node label of the gap element. Absolute improvements over the baseline are included in parentheses. Only cases with at least five occurrences are reported.

				PIPEbank					$\mathrm{CCG}^{\mathrm{gate}}$		
Right	Count	Exact match	Partial match	Precision	n Recall	\mathbf{F}_1	Exact match	Partial match	Precision	n Recall	\mathbf{F}_1
SBAR	13	23.1 (+7.7)	23.1 (+7.7)	100 (+60.0)	21.4 (+7.1)	35.3 (+14.2)	7.7 (-7.7)	7.7 (-7.7)	50.0 (+10.0)	7.1 (-7.2)	12.5 (-8.6)
S	11	72.7 (+9.1)	72.7 (+9.1)	88.9 (-11.1)	$ \begin{array}{c} 66.7 \\ (+8.4) \end{array} $	76.2 (+2.5)	63.6 (+0.0)	63.6 (+0.0)	$100 \\ (+0.0)$	58.3 (+0.0)	73.7 (+0.0)
\mathbf{PP}	7	0.0 (+0.0)	14.3 (+0.0)	50.0 (-50.0)	9.1 (+0.0)	15.4 (-1.3)	14.3 (+14.3)	28.6 (+14.3)	100 (+0.0)	27.3 (+18.2)	42.9 (+26.2)

Table 6.10: Labelled evaluation results for discontinuous dependency according to the node label of the right element. Absolute improvements over the baseline are included in parentheses. Only cases with at least five occurrences are reported.

It can be seen that the improvements in F-score for both models are caused by a solid increase in competence of the analysis of prepositional phrase gap elements (pipeline: +12.9, auxiliary: +21.9), with the improvement being strongly driven by better precision. Furthermore F-score for PP right elements increases significantly for the auxiliary model (+26.2). This confirms the previous finding from the error analysis in section 6.3.4 which showed reductions in PP attachment errors.

Both models experience a striking reduction in VP gap F-score (-18.2) resulting in a worstpossible score of 0.0. For CCG^{gate} this is combined with a reduction for SBAR right elements (-8.6) which is a common combination for discontinuous NP constituents as in figure 5.20.

A manual analysis showed that the drop in F-score for cases with VP nodes in the gap indeed corresponds to misleading supertag assignments such as in figure 5.21. The appended half of the discontinuous constituent is assigned a supertag in the CCGrebank that takes as argument the gap verb with all object slots filled $((S\NP)\(S\NP))$, i.e. it acts as a VP modifier. The parser learns to predict this supertag and is misled by the fact that standard projective VP modifiers carry the same supertag. Thus, it wrongly merges the gap element with the right constituent part, not constructing a discontinuity at all.

In other cases of discontinuous dependency, the CCG category assignments appear to be closer to the DPTB constituent structure. Take for instance (6.8a) which features a discontinuous NP category. Figure 6.21 shows part of its DPTB gold tree which is correctly retrieved by CCG^{gate} except that the topmost VP and its NP daughter are merged into a single VP node.

- (6.8) a. The institutions appeared confident Japanese regulators would step in to ensure orderly trading if necessary, and there was considerable speculation during the day that the Finance Ministry was working behind the scenes to do just that.
 - b. The need for hurried last-minute telephone negotiations among market officials will disappear once <u>rules</u> are in place that synchronize circuit breakers in all markets .



Figure 6.21: DPTB derivation including a discontinuous dependency with a PP gap element and an SBAR right element.

Figure 6.22 reproduces the corresponding CCGrebank category assignments. The appended part which can be reduced to S[em] is encoded as a core argument of *speculation* and not as a modifier of the gap. Thus, the relationship between the two parts of the discontinuous constituent has a direct correspondent in the supertag assignments. The parser is able to profit from the fact that the relative clause is encoded as a core argument of *speculation*. While *during the day* lies in between the two elements and has a category that suggests merging with *speculation* (function $(N/S)\setminus(N/S)$), the parser is able to identify that this category encodes an adjunct and should be merged after attaching the core argument *that the Finance Ministry* I assume that the distinction between arguments and adjuncts in CCG is a deciding factor for the model's ability to profit from supertag assignments.

there	was	considerable	speculation	during	the day	that	the Finance Ministry was working
NP[thr]	$\overline{(S[dcl] \setminus NP[thr])/NP}$	N/N	N/S[em]	$\overline{((N/S)(N/S))/NP}$	NP	$\overline{\mathrm{S[em]/S[dcl]}}$	S[dcl]
				(N/S)\(N/S	S)		> S[em]
				N/S[em]	<		
						Ν	>
					I	N	>
					Ν	IP	<u></u>
				S[dcl]\N	P[thr]		>
				S[dcl]			<

Figure 6.22: CCGrebank supertag assignment for a sentence that includes discontinuous dependency with a PP gap element and an SBAR right element.

For VP gap elements the problem lies in that the verb serves as the head of the whole clause in CCG and bridging it while retaining the relationship between the discontinuous constituent's substructures would necessitate encoding the appended S[em] constituent as an argument of *speculation*. This could be done by changing the assignments from those in figure 5.21 to figure 6.23 where N of *evidence* becomes NP/S[em] and is combined via crossed backward composition with *surfaced*, retaining the S[em] argument in the result. Alternatively, the argument and return type of *surfaced* could be changed to reflect the new S[em] argument of *evidence* which would give rise to an explosion of new verbal category types (figure 6.24). Both approaches have the problem that they come with the need of marking a new syntactic argument at the left NP category. One could justify this by viewing the right discontinuous part as an argument clause of the noun, but the linguistic motivation would definitely be questionable for relative clauses like in (6.8b). All in all,

In April 1987 ,	evidence	surfaced	that	commissions were p	aid
S/S	$\overline{NP/S[em]}$	$\overline{S[dcl] \setminus NP}$	$\overline{S[em]/S[dcl]}$	S[dcl]	
	S[dcl]	/S[em]		S[em]	->
			S[dcl]		->
		S[dcl]		->

cases like these will likely remain to pose a challenge in CCG-enriched models due to the properties of the CCG formalism.

Figure 6.23: Alternative CCG supertag assignment for discontinuous dependency using crossed composition.

In April 1987 ,	evidence	surfaced	that	commissions were paid		
S/S	$\overline{N/S[em]}$	$\overline{(S[dcl]/S) \setminus (N/S)}$	$\overline{S[em]/S[dcl]}$	S[dcl]		
S		[dcl]/S[em]		S[em]		
	S[dcl]					
		S[dcl		>		

Figure 6.24: Alternative CCG supertag assignment for discontinuous dependency with a verbal category that preserves the argument of its subject.

6.3.6 External Comparison

Table 6.11 shows a comparison between the pipeline model PIPE_{bank}, the auxiliary task model scoring best on discontinuities CCG^{gate} and results of previously proposed parsers on the DPTB. As can be seen, CCG^{gate} provides the highest F-score for fully supervised transition-based parsers on the DPTB test split reported so far, surpassing the previous state-of-the-art model Coavoux (2021), which exhibits wider word embeddings (32 vs. 300) and character biLSTM states (100 vs. 256), by 0.1 absolute points. However, with a difference of 1.2 absolute points, the model is still far off from the overall best-scoring fully supervised model, being the grammar-less chart-based approach of Corro (2020).

Unfortunately, discontinuous F-score turns out to be significantly worse on the test dataset than on the development split. This is surprising since neither of these datasets was used in training. One can see that this effect already occurred in the original stack-free parser of Coavoux and Cohen (2019) (70.9 dev DF, 67.3 test DF) while it is absent from the GAP models from Coavoux et al. (2019) and from Coavoux (2021) which have a similar network design.

Perhaps the discontinuous phenomena in the development split have an unusual composition that the stack-free transition system is able to treat unusually well. During training, the algorithm checks every four epochs if the current network iteration scores better on the development set than the previously reported best-score and if it does, saves it. This way, a model might have been selected that does not generalise well to normal compositions of discontinuous structures.

Table 6.11 also contains a comparison of speeds where they were reported by the respective authors. Note that parsing speed is to a wide extent dependent on the type of implementation and the hardware used to run the parser. I reran the evaluation for the original model of Coavoux and Cohen (2019) to validate their reported speed of 38 sent/s which resulted in a value of 31 sent/s, much closer to the 29 sent/s for CCG^{gate}. I suppose that deviating from the information in their paper, they actually use two CPU threads to run the evaluation. This is also the setting found in the evaluation script which they provide as supplementary material. I ran the original model with two CPU threads which produced a speed of 36 sent/s which is close to their reported speed.

Generally, the evaluation of parsing speeds shows that while training an auxiliary task model is computationally more expensive due to the increase in training instances and model parameters (3 LSTM stacks instead of 2, gated residual connections), it only suffers minimal computational penalties after deployment when compared to its single-task counterpart (delta of 2 sent/s). Pursuing a pipeline approach speeds up training since one can outsource supertagging for the training corpus to the pre-processing pipeline but results in a model that is computationally more costly in the test setting since for every sentence to predict, two full-fledged neural networks are run which cannot be parallelised (delta of 8 sent/s).

Unfortunately, up until now, English per-phenomenon scores computed by the testsuite provided by Coavoux (2021) have only been reported for an ML-GAP-proposal by Coavoux (2021). Table 6.12 contains a comparison of the phenomenon-specific labelled F-score of the models proposed here and the ML-GAP parser from Coavoux (2021) as well as the ML-GAP+BERT variant which is enriched with BERT contextual embeddings. It can be seen that the CCG^{gate} model outperforms ML-GAP in every type of discontinuity except for circumpositioned quotations. Indeed it is striking that all models from the work at hand, including the single-task baseline parser, perform only half as well on this class than ML-GAP and ML-GAP+BERT.

The improvements in F-score on discontinuous dependency attributed to the integration of BERT contextualised embeddings in the ML-GAP+BERT model remain unmatched. As Coavoux et al. (2019) note, most sentences in this class have no lexical trigger like *what* or *that* which makes recognising a discontinuity hard and requires world-knowledge. The ml-gap+bert model shows that this can be achieved to some extent by providing lexical knowledge to the model. Yet, the results of $PIPE_{bank}$ and CCG_{gate}^{gate} show that there is still some improvement to gain in this domain without using semi-supervised contextualised word-embeddings.

Phenomenon	Count	baseline	$\operatorname{Pipe}_{\operatorname{bank}}$	$\mathrm{CCG}^{\mathrm{gate}}$	ML-GAP	ML-GAP+BERT
Wh-extraction	91	84.0	80.8	86.4	79.5	89.8
Fronted quotation	71	94.3	92.2	93.6	92.3	95.8
Discontinuous dependency	37	37.9	41.4	42.1	33.3	71.1
Circumpositioned quotation	16	43.4	46.3	35.0	86.0	95.0
It-extraposition	12	58.8	66.7	66.7	55.6	81.8
Extraction+fronted quotation	7	84.2	84.2	100	97.3	100
Discontinuous dependency+extraction	5	62.9	62.1	68.8	64.5	88.2
Extraction+extraction	5	78.8	94.1	90.9	86.7	97.1
Subject inversion	5	75.0	75.0	75.0	75.0	100

Table 6.12: Comparison with previously published labelled per-phenomenon F-scores including models ML-GAP and ML-GAP+BERT from Coavoux (2021). Sentences where several discontinuous phenomena occur are treated separately. Only combinations with at least 5 occurrences in the DPTB development split are included.

Model	Type	Complexity	Dev		Test		sent/s	
			\mathbf{F}_1	DF_1	\mathbf{F}_1	DF_1	GPU	CPU
Fully supervised								
Evang and Kallmeyer (2011) , [‡] , < 25	GB	$\mathcal{O}(n^9)$	79^{\dagger}					
van Cranenburgh and Bod (2013), ≤ 40	GB	$\mathcal{O}(n^9)$	85.2		85.6			
van Cranenburgh et al. (2016), ≤ 40	GB	$\mathcal{O}(n^9)$	86.9		87.0		$<1^{\square}$	
Ruprecht and Mörbitz (2021)	GB	$\mathcal{O}(n^9)$			90.1	72.9	95^{γ}	
Coavoux and Cohen (2019)	TB	$\mathcal{O}(n^2)$	91.4	70.9	90.9	67.3		$_{38}$ VO
Coavoux et al. (2019)	TB	$\mathcal{O}(n^2)$	91.2	72.0	91.0	71.3		61^{\diamond}
Coavoux (2021)	TB	$\mathcal{O}(n^2)$	92.0	75.9	91.4	74.4		
This work, PIPE _{bank}	TB	$\mathcal{O}(n^2)$	91.5	70.2	91.4	66.8	31^{ϵ}	23 * °O
This work, CCG ^{gate}	TB	$\mathcal{O}(n^2)$	91.8	74.9	91.5	67.7	35^{ϵ}	29♣○
Corro (2020)	CB	$\mathcal{O}(n^3)$			92.7	64.2	355^{δ}	
Stanojević and Steedman (2020)	CB	$\mathcal{O}(n^6)$			90.5	67.1		
Corro et al. (2017)	DB	$\mathcal{O}(n^2)$			89.2			≈ 7.3
Semi-supervised (Pr	etraine	d embeddings)						
Ruprecht and Mörbitz (2021), fasttext+flair	\mathbf{GB}	$O(n^9)$			91.8	76.1	86^{γ}	
Coavoux (2021), fasttext	TB	$\mathcal{O}(n^2)$	92.7	78.1	92.3	76.5		
Corro (2020)	CB	$\mathcal{O}(n^3)$			92.9	64.9		
Vilares and Gómez-Rodríguez (2020), with Ling et al. (2015)	SL	$\mathcal{O}(n^2)$			89.3	45.2	572^{β}	104▲
Chen and Komachi (2023), fasttext	NC	$\mathcal{O}(n^3)$			92.1	78.1	970^{eta}	
Semi-supervise	ed (BEI	RT-base)						
Ruprecht and Mörbitz (2021)	GB	$\mathcal{O}(n^9)$			93.3	80.5	57^{γ}	
Coavoux (2021)	TB	$\mathcal{O}(n^2)$	95.0	85.8	95.0	82.5		
Corro (2020)*	CB	$\mathcal{O}(n^3)$			94.8	68.9		
Vilares and Gómez-Rodríguez (2020)	\mathbf{SL}	$\mathcal{O}(n^2)$			91.9	50.8	80^{β}	2▲
Fernández-González and Gómez-Rodríguez $\left(2021\right)$	RCB	$\mathcal{O}(n^3)$			94.0	68.9	231^{lpha}	
Fernández-González and Gómez-Rodríguez (2021)	RTB	$\mathcal{O}(n^2)$			94.1	67.2	152^{α}	
Semi-supervised (XLNet)								
Chen and Komachi (2023)	NC	$\mathcal{O}(n^3)$			95.0	83.0	375^{eta}	
Fernández-González and Gómez-Rodríguez $\left(2021\right)$	RCB	$\mathcal{O}(n^3)$			95.1	74.1	179^{α}	
Fernández-González and Gómez-Rodríguez (2021)	RTB	$\mathcal{O}(n^2)$			95.5	73.4	133^{α}	

Table 6.11: Global comparison of parsing results on the DPTB; rounded to the first decimal place. Complexity for grammar-based approaches assumes fan-out 3 as found in the DPTB. Types: GB: grammar-based, TB: transition-based, CB: grammar-less chart-based, SL: sequence-labelling based, DB: dependency-conversion based, NC: neural combinator, RCB: reorder and projective grammar-less chart-based, RTB: reorder and projective transition-based. *Size of BERT not reported. [†] Does not discount root symbols and punctuation, [‡] gold POS tags. Speeds are presented on the test set as reported by the respective authors and dependent on hardware and implementation. GPUs: ^{α}GeForce RTX 3090, ^{β}GeForce GTX 1080 Ti, ^{γ}GeForce RTX 2080, ^{ϵ}GeForce RTX 2060 SUPER 8GB, ^{δ}Tesla V100. CPUs: ^{\wedge}Intel i7-7700 @ 3.60GHz, ^{\vee}Intel i7, ^{\diamond}AMD Ryzen 5 3600 @ 3.60GHz. ^{\bigcirc}run on single core. ^{\diamond} reported on dev set. ^{\heartsuit}including supertagging. ^{\square}measured by Ruprecht and Mörbitz (2021).

7. Conclusion

This work explored the introduction of lexicalised syntactic information in the form of CCG supertags into discontinuous constituent parsing. To this end, several approaches for parsing discontinuous constituent trees have been discussed: grammar-based chart parsing for linear context-free rewriting systems (LCFRS) as well as neural transition-based parsers (SWAP, GAP, MERGE) among which the neural stack-free approach of Coavoux and Cohen (2019) was chosen as the base for the incorporation of supertags.

After introducing combinatory categorial grammar (CCG) and the supertags it gives rise to, I manually compared correlations between lexical category assignments found in the CCGrebank corpus and the way different types of discontinuities are analysed in the discontinuous Penn Treebank (DPTB). This analysis suggested close relationships for some forms of discontinuity like *wh*movement, fronted quotations and *it*-extrapositions while revealing potentially conflicting analyses for circumpositioned quotations, extraposed dependents and subject-verb inversion. Circumpositioned quotations showed to be especially troublesome since they do not seem to exhibit a coherent and transparent analysis in the CCGrebank. The manual analysis may be helpful for designing models in future works. It underlines the necessity to utilise architectures that can dynamically adapt to the considerably varying phenomenon-dependent task relatedness.

The main merit of this work lies in the implementation of CCG supertagging into the stack-free transition-based parser of Coavoux and Cohen (2019) following a pipeline and an auxiliary-task approach. For the latter, a variety of experimental settings was explored (gated residual connections, increased network width, deconstruction of supertags) showing that care must be exercised and adaptions must be made to benefit from shared representations for both parsing and supertagging. Both the PIPE_{bank} model and the best-scoring auxiliary model CCG^{gate} outperform the supertagless baseline in standard F-score. On discontinuities only, the pipeline model performs worse than the baseline while the auxiliary approach proofs to be an effective means of increasing a parser's syntactic competence regarding long-range dependencies. The auxiliary approach also outperforms its control model which confirms that the improvements can be attributed to the introduction of an auxiliary objective. It show that the complex relationship of seemingly incompatible syntactic descriptions can be effectively made use of to improve the quality of discontinuous constituent parsing. It also indicates that training a multi-task model can leverage statistical correlations between tasks to an extent a pipeline model is not able to. While the improvements are relatively modest, they suggest that research into more adapted specialised multi-task approaches for discontinuous constituent parsing that learn across representations is a worthwhile endeavour.

This result is closely connected with the utilisation of an alternative to the standard summative residual connection. Gated residual connections have been empirically shown to outperform their widely used counterpart in a hierarchical multi-task framework with stable results for biLSTM stacks up to four levels (CTR_4^{gate}).

The mixed results of networks with additional CCG supertag-based tasks in the form of decomposed supertags (CCG_{multi}^{gate}) and head-dependency information provided by the CCGrebank (CCG_{dep}^{gate}) indicate that care must be exercised when enriching a model with information that goes beyond the treatment of CCG supertags as discrete categories. Head-dependency information seems to be too close to the conflicting notion of CCG derivations. While subcomponent information does not beat the best-scoring multi-task model CCG^{gate} , it does improve the discontinuous metrics compared to its four-layer control model. Noteworthy about this result is the slight improvement in discontinuous recall which deviates from the trend of recall drops observable in most other multi-task models. This suggests that subcomponent training could actually be of value in a multi-task approach.

Additionally, I provided a first systematic investigation of the usefulness of four different syn-

tactic sequence labelling tasks as auxiliary objectives for transition-based parsing of discontinuous constituents. While CCG supertagging shows to be most beneficial for discontinuous constituents, the inclusion of chunking produces the best general F-score. Since the results may be influenced to some extent by the vastly differing numbers of distinct categories, it is difficult to draw conclusions about the individual tasks' informativeness for parsing. Nonetheless, the evaluation is a good starting point for the choice of auxiliary tasks for future projects and should be transferable to other neural parser types.

7.1. Open Questions

While this work has shown that introducing CCG supertagging as an auxiliary task for discontinuous constituent parsing can yield systematic improvements in parsing score, the experiments have not come close to the current state-of-the-art results which are dominated by models that use XLnet (Yang et al., 2019) and BERT (Devlin et al., 2019) contextual embeddings. Thus, the open question remains, whether CCG supertagging would provide any informative value to such models or if the syntactic competence induced by supertagging is already overtly encoded in the lexical information these embeddings provide.

Another question raised by this work deals with the viability of combining several types of unrelated auxiliary tasks. Here, I have only explored multi-task learning with one additional type of syntactic representation. The unsuccessful joint learning of CCG head-dependencies is the closest I have come to combining several representations. As explained in section 6.3, the different auxiliary tasks yield different types of improvements. For instance chunking improves general F-score while CCG supertagging affects discontinuous constituents more. Judging from this, combining several beneficial auxiliary tasks in a shared model could leverage additional synergies. However, this also raises the risk of overfitting and necessitates a computationally expensive training procedure. Furthermore, arranging several unrelated auxiliary tasks in a hierarchical architecture would pose a challenge. Overall, the potential of such an exploration is unclear.

A similar question stands for the deconstruction of supertags into subcomponent labelling tasks. Even though this approach does not perform better than the supertag-only model in this work, the unique improvement of discontinuous recall raises the question whether the kind of deconstructions or the model architecture can be improved upon to benefit from this effect.

In view of the recent successful discontinuous parsing approach using LCFRS-based supertags (Ruprecht and Mörbitz, 2021), I am looking forward to exploring ways to implement LCFRS supertagging as an auxiliary task in a neural framework effectively. Reducing the number of supertags in an informed manner as well as finding useful subtasks with smaller label set sizes might be the way to go.

Furthermore, questions regarding the neural architectures explored in this work remain. In future work, I want to explore using the gated residual connections to LSTM cell outputs proposed by Wu et al. (2016a) that inspired the simplified gated residual design in this work. Additionally, the comparison of training and development score of the CCG_{600}^{gate} model in figure 6.9 as well as the results of the CCG_{600}^{gate} model (cf. figure 6.10) suggested that there might be improvements to gain from better regularisation techniques. An interesting possibility is the application of variational dropout for recurrent connections as proposed by Gal and Ghahramani (2016).

Lastly, the exploration of different kinds of supertags as auxiliary tasks still lacks an important contender: lexicalised tree adjoining grammar. While the use of LTAG-spinal has not shown to be an effective means to improve discontinuous constituent parsing results, subcategorisation information as well as argument-adjunct distinctions inherent to traditional LTAG may be properties akin to the characteristics of CCG supertags that have proven helpful for the parser. I am looking forward to performing such an experiment after acquiring an LTAG-annotated corpus.

References

- Kazimierz Adjukiewicz. Die syntaktische Konnexität. Studia Philosophica, 1:1–27, 1935. English translation "Syntactic Connexion" by H. Weber in McCall, S. (Ed.) Polish Logic, pages 207–231, Oxford University Press, Oxford, United Kingdom, 1967.
- Krasimir Angelov and Peter Ljunglöf. Fast statistical parsing with parallel multiple context-free grammars. In Proceedings of the 14th Conference of the European Chapter of the Association for Computational Linguistics, Gothenburg, Sweden, pages 368–376, Stroudsburg, Pennsylvania, USA, April 2014. Association for Computational Linguistics. URL https://doi.org/10.3115/ v1/E14-1039.
- Miguel Ballesteros, Yoav Goldberg, Chris Dyer, and Noah A. Smith. Training with exploration improves a greedy stack LSTM parser. In Su et al. (2016), pages 2005–2010. URL https://doi.org/10.18653/v1/D16-1211.
- Srinivas Bangalore and Aravind K. Joshi. Supertagging: An approach to almost parsing. Computational Linguistics, 25(2):237-265, jun 1999. URL https://aclanthology.org/J99-2004.
- Srinivas Bangalore and Aravind Krishna Joshi, editors. Introduction, pages 1-31. In Bangalore and Joshi (2010b), 2010a. ISBN 978-0-262-01387-1. URL https://doi.org/10.7551/mitpress/ 8370.001.0001.
- Srinivas Bangalore and Aravind Krishna Joshi, editors. Supertagging: Using Complex Lexical Descriptions in Natural Language Processing. MIT Press, Cambridge, Massachusetts, USA, 2010b. ISBN 978-0-262-01387-1. URL https://doi.org/10.7551/mitpress/8370.001.0001.
- Yehoshua Bar-Hillel. A quasi-arithmetical notation for syntactic description. Language, 29(1): 47-58, 1953. ISSN 00978507, 15350665. URL http://www.jstor.org/stable/410452.
- Shai Ben-David and Reba Schuller. Exploiting task relatedness for multiple task learning. In Learning Theory and Kernel Machines, pages 567–580, Berlin, Germany, 2003. Springer. ISBN 978-3-540-45167-9. URL https://doi.org/10.1007/978-3-540-45167-9_41.
- Yoshua Bengio, Patrice Simard, and Paolo Frasconi. Learning long-term dependencies with gradient descent is difficult. *IEEE Transactions on Neural Networks*, 5(2):157–166, 1994. URL https://doi.org/10.1109/72.279181.
- Joachim Bingel and Anders Søgaard. Identifying beneficial task relations for multi-task learning in deep neural networks. In Lapata et al. (2017b), pages 164–169. URL https://aclanthology.org/E17-2026.
- E. Black, S. Abney, D. Flickenger, C. Gdaniec, R. Grishman, P. Harrison, D. Hindle, R. Ingria, F. Jelinek, J. Klavans, M. Liberman, M. Marcus, S. Roukos, B. Santorini, and T. Strzalkowski. A procedure for quantitatively comparing the syntactic coverage of English grammars. In Patti Price, editor, *Fourth DARPA Speech and Natural Language Workshop*, *Pacific Grove, California*, pages 306–311, Stroudsburg, Pennsylvania, USA, February 1991. Association for Computational Linguistics. URL https://aclanthology.org/H91-1060.
- Pierre Boullier. A generalization of mildly context-sensitive formalisms. In Proceedings of the Fourth International Workshop on Tree Adjoining Grammars and Related Frameworks (TAG+4), pages 17–20, University of Pennsylvania, August 1998a. Institute for Research in Cognitive Science. URL https://aclanthology.org/W98-0105.
- Pierre Boullier. Proposal for a natural language processing syntactic backbone. Research Report RR-3342, INRIA, Le Chesnay-Rocquencourt, France, 1998b. URL https://inria.hal. science/inria-00073347.
- Cem Bozsahin. Chapter 5. Combinatory Categorial Grammar, pages 61-86. De Gruyter Mouton, Berlin, Germany, 2013. ISBN 9783110296877. URL https://doi.org/10.1515/ 9783110296877.61.
- Sabine Brants, Stefanie Dipper, Peter Eisenberg, Silvia Hansen-Schirra, Esther König, Wolfgang Lezius, Christian Rohrer, George Smith, and Hans Uszkoreit. TIGER: Linguistic interpretation of a German corpus. *Research on Language and Computation*, 2:597–620, 2004. URL https: //doi.org/10.1007/s11168-004-7431-3.
- Joan Bresnan, Ronald M. Kaplan, Stanley Peters, and Annie Zaenen. Cross-serial dependencies in dutch. *Linguistic Inquiry*, 13(4):613–635, 1982. URL http://www.jstor.org/stable/4178298.
- Harry Bunt and Éric Villemonte de la Clergerie, editors. Proceedings of the 11th International Conference on Parsing Technologies (IWPT'09), Paris, France, Stroudsburg, Pennsylvania, USA, October 2009. Association for Computational Linguistics. URL https://aclanthology.org/ volumes/W09-38/.
- Marie Candito. Auxiliary tasks to boost biaffine semantic dependency parsing. In Findings of the Association for Computational Linguistics: ACL 2022, Dublin, Ireland, pages 2422-2429, Stroudsburg, Pennsylvania, USA, May 2022. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/2022.findings-acl.190.
- Rich Caruana. Multitask learning. Machine Learning, 28:41-75, July 1997. URL https://doi. org/10.1023/A:1007379606734.
- John Chen, Srinivas Bangalore, and Krishnamurti Vijay-Shanker. Automated extraction of treeadjoining grammars from treebanks. *Natural Language Engineering*, 12(3):251–299, 2006. URL https://doi.org/10.1017/S1351324905003943.
- Zhousi Chen and Mamoru Komachi. Discontinuous Combinatory Constituency Parsing. Transactions of the Association for Computational Linguistics, 11:267-283, 03 2023. ISSN 2307-387X. URL https://doi.org/10.1162/tacl_a_00546.
- Noam Chomsky. Three models for the description of language. IRE Transactions on Information Theory, 2(3):113-124, 1956. URL https://doi.org/10.1109/TIT.1956.1056813.
- Noam Chomsky. Formal properties of language. In R. Duncan Luce, Robert R. Bush, and Eugene Galanter, editors, *Handbook of Mathematical Psychology*, volume 2, pages 323–418. John Wiley & Sons., New York City, New York, USA, 1963.
- Noam Chomsky. Remarks on nominalization. In Roderick A. Jacobs and Peter S. Rosenbaum, editors, *Readings in English Transformational Grammar*, pages 184–221. Ginn and Company, A Xerox Company, Waltham, Massachusetts, USA, 1970.
- Noam Chomsky. The Logical Structure of Linguistic Theory. Springer US, New York City, New York, USA, 1975. ISBN 978-0-306-30760-7.
- Alonzo Church. A set of postulates for the foundation of logic. Annals of Mathematics, 34(4): 839–864, 1933. ISSN 0003486X. URL https://doi.org/10.2307/1968702.

- Stephen Clark and James R. Curran. Supertagging for efficient wide-coverage CCG parsing. In Bangalore and Joshi (2010b), pages 222–233. ISBN 978-0-262-01387-1. URL https://doi.org/ 10.7551/mitpress/8370.001.0001.
- Maximin Coavoux. BERT-proof syntactic structures: Investigating errors in discontinuous constituency parsing. In Findings of the Association for Computational Linguistics: ACL-IJCNLP 2021, Online, pages 3259–3272, Stroudsburg, Pennsylvania, USA, August 2021. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/2021.findings-acl.288.
- Maximin Coavoux and Shay B. Cohen. Discontinuous constituency parsing with a stack-free transition system and a dynamic oracle. In Jill Burstein (2019), pages 204–217. URL https://doi.org/10.18653/v1/N19-1018.
- Maximin Coavoux and Benoît Crabbé. Incremental discontinuous phrase structure parsing with the GAP transition. In Lapata et al. (2017a), pages 1259–1270. URL https://aclanthology.org/E17-1118.
- Maximin Coavoux and Benoît Crabbé. Multilingual lexicalized constituency parsing with wordlevel auxiliary tasks. In Lapata et al. (2017b), pages 331–336. URL https://aclanthology. org/E17-2053.
- Maximin Coavoux, Benoît Crabbé, and Shay B. Cohen. Unlexicalized Transition-based Discontinuous Constituency Parsing. Transactions of the Association for Computational Linguistics, 7: 73–89, 04 2019. ISSN 2307-387X. URL https://doi.org/10.1162/tacl_a_00255.
- Trevor Cohn, Yulan He, and Yang Liu, editors. *Findings of the Association for Computational Linguistics: EMNLP 2020, Online*, Stroudsburg, Pennsylvania, USA, November 2020. Association for Computational Linguistics. URL https://aclanthology.org/2020.findings-emnlp.
- Ronan Collobert and Jason Weston. A unified architecture for natural language processing: Deep neural networks with multitask learning. In *ICML '08: Proceedings of the 25th international conference on Machine learning, Helsinki, Finland*, pages 160—167, New York City, New York, USA, 2008. Association for Computing Machinery. ISBN 9781605582054. URL https://doi. org/10.1145/1390156.1390177.
- Caio Corro. Span-based discontinuous constituency parsing: a family of exact chart-based algorithms with time complexities from $\mathcal{O}(n^6)$ down to $\mathcal{O}(n^3)$. In Webber et al. (2020), pages 2753-2764. URL https://doi.org/10.18653/v1/2020.emnlp-main.219.
- Caio Corro, Joseph Le Roux, and Mathieu Lacroix. Efficient discontinuous phrase-structure parsing via the generalized maximum spanning arborescence. In Palmer et al. (2017), pages 1644–1654. URL https://doi.org/10.18653/v1/D17-1172.
- Benoit Crabbé. An LR-inspired generalized lexicalized phrase structure parser. In Proceedings of COLING 2014, the 25th International Conference on Computational Linguistics: Technical Papers, pages 541-552, Dublin, Ireland, August 2014. Dublin City University and Association for Computational Linguistics. URL https://aclanthology.org/C14-1052.
- Benoit Crabbé. Multilingual discriminative lexicalized phrase structure parsing. In Proceedings of the 2015 Conference on Empirical Methods in Natural Language Processing, Lisbon, Portugal, pages 1847–1856, Stroudsburg, Pennsylvania, USA, September 2015. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/D15-1212.
- James Cross and Liang Huang. Incremental parsing with minimal features using bi-directional LSTM. In Erk and Smith (2016), pages 32–37. URL https://doi.org/10.18653/v1/P16-2006.

- James Cross and Liang Huang. Span-based constituency parsing with a structure-label system and provably optimal dynamic oracles. In Su et al. (2016), pages 1–11. URL https://doi.org/10. 18653/v1/D16-1001.
- H. B. Curry. Grundlagen der kombinatorischen Logik. American Journal of Mathematics, 52(3): 509–536, 1930. ISSN 00029327, 10806377. URL https://doi.org/10.2307/2370619.
- Marie-Catherine de Marneffe, Bill MacCartney, and Christopher D. Manning. Generating typed dependency parses from phrase structure parses. In Proceedings of the Fifth International Conference on Language Resources and Evaluation (LREC'06), Genoa, Italy, Paris, France, May 2006. European Language Resources Association (ELRA). URL https://aclanthology.org/ L06-1260.
- Jacob Devlin, Ming-Wei Chang, Kenton Lee, and Kristina Toutanova. BERT: Pre-training of deep bidirectional transformers for language understanding. In Jill Burstein (2019), pages 4171–4186. URL https://doi.org/10.18653/v1/N19-1423.
- Greg Durrett and Dan Klein. Neural CRF parsing. In Zong and Strube (2015), pages 302-312. URL https://doi.org/10.3115/v1/P15-1030.
- Katrin Erk and Noah A. Smith, editors. Proceedings of the 54th Annual Meeting of the Association for Computational Linguistics (Volume 2: Short Papers), Berlin, Germany, Stroudsburg, Pennsylvania, USA, August 2016. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/P16-2.
- Kilian Evang. Parsing discontinuous constituents in English. Master's thesis, University of Tübingen, Tübingen, Germany, January 2011. URL https://kilian.evang.name/publications/ mathesis.pdf.
- Kilian Evang and Laura Kallmeyer. PLCFRS parsing of English discontinuous constituents. In Proceedings of the 12th International Conference on Parsing Technologies, Dublin, Ireland, pages 104–116, Stroudsburg, Pennsylvania, USA, October 2011. Association for Computational Linguistics. URL https://aclanthology.org/W11-2913.
- Daniel Fernández-González and Carlos Gómez-Rodríguez. Reducing discontinuous to continuous parsing with pointer network reordering. In Bonnie Webber, Trevor Cohn, and Yang Liu Yulan He, editors, Proceedings of the 2021 Conference on Empirical Methods in Natural Language Processing, Online and Punta Cana, Dominican Republic, pages 10570–10578, Stroudsburg, Pennsylvania, USA, November 2021. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/2021.emnlp-main.825.
- David Gaddy, Mitchell Stern, and Dan Klein. What's going on in neural constituency parsers? an analysis. In Marilyn Walker (2018), pages 999–1010. URL https://doi.org/10.18653/v1/ N18-1091.
- Yarin Gal and Zoubin Ghahramani. A theoretically grounded application of dropout in recurrent neural networks. In Proceedings of the 30th International Conference on Neural Information Processing Systems, Barcelona, Spain, pages 1027—-1035, Red Hook, New York, USA, 2016. Curran Associates Inc. ISBN 9781510838819.
- Gerald Gazdar. Applicability of indexed grammars to natural languages. In U. Reyle and C. Rohrer, editors, *Natural Language Parsing and Linguistic Theories*, pages 69–94, Dordrecht, The Netherlands, 1988. Springer Netherlands. ISBN 978-94-009-1337-0. URL https: //doi.org/10.1007/978-94-009-1337-0_3.

- Xavier Glorot and Yoshua Bengio. Understanding the difficulty of training deep feedforward neural networks. In Yee Whye Teh and Mike Titterington, editors, *Proceedings of the Thirteenth International Conference on Artificial Intelligence and Statistics*, volume 9 of *Proceedings of Machine Learning Research*, pages 249–256, Chia Laguna Resort, Sardinia, Italy, May 2010. Proceedings of Machine Learning Research (PMLR). URL https://proceedings.mlr.press/v9/glorot10a.html.
- Yoav Goldberg. Neural Network Methods for Natural Language Processing. Synthesis Lectures on Human Language Technologies. Springer Nature Switzerland, Cham, Switzerland, 1 edition, 2022. ISBN 978-3-031-01037-8. URL https://doi.org/10.1007/978-3-031-02165-7.
- Yoav Goldberg and Michael Elhadad. An efficient algorithm for easy-first non-directional dependency parsing. In Human Language Technologies: The 2010 Annual Conference of the North American Chapter of the Association for Computational Linguistics, Los Angeles, California, pages 742-750, Stroudsburg, Pennsylvania, USA, June 2010. Association for Computational Linguistics. URL https://aclanthology.org/N10-1115.
- Yoav Goldberg and Joakim Nivre. A dynamic oracle for arc-eager dependency parsing. In Kay and Boitet (2012), pages 959–976. URL https://aclanthology.org/C12-1059/.
- Yoav Goldberg and Joakim Nivre. A dynamic oracle for arc-eager dependency parsing. In Kay and Boitet (2012), pages 959–976. URL https://aclanthology.org/C12-1059.
- Carlos Gómez-Rodríguez, John Carroll, and David Weir. A deductive approach to dependency parsing. In *Proceedings of ACL-08: HLT*, *Columbus, Ohio, USA*, pages 968–976, Stroudsburg, Pennsylvania, USA, June 2008. Association for Computational Linguistics. URL https:// aclanthology.org/P08-1110.
- Carlos Gómez-Rodríguez, Marco Kuhlmann, Giorgio Satta, and David Weir. Optimal reduction of rule length in linear context-free rewriting systems. In Proceedings of Human Language Technologies: The 2009 Annual Conference of the North American Chapter of the Association for Computational Linguistics, Boulder, Colorado, pages 539–547, Stroudsburg, Pennsylvania, USA, June 2009. Association for Computational Linguistics. URL https://aclanthology.org/N09-1061.
- Alex Graves and Jürgen Schmidhuber. Framewise phoneme classification with bidirectional LSTM and other neural network architectures. *Neural Networks*, 18(5):602–610, 2005. ISSN 0893-6080. URL https://doi.org/10.1016/j.neunet.2005.06.042. IJCNN 2005.
- Dick Grune and Ceriel J. H. Jacobs. Parsing Techniques: A Practical Guide. Springer US, New York City, New York, USA, 2 edition, 2010. ISBN 978-1-4419-1901-4. URL https://doi.org/ 10.1007/978-0-387-68954-8.
- Carlos Gómez-Rodríguez. Finding the smallest binarization of a CFG is NP-hard. Journal of Computer and System Sciences, 80(4):796-805, 2014. ISSN 0022-0000. URL https://doi.org/ 10.1016/j.jcss.2013.12.003.
- David Hall, Greg Durrett, and Dan Klein. Less grammar, more features. In Proceedings of the 52nd Annual Meeting of the Association for Computational Linguistics (Volume 1: Long Papers), Baltimore, Maryland, pages 228–237, Stroudsburg, Pennsylvania, USA, June 2014. Association for Computational Linguistics. URL https://doi.org/10.3115/v1/P14-1022.
- Kaiming He, Xiangyu Zhang, Shaoqing Ren, and Jian Sun. Deep residual learning for image recognition. In 2016 IEEE Conference on Computer Vision and Pattern Recognition (CVPR), Las Vegas, Nevada, USA, pages 770–778, New York City, New York, USA, June 2016. Institute of Electrical and Electronics Engineers. URL https://doi.org/10.1109/CVPR.2016.90.

- Sepp Hochreiter and Jürgen Schmidhuber. Long Short-Term Memory. Neural Computation, 9(8): 1735–1780, 11 1997. ISSN 0899-7667. URL https://doi.org/10.1162/neco.1997.9.8.1735.
- Sepp Hochreiter, Yoshua Bengio, Paolo Frasconi, and Jürgen Schmidhuber. Gradient flow in recurrent nets: The difficulty of learning longterm dependencies. In John F. Kolen and Stefan C. Kremer, editors, A Field Guide to Dynamical Recurrent Networks, pages 237–243, New York City, New York, USA, 2001. Wiley-IEEE Press. ISBN 9780470544037. URL https://doi.org/ 10.1109/9780470544037.ch14.
- Julia Hockenmaier and Mark Steedman. CCGbank: A corpus of CCG derivations and dependency structures extracted from the Penn Treebank. *Computational Linguistics*, 33(3):355–396, 2007. URL https://doi.org/10.1162/coli.2007.33.3.355.
- Matthew Honnibal, James R. Curran, and Johan Bos. Rebanking CCGbank for improved NP interpretation. In Jan Hajič, Sandra Carberry, Stephen Clark, and Joakim Nivre, editors, Proceedings of the 48th Annual Meeting of the Association for Computational Linguistics, Uppsala, Sweden, pages 207–215, Stroudsburg, Pennsylvania, USA, July 2010. Association for Computational Linguistics. URL https://aclanthology.org/P10-1022.
- John E. Hopcroft, Rajeev Motwani, and Jeffrey D. Ullman. Introduction to Automata Theory, Languages, and Computation. Pearson Education, Boston, Massachusetts, USA, 3 edition, 2007. ISBN 0321455363.
- Joop Houtman. Coordination and constituency: a study in categorial grammar. dissertation, Rijksuniversiteit Groningen, Groningen, The Netherlands, 1994. URL https://research.rug. nl/en/publications/coordination-and-constituency-a-study-in-categorial-grammar.
- Alex Ivliev. Parsing of lexicalised linear context-free rewriting systems via supertagging, May 2020. URL https://iccl.inf.tu-dresden.de/web/Misc3072/en.
- Haoming Jiang, Pengcheng He, Weizhu Chen, Xiaodong Liu, Jianfeng Gao, and Tuo Zhao. SMART: Robust and efficient fine-tuning for pre-trained natural language models through principled regularized optimization. In *Proceedings of the 58th Annual Meeting of the Association for Computational Linguistics*, *Online*, pages 2177–2190, Stroudsburg, Pennsylvania, USA, July 2020. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/2020. acl-main.197.
- Thamar Solorio Jill Burstein, Christy Doran, editor. Proceedings of the 2019 Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies, Volume 1 (Long and Short Papers), Minneapolis, Minnesota, USA, Stroudsburg, Pennsylvania, USA, June 2019. Association for Computational Linguistics. URL https:// aclanthology.org/N19-1.
- Richard Johansson and Yvonne Adesam. Training a Swedish constituency parser on six incompatible treebanks. In Proceedings of the Twelfth Language Resources and Evaluation Conference, Marseille, France, pages 5219–5224, Paris, France, May 2020. European Language Resources Association. ISBN 979-10-95546-34-4. URL https://aclanthology.org/2020.lrec-1.642.
- Aravind K. Joshi. Tree adjoining grammars: How much context-sensitivity is required to provide reasonable structural descriptions? In David R. Dowty, Lauri Karttunen, and Arnold M. Zwicky, editors, *Natural Language Parsing: Psychological, Computational, and Theoretical Perspectives*, Studies in Natural Language Processing, pages 206–250, Cambridge, United Kingdom, 1985. Cambridge University Press. URL https://doi.org/10.1017/CB09780511597855.007.

- Aravind K. Joshi, Leon S. Levy, and Masako Takahashi. Tree adjunct grammars. Journal of Computer and System Sciences, 10(1):136–163, 1975. ISSN 0022-0000. URL https://doi.org/ 10.1016/S0022-0000(75)80019-5.
- Dan Jurafsky and James H. Martin. Speech and language processing. 3rd ed. draft, Appendix E (web only), https://web.stanford.edu/~jurafsky/slp3/, Jan 7 2023.
- Daniel Jurafsky and James H. Martin. Speech and Language Processing: An Introduction to Natural Language Processing, Computational Linguistics, and Speed Recognition. Prentice Hall Series in Artificial Intelligence. Pearson Educational International, Upper Saddle River, New Jersey, USA, 2 edition, 2009. ISBN 978-0-13-504196-3.
- Laura Kallmeyer. Parsing Beyond Context-Free Grammars. Cognitive Technologies. Springer Berlin, Heidelberg, 1 edition, 2010. ISBN 978-3-642-26453-5. URL https://doi.org/10.1007/ 978-3-642-14846-0.
- Laura Kallmeyer and Wolfgang Maier. An incremental Earley parser for simple range concatenation grammar. In Bunt and Éric Villemonte de la Clergerie (2009), pages 61-64. URL https://aclanthology.org/W09-3808.
- Laura Kallmeyer and Wolfgang Maier. Data-driven parsing with probabilistic linear contextfree rewriting systems. In Proceedings of the 23rd International Conference on Computational Linguistics, Beijing, China, COLING '10, pages 537—-545, Stroudsburg, Pennsylvania, USA, 2010. Association for Computational Linguistics. URL https://aclanthology.org/C10-1061. pdf.
- Laura Kallmeyer and Wolfgang Maier. Data-driven parsing using probabilistic linear context-free rewriting systems. *Computational Linguistics*, 39(1):87–119, March 2013. ISSN 0891-2017. URL https://doi.org/10.1162/COLI_a_00136.
- Jungo Kasai, Robert Frank, Pauli Xu, William Merrill, and Owen Rambow. End-to-end graphbased TAG parsing with neural networks. In Marilyn Walker (2018), pages 1181–1194. URL https://doi.org/10.18653/v1/N18-1107.
- Martin Kay and Christian Boitet, editors. *Proceedings of COLING 2012: Technical Papers, Mumbai, India, Mumbai, India, December 2012. The COLING 2012 Organizing Committee, Indian Institute of Technology Bombay. URL https://aclanthology.org/C12-1.*
- Paul Kiparsky. The shift to head-initial vp in germanic. In Samuel David Epstein Höskuldur Thráinsson and Steve Peter, editors, *Studies in Comparative Germanic Syntax II*, volume 38 of *Studies in Natural Language and Linguistic Theory*, pages 140–179, Cambridge, Massachusetts, USA, 1996. Kluwer, Dordrecht. ISBN 978-1-4020-0294-6. URL https://doi.org/ 10.7551/mitpress/8370.001.0001.
- Eliyahu Kiperwasser and Miguel Ballesteros. Scheduled Multi-Task Learning: From Syntax to Translation. *Transactions of the Association for Computational Linguistics*, 6:225–240, 04 2018. ISSN 2307-387X. URL https://doi.org/10.1162/tacl_a_00017.
- Eliyahu Kiperwasser and Yoav Goldberg. Simple and accurate dependency parsing using bidirectional LSTM feature representations. *Transactions of the Association for Computational Linguistics*, 4:313–327, 2016. URL https://doi.org/10.1162/tacl_a_00101.
- Donald E. Knuth. A generalization of dijkstra's algorithm. *Information Processing Letters*, 6(1): 1–5, 1977. ISSN 0020-0190. URL https://doi.org/10.1016/0020-0190(77)90002-3.

- Alexander Koller and Marco Kuhlmann. Dependency trees and the strong generative capacity of CCG. In Lascarides et al. (2009), pages 460–468. URL https://aclanthology.org/E09-1053.
- András Kornai and Geoffrey K. Pullum. The X-bar theory of phrase structure. Language, 66(1): 24–50, 1990. ISSN 00978507, 15350665. URL https://doi.org/10.2307/415278.
- Sandra Kübler, Ines Rehbein, and Josef van Genabith. TePaCoC a testsuite for testing parser performance on complex German grammatical constructions. In Frank Van Eynde, Anette Frank, Koenraad De Smedt, and Gertjan van Noord, editors, Proceedings of the 7th International Workshop on Treebanks and Linguistic Theories (TLT 7), Groningen, The Netherlands, pages 15 – 28, Utrecht, The Netherlands, January 2009. LOT. ISBN 978-90-78328-77-3.
- Marco Kuhlmann and Giorgio Satta. Treebank grammar techniques for non-projective dependency parsing. In Lascarides et al. (2009), pages 478–486. URL https://aclanthology.org/ E09-1055.
- Adhiguna Kuncoro, Miguel Ballesteros, Lingpeng Kong, Chris Dyer, Graham Neubig, and Noah A. Smith. What do recurrent neural network grammars learn about syntax? In Lapata et al. (2017a), pages 1249–1258. URL https://aclanthology.org/E17-1117.
- Mirella Lapata, Phil Blunsom, and Alexander Koller, editors. Proceedings of the 15th Conference of the European Chapter of the Association for Computational Linguistics: Volume 1, Long Papers, Valencia, Spain, Stroudsburg, Pennsylvania, USA, April 2017a. Association for Computational Linguistics. URL https://aclanthology.org/volumes/E17-1.
- Mirella Lapata, Phil Blunsom, and Alexander Koller, editors. Proceedings of the 15th Conference of the European Chapter of the Association for Computational Linguistics: Volume 2, Short Papers, Valencia, Spain, Stroudsburg, Pennsylvania, USA, April 2017b. Association for Computational Linguistics. URL https://aclanthology.org/E17-2.
- Alex Lascarides, Claire Gardent, and Joakim Nivre, editors. Proceedings of the 12th Conference of the European Chapter of the ACL (EACL 2009), Athens, Greece, Stroudsburg, Pennsylvania, USA, March 2009. Association for Computational Linguistics. URL https://aclanthology. org/E09-1.
- Mike Lewis and Mark Steedman. A* CCG parsing with a supertag-factored model. In Proceedings of the 2014 Conference on Empirical Methods in Natural Language Processing (EMNLP), Doha, Qatar, pages 990–1000, Stroudsburg, Pennsylvania, USA, October 2014. Association for Computational Linguistics. URL https://doi.org/10.3115/v1/D14-1107.
- Wang Ling, Chris Dyer, Alan W. Black, and Isabel Trancoso. Two/too simple adaptations of Word2Vec for syntax problems. In Proceedings of the 2015 Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies, Denver, Colorado, pages 1299–1304, Stroudsburg, Pennsylvania, USA, May 2015. Association for Computational Linguistics. URL https://doi.org/10.3115/v1/N15-1142.
- Wolfgang Maier. Discontinuous incremental shift-reduce parsing. In Zong and Strube (2015), pages 1202–1212. URL https://doi.org/10.3115/v1/P15-1116.
- Wolfgang Maier and Timm Lichte. Discontinuous parsing with continuous trees. In Proceedings of the Workshop on Discontinuous Structures in Natural Language Processing, San Diego, California, pages 47–57, Stroudsburg, Pennsylvania, USA, June 2016. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/W16-0906.

- Wolfgang Maier, Miriam Kaeshammer, and Laura Kallmeyer. PLCFRS parsing revisited: Restricting the fan-out to two. In Proceedings of the 11th International Workshop on Tree Adjoining Grammars and Related Formalisms (TAG+11), Paris, France, pages 126–134, September 2012. URL https://aclanthology.org/W12-4615.
- Wolfgang Maier, Miriam Kaeshammer, Peter Baumann, and Sandra Kübler. Discosuite a parser test suite for German discontinuous structures. In Proceedings of the Ninth International Conference on Language Resources and Evaluation (LREC'14), Reykjavik, Iceland, pages 2905-2912, Paris, France, May 2014. European Language Resources Association. URL http://www.lrec-conf.org/proceedings/lrec2014/pdf/230_Paper.pdf.
- Mitchell P. Marcus, Beatrice Santorini, and Mary Ann Marcinkiewicz. Building a large annotated corpus of English: The Penn Treebank. *Computational Linguistics*, 19(2):313–330, 1993. ISSN 0891-2017. URL https://aclanthology.org/J93-2004.
- Amanda Stent Marilyn Walker, Heng Ji, editor. Proceedings of the 2018 Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies, Volume 1 (Long Papers), New Orleans, Louisiana, USA, Stroudsburg, Pennsylvania, USA, June 2018. Association for Computational Linguistics. URL https://aclanthology.org/ volumes/N18-1.
- James D. McCawley. Parentheticals and discontinuous constituent structure. *Linguistic Inquiry*, 13(1):91-106, 1982. URL http://www.jstor.org/stable/4178261.
- Ryan McDonald and Joakim Nivre. Characterizing the errors of data-driven dependency parsing models. In *Proceedings of the 2007 Joint Conference on Empirical Methods in Natural Language Processing and Computational Natural Language Learning (EMNLP-CoNLL)*, *Prague, Czech Republic*, pages 122–131, Stroudsburg, Pennsylvania, USA, June 2007. Association for Computational Linguistics. URL https://aclanthology.org/D07-1013.
- Mark-Jan Nederhof. Weighted Deductive Parsing and Knuth's Algorithm. Computational Linguistics, 29(1):135-143, 03 2003. ISSN 0891-2017. URL https://doi.org/10.1162/ 089120103321337467.
- Joakim Nivre. Algorithms for deterministic incremental dependency parsing. Computational Linguistics, 34(4):513-553, 2008. URL https://doi.org/10.1162/coli.07-056-R1-07-027.
- Joakim Nivre. Non-projective dependency parsing in expected linear time. In Proceedings of the Joint Conference of the 47th Annual Meeting of the ACL and the 4th International Joint Conference on Natural Language Processing of the AFNLP, Suntec, Singapore, volume 1, pages 351–359, Stroudsburg, Pennsylvania, USA, August 2009. Association for Computational Linguistics. ISBN 9781932432459. URL https://aclanthology.org/P09-1040.
- Joakim Nivre, Marco Kuhlmann, and Johan Hall. An improved oracle for dependency parsing with online reordering. In Bunt and Éric Villemonte de la Clergerie (2009), pages 73–76. URL https://aclanthology.org/W09-3811.
- Martha Palmer, Rebecca Hwa, and Sebastian Riedel, editors. Proceedings of the 2017 Conference on Empirical Methods in Natural Language Processing, Copenhagen, Denmark, Stroudsburg, Pennsylvania, USA, September 2017. Association for Computational Linguistics. URL https: //doi.org/10.18653/v1/D17-1.
- Razvan Pascanu, Tomas Mikolov, and Yoshua Bengio. On the difficulty of training recurrent neural networks. In Sanjoy Dasgupta and David McAllester, editors, *Proceedings of the 30th International Conference on Machine Learning, Atlanta, Georgia, USA*, volume 28(3) of *Proceedings*

of Machine Learning Research, pages 1310–1318. Proceedings of Machine Learning Research (PMLR), Jun 2013. URL https://proceedings.mlr.press/v28/pascanu13.html.

- Adam Paszke, Sam Gross, Soumith Chintala, Gregory Chanan, Edward Yang, Zachary DeVito, Zeming Lin, Alban Desmaison, Luca Antiga, and Adam Lerer. Automatic differentiation in pytorch. In NIPS 2017 Workshop on Autodiff, Long Beach, California, USA, 2017. URL https: //openreview.net/forum?id=BJJsrmfCZ.
- Barbara Plank, Anders Søgaard, and Yoav Goldberg. Multilingual part-of-speech tagging with bidirectional long short-term memory models and auxiliary loss. In Erk and Smith (2016), pages 412–418. URL https://doi.org/10.18653/v1/P16-2067.
- Carl J. Pollard. Generalized Phrase Structure Grammars, Head Grammars, and Natural Language. dissertation, Stanford University, Stanford, California, USA, 1984.
- Boris T. Polyak and Anatoli B. Juditsky. Acceleration of stochastic approximation by averaging. SIAM Journal on Control and Optimization, 30(4):838–855, 1992. URL https://doi.org/10. 1137/0330046.
- Tapani Raiko, Harri Valpola, and Yann Lecun. Deep learning made easier by linear transformations in perceptrons. In Neil D. Lawrence and Mark Girolami, editors, *Proceedings of the Fifteenth International Conference on Artificial Intelligence and Statistics*, La Palma, Canary Islands, volume 22, pages 924–932. Proceedings of Machine Learning Research (PMLR), Apr 2012. URL https://proceedings.mlr.press/v22/raiko12.html.
- Raul Rojas. A tutorial introduction to the lambda calculus, 2015. URL https://doi.org/10. 48550/arXiv.1503.09060.
- Thomas Ruprecht and Richard Mörbitz. Supertagging-based parsing with linear context-free rewriting systems. In *Proceedings of the 2021 Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies, Online*, pages 2923–2935, Stroudsburg, Pennsylvania, USA, June 2021. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/2021.naacl-main.232.
- Kenji Sagae and Alon Lavie. A classifier-based parser with linear run-time complexity. In Proceedings of the Ninth International Workshop on Parsing Technology, Vancouver, British Columbia, pages 125–132, Stroudsburg, Pennsylvania, USA, October 2005. Association for Computational Linguistics. URL https://aclanthology.org/W05-1513.
- Itiroo Sakai. Syntax in universal translation. In Proceedings of the International Conference on Machine Translation and Applied Language Analysis, National Physical Laboratory, Teddington, United Kingdom, sep 1961. URL https://aclanthology.org/1961.earlymt-1.31.
- Victor Sanh, Thomas Wolf, and Sebastian Ruder. A hierarchical multi-task approach for learning embeddings from semantic tasks. *Proceedings of the AAAI Conference on Artificial Intelligence*, 33(01):6949–6956, Jul. 2019. URL https://doi.org/10.1609/aaai.v33i01.33016949.
- Yves Schabes and Aravind K. Joshi. Parsing with lexicalized tree adjoining grammar. In Masaru Tomita, editor, *Current Issues in Parsing Technology*, pages 25–47, Boston, Massachusetts, USA, 1991. Springer US. ISBN 978-1-4615-3986-5. URL https://doi.org/10. 1007/978-1-4615-3986-5_3.
- Hiroyuki Seki, Takashi Matsumura, Mamoru Fujii, and Tadao Kasami. On multiple contextfree grammars. *Theoretical Computer Science*, 88(2):191–229, 1991. ISSN 0304-3975. URL https://doi.org/10.1016/0304-3975(91)90374-B.

- Libin Shen, Champollion Lucas, and Aravind K. Joshi. Ltag-spinal and the treebank. Lang Resources & Evaluation, 42:1-19, October 2007. URL https://doi.org/10.1007/ s10579-007-9043-7.
- Stuart M. Shieber. Evidence against the context-freeness of natural language. Linguistics and Philosophy, 8:333—-343, august 1985. URL https://doi.org/10.1007/BF00630917.
- Klaas Sikkel and Anton Nijholt. Parsing of context-free languages. In Grzegorz Rozenberg and Arto Salomaa, editors, Handbook of Formal Languages: Volume 2. Linear Modeling: Background and Application, pages 61–100, Berlin, Germany, 1997. Springer Berlin Heidelberg. ISBN 978-3-662-07675-0. URL https://doi.org/10.1007/978-3-662-07675-0_2.
- Wojciech Skut, Thorsten Brants, Brigitte Krenn, and Hans Uszkoreit. A linguistically interpreted corpus of German newspaper text. In *Proceedings of the First International Conference on Language Resources and Evaluation, LREC, Granada, Spain*, pages 705–712, Paris, France, May 1998. European Language Resources Association. URL https://arxiv.org/abs/cmp-lg/ 9807008v1.
- Anders Søgaard and Yoav Goldberg. Deep multi-task learning with low level tasks supervised at lower layers. In Erk and Smith (2016), pages 231–235. URL https://doi.org/10.18653/v1/ P16-2038.
- Anders Søgaard and Wolfgang Maier. Treebanks and mild context-sensitivity. In Proceedings of FG 2008: The 13th conference on Formal Grammar, Hamburg, Germany, Stanford, California, USA, August 2008. CSLI Publications.
- Drahomíra Spoustová and Miroslav Spousta. Dependency parsing as a sequence labeling task. *The Prague Bulletin of Mathematical Linguistics*, 74:7–14, september 2010. URL https://ufal. mff.cuni.cz/pbml/94/art-johanka.pdf.
- Rupesh Kumar Srivastava, Klaus Greff, and Jürgen Schmidhuber. Training very deep networks, 2015. URL https://doi.org/10.48550/arXiv.1507.06228.
- Miloš Stanojević and Raquel G. Alhama. Neural discontinuous constituency parsing. In Palmer et al. (2017), pages 1666–1676. URL https://doi.org/10.18653/v1/D17-1174.
- Miloš Stanojević and Mark Steedman. Span-based LCFRS-2 parsing. In Proceedings of the 16th International Conference on Parsing Technologies and the IWPT 2020 Shared Task on Parsing into Enhanced Universal Dependencies, Online, pages 111–121, Stroudsburg, Pennsylvania, USA, July 2020. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/ 2020.iwpt-1.12.
- Mark Steedman. Constituency and coordination in combinatory grammar. In Mark R. Baltin and Anthony S. Kroch, editors, *Alternative Conceptions of Phrase Structure*, pages 201–231. The University of Chicago Press, Chicago, Illinois USA, 1989.
- Mark Steedman. Surface Structure and Interpretation. Linguistic Inquiry Monographs. MIT Press, Cambridge, Massachusetts, USA, 1996. ISBN 9780262193795.
- Mark Steedman. The Syntactic Process. Language, Speech, and Communication. MIT Press, Cambridge, Massachusetts, USA, 2000. ISBN 9780262692687.
- Michalina Strzyz, David Vilares, and Carlos Gómez-Rodríguez. Sequence labeling parsing by learning across representations. In Proceedings of the 57th Annual Meeting of the Association for Computational Linguistics, Florence, Italy, pages 5350–5357, Stroudsburg, Pennsylvania, USA,

July 2019a. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/P19-1531.

- Michalina Strzyz, David Vilares, and Carlos Gómez-Rodríguez. Viable dependency parsing as sequence labeling. In Jill Burstein (2019), pages 717–723. URL https://doi.org/10.18653/v1/N19-1077.
- Jian Su, Kevin Duh, and Xavier Carreras, editors. Proceedings of the 2016 Conference on Empirical Methods in Natural Language Processing, Austin, Texas, Stroudsburg, Pennsylvania, USA, November 2016. Association for Computational Linguistics. URL https://doi.org/10.18653/ v1/D16-1.
- Martin Sundermeyer, Ralf Schlüter, and Hermann Ney. LSTM neural networks for language modeling. In Proc. Interspeech 2012, Portland, Oregon, USA, pages 194–197, Baixas, France, September 2012. International Speech Communication Association. URL https://doi.org/ 10.21437/Interspeech.2012-65.
- Erik F. Tjong Kim Sang and Sabine Buchholz. Introduction to the CoNLL-2000 shared task chunking. In Fourth Conference on Computational Natural Language Learning and the Second Learning Language in Logic Workshop, 2000. URL https://aclanthology.org/W00-0726.
- Andreas van Cranenburgh. Efficient parsing with linear context-free rewriting systems. In Proceedings of the 13th Conference of the European Chapter of the Association for Computational Linguistics, Avignon, France, pages 460–470, Stroudsburg, Pennsylvania, USA, April 2012. Association for Computational Linguistics. URL https://www.aclweb.org/anthology/E12-1047.
- Andreas van Cranenburgh and Rens Bod. Discontinuous parsing with an efficient and accurate DOP model. In Proceedings of the 13th International Conference on Parsing Technologies (IWPT 2013), Nara, Japan, pages 7–16, Stroudsburg, Pennsylvania, USA, November 2013. Assocation for Computational Linguistics. URL https://aclanthology.org/W13-5701.
- Andreas van Cranenburgh, Remko Scha, and Rens Bod. Data-oriented parsing with discontinuous constituents and function tags. *Journal of Language Modelling*, 4(1):57–111, 2016. URL https://doi.org/10.15398/jlm.v4i1.100.
- Yannick Versley. Experiments with easy-first nonprojective constituent parsing. In Proceedings of the First Joint Workshop on Statistical Parsing of Morphologically Rich Languages and Syntactic Analysis of Non-Canonical Languages, pages 39–53, Dublin, Ireland, August 2014. Dublin City University. URL https://aclanthology.org/W14-6104.
- Krishnamurti Vijay-Shanker, David J. Weir, and Aravind K. Joshi. Characterizing structural descriptions produced by various grammatical formalisms. In 25th Annual Meeting of the Association for Computational Linguistics, Stanford, California, USA, pages 104–111, Stroudsburg, Pennsylvania, USA, July 1987. Association for Computational Linguistics. URL https: //doi.org/10.3115/981175.981190.
- David Vilares and Carlos Gómez-Rodríguez. Discontinuous constituent parsing as sequence labeling. In Webber et al. (2020), pages 2771-2785. URL https://doi.org/10.18653/v1/2020. emnlp-main.221.
- Éric Villemonte de la Clergerie. Parsing mildly context-sensitive languages with thread automata. In COLING 2002: Proceedings of the 19th International Conference on Computational Linguistics - Volume 1, Taipei, Taiwan, pages 1–7, Stroudsburg, Pennsylvania, USA, 2002. Association for Computational Linguistics. URL https://doi.org/10.3115/1072228.1072256.

- Alex Warstadt. The Syntax of Coordination and Discontinuity in a Combinatory Categorial Grammar. Presented at Tri-College Undergraduate Linguistics Conference, 09 2015. URL https://alexwarstadt.files.wordpress.com/2019/03/the_syntax_of_ coordination_and_discontin-1.pdf. Presented at Tri-College Undergraduate Linguistics Conference.
- Bonnie Webber, Trevor Cohn, and Yang Liu Yulan He, editors. Proceedings of the 2020 Conference on Empirical Methods in Natural Language Processing (EMNLP), Online, Stroudsburg, Pennsylvania, USA, November 2020. Association for Computational Linguistics. URL https://aclanthology.org/2020.emnlp-main.
- David Jeremy Weir. Characterizing Mildly Context-Sensitive Grammar Formalisms. PhD thesis, University of Pennsylvania, Philadelphia, Pennsylvania, USA, 1988.
- Huijia Wu, Jiajun Zhang, and Chengqing Zong. An empirical exploration of skip connections for sequential tagging. In Proceedings of COLING 2016, the 26th International Conference on Computational Linguistics: Technical Papers, Osaka, Japan, pages 203-212. The COLING 2016 Organizing Committee, December 2016a. URL https://aclanthology.org/C16-1020.
- Yonghui Wu, Mike Schuster, Zhifeng Chen, Quoc V. Le, Mohammad Norouzi, Wolfgang Macherey, Maxim Krikun, Yuan Cao, Qin Gao, Klaus Macherey, Jeff Klingner, Apurva Shah, Melvin Johnson, Xiaobing Liu, Łukasz Kaiser, Stephan Gouws, Yoshikiyo Kato, Taku Kudo, Hideto Kazawa, Keith Stevens, George Kurian, Nishant Patil, Wei Wang, Cliff Young, Jason Smith, Jason Riesa, Alex Rudnick, Oriol Vinyals, Greg Corrado, Macduff Hughes, and Jeffrey Dean. Google's neural machine translation system: Bridging the gap between human and machine translation, 2016b. URL https://doi.org/10.48550/arXiv.1609.08144.
- Zhilin Yang, Zihang Dai, Yiming Yang, Jaime Carbonell, Russ R Salakhutdinov, and Quoc V Le. Xlnet: Generalized autoregressive pretraining for language understanding. In H. Wallach, H. Larochelle, A. Beygelzimer, F. d'Alché-Buc, E. Fox, and R. Garnett, editors, Advances in Neural Information Processing Systems, Vancouver, Canada, volume 32, Red Hook, New York, USA, 2019. Curran Associates, Inc. URL https://proceedings.neurips.cc/paper_files/paper/2019/file/dc6a7e655d7e5840e66733e9ee67cc69-Paper.pdf.
- Masashi Yoshikawa, Hiroshi Noji, and Yuji Matsumoto. A* CCG parsing with a supertag and dependency factored model. In Proceedings of the 55th Annual Meeting of the Association for Computational Linguistics (Volume 1: Long Papers), Vancouver, Canada, pages 277– 287, Stroudsburg, Pennsylvania, USA, 2017. Association for Computational Linguistics. URL http://doi.org/10.18653/v1/P17-1026.
- Poorya Zaremoodi and Gholamreza Haffari. Adaptively scheduled multitask learning: The case of low-resource neural machine translation. In *Proceedings of the 3rd Workshop on Neural Generation and Translation, Hong Kong*, pages 177–186, Stroudsburg, Pennsylvania, USA, November 2019. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/ D19-5618.
- Junru Zhou, Zuchao Li, and Hai Zhao. Parsing all: Syntax and semantics, dependencies and spans. In Cohn et al. (2020), pages 4438-4449. URL https://doi.org/10.18653/v1/2020. findings-emnlp.398.
- Junru Zhou, Zhuosheng Zhang, Hai Zhao, and Shuailiang Zhang. LIMIT-BERT : Linguistics informed multi-task BERT. In Cohn et al. (2020), pages 4450-4461. URL https://doi.org/ 10.18653/v1/2020.findings-emnlp.399.

- Zhenqi Zhu and Anoop Sarkar. Deconstructing supertagging into multi-task sequence prediction. In Proceedings of the 23rd Conference on Computational Natural Language Learning (CoNLL), Hong Kong, pages 12–21, Stroudsburg, Pennsylvania, USA, November 2019. Association for Computational Linguistics. URL https://doi.org/10.18653/v1/K19-1002.
- Chengqing Zong and Michael Strube, editors. Proceedings of the 53rd Annual Meeting of the Association for Computational Linguistics and the 7th International Joint Conference on Natural Language Processing (Volume 1: Long Papers), Beijing, China, Stroudsburg, Pennsylvania, USA, July 2015. Association for Computational Linguistics. URL https://doi.org/10.3115/ v1/P15-1.

Statutory Declaration

I affirm that I have authored the thesis titled

Integrating Supertag Features into Neural Discontinuous Constituent Parsing

independently and without illicit assistance from third parties. All parts of the text, figures, examples and other graphical representations that were taken from the work of others by wording or spirit are marked explicitly with regard to the sources. I have not used any sources other than those specified.

Ich versichere an Eides statt, dass die Arbeit mit dem Titel

Integrating Supertag Features into Neural Discontinuous Constituent Parsing

von mir selbständig und ohne unzulässige fremde Hilfe verfasst wurde. Alle Stellen der Arbeit sowie beigefügte Zeichnungen, Skizzen und andere graphische Darstellungen, die anderen Werken dem Wortlaut oder dem Sinn nach entnommen sind, habe ich unter Angabe der Quellen kenntlich gemacht. Ich habe keine anderen als die angegebenen Quellen genutzt.

Lukas Mielczarek